Remote Working and Innovative Work Behaviour: A Case of Australian Employees

Snjezana Ahlgren

Bachelor of Business – Applied Economics/International Trade Victoria University

> Master of Applied Economics Monash University

Thesis submitted for the fulfilment of the requirements for the degree of Doctor of Philosophy

Victoria University, Australia Institute of Sustainable Industries and Liveable Cities

August 2024

Abstract

This thesis explores the relationship between innovative work behaviour (IWB) and remote work among Australian employees, acknowledging innovation as a pivotal strategy for organisational competitiveness and recognising that those innovative ideas often stem from employees. The research problem stems from the contemporary landscape, characterised by the rise of remote working, and accentuated by the COVID-19 pandemic, necessitating an exploration of how remote work may impact an employee's ability to engage in innovative practices.

Employing a comprehensive mixed-methods approach encompassing quantitative surveys and qualitative in-depth interviews, this study delves deep into the employees' experiences of IWB. Guided by the Ability, Motivation, and Opportunity framework (Baily, 1993), it endeavours to understand the ways in which remote work influences IWB, from the perspectives of the employees themselves. Thus, the central question guiding this research is: How do employees perceive the relationship between remote work and their innovative work behaviour? In particular, what is the extent of IWB among employees engaged in remote work, how do demographics influence the outcomes of employees' IWB when engaged in remote work, what factors contribute to fostering or inhibiting IWB in remote working environments, and what are the distinctions in employee experience regarding IWB when comparing remote and office-based working environments.

There are several findings from this study. However, the key findings are as follows. Employees exhibit more IWB with greater frequency of remote work, as long as remote work does not encompass their entire work schedule. In the context of demographic factors, full-time employment, longer tenure, and greater seniority are associated with higher IWB among remote workers. The findings also show that remote working is a double-edged sword when it comes to employee experience. While remote working affords employees autonomy, flexibility, and uninterrupted time for ideation, it can also lead to isolation, over-working, and reduced opportunities for spontaneous idea exchange.

The study makes a significant contribution by clarifying the paradoxes associated with remote work and IWB as highlighted within existing literature and makes attempts to connect remote work research with innovation research, effectively bridging a crucial gap in understanding. By enriching the existing body of knowledge, the findings provide valuable insights for both organisations and policymakers. These insights can guide the development of remote work policies and practices aimed at cultivating an environment conducive to innovation among employees.

Student Declaration

I, Snjezana Ahlgren, declare that the PhD thesis entitled Remote Working and Innovative Work Behaviour: A Case of Australian Employees, is no more than 80,000 words in length including quotes and exclusive of tables, figures, appendices, bibliography, references and footnotes. This thesis contains no material that has been submitted previously, in whole or in part, for the award of any other academic degree or diploma. Except where otherwise indicated, this thesis is my own work.

I have conducted my research in alignment with the Australian Code for the Responsible Conduct of Research and Victoria University's Higher Degree by Research Policy and Procedures.

Signature:

Date: August 2024

Ethics Declaration

All research procedures reported in the thesis were approved by the Victoria University Human Research Ethics Committee [Application ID: HRE21-055].

Signature:



Date: August 2024

Dedication

This thesis is dedicated to Mathias Torsten Ahlgren and my two greatest blessings, Skye and Anya.

Acknowledgements

I would like to sincerely thank all those who have supported and helped me in my journey to successfully complete this thesis. Throughout this long journey, I have met many individuals who inspired, encouraged, and pushed me. I wish to acknowledge the efforts of many people, without whom this thesis would not be published.

Firstly, I would like to thank my Principal Supervisor Associate Professor Shahnaz Naughton who is an extraordinary supervisor. She is passionate, funny, encouraging, and above all else, highly committed to all her students. She wants each and every one of us to succeed, and it was with her guidance, intelligence and mentoring I persevered. She supported me from the very start, with my minor thesis in pre-candidature. I hold her in immense admiration; not only has she been my supervisor, but she has also become a mentor and a lifelong friend. Despite my own doubts, she believed I would finish my thesis on time. Though she described herself as "tough," she was always fair and always kind in her approach. She gave me the autonomy to do my research and craft my ideas, and with that she taught me how to be an independent researcher.

I would also like to thank Dr Wesley McClendon who came onboard as my associate supervisor. What a privilege it was to have him in my corner. An experienced, articulate, passionate academic and industry professional, our conversations were always so exciting and motivating. He encouraged me to pursue a quantitative study to complement the qualitative, and whilst I was nervous, I never felt unsupported or alone. It was definitely the right choice to add this element to my study. His stories and anecdotes added so much richness to my own thoughts and ideas, and he always gave me more to think about.

I would also like to thank my parents, Mirko and Ljubica Bogve who have always supported and carried me through. Their unconditional love and support have allowed me to pursue my dreams. I don't think you could ask for better parents, I am truly blessed. Although my father never had the opportunity to pursue higher education for himself, he is the most intelligent person I know. His knowledge is wide ranging, and our robust conversations and debates have always inspired me. I would also like to express my gratitude to my sister Spomenka, who has always been a source of encouragement, cheering me on with love and belief in my journey.

I would also like to take this opportunity to express my sincere gratitude and thanks to several of my fellow research students, in particular Mohammed Alhamami, who gave me advice, encouragement, and motivation. Along with fellow students, Dr Christine Millward was an incredible teacher who gave her time and helped me during the quantitative analysis.

Lastly, I owe everything to my family. To my husband Mathias, I thank you for your love, encouragement and your unwavering support. You gave me the space and time to do my work and when I wanted to quit, you encouraged me to keep going. Special thanks to my wonderful daughters Skye and Anya. I know you have made sacrifices for me to pursue this PhD, often not having my time or attention. My hope is that I've not only made you proud but also demonstrated the power of perseverance and determination. You are my motivation, and I'm endlessly proud to be your mother.

Publications and Presentations Associated with this Thesis

Conference Posters

Ahlgren, S. (2020) *Innovative Work Behaviours: Self-organisation in Flexible Working Environments*. Poster presented at the Twentieth International Conference on Knowledge, Culture, and Change in Organizations. University of Illinois at Chicago, Chicago, USA, 27–28 May 2020.

Conference Presentations

Ahlgren, S. (2020) *Innovative Work Behaviours: Self-organisation in Flexible Working Environments.* Paper presented at the Twenty First International Conference on Knowledge, Culture, and Change in Organizations. Auckland University, Auckland, New Zealand, 16–17 January 2021.

Ahlgren, S. and Naughton, S, (2022) *Do employees remain innovative when working remotely?* Paper presented at the 2022 Higher Degree by Research (HDR) Conference, Victoria University Business School, Melbourne, Australia, 4 November 2022.

Ahlgren, S. and Naughton, S, (2022) *Do employees remain innovative when working remotely?* Paper presented at the Thirty Fifth ANZAM Conference. Griffith University, Gold Coast, Australia, 5–7 December 2022.

Abstract	t	ii
Student	Declaration	iv
Dedicati	ion	v
Acknow	ledgements	vi
Publicat	ions and Presentations Associated with this Thesis	viii
Table of	f Contents	ix
List of T	Tables	xvi
List of F	igures	xix
Abbrevi	ations	xx
CHAPT	ER 1 - INTRODUCTION	1
1.1.	Introduction	1
1.2.	Background	2
1.3.	The research problem	4
1.4.	Thesis objectives	5
1.5.	Research questions	6
1.6.	Thesis scope and definitions	7
1.7.	The significance of the thesis	8
1.8.	Thesis method	9
1.9.	Thesis structure	10
1.10.	Chapter summary	12
CHAPT	ER 2 - LITERATURE REVIEW	13
2.1.	Introduction	13
2.2.	Innovation	13
2.2.1.	Innovation and Innovative Work Behaviour	15
2.3.	Innovative Work Behaviour	16
2.3.1.	Definitions of Innovative Work Behaviour	18
2.3.2.	Phases of Innovative Work Behaviour	19
2.3.3.	Measurement of Innovative Work Behaviour	23
2.3.4.	Significance of Innovative Work Behaviour	25
2.3.5.	Influences on Innovative Work Behaviour	26
2.3.6.	Innovative Work Behaviour in the Australian context	28
2.4.	Remote working	31
2.4.1.	Definitions of remote working	31
		Page ix

Table of Contents

2.4.2.	Outcomes of remote working	32
2.4.3	Remote working in Australia	34
2.5.	Innovative Work behaviour and remote working	35
2.5.1	Autonomy and flexibility	36
2.5.2	Work-life balance and isolation	37
2.5.3	Communication	38
2.5.4	Organisational culture, support and trust	39
2.5.5.	Collaboration	41
2.5.6	Physical proximity	41
2.6.	Hypothesis development	43
2.6.1	Remote working frequency and IWB	. 44
2.6.2	Age of remote working employees and IWB	45
2.6.3	Gender of remote working employees and IWB	46
2.6.4	Tenure of remote working employees and IWB	49
2.6.5	Employment Status of remote working employees and IWB	50
2.6.6	Employment level of remote working employees and IWB	51
2.7.	Theoretical framework	52
2.7.1	Ability, Motivation and Opportunity framework	52
2.8.	Conceptual framework	56
2.9.	Gaps in the literature	57
2.10.	Chapter summary	58
СНАРТ	ER 3 - RESEARCH METHODOLOGY	59
3.1.	Introduction	59
3.2.	Research paradigm	59
3.2.1	Interpretivism	61
3.3.	Rationale for mixed methods approach	64
3.4.	Research design	65
3.5.	Research questions	67
3.6.	Population sample	69
3.7.	Ethics	69
3.8.	Chapter summary	69
CHAPT	ER 4 - THE QUANTITATIVE RESEARCH METHOD	70
4.1.	Introduction	70
4.2.	Quantitative research methods	70
	Pa	ige x

4.2.1. Sampling method and sample size	
4.2.2. Participant recruitment	
4.3. Data collection	
4.3.1. Questionnaire design	
4.3.2. Likert scale	
4.3.3. Section 1: Demographics	
4.3.4. Sections 2–5: Idea Exploration, Generation, Championing, and Imp	lementation
75	
4.3.5. Section 6: Participation in the research interview	
4.4. Quantitative data analysis method	
4.1.1. Univariate Analysis	
4.1.2. Bivariate and Multivariate Analysis	
4.1.3. Significance level	
4.1.4. Variables	
4.2. Hypothesis	
4.3. Analysis – ANOVA	80
4.3.1. Testing for normality	
4.3.1.1. Kruskal-Wallis H Test	
4.3.2. Homogeneity of variances	
4.3.3. Post-hoc testing	
4.4. Analysis – Logistic regression	
4.5. Chapter summary	86
CHAPTER 5 - THE QUALITATIVE RESEARCH METHOD	
5.1. Introduction	
5.2. Qualitative research methods	
5.2.1. Sampling method and sample size	
5.2.2. Participant recruitment	89
5.3. Data collection	
5.3.1. Interview design	
5.3.1.1. Section 1: Participant background	
5.3.1.2. Section 2: Usage of remote working	
5.3.1.3. Section 3: Ability, Motivation, Opportunity Factors	
5.4. Qualitative data analysis method	
5.4.1. Thematic analysis steps	
	Page xi

5.4.1	1.1.	Data familiarisation	
5.4.1	1.2.	Generate initial codes	
5.4.1	1.3.	Search for themes	
5.4.1	1.4.	Review themes	
5.4.1	1.5.	Define themes	
5.4.1	1.6.	Write-up	
5.5.	Chap	ter summary	
CHAP	TER 6	- QUANTITATIVE DATA ANALYSIS	
6.1.	Intro	duction	
6.2.	Data	screening and transforming	
6.3.	Desci	riptive statistics	
6.3.1	l. Remo	ote working frequency (RWF)	
6.3.1	1.1.	RWF by age	
6.3.1	1.2.	RWF remotely by gender	
6.3.1	1.3.	RWF by employment status	
6.3.1	1.4.	RWF by tenure	
6.3.1	1.5.	RWF by employment level	
6.3.1	1.6.	Summary of RWF	
6.3.2	2. Innov	vative Work Behaviour	
6.3.2	2.1.	IWB by age	
6.3.2	2.2.	IWB by gender	
6.3.2	2.3.	IWB by employment status	
6.3.2	2.4.	IWB by tenure	
6.3.2	2.5.	IWB by employment level	
6.3.2	2.6.	IWB by <i>RWF</i>	
6.3.2	2.7.	Summary of Innovative Work Behaviour	
6.4.	ANO	VA and Logistic regression testing	
6.4.1	l. Relia	bility analysis	
6.5.	ANO	VA analysis	
6.5.1	I. IWB	among RWF groups (H1)	
6.5.2	2. IWB	among age groups (H2)	
6.5.3	3. IWB	among gender groups (H3)	
6.5.4	4. IWB	among tenure groups (H4)	
6.5.5	5. IWB	among employment status groups (H5)	
			Page xii

6.5.6. IWB among employment level groups (H6)	
6.5.7. Summary of ANOVA tests	
6.6. Logistic regression analysis	
6.6.1. RWF, demographics and IWB (H7)	
6.6.2. RWF, demographics and Idea Exploration phase (H8)	
6.6.3. RWF, demographics and Idea Generation Phase (H9)	
6.6.4. RWF, demographics and Idea Championing Phase (H10)	141
6.6.5. RWF, demographics and Idea Implementation Phase (H11)	
6.6.6. Summary of logistic regression tests	145
6.7. Summary of ANOVA and Logistic Regression tests	145
6.8. Chapter summary	146
CHAPTER 7 - QUALITATIVE DATA ANALYSIS	148
7.1. Introduction	
7.2. Organisation of data	
7.2.1. Data management	
7.2.2. Generate initial codes	
7.2.3. Search for themes	149
7.3. Thematic analysis	149
7.3.1. Resources and tools	151
7.3.2. Perspectives and feelings	153
7.3.3. Support and culture	
7.3.4. Workspaces and tasks	
7.3.5. Communication and interactions	
7.3.6. Collaboration and relationships	
7.4. Chapter summary	181
CHAPTER 8 - FINDINGS AND DISCUSSION	
8.1. Introduction	
8.2. Revisiting the research questions	
8.3. SQ1: The extent of IWB among employees engaged in remote wor	k 183
8.3.1. Remote Working Frequency (RWF) and IWB	
8.4. SQ2: Demographics and IWB	
8.4.1. Demographics and IWB overall	
8.4.2. Demographics and each phase of IWB	
8.4.2.1. Demographics and idea exploration	197 Page xiii

8.4.2.	2.	Demographics and idea generation	. 198
8.4.2.	3.	Demographics and idea championing	. 200
8.4.2.	4.	Demographics and idea implementation	. 203
8.5.	Sumr	nary of quantitative study findings	. 206
8.6.	SQ3:	Factors of remote working that foster and hinder IWB	. 208
8.6.1.	Abili	ty of remote working employees to demonstrate IWB	. 209
8.6.1.	1.	Tangible resources for remote work	. 209
8.6.1.	2.	Exploring and generating new ideas	. 210
8.6.1.	3.	Creative work	. 212
8.6.1.	4.	Innovation	. 214
8.6.1.	5.	Connections in the digital realm	. 217
8.6.2.	Motiv	vation of remote working employees to demonstrate IWB	. 220
8.6.2.	1.	Positive side to remote working	. 220
8.6.2.	2.	Negative side to remote working	. 223
8.6.2.	3.	Importance of remote working to employees	. 226
8.6.3.	Oppo	rtunity of remote working employees to demonstrate IWB	. 228
8.6.3.	1.	Intangible support for remote working	. 228
8.6.3.	2.	Organisational support for innovation	. 230
8.6.3.	3.	Impromptu connections	. 232
8.6.4.	Sumr	nary of factors that foster and hinder IWB	. 234
8.7.	SQ4:	The employee experience between remote and office-based working	. 235
8.7.1.	1.	Frequency of working remotely	. 235
8.7.1.	2.	High focus work	. 237
8.7.1.	3.	Work environments	. 239
8.7.1.	4.	Collaborating in different contexts	. 241
8.7.1.	5.	In person communications	. 244
8.7.2.	Sumr	nary of employee experience between remote and office-based working	ng
	246		
8.8.	Sumr	nary of qualitative study findings	. 247
8.9.	Sumr	nary of findings by research question	. 248
8.10.	Chap	ter summary	. 252
CHAPT	ER 9 -	- CONCLUSION	. 253
9.1.	Intro	duction	. 253
9.2.	Key o	contributions and implications	. 253
		Pa	age xiv

9.2.1. Remote working frequency and demographics influence on IWB	253
9.2.2. Employee experience and perspective	254
9.2.3. Four phase model of IWB	256
9.2.4. Ability, Motivation, Opportunity (AMO) framework	259
9.3. Conclusions from research findings	260
9.4. Limitations of the study	261
9.5. Future areas of research	263
SUMMARY	264
REFERENCES	265
APPENDIX A - Ethics approval	338
APPENDIX B – Questionnaire protocol	340
APPENDIX C – Information to survey participants	345
APPENDIX D – Information to interview participants	347
APPENDIX E – Consent form for interview participants	349
APPENDIX F – Interview protocol	351

List of Tables	
----------------	--

Table 1-1: Research questions and research objectives	7
Table 2-1: Definitions of IWB	. 18
Table 2-2: Phases of IWB	. 20
Table 2-3: Studies that developed IWB Questionnaires	. 24
Table 2-4: Studies of Innovative Work Behaviour	. 27
Table 2-5: Hypothesis summary	. 44
Table 3-1: Research sub-questions	. 68
Table 4-1: Questionnaire sections and items	. 74
Table 4-2: Five-point Likert scale	. 75
Table 4-3: Variables	. 78
Table 4-4: Hypothesis table	. 79
Table 5-1: Summary of participant recruitment	. 90
Table 5-2: Interview sections and items	. 92
Table 5-3: Example questions from interview protocol	. 93
Table 6-1: Data categories following transformation	100
Table 6-2: Continuous dependent variables (sum)	100
Table 6-3: Continuous dependent variables (avg)	101
Table 6-4: Binary dependent variables	101
Table 6-5: Respondent demographics	102
Table 6-6: Overall frequency of working remotely	103
Table 6-7: Cross tabulation of RWF by age	103
Table 6-8: Cross tabulation of RWF by gender	104
Table 6-9: Cross tabulation of RWF by employment status	105
Table 6-10: Cross tabulation of RWF by tenure	105
Table 6-11: Cross tabulation of RWF by employment level	106
Table 6-12: Means and standard deviation	107
Table 6-13: IWB frequency table – percentages of rounded averages	108
Table 6-14: Frequency of IWB by age	109
Table 6-15: Frequency of Idea exploration by age	109
Table 6-16: Frequency of Idea generation by age	110
Table 6-17: Frequency of Idea championing by age	110
Table 6-18: Frequency of Idea implementation by age	111
Table 6-19: Frequency of IWB by gender Pag	112 ge xvi

Table 6-20: Rate of Idea exploration by gender	112
Table 6-21: Frequency of Idea generation by gender	113
Table 6-22: Frequency of Idea championing by gender	113
Table 6-23: Frequency of Idea implementation by gender	114
Table 6-24: Rate of IWB by employment status	114
Table 6-25: Rate of Idea exploration by employment status	115
Table 6-26: Rate of Idea generation by employment status	115
Table 6-27: Rate of Idea championing by employment status	116
Table 6-28: Rate of Idea implementation by employment status	116
Table 6-29: Rate of IWB by tenure	117
Table 6-30: Rate of Idea exploration by tenure	117
Table 6-31: Rate of Idea generation by tenure	118
Table 6-32: Rate of Idea championing by tenure	118
Table 6-33: Rate of Idea implementation by tenure	119
Table 6-34: Rate of IWB by employment level	119
Table 6-35: Rate of Idea exploration by employment level	120
Table 6-36: Rate of Idea generation by employment level	120
Table 6-37: Rate of Idea championing by employment level	121
Table 6-38: Rate of Idea implementation by employment level	121
Table 6-39: Rate of IWB by RWF	122
Table 6-40: Rate of Idea Exploration by RWF	122
Table 6-41: Rate of Idea Generation by RWF	123
Table 6-42: Rate of Idea Championing by RWF	123
Table 6-43: Rate of Idea Implementation by RWF	124
Table 6-44: Reliability test	125
Table 6-45: Normality test for IWB and RWF	126
Table 6-46: One-way ANOVA – Comparison of IWB across RWF	127
Table 6-47: Normality test for IWB and age	128
Table 6-48: One-way ANOVA – Comparison of IWB across Age	128
Table 6-49: Normality test for IWB and gender	129
Table 6-50: T-Test – Comparison of IWB between genders	130
Table 6-51: Normality test for IWB and tenure	130
Table 6-52: Welch ANOVA – Comparison of IWB across tenure	131
Table 6-53: Normality test for IWB and employment status	132
	I age AVII

Table 6-54: One-way ANOVA - Comparison of IWB across employment status 133
Table 6-55: Normality test for IWB and employment level
Table 6-56: One-way ANOVA - Comparison of IWB across employment level 135
Table 6-57: Summary of ANOVA tests 136
Table 6-58: Logistic Regression predicting likelihood of IWB based on RWF and
demographics
Table 6-59: Logistic Regression predicting likelihood of Idea Exploration based on
RWF and demographics140
Table 6-60: Logistic Regression predicting likelihood of Idea Generation based on RWF
and demographics
Table 6-61: Logistic Regression predicting likelihood of Idea Championing based on
RWF and demographics142
Table 6-62: Logistic Regression predicting likelihood of Idea Implementation based on
RWF and demographics144
Table 6-63: Summary of logistic regression tests 145
Table 6-64: Summary of ANOVA and Logistic Regression tests
Table 7-1: Themes and codes
Table 7-2: Theme: Resources and tools 151
Table 7-3: Theme: Perspectives and feelings
Table 7-4: Support and culture theme: Intangible support for remote work 159
Table 7-5: Theme: Workspaces and tasks 162
Table 7-6: Theme: Communication and interactions 167
Table 7-7: Theme: Collaboration and relationships 177
Table 8-1: Research questions and research objectives
Table 8-2: Themes, codes and the AMO Framework 208

List of Figures

Figure 1-1: Thesis structure	. 12
Figure 2-1: Four-phase model of Innovative Work Behaviour	. 21
Figure 2-2: Hypothesis	. 43
Figure 2-3: The AMO model of performance	. 54
Figure 2-4: Conceptual framework	. 57
Figure 3-1: Methodological Pyramid	. 60
Figure 3-2: Reflective Process	. 60
Figure 3-3: Stages of research design	. 67
Figure 4-1: Calculation of sample size	. 72
Figure 5-1: Steps of thematic analysis	. 95

Abbreviations

AMO	Ability, Motivation, and Opportunity Framework	
ES	Employment Status	
EL	Employment Level	
HPWS	High-performance Work Systems	
HRM	Human Resource Management	
IWB	Innovative Work Behaviour	
RWF	Remote Working Frequency	

CHAPTER 1 - INTRODUCTION

1.1. Introduction

In recent years, the nature of work has undergone a transformative shift, propelled by technological advancements and changing societal norms (Jetha et al., 2021; Trenerry et al., 2021). One of the most notable changes is the widespread adoption of remote working, where individuals can carry out their professional responsibilities from locations outside the traditional office setting (Chafi et al., 2021; Cuel et al., 2022; Tomić & Vizinger, 2023). This arrangement is made possible with technology and collaboration tools, which enable individuals to communicate and collaborate with colleagues and access necessary resources remotely. The COVID-19 pandemic saw a record increase in employees working remotely (Adekoya et al., 2022; Beck & Hensher, 2022; Vij et al., 2023). In Australia during the pandemic, the Australian Bureau of Statistics found that at least 40 per cent of the Australian workforce reported working remotely one or more times a week during the pandemic's peak (ABS, 2020). Post-pandemic, it is suggested that working remotely will continue (Barrero et al., 2021; Brown & Tousey, 2023). Australian employees are willing to forgo up to 8 percent of their annual wages for the option to work remotely in a post-pandemic world (Vij et al., 2023) .

Remote working offers numerous benefits, such as increased flexibility, improved work-life balance, and reduced commuting time (Al-Madadha et al., 2022; Nizar et al., 2021; W. Wang et al., 2020). However, as this new paradigm becomes increasingly prevalent, assessing its impact on vital work-related outcomes is crucial, particularly innovation and innovative work behaviour (IWB).

Innovation is a cornerstone of long-term organisational success (Hashim et al., 2022; Lopes et al., 2021; Shanker et al., 2017), demanding careful consideration within this evolving remote work landscape. IWB refers to the generation, implementation, and promotion of new ideas, processes, products, or services within the workplace, by the employees of the organisation. West and Farr (1990) define IWB as the '…intentional introduction and application within a role, group or organisation, of ideas, processes, products or procedures, new to the relevant unit of adoption, designed to significantly benefit the individual, the group, organisation or wider society.' This definition highlights the generation and promotion of creative ideas and their successful implementation,

individually or collaboratively, within work contexts. Scott and Bruce (1994) define IWB as a '...multistage process, with different activities and individual behaviour necessary at each stage'. This definition emphasises that innovative work behaviour involves deliberate actions to introduce and implement new ideas or practices that improve work outcomes at various levels. IWB involves activities beyond routine tasks and requires employees to engage in creative problem-solving, experimentation, and risk-taking. Both definitions underscore the importance of novelty, creativity, and the application of ideas in the work setting. IWB involves actively engaging in activities beyond the routine and contributing to positive changes, whether at the individual, team, or organisational level.

Understanding how remote working influences IWB has become a topic of great interest for both researchers and practitioners seeking to optimise work arrangements and foster a culture of innovation (Ferreira et al., 2022; Meiryani et al., 2022; Zappalà et al., 2021). This thesis aims to contribute to the existing body of knowledge by exploring the relationship between remote working and IWB. By examining the experiences and perceptions of individuals engaged in remote work, we seek to uncover the potential drivers and barriers that may impact an employee's ability to engage in innovative behaviours.

A mixed-methods approach was employed to conduct the research, incorporating both quantitative surveys and qualitative interviews. The survey gathered self-reported data on the frequency of innovative behaviour exerted by remote workers across the fourphase IWB model developed by De Jong and Den Hartog (2010). The qualitative interviews provide in-depth insights into the lived experiences of remote workers, shedding light on the challenges and facilitators they encounter in fostering innovation while working remotely – understanding perceptions of their work arrangement, levels of IWB, and relevant contextual factors that influence this relationship.

1.2. Background

Innovation is imperative for organisations to maintain their competitive advantage and succeed (Chatzoglou & Chatzoudes, 2018; Edison et al., 2018; Tidd & Bessant, 2018). Increasingly, the sources of competitive advantage are found within an organisation's intellectual capabilities, where creativity and innovation are the foundations for success and firm performance (Ali, 2021; Sheth & Sinfield, 2022). 'One option for organisations to become more innovative is to encourage their employees to be innovative' (Agarwal, 2014, p. 43). Organisations cannot be innovative without their employees, as individual employees are often the source of new or novel ideas (De Jong & Den Hartog, 2007).

The emergence of the COVID-19 pandemic triggered an unprecedented set of government actions in Australia, which included restrictions on movement, quarantine requirements, restrictions on gatherings and non-essential services and the closing of Australia's borders to non-residents (Evans, 2022; Nguyen et al., 2021; Price et al., 2020). In Australia, a federal government-issued public health order was issued in all Australian states and territories urging employees to work from home where reasonably practical (Beck & Hensher, 2022; Evans, 2022; Mayangsari, 2020). As a result, there was a fundamental shift in how organisations did business. Employees moved from the office to their homes, and online working, video conferencing and teleconferencing rapidly became the new normal (Merone & Whitehead, 2021). Up to 88 percent of organisations in Australia either encouraged or required employees to work from home during the pandemic (Mitchell, 2020). The Families in Australia Survey: Towards COVID Normal found that 67 percent of survey respondents sometimes or always worked from home during COVID-19, compared to 42 percent pre-COVID (Baxter; & Warren, 2021). In the second year of the pandemic, the Australian Bureau of Statistics stated that 41 percent of employed people regularly worked from home. This was an increase of 32 percent from the previous year (ABS, 2021).

Australian organisations faced the challenge of managing increased levels and, at times, mandatory remote working while ensuring they remain innovative and competitive. 'Research shows that innovation is the key driver of success for many organizations...Work-from-home conditions only exacerbate this need, as organizations are required to find novel digital means of not only surviving but maintaining a competitive edge under unprecedented uncertainty' (Abhari et al., 2023, p. 2031). This challenge continues as employees will be unlikely to revert to pre-pandemic working arrangements (Carroll & Conboy, 2020; Phillips, 2020; Vincenzi et al., 2022; Williamson & Pearce, 2022). Many organisations experienced performance gains due to the shift to remote work, to the extent that numerous organisations intend to maintain remote work even though the pandemic has subsided (Loignon et al., 2022).

1.3. The research problem

The primary focus of this thesis is understanding the relationship between remote working and IWB from the employee perspective. Innovation is crucial in enabling organisations to sustain a competitive advantage over their counterparts (Fatonah & Haryanto, 2022; Ferreira et al., 2020; Nangoy et al., 2020). Employees' IWB represents a significant factor conducive to fostering the innovation process within organisations. IWB encompasses the willingness of employees to proactively introduce novel and creative ideas relevant to their tasks in the workplace, ultimately contributing to the advancement and success of the organisation.

Despite the abundance of literature on IWB and remote working separately, studies connecting IWB specifically with remote working are scarce (Coenen & Kok, 2014; George et al., 2020; Giannetti & Madia, 2013; Gilson et al., 2015; Martínez-Sánchez et al., 2011; Moll & de Leede, 2016; Sarbu, 2022). Notwithstanding the substantial transition to remote work prompted by the COVID-19 pandemic, extant literature addressing the nexus between these two phenomena remains limited and the gap is apparent as highlighted by Becker et al. (2022) 'While it is apparent that increased remote work is not going away anytime soon, if ever, the impact of this mandatory shift to remote work on employee perceptions, wellbeing and work-related behaviours remains underexplored' (Becker et al., 2022, p. 450). There is abundant research on innovation at a firm level, but there is a lack of knowledge about encouraging innovation at an individual level especially when relating to remote work (Bos-Nehles, Renkema, et al., 2017). In addition, the presence of conflicting knowledge regarding IWB coupled with the ambiguity surrounding how workplace policies and structures impact it, makes it challenging to discern the most effective methods for stimulating employees to actively participate in IWB (Bos-Nehles, Renkema, et al., 2017).

The research problem this study aims to address stems from an apparent gap in the literature. Despite the increasing prevalence of remote work (Helmy et al., 2023), there is limited understanding of its impact on IWB. Research on IWB and remote work is still evolving, and more studies are needed to fully understand the relationship between the two. This encompasses an exploration of the various facets of remote working, including but not limited to communication, collaboration tools, role of technology, role of face-to-face interaction, organisational culture, and job autonomy, and how these factors either stimulate or hinder employees' inclination and capacity for innovative thinking and action. Further, it aims to increase our understanding of how remote work compares to office-bound working arrangements in terms of innovative behaviours. While empirical literature on the relationship is limited, extant literature offers mixed findings.

For instance, remotely interacting team-mates appear to miss the creative benefits that can flow from frequent face-to-face interactions (Allen et al., 2015). While remote working has a positive effect on task performance, it can negatively affect innovation-related behaviours, suggesting that there are potential limitations in generating innovative ideas (Gajendran & Harrison, 2007). On the other hand, remote working was found to be positively associated with employee creativity since this type of work arrangement allows for autonomy, flexibility, comfort and reduced distractions in the working environment (Fukumura et al., 2021; Tønnessen et al., 2021; Vega et al., 2015). Remote work can enhance innovative behaviours if job design includes autonomy, skill variety, and feedback (Grant et al., 2011). When employees have greater autonomy over their work and are given opportunities for skill development, remote work arrangements can facilitate innovative thinking and behaviours.

1.4. Thesis objectives

The thesis aims to achieve several objectives in exploring the relationship between remote working and IWB. These objectives provide a clear direction for the research and guide the investigation into understanding the influence of remote working on IWB.

The first research objective of the thesis is to assess the extent to which remote workers engage in IWB. This aspect involves measuring the self-reported frequency of innovative behaviours demonstrated by remote-working employees, such as generating new ideas, promoting new ideas and implementation. By quantifying the level of IWB, the thesis seeks to understand how remote working may influence the extent to which employees exhibit innovative behaviours.

The second research objective is to explore demographic variables to ascertain their potential differential impact on the degree of IWB employees report when working remotely. The third research objective seeks to explore lived experiences of remote workers. This objective entails gathering qualitative insights through interviews to elucidate remote workers' experiences, perceptions, and perspectives regarding the supportive elements and impediments encountered in engaging in innovative practices while working remotely. The thesis examines various individual, organisational, and contextual factors facilitating or impeding innovative behaviours within remote work contexts. Variables such as communication channels, technological infrastructure, autonomy, and team collaboration are analysed to discern their influence on the innovative behaviours of remote workers.

The final objective is to understand from the employees' viewpoint, the distinctions in their innovative behaviour when contrasting remote work with in-office work. This involves a targeted exploration into employees' experiences and viewpoints to discern the effectiveness of various factors in fostering IWB within these distinct work environments.

By achieving these objectives, the thesis seeks to contribute to understanding the relationship between remote working and IWB. The findings could provide valuable contributions towards current and future research, as well as insights for organisations, people managers, and remote workers themselves, enabling them to make informed decisions and implement strategies to foster a culture of innovation in remote work settings.

1.5. Research questions

To meet the thesis objectives, the central question guiding this research is: *How do employees perceive the relationship between remote work and their innovative work behaviour (IWB)*? Table 1.1 outlines the research sub-questions and the corresponding research objective.

	Research Sub-Questions	Research Objectives
SQ1	What is the extent of IWB among employees engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?	RO1: Understand the extent to which remote workers engage in innovative work behaviour.
SQ2	How do demographics influence the outcomes of employees' IWB when engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?	RO2: Explore the impact of demographic variables of remote working employees to discern their influence on innovative work behaviours.
SQ3	What factors contribute to fostering or inhibiting innovative work behaviours in remote working environments?	RO3: Explore the factors of remote working that foster or hinder innovative work behaviours.
SQ4	What are the distinctions in employee experience regarding innovative work behaviour when comparing remote and office-based working environments?	RO4: Explore the employee's lived experience between remote and office-based working in terms of innovative behaviour.

1.6. Thesis scope and definitions

The scope of this thesis is focused on Australian-based employees who have experience with spending some or all of their working hours away from their employer's main premises.

IWB is a multifaceted concept encompassing the intentional and proactive generation, introduction, and application of new ideas within a work role, group, or organisation. It involves employees' willingness and capacity to engage in creative problem-solving and idea generation relevant to their tasks to enhance role performance, group effectiveness, or organisational outcomes. The thesis uses the West and Farr (1990) definition of IWB: '...the intentional introduction and application within a role, group or organisation, of ideas, processes, products or procedures, new to the relevant unit of adoption, designed to significantly benefit the individual, the group, the organisation or

wider society'. The four-phase IWB model discussed by De Jong and Den Hartog (2010) is used as the basis for the research, which includes idea exploration, idea generation, idea championing and idea implementation.

The term 'telecommuting' emerged approximately fifty years ago during the oil crisis of the 1970s, credited to Jack Nilles, who coined the term while working remotely as an engineer on NASA's communications systems (Maghlaperidze et al., 2021). Initially conceived as a remedy for traffic congestion and a means to curtail the utilisation of fossil fuels, telecommuting emerged as a strategic response to the challenges posed by traditional commuting practices. During this period, information and technology firms were among the early adopters of telecommuting practices (Allen et al., 2015). Numerous terms are employed interchangeably to denote the same phenomenon, including telework, telecommuting, work from home (WFH), and mobile work (Allen et al., 2015; Bick et al., 2023).

In the examination of the telecommuting literature, all definitions shared the fundamental concept that telecommuting entails conducting work from a location distant from a central office. Moreover, most definitions also incorporate the essential element of utilising technology to accomplish tasks while operating outside the confines of the central office. This thesis uses the definition of Allen et al. (2015, p. 44) as '...a work practice that involves members of an organisation substituting a portion of their typical work hours (ranging from a few hours per week to nearly full-time) to work away from a central workplace – typically principally from home – using technology to interact with others as needed to conduct work tasks.'

1.7. The significance of the thesis

The significance of this thesis lies in its exploration of the influence of remote working on IWB, contributing valuable new knowledge to a rapidly evolving field. With the trend of remote working here to stay, driven by shifting work preferences and technological advancements (Chafi et al., 2021; Costa et al., 2022; Tomić & Vizinger, 2023), it is imperative to understand how this mode of work impacts innovation within organisations. Innovation is critical for organisational success, growth, and competitiveness (Hashim et al., 2022; Lopes et al., 2021; Shanker et al., 2017) making it essential to identify the factors that support innovative behaviours in remote settings. By identifying these factors, this research provides actionable insights for organisations to enhance their remote work setups, ensuring that employees can generate and implement new ideas effectively. Additionally, by offering a comprehensive understanding of how remote work impacts each phase of IWB, this research contributes to knowledge around cultivating innovation across all stages of the innovation process in remote settings. Specifically, how to facilitate brainstorming and creativity among remote teams (idea exploration), remote collaboration (idea generation), support and advocacy for new ideas within dispersed teams (idea championing) and translating innovative ideas into tangible outcomes in remote work environments (idea implementation).

Moreover, this thesis delves into the impacts of demographics on IWB in remote settings. Understanding how factors such as age, gender, tenure, employment work arrangements and level may influence IWB is a significant contribution. Different demographic groups may experience remote working differently, and recognising these differences can help in creating more inclusive and supportive work environments that promote innovation across all segments of the workforce.

Additionally, comparing innovative behaviour in remote working settings with office-based working is crucial for gaining a comprehensive understanding of the advantages and challenges associated with each mode of work. This comparison provides insights into fostering innovation regardless of the work location and within dispersed teams.

In summary, this thesis makes a significant contribution to the fields of organisational behaviour, human resource management, and innovation theory by filling a critical gap in the literature and offering new insights pertinent to the evolving landscape of today's work environment.

1.8. Thesis method

This thesis employed a mixed-methods approach combining qualitative and quantitative methods to gather and analyse data. Mixed-methods research is '...the type of research in which the researcher or team of researchers combines elements of qualitative and quantitative research approaches...for the purposes of breadth and depth of understanding and corroboration.' (Johnson et al. (2007, p. 123). This approach allows researchers to gain a deeper understanding of the research topic by capturing both Page 9

numerical data and rich qualitative insights. By integrating qualitative and quantitative data collection and analysis techniques, a mixed-methods study can provide a more holistic view, complementing each other's strengths and compensating for their limitations (Dawadi et al., 2021; Green et al., 2014). This approach allowed the exploration of the research objectives from multiple angles, integrating different perspectives and shedding light on the 'what' and the 'why' of the research topic.

Quantitative data was captured using an online survey, and qualitative data was gathered using semi-structured interviews. Descriptive statistics, variance analysis and regression models, were used to analyse the quantitative data to identify potential relationships. The qualitative data was analysed through thematic analysis, where themes, patterns, and unique perspectives were identified. The qualitative data provided an indepth understanding of the context, experiences, and perspectives of individuals involved.

Combining qualitative and quantitative methods in a mixed-methods study offered several advantages. Qualitative data can provide context and depth to quantitative findings, while quantitative data can lend credibility and generalisability to qualitative insights. Furthermore, a mixed-methods study allowed for triangulation of data, where findings from different sources are compared, corroborated, or merged to enhance the validity and reliability of the results (Denzin & Lincoln, 2017).

1.9. Thesis structure

This thesis has nine chapters as follows:

Chapter 1 provides the research context, research problem, significance of the thesis, research objectives, and research questions.

Chapter 2 synthesises the current literature on remote working and IWB. Its purpose is to explore the theoretical concepts used throughout this thesis and the gap in knowledge in this field.

Chapters 3 to 5 discuss the overarching research design. This includes the research methods used for the thesis and a justification for the research paradigm and mixed-

methods methodology. The methods of quantitative and qualitative data collection and analysis used are also discussed.

Chapter 6 discusses the results of the quantitative study. The data explores the relationship between IWB and remote working. The aim is to understand how often employees exert innovative behaviour overall and during the four-phase model of IWB. Also to understand which demographic variables have an influence on IWB when working remotely.

Chapter 7 presents the results of the qualitative study. The data explores Australian-based employees' perceptions of the influence remote working may have on their IWB. Additionally, it seeks to juxtapose their experiences between working in-office and working remotely.

Chapter 8 discusses the study's major findings from the quantitative analysis and qualitative analysis, providing answers to the thesis research questions.

Finally, Chapter 9 addresses the thesis's contributions and limitations, as well as addressing future areas of research.

Figure 1-1: Thesis structure



1.10. Chapter summary

As organisations continue to navigate the evolving landscape of work arrangements where remote working will continue to prevail, it becomes increasingly important to delve into the nuances of remote work's impact on innovation. This introductory chapter has provided an overview of the thesis and laid the foundation for the research. It introduced the background of the research, followed by the research problem and objectives. A central research question and four sub-questions were proposed in accordance with the research aims. The next chapter provides a literature review with the aim of identifying research gaps.

CHAPTER 2 - LITERATURE REVIEW

2.1. Introduction

This thesis aims to increase our understanding of innovative work behaviour (IWB) in remote working settings. Thus, this chapter reviews the literature to position this research within the existing knowledge about IWB and remote working. It is noted that prior to the COVICD-19 pandemic, few studies explored the combined topics of this research. However, with the onset of COVID-19 and the rise in remote working, there has been a growing body of literature in recent years.

Numerous studies have delved into the multifaceted realms of IWB, alongside investigations into the dynamics of remote work. Scholars from disciplines such as organisational behaviour, human resource management, psychology, and innovation management have contributed to this research.

While there has been an increase in research on innovative behaviour and remote working, it is still a developing field, and there is scope for further exploration. The dynamic nature of remote work and the ongoing changes in work arrangements necessitate continuous research to understand its impact on innovation and innovative behaviour in remote settings.

This chapter begins with a review of the literature about the dimensions of innovation. Section 2.3 introduces the concept of IWB. Section 2.4 introduces the concept of remote working, and Section 2.7 discusses the theoretical frameworks that underpin these combined concepts. The conceptual framework and gaps in the literature are presented in 2.8 and 2.9.

2.2. Innovation

'Innovation and creativity in the workplace have become increasingly important determinants of organisational performance, success and longer-term survival.' (Anderson et al., 2014)

Innovation is a key driver of economic growth and competition (Bilbao - Osorio & Rodríguez - Pose, 2004; Hasan & Tucci, 2010; LeBel, 2008). With increasing global competition, rapid technological changes and changing business environments, Page 13

organisations need to be innovative to sustain their competitive advantage and thrive (Abstein & Spieth, 2014; Ganguly et al., 2019; Hitt et al., 1997). Innovations provide organisations with a source of competitive advantage, allowing them to differentiate themselves from competitors and attract customers (Wang et al., 2022). By developing and implementing creative ideas, organisations can meet customers' changing needs. They can contribute to developing new products and services that are environmentally friendly, address societal challenges and meet customer demand (Javed, Y. A. Rawwas, et al., 2018; Soewarno et al., 2019). Innovation within an organisation improves its internal functioning, creates solutions to problems or challenges, creates market share, increases profits, and creates value for the organisation (Hammond et al., 2011; Messmann & Mulder, 2012; West & Farr, 1990). Moreover, innovation can improve organisational performance (Ghasemi et al., 2020; Nawab et al., 2015; Panuwatwanich et al., 2008), as organisations with increased innovation capability tend to have better overall performance. In addition to the benefits for the organisation, the benefits to society include economic growth, improved products and services, and employment (Ahlstrom, 2010; Baumol & Strom, 2007; Stam & Wennberg, 2009; Wang, 2022).

Innovation has been studied in various disciplinary literatures (Drucker, 2015; Fagerberg & Verspagen, 2009; Kline & Rosenberg, 2010; Shafique, 2013). This includes economics, organisational studies, business and management, technology, science and engineering and marketing. Scholars from different disciplines provide or use different definitions. However, they all commonly reference the creation of something new or novel (Mancini & González, 2021; Taylor, 2017). Despite the overlap between various definitions, there is no single authoritative definition (Baregheh et al., 2009; Kooij, 2017; Quintane et al., 2011). The seminal work of Schumpeter (1934) on innovation and entrepreneurship provides one of the earliest definitions of innovation as the creation and implementation of new products, new methods of production, and opening up new markets or services. A similar definition was provided by Thompson (1965) as 'the generation, acceptance and implementation of new ideas, processes, products or services' (Baregheh et al., 2009, p. 1325). Damanpour and Evan (1984) and Zaltman et al. (1973, p. 656) built on these early definitions by arguing the concept of being 'new' to the adopting unit (being individual or organisation), not necessarily to the world, stating that innovation is 'any idea, practice, or material artefact perceived to be new by the relevant unit of adoption.' Even though a similar idea could be seen elsewhere or considered an imitation, as long as it is perceived as new to the people involved, it is considered innovation (Van de Ven, 1986).

In addition to the economic and management literature, the Oslo Manual (4th Edition, 2018), produced by the Organisation for Economic Cooperation and Development (OECD), has a definition used by national statistical agencies, government bodies, and industry and organisations. The manual provides guidelines for collecting, interpreting, and reporting innovation data and setting an appropriate benchmark. In earlier editions, the manual focused on the first two of Schumpeter's categories (product and process innovation), as these were easier to define and measure. Later editions of the Oslo manual expanded on the categories to include 'new markets', supporting Schumpeter's definition. The Fourth Edition defines innovation as '...a new or improved product or business process (or a combination thereof) that differs significantly from the firm's previous products or business processes, and that has been introduced on the market or brought into use by the firm' (OECD/Eurostat, 2018, p. 68). The manual emphasises the concept of 'significant', excluding minor changes or enhancements.

2.2.1. Innovation and Innovative Work Behaviour

The concept of innovation is broad and can be viewed at different levels: organisation, team and individual (Anderson et al., 2014; Camisón-Zornoza et al., 2004). Innovation and IWB are related concepts, but they differ in their scope and application within the context of work and organisations (Awan & Jehanzeb, 2022; Nguyen et al., 2020; Wu et al., 2011).

As discussed, innovation refers to creating and implementing new ideas, products, processes, or services that bring about significant positive change and provide value. It involves introducing something new or improving existing practices, enhancing efficiency, effectiveness, or competitiveness. Various factors, such as technological advancements, market demands, or organisational strategies, can drive innovation. It is often seen as a strategic imperative for organisations to stay relevant, differentiate themselves, and adapt to changing environments.

On the other hand, IWB focuses on the individual level and refers to the behaviours exhibited by employees that contribute to the generation, promotion, and implementation of innovative ideas and practices within the workplace (Janssen, 2000; Scott & Bruce, 1994). IWB encompasses a range of actions, including idea generation, problem-solving, knowledge sharing, collaboration, experimentation, risk-taking, and proactive behaviour (De Jong & Den Hartog, 2010; Janssen, 2000). It involves individuals going beyond their assigned roles and responsibilities to contribute to the creative process and drive innovation within their teams or organisations.

In summary, innovation is a broader concept that encompasses the overall process of creating and implementing new ideas or practices, bringing about organisational and societal change, while IWB zooms in on the specific behaviours exhibited by individuals that contribute to innovation and new ideas within the workplace to improve individual, group or organisational performance (Hirst et al., 2009; Li et al., 2020). The following section will discuss innovation at the individual level, which is characterised as Innovative Work Behaviour.

2.3. Innovative Work Behaviour

Fostering innovation within employees is crucial to improved organisational performance and sustained competitive advantage.

(Axtell et al., 2000; Oldham & Cummings, 1996; Ramamoorthy et al., 2005)

This study is conducted within the Australian workplace context and specifically focuses on innovation at the individual rather than the organisational level. As such, an operational definition of this concept is required for this thesis, which is IWB as defined by West and Farr (1990, p. 9) being '...the intentional introduction and application within a role, group or organisation, of ideas, processes, products or procedures, new to the relevant unit of adoption, designed to significantly benefit the individual, the group, the organisation or wider society'. IWB refers to intentional behaviour from an employee towards developing new or improved products, new markets, or improving business routines in their employing organisation. This infers that innovative efforts and resulting outcomes arise from activities undertaken by individuals. These activities involve exploring and proposing new ideas, testing ideas, building support for the idea, and ultimately implementing these ideas.
Organisations cannot be innovative without their employees, who are encouraged, motivated and willing to innovate (Agarwal, 2014; De Jong & Den Hartog, 2007; Janssen, 2000). As these employees are closer to the customer or the process, individual employees are the primary resource available to organisations to look for ideas and opportunities and will determine how to implement them (Agarwal, 2014; Carmeli & Spreitzer, 2009; De Jong & Den Hartog, 2007; Kozlowski & Klein, 2000). Carmeli and Spreitzer (2009, p. 171) argued that employees 'are often on the front line with customers and see opportunities for change and improvement in work processes and procedures that are invisible to managers or others formally responsible for innovation in the organisation'.

The concepts of employee creativity and innovation are often discussed interchangeably (Anderson et al., 2004; Hülsheger et al., 2009), and while they are related concepts, they have distinct meanings. Creativity refers to generating new and valuable ideas, concepts, or solutions that are original, novel and imaginative (Shai et al., 2009; Weisberg, 2015). It involves thinking in unconventional ways, making connections between seemingly unrelated concepts, and coming up with unique and imaginative ideas. Creativity is, therefore, the foundation of innovation.

On the other hand, innovation is the process of implementing and applying creative ideas to create new products, services, processes, or improvements in existing ones. It goes beyond generating ideas and involves successfully implementing and commercializing those ideas to deliver value. Innovation involves taking creative ideas and turning them into practical and tangible outcomes that bring about positive change.

Innovative behaviour includes generating and implementing ideas, whereas creativity focuses on creating new ideas but not necessarily producing or implementing them. 'Unlike creativity, IWB has a clearer applied component and is expected to produce some kind of innovative output or benefit' (Afsar et al., 2014, p. 1273). Creativity is about generating novel and valuable ideas, while innovation focuses on the practical application and implementation of those ideas to create tangible outcomes. Creativity is the starting point, and innovation is the process of bringing creative ideas to life. Therefore, while creativity is an essential function of the innovation process, usually occurring at the beginning, innovative behaviour builds on creativity and goes a step beyond to include implementation.

2.3.1. Definitions of Innovative Work Behaviour

For an organisation to realise innovations, the employee will generate new ideas by displaying behaviours and undertaking activities that include exploring new opportunities, looking for performance gaps, or implementing solutions to problems. Innovative work behaviour (IWB) has been conceptualised in the literature in various ways. Several authors adopted a behavioural perspective (Abstein & Spieth, 2014; Janssen, 2000; Prieto & Pérez-Santana, 2014), while others focused on outputs (Park et al., 2014; West & Farr, 1990; M. J. Xerri & Y. Brunetto, 2013).

A literature review found commonality in the definitions of IWB, which describe a multistage process of different phases of innovation. Table 2.1 provides a succinct presentation of select definitions for reference.

Author(s)	Definition of IWB
Kanter (1988)	'Innovative work behaviour is: (a) idea generation and
	activation of the drivers of the innovation; (b) coalition
	building and acquisition of the power necessary to move
	the idea into reality; (c) idea realisation and innovation
	production, turning the idea into a model – a product or
	plan or prototype that can be used; (d) transfer or
	diffusion, the spreading of the model – the
	commercialisation of the product, the adoption of the
	idea'
West and Farr (1990)	'The intentional introduction and application within a
	role, group or organisation of ideas, processes, products
	or procedures, new to the relevant unit of adoption,
	designed to significantly benefit the individual, the
	group, the organisation or wider society'
Scott and Bruce (1994)	'Innovation is viewed as a multistage process, with
	different activities and different individual behaviour
	necessary at each stage'

Table 2-1: Definitions of IWB

(Source: compiled by author)

Jansen (2000)	'We conceive innovative work behaviour in the
	workplace as complex behaviour consisting of a set of
	three different behavioural tasks: idea generation, idea
	promotion and idea realization'
Park, Song, Yoon and	'A multistage process, innovative behaviour is defined as
Kim (2003)	actions that search for, develop, and apply new ideas and
	solutions in the current situation'
De Jong and Den Hartog	'An individual's behaviour that aims to achieve the
(2010)	initiation and intentional introduction (within a work
	role, group, or organisation) of new and useful ideas,
	processes, products or procedures'
Xeric and Brunetto	'Innovative behaviour is a process that should improve
(2013)	the efficiency and effectiveness of problem-solving in
	the workplace'
Prieto and Perez-Santana	'Innovative work behaviour as a complex behaviour
(2014)	consisting of generation, introduction, or application of
	novel appropriate ideas, processes, and solutions'
Abstein and Speith	'Innovative behaviour at work as a multistage process
(2014)	comprising different behaviours that can be linked to
	distinct stages of the innovation process'

2.3.2. Phases of Innovative Work Behaviour

Innovative Work Behaviour typically involves a series of phases that individuals go through as they generate and implement new ideas. Table 2.2 below describes the various phases of innovation as adopted in key IWB literature. It shows that, for the most part, they describe the same process of activities being undertaken but with slight differences in the number of phases and in the terminology used.

Table 2-2: Phases of IWB

(Source: compiled by author)

	IWB Phase				
	Exploring ideas	Generating ideas	Gathering support for ideas	Implement	ting ideas
Kanter (1988)	Idea Generation and Activation		Coalition Building	Idea Realisation	Innovation Production
Scott and Bruce (1994)	Problem Recognition and Idea Generation		Sponsor seeking	Idea Pro	duction
Janssen (2000)	Idea Generation and Formative Investigation		Idea Promotion	Idea Rea	alisation
Kleyson & Street (2001)	Opportunity Exploration	Generativity	Championing	Applic	ation
De Jong & Den Hartog (2010)	Idea Exploration	Idea Generation	Idea Championing	ldea Imple	mentation

These studies describe a process by which employees generate new ideas, champion ideas within the organisation, and then develop ideas into valuable output. While the various phases include different activities and may include different behaviours, each phase builds on its predecessor and is highly interrelated and connected by feedback loops (Dorenbosch et al., 2005; Scott & Bruce, 1994).

This thesis adopted the four-phase model shown below in Figure 2.1 developed by De Jong and Den Hartog (2010) to explore the activities and behaviours demonstrated in each phase. The process starts with the employee actively seeking out opportunities to improve a product or process before they generate the idea. 'The start of an innovation process often has an element of chance: the discovery of an opportunity or some problem arising' (De Jong & Den Hartog, 2010, p. 24). The last phase is the transfer or diffusion of the innovation, referring to commercialising the idea (Kanter, 1988). The decision to utilise the four-phase model stems from recognising the crucial significance of acknowledging the preliminary exploration stage and openness to new idea opportunities. This stage plays a pivotal role in the innovation process, as it serves as a foundation for generating novel ideas. By embracing an open mindset and actively exploring new avenues, individuals broaden their perspectives and expose themselves to diverse sources of inspiration. This initial phase sets the stage for subsequent phases, facilitating the generation of innovative concepts. Further, these phases are not always linear or strictly sequential. They can overlap, and individuals may move back and forth between phases depending on factors such as feedback, organisational support, and changes in the external environment.

Figure 2-1: Four-phase model of Innovative Work Behaviour

(Source: adapted from De Jong and Den Hartog (2010))



Innovative Work Behaviour starts with *idea exploration (IE)*. This refers to employees looking for opportunities in their organisation to generate ideas. It may be the result of a problem that has arisen, inconsistencies or gaps, new emerging trends or an envisaged opportunity to improve effectiveness, often arising from questions and problems that occur during routine tasks (Bygballe & Ingemansson, 2014; Z. Wang et al., 2020). Basadur (2004) states that the exploration phase involves the employee observing, diagnosing, and gathering information and data to look for ways to improve current products, services, or work processes and to find a solution. Thurlings et al. (2015) explained that the employee will not only observe but listen to others and adapt ideas to evaluate their potential. Employees in the IE phase may engage in questioning, problemsolving, and seeking alternative approaches to address challenges (Bygballe & Ingemansson, 2014). Four behaviours are common in this phase: (1) paying attention to opportunity sources, (2) looking for opportunities to innovate, (3) recognising opportunities, and (4) gathering information about opportunities.

The next phase is *idea generation (IG)*. This refers to the intentional creation and development of new and creative ideas within a work role, work-group, or organisation (Wang et al., 2015). This phase aims to generate ideas and solutions based on the problem or opportunity. 'The key to idea generation appears to be the combination and reorganisation of information and existing concepts to solve problems or to improve performance' (De Jong & Den Hartog, 2010, p. 24). According to Kanter (1988), successful idea generation is when an individual looks for opportunities or problems from a different standpoint, referring to this as 'kaleidoscopic thinking', the process of rearranging and combining already known aspects into something new. Employees will brainstorm, explore, and gather information to develop novel ideas. Six behaviours are common in this phase: (1) generating ideas and solutions to opportunities; (2) generating

representations and categories of opportunities; (3) generating associations and combinations of ideas and information; (4) formulating ideas and solutions; (5) experimenting with ideas and solutions; and (6) evaluating ideas and solutions.

The next phase is *idea championing (IC)*. The employee seeks support for the generated idea and assembles a group of advocates to 'sell' the idea across the organisation (Kanter, 1988; Scott & Bruce, 1994). Employees take the generated ideas and promote them within their organisation or team. They present their ideas to colleagues, managers, or other stakeholders, highlighting the potential benefits and addressing concerns or questions (Perry-Smith & Mannucci, 2017). The idea needs to be advocated by the appropriate employees in an organisation who have the 'expertise, resources, contacts, influence and formal power to drive a successful implementation' (Moll & de Leede, 2016, p. 99).

This phase is crucial for gaining support and resources to move ideas forward and bring them to fruition. The need for coalition building is because ideas will signify a change for employees, teams, or the organisation. So, while the idea may address an issue or problem or improve performance, there may still be a reluctance to change. Kanter (1988) claimed that resistance to change often occurs for most ideas due to uncertainty about whether the benefit will exceed the cost of implementing the idea. Idea champions must navigate organisational politics and overcome resistance to change by addressing concerns and demonstrating the value and feasibility of the idea (Perry-Smith & Mannucci, 2017). Therefore, the idea must be promoted, and to do this successfully, the employee who generated the idea must go out and mobilise backers and get sponsorship to champion the idea to convince decision-makers to support the implementation of the idea.

Everyday activities include building coalitions, networking and forming alliances with others who can help advance the idea (Kleysen & Street, 2001). The employee needs to display confidence, enthusiasm, risk-taking, persistence, conviction, and energy to find and select the right people to be involved (Carmeli & Spreitzer, 2009; Howell et al., 2005). They use their influence and persuasive skills to gain support from key stakeholders and decision-makers within the organisation (Perry-Smith & Mannucci, 2017). There are four behaviours common to this phase: (1) mobilising resources, (2)

persuading and influencing, (3) pushing and negotiating, and (4) challenging and risk-taking.

The last phase is *idea implementation (II)*. Once an idea receives support and approval, the focus shifts to its implementation. This phase refers to the process of putting innovative ideas into action and bringing them to fruition (Khan et al., 2021), taking proactive action to execute and implement the idea (Parker et al., 2006). Ideas must be developed into valuable outputs or outcomes that can be used or implemented in regular, business-as-usual processes. The idea has been created, supported, and then approved for implementation, including developing, testing, or modifying the new process or product.

This phase involves planning, organising resources, coordinating with relevant stakeholders and executing the necessary steps to bring the idea to life (Déprez et al., 2021). It may include prototyping, testing, refining, and eventually integrating the innovation into existing processes or systems. Therefore, 'The production of a prototype or model of the innovation...that can be touched or experienced, that can now be diffused, mass-produced, turned to productive use, or institutionalised' (Kanter, 1988, p. 191). De Jong and Den Hartog (2010) stated that innovative employees must display a results-oriented attitude combined with considerable effort to realise the idea to overcome organisational constraints such as funding, scarce resources, resistance to change or bureaucracy.

Moreover, successfully implementing innovative ideas often requires collaboration and teamwork (Johari, Wahat, et al., 2021). Idea implementers may need to work closely with colleagues, departments, or external partners to ensure the smooth execution of the idea. Therefore, effective communication, coordination, and collaboration are essential behaviours in the idea implementation phase (Johari, Wahat, et al., 2021). There are three behaviours common to this phase: (1) implementing, (2) modifying, and (3) routinising.

2.3.3. Measurement of Innovative Work Behaviour

Several studies have proposed a method to measure IWB. Table 2.3 below provides a selection of key studies that developed a questionnaire to measure IWB. These have been developed as both one-dimensional studies, whereby IWB was measured as one phase overall (Basu & Green, 1997; Scott & Bruce, 1994) and multi-dimensional (De

Jong & Den Hartog, 2010; Dorenbosch et al., 2005; Janssen, 2000; Kleysen & Street, 2001; Krause, 2004), where they attempted to measure the individual phases within IWB. However, a number of the latter studies found a strong correlation between the phases, arguing that the 'distinctiveness of the four dimensions was, however weak, suggesting that IWB is one-dimensional' (De Jong & Den Hartog, 2010, p. 23).

'	Table 2-3: Studies that developed IWB Questionnaires
((Source: compiled by author)

Author(s)	IWB	Country	Type of
	Questions		organisation
Scott & Bruce (1994)	6 items	United States	Industrial
Spreitzer (1995)	4 items	United States	Industrial
Basu & Green (1997)	4 items	United States	Manufacturing
			Printing
Janssen (2000)	9 items	Netherlands	Food manufacturer
Kleysen & Street (2001)	14 items	United States	Various
Krause (2004)	8 items	Germany	Various
Dorenbosch, van Engen &	16 items	Netherlands	Local Government
Verhagen (2005)			
Reuvers et al. (2008)	4 items	Australia	Hospitals
De Jong & Den Hartog (2010)	10 items	Netherlands	Various service
			firms
Messmann & Mulder (2012)	24 items	Germany	Car manufacturer
Messmann & Mulder (2020)	8 items	Germany	Various domains

Taking the view that IWB is a one-dimensional construct, these studies have not acknowledged the differences in the activities and behaviours that may occur in each of the IWB phases. Wicaksono and Pusparini (2022) acknowledge the multi-dimensional nature of innovative behaviour but highlight the need for more substantial evidence supporting the multi-dimensional concept. It suggests that previous research has often unidimensionally measured and analysed IWB. However, a sound argument exists for taking a multi-dimensional approach to assessing IWB (Birdi et al., 2014; Duradoni & Fabio, 2019; Sadia Tabassam & Aman, 2018). While employees could be engaged in various phases simultaneously and similar behaviours could be exhibited in each phase, a deeper investigation of each phase separately warrants investigation.

2.3.4. Significance of Innovative Work Behaviour

IWB is of significant importance to organisations in various ways. IWB is crucial for organisational sustainability and effectiveness. Organisational leaders recognise the need for continuous innovation in products and internal processes to stay competitive in the market (Agarwal et al., 2012). IWB contributes to the development of new ideas, processes, products, or procedures that can enhance the quality, efficiency, and competitiveness of an organisation, enabling organisations to adapt to changing environments, meet customer needs, and stay ahead of the competition (Fadhil et al., 2022; Klaeijsen et al., 2017; Wang et al., 2019). Organisations can gain a competitive edge in the market by continuously generating innovative ideas and implementing them effectively.

IWB is also associated with improved organisational performance (Baety & Rojuaniah, 2022; Shanker et al., 2017; Yang & Wu, 2021). Studies have shown that organisations with a higher level of IWB tend to outperform their competitors (Dzimbiri & Molefi, 2022; Shanker et al., 2017). Wu et al. (2022) propose that this is due to the organisation's enhanced creativity, problem-solving and continuous improvements. Innovative behaviour involves thinking outside the box and finding novel solutions to problems (Gumusluoğlu & İlsev, 2009; G. Li et al., 2023). Organisations can address complex issues and navigate the evolving business landscape more effectively by fostering creativity and problem-solving to encourage employees to generate new ideas. This enables them to identify untapped opportunities and develop unique approaches to stay competitive. Organisations can continually adapt and optimise their operations by encouraging employees to experiment, learn from failures, and continuously refine their ideas. Černe et al. (2017) found that creating an environment that supports IWB can positively impact organisational performance, which is supported by Akbari et al. (2020), who suggested that leadership that fosters IWB is equally vital in ensuring organisational performance.

IWB also promotes a culture of learning and growth within organisations (Afsar & Afsar, 2019; Lin & Lee, 2017; Park et al., 2014). It encourages employees to think outside the box, take risks, and explore new possibilities while developing their skills and knowledge (Cangialosi et al., 2021; Harun et al., 2022). As employees acquire new skills and stay updated with industry trends, they can contribute to increased productivity and effectiveness within the organisation. Klaeijsen et al. (2017) argue that when employees display IWB, it fosters a sense of autonomy, empowerment, and ownership, as they are encouraged to contribute their ideas and make a meaningful impact on the organisation.

Moreover, IWB is positively related to employee engagement, job satisfaction, and organisational commitment, further contributing to organisational success (Meng et al., 2022; Osayande & Okolie, 2019). Employees who are allowed to engage in IWB feel valued, recognised, and empowered, enhancing employee motivation and job satisfaction (Shailja et al., 2023; Zhang & Liu, 2022). It gives them a sense of purpose and fulfilment in their work, as they can contribute their unique ideas and make a difference. Employees are more likely to be satisfied and committed when they have opportunities to contribute their ideas and be part of innovative initiatives, which can lead to higher retention rates. Further, IWB often requires giving employees the autonomy and empowerment to make decisions and take risks. When employees are trusted and empowered to contribute their ideas, they feel greater ownership and control over their work (Cheng et al., 2021; Javed, Abdullah, et al., 2018; Spiegelaere et al., 2014). This autonomy fosters a positive work environment and promotes engagement (Foss et al., 2009; Nahrgang et al., 2011; Spiegelaere et al., 2014).

2.3.5. Influences on Innovative Work Behaviour

Organisations that can innovate are more likely to retain competitive advantage and survive. The IWB literature has established that IWB contributes positively to organisational outcomes and has sought to understand which factors influence IWB through shaping individuals' attitudes, behaviours and knowledge (Beugelsdijk, 2008; Jimenez-Jimenez & Sanz-Valle, 2008; Laursen & Foss, 2003). Studies have been undertaken at both an organisational and individual levels (Afsar et al., 2014; Carmeli & Spreitzer, 2009; Odoardi et al., 2015; Ramamoorthy et al., 2005). For instance, research has shown that transformational leadership positively influences employees' IWB, with factors such as motivation to learn, task complexity, and innovation climate mediating this relationship (Afsar et al., 2014; Reuvers et al., 2008; Sadia Tabassam & Aman, 2018). Other research has examined the role of psychological empowerment, creative self-efficacy, and trust (Afsar & Badir, 2016; Agarwal, 2014; Godart et al., 2017; Singh & Sarkar, 2012) in fostering IWB.

Table 2.4 below provides a sample of studies that have explored IWB and sought to understand the various factors that may influence it. Most of these studies, as well as the studies that have created a measure for IWB, have essentially been quantitative studies in the context of manufacturing and industrial entities, predominantly within the United States and Europe.

Table 2-4: Studies of Innovative Work Behaviour

(Source: compiled by author)

Author(s)	Measures of IWB	Factors
Ramamoorthy et al. (2005)	9 questions from Janssen (2000)	Organisational processes, psychological contract, job autonomy
Carmeli and Spreitzer (2009)	6 questions from Scott & Bruce (1994)	Trust, connectivity and thriving
Afsar, Badir and Saeed (2014)	10 questions from De Jong & Den Hartog (2010)	Transformational leadership, psychological empowerment
Prieto and Pilar Pérez-Santana (2014)	5 questions from Scott & Bruce (1994) and Janssen (2000)	High involvement HR practices, management support and coworker support
Park et al. (2014)	6 questions from Scott & Bruce (1994)	Learning organisation, work engagement
Veenendaal and Bondarouk (2015)	6 questions from De Jong & Den Hartog (2010) and 6 items from Kleysen & Street (2001)	Supervisory support, Training & Development, information sharing, compensation

Odoardi et al. (2015)	9 questions from Janssen (2000)	Participative leadership, teamwork, information sharing, group processes, psychological empowerment
Taştan and Davoudi (2015)	10 questions from De Jong & Den Hartog (2010)	Leader-member exchange, Trust
Widmann and Mulder (2018)	28 questions from Messmann & Mulder (2012)	Team Learning
Stoffers, Van der Heijden and Jacobs (2019)	9 questions from Janssen (2000)	Employability

2.3.6. Innovative Work Behaviour in the Australian context

According to the Global Innovation Index 2022, which assesses countries' innovation performance worldwide, Australia ranked 25th out of 132 economies overall and 7th among the 17 economies in South East Asia, East Asia and Oceania (Dutta et al., 2022). Switzerland is the most innovative economy in the world. Making up the top 10, this is followed by the United States, Sweden, the United Kingdom, Netherlands, the Republic of Korea, Singapore, Germany, Finland, and Denmark. The index evaluates 80 indicators grouped in innovation inputs and outputs, including institutions, human capital and research, infrastructure, market sophistication, business sophistication, knowledge and technology outputs, and creative outputs. According to the index, Australia's innovation strengths include regulatory quality, tertiary enrolments, school life expectancy, new businesses, and university rankings. Australia's innovation weaknesses include entrepreneurship policies and culture, government funding for secondary pupils, the number of graduates in science and engineering, and ICT services exports as a proportion of total trade.

Patents can also measure technological innovation. According to IP Australia, 2,996 Australian applicants filed patents in 2021, representing an increase of 25% (IPAustralia, 2022). The five leading classes for standard patent filings were

Pharmaceuticals, Medical Technology, Biotechnology, Organic fine chemistry, and Computer technologies. In particular, audio-visual technology patents grew by 85%.

Startups and Entrepreneurship are another good indicator of innovation performance. Startups are critical in bringing new products and services to the market. Australia has a thriving startup ecosystem with a growing number of innovative ventures. Startup Muster is the largest survey on the Australian startup ecosystem. According to Startup Muster's 2018 Annual Report, there were an estimated 1,465 startups in 2018, covering a wide range of industries and representing a significant increase from 954 in 2015 (Hurps et al., 2018). The report also highlighted sectors such as artificial intelligence, fintech, and education as significant industries in startup activity.

Numerous studies have been conducted on IWB and innovativeness in Australia. These studies contribute to the global body of knowledge on IWB while also providing insights specific to the Australian context. Below are seven exemplar studies using data from Australian employers and employees.

Reuvers et al. (2008) examined the relationship between transformational leadership and employee IWB, specifically focusing on the moderating effect of gender. The researchers collected data from four Australian hospitals, resulting in a sample of 335 respondents. The study's findings indicate a positive and significant relationship between transformational leadership and IWB. However, the study also found that the manager's gender moderates this relationship. Specifically, employees reported more innovative behaviour when male managers displayed transformational leadership compared to female managers. This finding supports the hypothesis of gender bias in the relationship between transformational leadership and IWB. However, the study did not find a significant effect for the three-way interaction of transformational leadership, the gender of the manager, and the gender of the employee. This suggests that the employee's gender does not significantly influence the relationship between transformational leadership and IWB.

Nsenduluka and Shee (2012) explored the factors influencing work group service innovativeness. They argue that the context in which a group operates plays a significant role in determining its effectiveness and innovation. The study examines the individual and combined effects of various organisational-level and group-level variables on workgroup service innovativeness. The study collected data from four hotels in Melbourne, Australia. The results indicate that organisational climate, task design, group citizenship behaviour, group self-efficacy and group climate for innovation are significant predictors of work group service innovativeness. However, market orientation was not found to be a significant predictor. Their findings were consistent with previous research on service innovation performance.

Brunetto et al. (2020) compared the impact of personal and organisational support on the innovative behaviour of frontline healthcare workers in Australia and the United States. The study found that both personal support, such as psychological capital (PsyCap), and organisational support, such as manager-subordinate relationships, significantly influenced the innovative behaviour of frontline healthcare employees. The study also highlighted the importance of reciprocal social exchanges in facilitating the growth of healthcare employees' PsyCap, contributing to their resilience and ability to innovate in the workplace.

Xerri and Reid (2018) investigated the relationship between human resource management factors (satisfaction with teamwork and satisfaction with training opportunities) and the innovative behaviour of nurses in Australian public and private hospitals. The study proposed that fostering innovative behaviour is a way for hospitals to improve efficiency and effectiveness in the face of nurse shortages and limited budgets. The findings suggested that employees' perception of wellbeing mediated the relationship between human resource management factors and innovative behaviour.

Xerri and Brunetto (2018) focused on the importance of employee commitment and organisational citizenship behaviour (OCB) in fostering innovative behaviour. The researchers examined the relationship between nurses' organisational commitment, OCB, and innovative behaviour in the workplace with data collected from 210 nursing employees in Australia. The findings revealed positive and significant paths from affective commitment to innovative behaviour and from OCB with an individual focus on innovative behaviour.

Luksyte et al. (2017) examined gender differences in perceptions and evaluations of IWB. Data was collected through three complementary field and experimental studies, with studies 2 and 3 collected in Australia. Study 3 included undergraduate students from a business school in one of the largest universities in Australia. Luksyte et al. (2017) argue that Australia has one of the world's largest proportions of innovative workers. The research found that IWBs are stereotypically associated with men more than women. Furthermore, favourable performance evaluations are associated with IWBs for men but not for women. Women who engage in IWBs may not receive better performance evaluations than those who do not innovate, whereas engaging in IWBs benefits men. The study highlights the presence of sex bias in perceptions and evaluations of IWB.

Newman et al. (2017) examined the influence of servant and entrepreneurial leadership on employees' work outcomes in social enterprises. The research analyses data from employees and social entrepreneurs in the social enterprise sector in Australia, the UK, and Canada. The findings indicate that servant leadership positively relates to followers' organisational commitment, while entrepreneurial leadership positively relates to innovative behaviour.

The literature on IWB in Australia has identified a lack of studies in this specific context of remote working. Further research is needed to fill these gaps and enhance our understanding of IWB in Australia.

2.4. Remote working

2.4.1. Definitions of remote working

'...a work practice that involves members of an organisation substituting a portion of their typical work hours (ranging from a few hours per week to nearly full-time) to work away from a central workplace – typically principally from home – using technology to interact with others as needed to conduct work tasks.'

(Allen et al., 2015, p. 44)

The term 'remote working' is used interchangeably with telecommuting, teleworking, working remotely, and working from home. According to Parris (2017), all five terms are comparatively synonymous. The term 'telecommuting' was introduced by Jack Nilles in 1973 (Nilles, 1994), an engineer working for the National Aeronautics and Space Administration. He proposed to move the work to workers rather than move workers to the work, aiming to reduce energy consumption and traffic congestion (Allen

et al., 2015). Organisations now prefer to use the terms 'remote working' or 'working remotely' as they are seen as more modern than telecommuting (Parris, 2017). For this reason, this study will use the terms 'remote work', 'remote working', and 'working remotely' and will be based on literature that uses those three terms in addition to teleworking and telecommuting as their choice in terminology.

Remote working allows employees to work partly from home or elsewhere outside the official office building. At the same time, the extensive use of ICT applications enables interaction and coordination with colleagues inside and outside the organisation (Gajendran & Harrison, 2007). Although remote working is independent of an employee's work arrangement (part-time or full-time), part-time employees tend to work remotely more frequently than full-time employees (Bailey & Kurland, 2002; Gajendran & Harrison, 2007; Sener & Bhat, 2011).

Remote working practices are predominantly suited to white-collar and knowledge workers, i.e. employees whose work does not require extensive physical efforts such as managerial, clerical and administration (Bailey & Kurland, 2002; Eraso & Erro-Garcés, 2020; Tramontano et al., 2021; Vandelanotte et al., 2013). Their work involves tasks that can be performed digitally on a laptop or mobile device, allowing for freedom in how the work will be done provided the goal or task outcome is achieved. The rise of digitalisation and non-physical, knowledge-based work has decoupled the place of work from the location of the employer, enabling more remote work and telecommuting. The COVID-19 pandemic has further highlighted the importance and possibility of working from home, especially for white-collar employees (Degerli, 2023; Risi & Pronzato, 2021). Employees no longer need to conduct their work in the main office building; likewise, many tasks and processes no longer must be done at specific times.

2.4.2. Outcomes of remote working

Remote working has become increasingly prevalent, especially in the context of the COVID-19 pandemic (Adekoya et al., 2022; Galanti et al., 2021; Shimura et al., 2021). This work arrangement allows employees to perform their job duties outside of the traditional office setting. This section will provide an overview of the key outcomes associated with remote working, drawing upon relevant research studies.

One key outcome of remote working is the potential for improved work-life balance. Research has shown that remote work can provide employees greater flexibility in managing their personal and professional responsibilities (Anderson & Kelliher, 2020; Chafi et al., 2021). This flexibility allows individuals to integrate their work and personal lives better, leading to increased satisfaction and reduced stress levels (B. Wang et al., 2021). Additionally, remote work can offer employees increased autonomy in managing their tasks and schedules, which has been associated with positive individual outcomes such as job satisfaction and task performance (B. Wang et al., 2021). With greater autonomy and control over their work, employees can manage their tasks and schedules more independently (Anderson & Kelliher, 2020; Chafi et al., 2021). Higher levels of job satisfaction are potentially due to increased flexibility, reduced commuting time, and improved work-life balance (Anderson & Kelliher, 2020).

However, remote working has its challenges. Communication and collaboration can be more difficult in a remote work environment, as employees may face barriers to effective communication and collaboration and maintaining social connections (Chafi et al., 2021; B. Wang et al., 2021). Social isolation from colleagues and the workplace can also result from remote work, particularly during periods of enforced remote work such as the COVID-19 lockdowns (Toscano & Zappala, 2020). While it can provide increased flexibility and reduce the stress associated with commuting, isolation resulting from remote work can lead to feelings of loneliness, lost comradery and decreased employee wellbeing (Chafi et al., 2021; Toscano & Zappala, 2020; B. Wang et al., 2021). Managing the work-family interface can also be more complex for remote workers, as work can become embedded in the family domain, leading to difficulties in disengaging from work and increased work-family conflict (Eddleston & Mulki, 2015).

Another important outcome of remote working is its potential impact on productivity. While remote work can provide employees with fewer distractions, a quieter workplace and more focused work time to focus better on their tasks, thereby increasing productivity (Adekoya et al., 2022; Allen et al., 2015; Chafi et al., 2021), it can also present challenges in terms of maintaining productivity and managing workloads (Shimura et al., 2021). Factors such as job stressors, the surrounding environment, and personal circumstances can influence how remote work affects individuals' productivity (Shimura et al., 2021). Distractions during remote work, such as family disruptions and

caring responsibilities, can have a negative impact on productivity (Ali et al., 2022; Stanisçuaski et al., 2021). Otonkorpi-Lehtoranta et al. (2021) found that remote work can increase work-family conflict, particularly for women and parents, due to the blurred lines between work and home.

Remote work can blur the boundaries between work and personal life, making it harder for employees to switch off from work (Charalampous et al., 2018; Rauch & Ansari, 2022). The use of remote technologies, particularly the use of smartphones, and being accessible and continuously contactable can dissolve the separation between work and home, leading to an expansion of work scope and duration (Rauch & Ansari, 2022). Blurred work-life boundaries can result in work intensification, increased work-related stress, and a negative impact on work-life balance (Palumbo et al., 2020; Sandoval-Reyes et al., 2021). Sandoval-Reyes et al. (2021) argue that remote work leads to overworking, with employees extending their working hours and experiencing difficulties in achieving work-life balance, while Chan et al. (2023) argue that the intensive use and reliance on information and communication technology can contribute to emotional, psychological, and physical strains.

The significance of remote working has been widely recognised, particularly in the context of the COVID-19 pandemic and the future of workplaces. Remote work offers both benefits and challenges that impact various aspects of work and employees.

2.4.3. Remote working in Australia

According to the Australian Bureau of Statistics (ABS), more than 75% of large companies in Australia and about one-third of micro businesses have the facility for staff to work remotely (Houghton et al., 2018). As far back as 2013, the Australian Communication and Media Authority found that approximately 51% of the Australian workforce, or 5.6 million adult Australians, were utilising the internet to work away from the office outside 'standard working hours' or substituting coming into the office for part or all of the day (Houghton et al., 2018). The Families in Australia Survey: Towards COVID Normal found that 67% of survey respondents sometimes or always worked from home during COVID-19, compared to 42% pre-COVID (Baxter; & Warren, 2021). In the second year of the pandemic, the Australian Bureau of Statistics stated that 41% of employed people regularly worked from home. This increased by 32% from the previous

year (ABS, 2021). Furthermore, the COVID-19 pandemic led to an increased interest in working from home (Guaralda et al., 2020) and found many employees relocating to regional towns in Australia enabled by the increased flexibility (Gurran et al., 2021; P. Li et al., 2023).

While this suggests that remote working is a common practice in some sectors of the Australian workforce, Beck et al. (2020) argue there was a low incidence of working from home in Australia prior to COVID-19, with the number of people who worked from home regularly since 2001 only 4.6%, and only for an average of 11 hours per week. The decision to prevent remote work was often a managerial decision rather than a function of the type of work (Hopkins & McKay, 2019), as well as low managerial confidence in managing employees remotely (Parker et al., 2020).

Remote work during the COVID-19 pandemic has paved the way for the transition to hybrid work post-pandemic, where employees spend their working time both at the office and remotely, usually at home. As organisations adapted to remote work to ensure business continuity and employee safety, they gained valuable insights into the feasibility and benefits of flexible work arrangements. According to the Australian Bureau of Statistics, data covering 7,000 organisations and a recent survey of 1,500 business leaders and employees, pre-COVID-19, 19% of employees were hybrid working, expected to rise to 35% post-COVID-19 (Telstra, 2021). This shift towards hybrid work is driven by the desire to balance the productivity and collaboration benefits of in-person interactions and the flexibility and autonomy of remote work (PwC, 2022).

Many organisations in Australia, such as ANZ and Atlassian, are supporting hybrid work models (Rauv, 2022). A 2020 Gartner survey found that 82% of company leaders plan to support employees working remotely some of the time (Baker, 2020). As technological advancements continue to reshape the modern workplace, employers and employees recognise the benefits and flexibility of remote and hybrid work arrangements. Consequently, remote and hybrid work is poised to become more prevalent in Australia in the coming years.

2.5. Innovative Work behaviour and remote working

Many factors may influence the relationship between remote working and innovative behaviour. On the one hand, remote working allows for flexibility and Page 35

autonomy (Gajendran & Harrison, 2007; B. Wang et al., 2021; Wood et al., 2018), which can enhance creativity and innovation (Beugelsdijk, 2008; Liu et al., 2011; Rachman et al., 2022; Slåtten et al., 2020). Remote working allows employees to have a greater work-life balance, which can lead to increased job satisfaction and engagement (Gillet et al., 2021). Additionally, remote working can provide employees with more time for reflection and independent thinking, which can stimulate creative ideas (Pfund et al., 2020). On the other hand, remote working can also adversely affect innovative behaviour and creativity. The lack of face-to-face interaction and spontaneous collaboration that comes with remote working can hinder the exchange of ideas and the generation of new insights (Katrahmani et al., 2022). Remote working can also lead to feelings of social isolation, negatively impacting creativity (Costa et al., 2022). Furthermore, the apparent increase in autonomy that comes with remote working can sometimes lead to lower work engagement and decreased motivation (Costantini & Rubini, 2021).

This section examines existing literature on remote working and its impact on IWB. It will encompass research on various aspects, such as the effects of remote work on employees' ability, motivation, and opportunity to engage in innovative behaviours. By synthesising and analysing the existing literature, this section will establish a foundation for understanding the current state of knowledge on the topic and identify gaps that this research aims to address.

2.5.1. Autonomy and flexibility

Remote work often gives employees greater autonomy over their work schedules and environments (Chafi et al., 2021; Neidlinger et al., 2022). This freedom can lead to increased creativity and innovative thinking as individuals have the space to explore new ideas and approaches without constant supervision and may also feel more empowered to think creatively (Gajendran & Harrison, 2007; Maitlo et al., 2022; Shafique, 2013). It also allows individuals to work in their preferred environment, minimising distractions and promoting focus (Kowalski & Ślebarska, 2022; Lee & Brand, 2010). Employees may find it easier to focus and concentrate on their work when not in a bustling office environment (Charalampous et al., 2018; S. Leroy et al., 2021; van der Meulen et al., 2012).

Autonomy also enables flexibility for the employee (Aziz-Ur-Rehman & Siddiqui, 2019; Kim & Cho, 2020) and improved work-life balance (Felstead & Henseke,

2017; Neidlinger et al., 2022) as they can strike a better balance between work and personal life and have more control over their work hours. Remote work allows employees to balance their professional and personal lives more effectively. This flexibility can reduce stress and enhance overall job satisfaction, which can be conducive to creativity and innovation.

2.5.2. Work-life balance and isolation

Striking a healthy work-life balance can contribute to employees' overall wellbeing and job satisfaction (Haider et al., 2018; Mansor et al., 2022), which are essential for job performance and, therefore, maintaining creativity and preventing burnout. Employees with time for relaxation or personal pursuits are more likely to approach work with a refreshed and creative mindset (Dul & Ceylan, 2014). Both Beaty et al. (2018) and Fink et al. (2008) suggest that taking breaks, relaxing, and allowing the brain to rest can enhance creative capabilities and generate innovative solutions.

However, remote working can lead to overworking and less work-life balance (Hayes et al., 2021; Toscano & Zappala, 2020). Palumbo et al. (2020) found that remote work can increase work-related fatigue, leading to physical and emotional exhaustion for home-based workers. Similarly, Felstead and Henseke (2017) noted that remote work comes at the cost of work intensification and a greater inability to switch off. Remote workers may experience difficulties in maintaining a healthy work-life balance due to the blurred boundaries between work and personal life (Hayes et al., 2021; Irawanto et al., 2021). Golden et al. (2006) found that remote working can blur the boundaries between work and family, leading to increased work-family conflict. Such conflicts may decrease individuals' capacity to engage in innovative behaviour due to competing demands and distractions from the home environment (Utama & Soetjipto, 2022; X. Wang et al., 2021).

Creativity and innovative behaviours are enhanced when work-home conflict and social isolation are minimised (Costa et al., 2022; Delanoeije et al., 2019; Johnson et al., 2016). However, social isolation can be a significant issue when working remotely (Savolainen et al., 2021; Toscano & Zappala, 2020; B. Wang et al., 2021). Remote work often means working alone, without the physical presence of colleagues. This lack of face-to-face interaction can lead to feelings of isolation and loneliness, as humans are social creatures and need social interactions to thrive.

2.5.3. Communication

Remote work heavily relies on digital communication tools such as emails, chat apps, and video conferences (Jarvenpaa & Leidner, 2006; Neumann & Bogdanov, 2022; C.-L. Yang et al., 2022). Digital communication tools are pivotal in facilitating employee collaboration and interaction, fostering IWB across diverse locations (Kingma, 2019; Kohler et al., 2011; Yokoi et al., 2021). Virtual brainstorming sessions, online collaboration platforms, and video conferences facilitate the exchange of ideas, fostering innovation. While these tools facilitate work-related communication, they may not always provide the same level of social connection as in-person conversations. Nonverbal cues and casual interactions are often missed, making the work environment feel more isolated.

It is both the availability and effectiveness of communication and collaboration tools that play a crucial role in remote work (Chantziaras et al., 2021; C.-L. Yang et al., 2022). Teams need reliable platforms for video conferencing, instant messaging, file sharing, and project management to facilitate seamless communication, working relationships, sharing ideas and feedback, and collaboration.

With ineffective communication and team communication tools, remote workers can face social isolation and have difficulties in maintaining social relationships and forming connections with their remote colleagues while also affecting workers' perceptions of remote colleagues, particularly with colleagues who may have limited interaction (Grant et al., 2013; C.-L. Yang et al., 2022). Misinterpretation of messages, lack of non-verbal cues, and asynchronous communication can hinder effective information sharing, collaboration, and brainstorming, which are crucial for fostering innovative behaviour. Allen et al. (2015) pointed out that gestures and emotions are difficult to transmit through email, for example, which can affect the quality of communication in remote work settings. Delayed feedback and response times can also slow down the iterative idea development and refinement process, potentially hindering innovation. Therefore, in settings with reduced collaboration and decreased communication, innovative behaviour is challenged by limiting the exchange of ideas and impeding collective creativity (Allen et al., 2015).

Conversely, communication and collaboration can be improved in virtual environments, where employees can share their ideas and present their work from different places while being virtually located (Neumann & Bogdanov, 2022; Öztürk et al., 2021). These communication systems provide a highly cost-effective method of facilitating multidisciplinary interactions between remotely located employees, enabling remote collaboration (Dulai et al., 2020). Further, in remote work settings, establishing creative group techniques, such as remote brainstorming, can encourage creative thinking, promote psychological safety within teams and enable routine innovation (Chen et al., 2020; Domingo et al., 2021; Paulus & Brown, 2007).

2.5.4. Organisational culture, support and trust

There is a well-established relationship between organisational culture and innovation (Baer & Frese, 2002; Imran et al., 2010; Uzkurt et al., 2013). An organisation's existing culture can either support or hinder innovative behaviour in a remote work setting. Organisational cultures that promote innovation and value experimentation, creative thinking, risk-taking, and openness to new ideas are more likely to foster innovative behaviours (He et al., 2019; Katrahmani et al., 2022; Litchfield et al., 2014). In addition, organisations that foster a culture of trust, psychological safety, and open communication are likely to have more IWB among employees (Agarwal, 2014; Akhtar et al., 2019; Carmeli & Spreitzer, 2009; Godart et al., 2017; Kmieciak, 2020; Xu & Suntrayuth, 2022).

Trust is a crucial element in remote work, as it impacts various aspects of innovation. Where face-to-face communication is limited, building trust may be challenging. It has been noted that trust may be lower in virtual teams than in co-located teams and more challenging to achieve (Ford et al., 2020; Jensen et al., 2020; Soomar, 2020). Nevertheless, remote interactions between employees require trust for successful communication and effective work (Kähkönen, 2023; Krzyżowska, 2022; Tramontano et al., 2021; B. Wang et al., 2021). Remote work can impact employees' trust in their organisation, supervisors, and colleagues (Juchnowicz & Kinowska, 2021; Stavrova et al., 2023; Zheng et al., 2023). Likewise, remote work has implications for leaders' trust in their employees (Stavrova et al., 2023; Zheng et al., 2023; Zheng et al., 2023; Zheng of virtual teams and for supervisors to trust remote workers to remain productive (Stavrova et al., 2023). The importance of trust in fostering

cooperation and knowledge sharing within a team is well-supported by various studies. Trust has been found to positively impact team effectiveness, communication, information sharing, and collaboration, all of which are essential for fostering innovation (Breuer et al., 2016; Bulińska-Stangrecka & Bagieńska, 2019; Lee et al., 2015). Furthermore, the lack of trust can lead to a decrease in cooperation, defensive behaviour, and a reduction in team cohesion, ultimately hindering knowledge sharing and innovation (Buvik & Tvedt, 2016; Cheung et al., 2016; Chiu & Chiang, 2019; Yousaf, 2018). Moreover, the literature emphasises the role of trust in virtual teams, where it is positively related to team effectiveness criteria, including team performance and project success, further underlining its significance in promoting cooperation and knowledge sharing in innovative team settings (Bierly et al., 2009; Breuer et al., 2016; Dorairaj et al., 2012; Lukić & Vračar, 2018)

Psychological safety refers to the shared belief within a team or organisation that it is safe to take interpersonal risks, such as speaking up, asking questions, and expressing ideas or concerns without fear of negative consequences or judgment (Edmondson & Lei, 2014). When employees feel psychologically safe, they are more likely to take risks, share ideas, and engage in innovative behaviours (Nguyen, 2021; Tkalich et al., 2023; Xu & Suntrayuth, 2022).

Another factor in the influence of remote working on IWB is organisational support in the form of encouragement, incentives, and recognition for innovative initiatives, which can influence employee engagement and job satisfaction as well as the employees' motivation to innovate while working remotely (Afsar & Afsar, 2019; Afsar & Umrani, 2019; Singh et al., 2020; Ude & Diala, 2015). Afsar and Umrani (2019) suggest that when leaders recognise and reward innovative behaviour, it creates a climate that encourages employees to engage in further innovative behaviours. Anindita (2023) and Bakker and Demerouti (2008) argue that employee engagement is crucial in fostering creativity, leading to IWB. However, the latter study found that the relationship between employee engagement, creativity, and IWB holds true regardless of whether employees work remotely or in a traditional office setting.

Studies have also shown that employees who have the option to work remotely, provided through organisational support for such practices, tend to have higher job satisfaction (Irawanto et al., 2021; Jamal et al., 2023). Employees with higher job

satisfaction are more likely to be engaged and motivated, which can positively impact their IWB (Mustafa et al., 2021; Tang et al., 2019). Research by Niu (2014) suggests that employees with high job satisfaction are more likely to have greater job embeddedness, strengthening their motivation to generate, spread, and implement innovative ideas within their organisations. Additionally, a study by Park et al. (2015) found that higher levels of job satisfaction are associated with a positive attitude toward the job and increased commitment to the organisation, both of which can contribute to a more innovative work environment (Nikpour, 2018; Wahyuni et al., 2021; M. Xerri & Y. Brunetto, 2013).

2.5.5. Collaboration

The cohesion and collaboration within remote teams play a crucial role in fostering innovative behaviours (Akram et al., 2020; Mehmood et al., 2022; Rachman et al., 2022). Teams that communicate effectively, trust each other, and collaborate seamlessly are more likely to exhibit innovative behaviour (Gundry et al., 2015; Litchfield et al., 2017). Collaboration fosters the exchange of diverse perspectives, knowledge sharing, and collective problem-solving, which are conducive to innovative thinking and behaviour (Majchrzak et al., 2012; Sun et al., 2020). However, collaboration can be challenging when working remotely due to team members' physical separation and communication technologies' limitations. For instance, Hinds and Bailey (2003) emphasised that technology mediation threatens a team's ability to share information, which is crucial for collaborative work. B. Wang et al. (2021) emphasised the limitations of ICT-mediated communication and the potential hindrance relative to face-to-face interaction. Further, most systems used by remote users to collaborate have limited ability to gesture or adapt interaction and awareness features, which can hinder effective communication and collaboration (Chénéchal et al., 2019; Fussell et al., 2004)

2.5.6. Physical proximity

Physical proximity, reduced face-to-face interaction, and difficulties in coordinating activities brought about by working remotely can hinder collaborative processes, which in turn may hinder IWBs (Driskell et al., 2003; Hertel et al., 2005; B. Wang et al., 2021). Ceci et al. (2021) suggested the importance of face-to-face interaction, particularly in the early stages of innovation, which implies that a lack of face-to-face interaction may hinder the innovation process.

Strong social connections are also known to facilitate knowledge sharing, trust, and collaborative creativity (Huo & Photchanachan, 2021; Nahapiet & Ghoshal, 1998), which are essential for innovative behaviour. However, the absence of face-to-face interactions and physical proximity in remote work setups can lead to reduced social cohesion, employee isolation and weaker social ties among team members, affecting team dynamics and collaboration (Handke et al., 2020; Tkalich et al., 2023). For instance, Tkalich et al. (2023) highlighted that remote workers tend to have significantly reduced team cohesion, poor awareness of 'who did what' and 'who knows what', divergent viewpoints, conflicts, and team coordination problems.

Maintaining good team dynamics and building shared understandings, such as encouraging virtual team-building activities and regular social interactions, can help maintain a sense of camaraderie and stimulate collaboration and innovative thinking (Okpara et al., 2022; Treacy, 2022). Treacy (2022) further emphasised the need for organisations to have strategies to maintain an innovative culture in a remote working setup.

Traditional beliefs may suggest that physical proximity is crucial for innovative behaviour. For instance, Tkalich et al. (2023) highlight the limitations of remote working, such as a limited ability to brainstorm. Post-COVID-19 pandemic, many organisations are looking for their employees to return to the office, citing reduced collaboration and innovation, among other reasons. Remote working can reduce unplanned and spontaneous interactions among team members (Ford et al., 2020; Gajendran & Harrison, 2007), which are often catalysts for generating new ideas and fostering innovation. These impromptu discussions and informal collaborations are more common in traditional office environments. In a physical office environment, colleagues can easily exchange ideas, share knowledge, and engage in impromptu brainstorming sessions. Remote work limits the occurrence of these unplanned encounters, reducing the potential for unexpected insights and creative connections. There is also the reduced opportunity for informal knowledge sharing and learning. However, some research provides competing evidence for this claim. Thoring et al. (2018) highlight that virtual spaces can facilitate creative activities such as brainstorming, challenging the notion that physical presence is essential for effective brainstorming. Additionally, Gidel et al. (2020) present a collaborative interaction model and behaviour rules that could enhance brainstorming results, suggesting that the effectiveness of brainstorming may not be solely dependent on physical proximity.

The following section will delve into the development of the hypotheses based on further literature review. Building upon the foundation laid by the initial literature review, this examination of relevant studies enhances our understanding of the research area.

2.6. Hypothesis development

Drawing upon prior theoretical examinations, this thesis posits the following hypotheses concerning IWB, as influenced by the extent of remote work engagement and various demographic characteristics among employees participating in remote work arrangements within the context of Australia. The overall hypothesis development for this thesis is summarised in Figure 2.2, and a table of all hypotheses is shown in Table 2.5. The literature to support the hypotheses development is explained in Sections 2.6.1 to 2.6.6.





Table 2-5: Hypothesis summary

Code	Description
H1	RWF has a significant effect on IWB
H2	Age has a significant effect on IWB when working remotely
H3	Gender has a significant effect on IWB when working remotely
H4	Tenure has a significant effect on IWB when working remotely
H5	Employment Status (ES) has a significant effect on IWB when working remotely
H6	Employment Level (EL) has a significant effect on IWB when working remotely
H7	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of IWB occurring
H8	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Exploration occurring
H9	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Generation occurring
H10	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Championing occurring
H11	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Implementation occurring

2.6.1. Remote working frequency and IWB

The impact of remote work on IWB is shaped by a complex interplay of factors encompassing individual, organisational, and contextual dimensions. However, there remains a dearth of studies that specifically explore the intersection of remote working and IWB. Different outcomes of remote working, such as communication, technology, and culture, can influence IWB. Some of these factors may not directly influence innovative behaviours but rather have some impact indirectly.

Research discussed in previous sections has shown that leadership style is essential in promoting IWB in remote work contexts (Afsar & Afsar, 2019; Coun et al.,

2021; Khan et al., 2020). Factors such as work independence and autonomy (Ashlan & Bahri, 2023; Battistelli et al., 2013; Boskovic, 2021; Wicaksono & Pusparini, 2022), clarity of job criteria (Hudson et al., 2017; Malhotra et al., 2016), interpersonal trust (Carmeli & Spreitzer, 2009; Clegg et al., 2002; Parker et al., 2020; Rune Ellemose, 2016), and social isolation can also impact employees' adjustment to remote work and, consequently, their IWB (Garlatti Costa et al., 2022; Golden et al., 2008; Hu & Subramony, 2022; Toscano & Zappala, 2020; Zoonen et al., 2021).

Additionally, remote work can present challenges in maintaining efficiency, teamwork (De Leede & Nijland, 2016; Johari, Abdul Wahat, et al., 2021; Voordt, 2004), and use of and quality of communication technology, which can affect IWB (Assarlind et al., 2013; Gorokhova et al., 2021; Jarvenpaa & Leidner, 2006; Nguyen et al., 2020).

Based on the literature, the following hypotheses were developed:

H1: Remote working frequency (RWF) has a significant effect on Innovative Work Behaviour (IWB).

H7: RWF, Age, Gender, Tenure, ES, and EL have a significant effect on the odds of Innovative Work Behaviour (IWB) occurring.

H8: RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Exploration occurring.

H9: RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Generation occurring.

H10: RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Championing occurring.

H11: RWF, Age, Gender, Tenure, ES, and EL have a significant effect on the odds of Idea Implementation occurring.

2.6.2. Age of remote working employees and IWB

Age can significantly influence IWB within organisational contexts. Younger employees often bring fresh perspectives and a natural affinity for technology, enabling them to adapt quickly to new tools and platforms (Kim et al., 2011). Younger employees,

referred to as digital natives, exhibit higher creative behaviours than older employees, referred to as digital immigrants (Buhl et al., 2016). The authors suggest that digital natives' familiarity with technology and their comfort in using digital tools contribute to their higher levels of innovative behaviour. Kim et al. (2011) found that younger employees' enthusiasm and willingness to take risks can drive creative thinking and experimentation, leading to innovative solutions. Fasbender et al. (2021) suggest that emotional intelligence positively relates to creativity and innovation. Their study found that younger employees, who often exhibit higher levels of emotional intelligence, may be more inclined to engage in creative thinking and experimentation, leading to innovative solutions.

Moreover, younger workers tend to be more open to change and may challenge traditional methods, fostering a culture of innovation. However, with their extensive experience and deep industry knowledge, older employees bring valuable insights to the table (Kim et al., 2011; Meyer, 2008). Their years of expertise often translate into a nuanced understanding of market trends and customer needs, allowing them to develop innovative products and services tailored to specific demands. Additionally, older workers tend to excel in problem-solving and decision-making, crucial skills in the innovation process (Guillén & Kunze, 2019; Meyer, 2008).

Based on the literature, the following hypothesis was developed:

H2: Age has a significant effect on Innovative Work Behaviour (IWB) when working remotely.

2.6.3. Gender of remote working employees and IWB

Gender can play a significant role in shaping IWB within organisational contexts. Diverse teams, including gender diversity, tend to generate a broader range of ideas and perspectives, leading to increased innovation (Hülsheger et al., 2009; Vegt & Janssen, 2003). Further, these teams tend to outperform homogenous teams regarding creativity and problem-solving. Several studies support this claim. Salazar et al. (2017) found that diverse teams have increased cognitive flexibility and a willingness to consider a wide variety of perspectives, generating more novel ideas. Similarly, Dutcher and Rodet (2018) found that experientially diverse teams create more ideas. Gender diversity improves participative communication, coordination, cohesion, and mutual support within a team, which can enhance innovative activities (Dai et al., 2018; Rejeb et al., 2019). Li et al. (2018) highlighted that members of diverse teams are likelier to contribute unique opinions and perspectives, facilitating idea generation. Furthermore, research by H. Leroy et al. (2021) demonstrated that teams with high inclusion levels are more prone to discuss, recombine, and integrate novel ideas and perspectives.

Gender diversity in teams can enhance innovative behaviours by promoting diverse perspectives, critical thinking, participative communication, coordination, and creativity. It further contributes to a more inclusive and innovative work environment, improving decision-making and innovation outcomes. Afsar and Badir (2017) argue that creating an inclusive and supportive work environment for both men and women can enhance IWB.

However, gender disparities persist in many workplaces, with males often dominating leadership positions and decision-making roles. This unequal distribution of power and influence can impact the IWBs of both men and women. Women, especially in male-dominated fields, may face challenges in having their innovative ideas recognised and accepted. Martin and Barnard (2013) found that women in male-dominated fields face unique challenges, including gender stereotypes, biases, and a lack of recognition for their ideas and contributions. These challenges can lead to hesitancy in voicing their opinions and innovative ideas.

Moreover, workplace cultures that perpetuate gender stereotypes or biases may limit the opportunities for women to participate in innovation-focused activities or projects. Concerns about negative gender stereotypes in professional contexts can hinder women's motivation, performance, and engagement, ultimately contributing to workplace gender inequity. This suggests that gender stereotypes and biases can hinder women from participating in innovation-focused activities (Cortland & Kinias, 2019). Heilman and Eagly (2008) found that gender stereotypes can influence perceptions of women's competence and limit their opportunities to participate in innovation-focused projects. Gender stereotypes may also impact perceptions and evaluations of IWB. Studies have found that IWBs are stereotypically associated with men more than women (Luksyte et al., 2017). It argues that IWB is viewed as a prototypically masculine activity associated with risk-taking, initiative, and embracing change. This supports the idea that IWB is expected more from men than women, leading to gender-based stereotypes. Women may face barriers and biases in accessing opportunities for IWB due to gender-based stereotypes. Carpini et al. (2023) discuss how gender stereotypes can interfere with the perception and evaluation of various work behaviours, including innovative behaviours. It suggests that women engaging in stereotype-inconsistent behaviour may not be recognised for their contributions, while men are noticed and rewarded for demonstrating stereotype-consistent behaviour, such as innovation.

Research has also shown that gender differences may account for variations in the degree to which men and women are involved or recognised in innovative work (Carmeli & Spreitzer, 2009). For example, it was found that employees reported more innovative behaviour when transformational leadership was displayed by male managers compared to female managers, suggesting that gender may play a role in how employees perceive and respond to leadership styles, which can in turn influence their IWB (Nusair et al., 2012). Reuvers et al. (2008) discuss the relevance of gender differences in transformational leadership and IWB. They suggest that male and female superiors exhibit a positive bias towards subordinates of the same gender, which can impact performance appraisals and contribute to the perception that IWBs are more associated with specific genders. J. Wang et al. (2017), on the other hand, examined the relationship relating to subjective wellbeing and knowledge sharing, the latter pivotal to innovation, and found that gender had no significant impact. The results suggested that employees of different genders are similarly willing to share knowledge and engage in innovation.

Providing equal access to opportunities for all genders can also influence IWB (Kim et al., 2018; Mehmood et al., 2022; Prieto & Pérez-Santana, 2014). Afsar and Afsar (2019) emphasise the importance of creating an organisational climate that supports risk-taking and motivates employees to apply initiative, which are crucial factors for fostering IWB. This suggests that providing equal access to opportunities for both men and women, such as training programs, mentorship, challenging assignments, an innovative climate, and access to opportunities to innovate, can enhance their IWB.

Work-family balance significantly influences IWB, suggesting that having access to work-family balance opportunities can positively impact an employee's ability to engage in IWB (Damayanti & Kurniawan, 2023; Li & Liu, 2023; Mishra et al., 2017). This can be achieved through opportunity-enhancing HR practices. Opportunityenhancing HR practices are significantly related to IWB (Bos-Nehles, Renkema, et al., 2017; Prieto & Pérez-Santana, 2014). This includes HR practices that impact autonomy, task composition, feedback, job demands and time pressure (Berkel et al., 2021; Bos-Nehles, Bondarouk, et al., 2017; Singh et al., 2020).

Based on the literature, the following hypothesis was developed:

H3: Gender has a significant effect on Innovative Work Behaviour (IWB) when working remotely.

2.6.4. Tenure of remote working employees and IWB

The length of tenure an employee has within a specific organisation can significantly influence their IWB (Woods et al., 2018). Long-tenured employees often deeply understand the organisation's culture, history, and existing processes which can foster IWB (Ng & Feldman, 2010). Employees with longer tenure often exhibit more innovation due to the networks and relationships they have developed over time (Ng & Feldman, 2010). Older and longer-tenured workers tend to have broader external networks and hold more central positions in internal networks, allowing them to disseminate and implement innovation within organisations effectively (Liu et al., 2016; Ng & Feldman, 2010). Additionally, longer tenure can enhance communication between leaders and employees, promoting high-quality exchange relations that contribute to innovation (Yuan, 2021). This institutional knowledge can be a double-edged sword concerning innovation. On one hand, it provides valuable insights into what has been attempted before, enabling employees to build upon past initiatives and avoid repeating unsuccessful endeavours. On the other hand, long-tenured employees might become entrenched in established ways of thinking and resist change, hindering the adoption of innovative practices. (Dam et al., 2008; Lauterbach & Kunze, 2023).

Conversely, with shorter tenures, new employees bring fresh perspectives and a willingness to challenge existing norms (Liu et al., 2016; Woods et al., 2018). Their lack of historical attachment to established processes can drive them to propose innovative Page 49

solutions and explore uncharted territories. Ng and Feldman (2010) found that for those employees with shorter average tenure, there is a positive relationship between tenure and job performance, while for employees with longer average tenure, the relationship becomes negative. Liu et al. (2016) found that factors such as status hierarchy and position tenure influence the relationship between organisational tenure and innovative behaviour. The relationship can be positive for individuals with low-status hierarchy and short position tenure but negative for those with longer tenure. Similarly, Yuan (2021) found that as tenure increases, employees with higher levels of conscientiousness tend to exhibit less innovative behaviour.

Based on the literature, the following hypothesis was developed:

H4: Tenure has a significant effect on Innovative Work Behaviour (IWB) when working remotely.

2.6.5. Employment Status of remote working employees and IWB

The nature of employment, whether part-time, full-time, temporary, or contractbased, can significantly influence IWB within organisations. Full-time employees often enjoy greater job security and benefits, which can foster a sense of stability and long-term commitment to their roles (Kalmi & Kauhanen, 2008; Peng & Li, 2021). This security can give employees the peace of mind to explore innovative ideas, take risks, and invest time in creative problem-solving (Kalmi & Kauhanen, 2008; Loi et al., 2011). On the other hand, part-time, temporary, or contract workers might experience higher levels of job insecurity, which can lead to a lack of motivation for long-term innovation (Montani et al., 2021; Probst et al., 2019; Spiegelaere et al., 2014). Such employees may focus on short-term tasks to secure their immediate position rather than dedicating time and effort to innovative projects that could benefit the organisation in the long run. Additionally, full-time employees typically have more opportunities for skill development and continuous learning within the organisation, which can enhance their innovative capabilities. In contrast, temporary or part-time workers might have limited access to training programs and resources, hindering their ability to contribute meaningfully to innovative initiatives.

Based on the literature, the following hypothesis was developed:

H5: Employment Status (ES) has a significant effect on Innovative Work Behaviour (IWB) when working remotely.

2.6.6. Employment level of remote working employees and IWB

The hierarchical employment level within an organisation, ranging from junior positions to administrators, managers, and leaders, can significantly impact IWBs among employees.

Junior employees, often at the beginning of their careers, may possess fresh perspectives and creative ideas driven by their enthusiasm and willingness to explore new concepts (Binnewies et al., 2007; Liu et al., 2016; Miron-Spektor et al., 2022). However, they might lack the experience and confidence to implement their innovations effectively. Nevertheless, employees in lower hierarchical positions may be more motivated to prove themselves and make a positive impact, leading to a proactive and innovative approach to their work (Dai et al., 2022). These employees, as well as administrators, are typically tasked with operational duties, may be more involved in day-to-day operations and have a deeper understanding of operational challenges and customer needs, which can inspire innovative solutions.

With their decision-making authority, managers play a crucial role in fostering innovation within their teams. They can empower employees by providing resources, support, and a conducive work environment, encouraging innovative thinking and risk-taking. At the top of the organisational hierarchy, leaders set the tone for innovation by promoting a culture of creativity, celebrating experimentation, and tolerating failure. Both of these higher hierarchical positions generally have access to more resources, decision-making authority, and access to information, which can facilitate their engagement in innovative behaviours (Niranga & Dharmadasa, 2021). Their vision and strategic direction guide the organisation toward innovative initiatives that align with the company's goals. Junior employees who lack this exposure to strategic initiatives may not receive the necessary support and recognition to engage in innovative behaviour (Shipton et al., 2006). Therefore, the employment level within an organisation not only reflects the individual's role and responsibilities but also influences their capacity to engage in IWBs,

with leaders playing a pivotal role in shaping a culture of innovation within the workplace.

Based on the literature, the following hypothesis was developed:

H6: Employment Level (EL) has a significant effect on Innovative Work Behaviour (IWB) when working remotely.

The following section introduces the theoretical framework guiding the study. The framework is grounded in established theory and provides a structured approach to understanding the phenomena under investigation. It not only outlines the key variables but also informs the analysis to contribute to the broader theoretical discourse within the field. Through the lens of this theoretical framework, we will explore the research questions and objectives, paving the way for a rigorous and insightful examination of the topic at hand.

2.7. Theoretical framework

Theoretical frameworks provide a conceptual foundation for the research study (Grant & Osanloo, 2014). These frameworks guide and structure the study, helping frame the research questions, define key variables, and establish the relationships between them. Theoretical frameworks draw upon existing theories, models, and concepts from relevant fields to inform the research design and analysis. They offer a lens through which the researcher can interpret and make sense of the empirical data, providing a framework for understanding the phenomenon under investigation. By employing a theoretical framework, the researcher seeks to develop a comprehensive understanding of the existing literature, integrate prior knowledge with new findings, and contribute to the field's theoretical advancements.

In the context of this study, there are several theoretical frameworks which could be used to study IWB and remote working. This thesis has adopted the Ability-Motivation-Opportunity (AMO) framework to guide the study.

2.7.1. Ability, Motivation and Opportunity framework

This thesis aims to explore the relationship between IWB and remote working, guided by the Ability-Motivation-Opportunity (AMO) framework. Since its emergence,
the framework has been largely accepted for explaining the link between human resource management principles and firm performance. The AMO framework provides a comprehensive theoretical framework that explains the relationship between individual abilities, motivation, and opportunities in influencing their performance and outcomes (Edgar et al., 2020; Obaid et al., 2022; Tabiu et al., 2016). As a theoretical model, it will allow us to understand and analyse employee performance and behaviour within organisations, namely the factors influencing employees' ability, motivation, and opportunity to engage in innovative behaviours. When employees have the necessary abilities, high levels of motivation, and favourable opportunities, they are more likely to perform well, achieve desired outcomes, and engage in desired behaviours, in this case, innovative behaviours. By applying this framework, the study will examine how remote working may influence each component of the AMO framework and its subsequent influence on IWB.

The AMO framework was initially proposed by Bailey (1993). He advocated the notion that cultivating employees' discretionary effort necessitates the integration of three fundamental components: firstly, employees must possess the necessary skills to contribute effectively (Ability); secondly, they must be imbued with suitable motivation to channel their efforts purposefully (Motivation); and lastly, employers must extend opportunities that enable and encourage their active participation (Opportunity) (Appelbaum et al., 2000). Building upon this theoretical model and incorporating insights from high-performance work systems (HPWS), Appelbaum et al. (2000) further refined the framework (Bayo-Moriones & Galdon-Sanchez, 2010; Knies & Leisink, 2013).

The AMO model of performance in Figure 2.3 below suggests that effective HPWS require three essential components to effectively use employees' discretionary effort: the opportunity to participate, appropriate incentives, and the development of employees' abilities and skills. This framework focuses on an individual's ability, motivation, and opportunity as key factors influencing behaviour and performance.

According to the AMO framework, discretionary effort will positively affect organisational performance (Boxall & Purcell, 2003). This is achieved by (a) developing employee skills and capabilities through knowledge, education and wellbeing (i.e. the ability to perform); (b) increasing an employee's motivation for the discretionary effort through job satisfaction, values and expectations (i.e. willingness to perform); and (c) providing employees with the opportunity to perform through autonomy, working conditions and ICT tools (i.e. the opportunity to perform). In summary, employees perform well when they have the capabilities to have adequate motivation, and their work environment provides opportunities to participate (Marin-Garcia & Tomas, 2016; Pak et al., 2019; Sterling & Boxall, 2013; Tabiu et al., 2016).

Figure 2-3: The AMO model of performance

(Adapted from Appelbaum et al. (2000))



Throughout successive studies, the model has undergone refinement and enhancement. Indeed, some scholars contend that the influence of AMO on performance is more multifaceted than initially anticipated, owing to the intricate interplay of factors. This complexity arises not solely from the mere presence of a set of practices (HRM content) but is further compounded by the subjective perceptions of these practices held by individual employees (Boxall & Macky, 2009; Ehrnrooth & Björkman, 2012).

The thesis will explore the ability aspect by assessing the skills, knowledge, competencies, and resources required to enable remote employees to engage in innovative behaviour (Bhatti et al., 2020). The ability component is crucial as it provides the foundation for individuals to perform tasks successfully (Obaid et al., 2022). Ability refers to an individual's knowledge, skills, and capabilities to effectively perform a specific task or behaviour. It encompasses both the inherent qualities and acquired competencies. Employees need the necessary abilities to engage in IWB. For example, they have knowledge of relevant tools or techniques, problem-solving skills, and domain expertise. In the context of this study, we are looking to explore how remote working impacts employees' ability to engage in IWB. Specifically, how does the availability of resources, tools, and technical skills required for innovation differ in remote work settings compared to traditional office environments?

The motivation aspect will be investigated by examining the motivational factors that drive employees to innovate while working remotely. It refers to individuals' internal drive, goals, and desires that influence their engagement, effort, and persistence in performing tasks. It includes intrinsic motivation, which arises from internal factors such as personal interest and enjoyment, and extrinsic motivation, which is driven by external rewards or pressures. Motivation plays a significant role in determining individuals' level of engagement and commitment to tasks (Knies & Leisink, 2013; Obaid et al., 2022). In the context of IWB, motivation involves the desire to generate new ideas, experiment, take risks, and contribute to the organisation's success. This thesis explores which factors may influence remote workers' motivation to engage in IWB. Aspects such as autonomy, recognition, and remote team dynamics may affect employees' motivation to generate and implement creative ideas while working remotely.

Finally, the opportunity aspect will be explored by analysing the remote work environment's supportive structures and processes that facilitate or hinder IWB. The opportunity component refers to the external factors and conditions that enable individuals to perform tasks effectively. In the context of IWB, opportunity includes factors such as organisational support, leadership encouragement, access to information and resources, and a culture that fosters experimentation and learning to provide employees with the necessary conditions and resources to effectively utilise their abilities and motivation. Organisations should create a supportive climate for innovation, encourage learning, and provide opportunities for employees to use their knowledge and skills (Afsar & Afsar, 2019).

The AMO framework has found wide-ranging application in diverse research domains such as human resource management, innovation and organisational behaviour (Bos-Nehles et al., 2013; Juan & Juan Martinez, 2016; Knies & Leisink, 2013; Rahman et al., 2023; Rayner & Morgan, 2017; Shahzad et al., 2019).

Fu et al. (2015), Shahzad et al. (2019) and Bhatti et al. (2020) have delved into the examination of the correlation between high-performance work systems (HPW) and innovation. In contrast, Radaelli et al. (2014) used the AMO framework to explore the relationship between knowledge sharing and IWB. Furthermore, Bos-Nehles, Renkema, et al. (2017) undertook a systematic literature review concerning the intersection of human resource management (HRM) and IWB, and concurrently, Shipton et al. (2017) investigated the linkage between HRM practices and innovation. More recently, Cui and Gui-lan (2021) and Ferrarini and Curzi (2022) both used an AMO theory perspective to explore innovation performance. Collectively, these studies substantiate the appropriateness of utilising the AMO theoretical framework as the underpinning for the present investigation.

In the next section, the conceptual framework that underpins this thesis is introduced, drawing from the literature review conducted and hypothesis development. This conceptual framework serves as the foundation for constructing the research questions and objectives. This section aims to establish a theoretical foundation that informs the research methodology and guides the exploration of the research questions and hypotheses. By aligning the study with the existing body of knowledge, we can identify the gaps in understanding.

2.8. Conceptual framework

The literature review results summarised the literature on IWB and remote working and discussed the relationship between them. A conceptual model that aggregates the two constructs is shown in Figure 2.4 below. The conceptual framework outlines the research study's key concepts, variables, relationships, and questions. Firstly, the self-reported level of IWB among Australian remote working employees (research objective one). Second, the impact of selected demographic factors (age, gender, tenure, employment status, employment title, and rate of working remotely) on the IWB of remote working employees (research objective two). Third, the factors of remote working that foster or hinder IWBs using the Ability, Motivation and Opportunity framework from the employee perspective (research objective three). Finally, the differences between remote work and office-based work on IWB from the employee's perspective (research objective four).

Figure 2-4: Conceptual framework



2.9. Gaps in the literature

The literature review has revealed several discernible gaps within the existing body of knowledge. The most significant gap lies in the need for more studies investigating the relationship between remote working and IWB. While extensive research has been conducted on remote working and IWB separately, there is a notable deficiency in the literature concerning studies that specifically explore the nuanced relationship between these two variables.

Secondly, most of the research has concentrated on investigating IWB and its connection to remote working within the European and United States contexts, leaving a notable dearth of research in the Australian context, which this study is centred on.

Thirdly, the literature is short on studies that have scrutinised each phase of IWB separately in a multidimensional manner. Most studies have treated IWB as a single dimension, whereas this study examines the individual phases of IWB along with IWB overall.

Additionally, a research gap exists in examining IWB and the influence of demographic variables (e.g., age, gender, employment status, tenure, employment level) within the context of remote work. This study aims to address this gap by investigating the impact of demographic factors on the IWB of remote-working employees in Australia.

Lastly, there is a paucity of qualitative research specifically examining the employee experience of remote working and its potential effects on innovative behaviours. Furthermore, there is a lack of studies that compare the lived experiences of employees working in an office environment versus when they are working remotely. This study addresses these gaps by including a qualitative investigation aimed at understanding the employee experience.

2.10. Chapter summary

This chapter comprehensively examines the existing literature on IWB and remote working. Additionally, the theoretical framework that guided the research study is explained in relation to the conceptual framework. The literature review reveals notable gaps in knowledge, thereby warranting an investigation into the employees' perceptions of the impact of remote working practices on their IWB, specifically within the Australian context. The identified gaps from the literature review inform the overarching methodology and research design of the study discussed in the next chapter.

CHAPTER 3 - RESEARCH METHODOLOGY

3.1. Introduction

Chapter 2 of the thesis reviewed the existing literature on innovative work behaviour (IWB) and remote working. It provided a comprehensive understanding of the previous research conducted in these areas. Furthermore, the chapter introduced the conceptual framework that forms the basis of the thesis, incorporating the Ability, Motivation, and Opportunity (AMO) framework to guide the research and establish a theoretical framework.

Moving forward, Chapter 3 presents the research design employed in the thesis to investigate the impact of remote working on the IWBs of Australian employees. The chapter begins with a discussion of the research philosophy underpinning the thesis, providing insight into the choice of a mixed methods research design. To ensure ethical considerations are addressed appropriately, the chapter concludes with a section dedicated to ethical considerations.

3.2. Research paradigm

'The philosophical framework is 'the world view within which the research is situated.'

(Quinlan et al., 2019, p. 56)

Before embarking on any research project, an essential decision for the researcher is the research methodology (Fawcett & Hearn, 2004; Zikmund et al., 2019). However, before a decision can be made on methodology, the researcher needs to consider which research paradigms will support the research methodology (Mackenzie & Knipe, 2006; Mattar & Ramos, 2022). Referring to Figure 3.1, the research philosophy supports the research methodology, which in turn supports the data-gathering methods. Defining the research paradigm is important, as it will guide the overall design of the research project and form the basis of the methodology, methods, types of research questions and data collection (Fawcett & Hearn, 2004; Lim, 2023; Wilson, 2014). Mackenzie and Knipe (2006, p. 194) emphasised, 'It is the choice of paradigm that sets down the intent, motivation and expectations for the research'.

Figure 3-1: Methodological Pyramid

(Source: adapted from Zikmund et al. (2019))



Researchers have different philosophical beliefs about how research should be conducted and what the research results should achieve (Bleiker et al., 2019; Cavana et al., 2001). How researchers interpret the world can be explained by research paradigms, which is the research philosophy. These define a set of assumptions and beliefs regarding how the world is perceived (Jonker & Pennink, 2010) and how knowledge is developed. Saunders et al. (2019) encourage researchers to undertake a reflective process to determine their philosophical position. This reflection process is depicted in Figure 3.2 below.

Figure 3-2: Reflective Process

(Source: adapted from Saunders et al. (2019, p. 126))



3.2.1. Interpretivism

'The social world is observed by seeing what meanings people give to it and interpreting these meanings from their viewpoint.'

(Blumberg et al., 2008, p. 21)

There are five major research philosophies: positivism, interpretivism, realism, pragmatism and postmodernism (Saunders et al., 2019). In this thesis, the researcher applied the research paradigm of interpretivism and employed a mixed-methods approach. Interpretivism, as a research philosophy, emphasises the understanding of the social world through the subjective experiences and meanings that individuals ascribe to it (Chen et al., 2011; Pathak & Thapaliya, 2022). Interpretivism is deeply rooted in the qualitative approach to understanding phenomena by focusing on subjective interpretations rather than objective realities (Chen et al., 2011). In a mixed methods study that incorporates both quantitative and qualitative elements, the research paradigm of interpretivism plays a crucial role in providing a comprehensive understanding of the research topic. By combining interpretivism with quantitative methods, researchers can delve into the subjective interpretations and meanings individuals attribute to their experiences while also quantifying certain aspects of the phenomenon under study (Quigley et al., 2022).

Philosophical perspectives can be seen as a collection of assumptions that shape the approach to and execution of research. Throughout the research project, the researcher will make several assumptions. Assumptions about the nature of knowledge and the realities a researcher encounters in their research (ontology), the use of knowledge (epistemology), and the extent to which their values influence their research process (axiology) (Saunders et al., 2019).

Ontology refers to the position taken on the nature of reality (Crowley et al., 2016; Scotland, 2012). Ontology serves as the starting point for research, guiding researchers in understanding the nature of reality and truth, which in turn influences the choice of research methods and approaches (Bleiker et al., 2019).

Interpretivism assumes a relativist ontology, where the researcher believes that social actors construct reality and that people's perceptions of reality differ from person

to person (Denzin & Lincoln, 2011; Scotland, 2012). Ontological assumptions underpinning an interpretivist paradigm are that reality is subjective, socially constructed, and will involve multiple perspectives and experiences (Rashid et al., 2019). The researcher relies upon the participant's perceptions of the situation and their version of 'reality' (Creswell, 2007). As the thesis seeks to understand how employees perceive that remote working may influence their innovative behaviours, adopting an interpretivist approach is well suited to uncover the reality as it is understood by the individual (Cavana et al., 2001; Chowdhury, 2014).

Epistemology concerns acceptable knowledge between the researcher and the participant (Hallebone & Priest, 2008). It refers to the manner of generating, using, and understanding knowledge that is deemed acceptable and valid. Epistemological assumptions underpinning an interpretivist paradigm are that knowledge is made up of subjective meanings and experiences. This thesis focuses on the lived experiences of employees working remotely. It focuses on the details and reality of the situation, including motivating actions.

Axiology in research refers to the roles of values in the research project and the researcher's position concerning the subject (Howes, 2015; Parter & Wilson, 2021). Researchers are part of the research, and their interpretation is crucial to the contribution. The interpretivist researcher explores and takes on reality from the subject's perspective, which is referred to as an insider perspective. Therefore, the collected and analysed data will be influenced by the experiences and values of the researcher and the subjects.

The notions of interpretivism aligned with the proposed research objectives and questions and guided the methodology undertaken in this thesis. 'The interpretivist researcher is interested in understanding the lived experience of human beings' (Cavana et al., 2001, p. 9) and is concerned with how humans make sense of the world around them. This philosophical stance is heavily focused on perceptions and interpretations and argues that humans are different from physical phenomena in that they create meanings. An interpretivist approach is suitable for this thesis as it seeks to understand the reality of working remotely and the influence this may have from an employee's perspective. This approach recognises that each employee will have their background and experiences that will influence their reality. Therefore, this approach accepts multiple realities because experiences are subjective.

Methods using an interpretivist philosophical position are typically qualitative data, which includes small-scale samples, and in-depth investigations such as interviews (Hammond & Wellington, 2021). This provides the rich, highly detailed accounts of individual experiences sought by this approach. While positivist researchers believe in replication research to test theories using quantitative data, interpretivist researchers try to understand inside perspectives or the real meaning of social phenomena. Interpretivists use an inductive process to develop a theory or patterns of meaning.

It has previously been thought that the philosophical basis of quantitative and qualitative research are different (Walliman, 2018) and cannot be combined. However, researchers now agree that the 'distinction between the methods used for quantitative and qualitative research are not as distinct as previously believed, that many data collection methods and data types can be used for both approaches and that the paradigms behind them can coexist in a single study' (Walliman, 2018, p. 168). Instead of conflicting with the various paradigms, a mixed-methods approach combines the differences to achieve a deeper understanding (Walliman, 2018). Therefore, an interpretivist research study can utilise a range of data including both qualitative and quantitative, including case studies, document analysis, and questionnaires (Cavana et al., 2001; Saunders et al., 2019).

In this thesis, the interpretivist paradigm guided qualitative and quantitative methods in collecting, analysing, and interpreting data to understand remote working and IWB better. The quantitative data collection phase commenced first, closely followed by the start of the qualitative phase. The quantitative phase also provided an opportunity to recruit participants for the qualitative phase by asking respondents if they wanted to participate in an interview. Both data collection methods were being conducted simultaneously during the overall data collection period.

A survey of Australian-based employees was used to collect quantitative data to test the hypotheses. The quantitative method was conducted as an exploratory study to understand how often innovative behaviours are exerted by employees when they work remotely. The survey sought to understand how often these behaviours were demonstrated overall and within each of the four phases of innovative working behaviour.

In-depth, semi-structured interviews were used to collect qualitative data. Thematic data analysis was employed to analyse the emerging themes. The qualitative method was conducted as an exploratory study to understand the perceptions, from the employees' point of view, of how and in what way working remotely may influence their innovative working behaviours. The literature review identified a need for qualitative studies exploring remote working and IWB, particularly from the employee context. Therefore, the qualitative study was vital in exploring employee perceptions and understanding the role of ability, motivation and opportunity to exert innovative behaviours while working remotely.

The findings from the survey used in the quantitative phase and the insights that emerged from the interviews during the qualitative phase contributed to the overall thesis findings.

3.3. Rationale for mixed methods approach

The mixed methods approach combines both quantitative and qualitative data collection methods. This research study adopted a mixed methods approach for several reasons, including to:

- explain and interpret a situation
- explore a phenomenon
- address the research questions at different levels
- utilise the advantages (strengths) and minimise each approach's disadvantages (weaknesses) (Bracio & Szarucki, 2020; Walliman, 2018).

A mixed-methods approach was chosen because it can provide a greater understanding of the research problem than could be achieved using a single approach (i.e. quantitative or qualitative alone) (Creswell, 2007; Johnson & Onwuegbuzie, 2004; Pluye & Hong, 2014). This thesis aims to not only understand how often an employee demonstrates innovative behaviours when working remotely, which can be analysed using quantitative approaches, but also to explore the factors that may support or hinder the employee exerting such behaviours, which lends itself to qualitative research. It further provides confirming, complementary and contrasting sources of data (Hammond & Wellington, 2021). Using a combination of methods in a single study improves the effectiveness of the research model in answering the research questions (Johnson et al., 2007; Shorten & Smith, 2017). Researchers can combine quantitative observation data with qualitative perception data to address the research aims. The research questions in this thesis relate to what is happening (i.e. how often the employee exerts innovative behaviours) as well as the perceptions of what is happening (i.e. what influence the remote working is having on innovative behaviour), questions that can be answered using both quantitative and qualitative methods.

3.4. Research design

This section discusses the research methods and overarching design of the study. The research methods, types, and sources of data collected are discussed in the subsequent two sections, first the quantitative study and then the qualitative study.

The research design is the plan of how the study will answer the research questions and methods used, including the sample and measures (Creswell, 2007; Trochim et al., 2016). All research studies should begin with a research problem. The researcher should then review empirical literature related to the topic, generate the research questions, collect data, analyse the data, and generate findings.

The first step of the mixed methods approach is deciding the balance between the quantitative and qualitative methods used within the research study (Johnson et al., 2007). Researchers can choose between quantitative dominant, qualitative dominant or equal status approach. The quantitative dominant approach is based on a post-positivist quantitative paradigm, with some qualitative aspects also researched. A qualitative dominant approach is based on realism, interpretivism or constructivist paradigms, with the addition of quantitative methods. An equal status approach is a pragmatic approach with complementary and equal qualitative and quantitative elements (Walliman, 2018).

For this thesis, a qualitative dominant approach was chosen, as the primary focus of the study to understand the employees' perceptions and lived experiences. The overarching research aim in this thesis is how employees perceive that their IWB may be shaped by working remotely. The interpretivist paradigm is, therefore, a suitable philosophical position for a qualitative dominant mixed methods approach. 'Understanding why individuals and groups think and behave as they do lies at the heart of qualitative research' (Gill et al., 2008, p. 292). Therefore, the research study employed a qualitative dominant approach because of its capacity to explain the employee experiences of remote working and their meaning for participants. Quantitative methods were included because of their capacity to measure the level of innovative behaviour exerted by employees working remotely to add greater depth and understanding of whether remote working influences innovative behaviours.

Once a researcher has decided on the balance between methods, a decision on collection strategy, or ordering, needs to be made between sequential or concurrent (Clark & Creswell, 2008; Stentz et al., 2012). Sequential designs involve researchers collecting either quantitative or qualitative data first, followed by the other method in a subsequent phase. According to Clark and Creswell (2008), there are three sequential strategies: explanatory, exploratory or transformative. Concurrent designs involve researchers collecting both quantitative and qualitative data simultaneously and can employ triangulation, nested or transformative strategies.

This thesis adopted a concurrent nested mixed methods research design, using the appropriate research method for each research question. The concurrent nested strategy is where the researcher uses appropriate data collection and analysis methods to answer each research question. The nested strategy is used to address different research questions or objectives within the same study, with one form of data providing a supportive, supplementary role to the other. One type of data (quantitative or qualitative) is predominant, while the other is nested within it to provide additional insights. This secondary data type might help explain the primary data or provide context. In this thesis, quantitative to measure IWB when remote working and qualitative to explore employee perceptions of their innovative behaviour when working remotely. The findings from each method are combined to answer the overarching research question.

Figure 3.3 shows the stages of the thesis research design, including the processes and outcomes. The two types of data collected were semi-structured interviews with Australian-based employees undertaken via Zoom and an online questionnaire of Australian-based employees using Qualtrics software.

Figure 3-3: Stages of research design



3.5. Research questions

The increased availability and use of remote working presents both a challenge and an opportunity for organisations. The research gap was identified through the literature review on IWB and remote working literature. This knowledge gap relates to how the usage of remote working in Australian organisations influences employees' innovative behaviour in each of the four phases of the IWB model (De Jong & Den Hartog, 2010). To address this research gap, the research questions were designed to gather deep insights into the employee's 'reality' and are consistent with the interpretivist paradigm.

This thesis's primary research question aims to explore IWB and remote working from the perspective of Australian employees. The key research question is: *How do employees perceive the relationship between remote work and their innovative work behaviour (IWB)*?

The research sub-questions shown in Table 3.1 were as follows:

 Table 3-1: Research sub-questions

	Research sub-questions	Type of study
SQ1	What is the extent of IWB among employees engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?	Quantitative
SQ2	How do demographics influence the outcomes of employees' IWB engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?	Quantitative
SQ3	What factors contribute to fostering or inhibiting innovative work behaviours in remote working environments?	Qualitative
SQ4	What are the distinctions in employee experience regarding innovative work behaviour when comparing remote and office-based working environments?	Qualitative

The first and second sub-questions explore how often innovative behaviour is exerted when working remotely. This will allow us to examine if there are differences in the level of self-reported frequency subject to how often the employee works remotely. It also allows us to explore the relationship between employee profiles (e.g. level at organisation, tenure at organisation, gender and age) and the amount of IWB exerted.

The third sub-question has been designed to examine the features of remote working that can foster or hinder the employees' innovative behaviour. The factors explored relate to the employees' ability, motivation, and opportunity to demonstrate innovative behaviours when working remotely. The fourth sub-question aims to allow employees to reflect and provide their views on their experience and innovative behaviour when working at the office compared to when working remotely.

3.6. Population sample

The theoretical population for this study is Australian-based employees who currently work some or all of their time remotely, away from the main office building.

The Australian Bureau of Statistics (ABS) defines an employee as people who work for a public or private employer and received remuneration in wages or salary or are paid a retainer fee by their employer or worked on a commission basis for tips, piece rates or payment in kind (ABS, 2022b).

The ABS defines people as 'employed' if they are aged 15 years and over and work one hour or more in a reference week. The 'one-hour rule' is used internationally and allows employment figures to be compared with other countries.

The population sample was based on the 2021 Australian Census which recorded that of the 12 million people employed in Australia on 10 August 2021, 2.5 million people worked from home (ABS, 2022a).

3.7. Ethics

An ethics application was submitted to the Victoria University Human Research Ethics Committee (VUHREC) in April 2021 and formally approved in July 2021. The approved ethics application number is HRE21-055 (see Appendix A). The ethics application described the research design, methodology, data collection procedures, how participants would be recruited and the survey and interview questions. The consent form, information sheet for participants, interview protocol and survey questions are included in Appendices B, C, D, E, and F.

3.8. Chapter summary

In summary, this chapter provides an overview of the thesis methodology, research design, and approach. It provides the rationale behind adopting a mixed methods approach and research paradigm. The next two chapters discuss the data collection methods, sampling techniques, and instruments utilised in the quantitative and qualitative studies.

CHAPTER 4 - THE QUANTITATIVE RESEARCH METHOD

4.1. Introduction

This chapter provides an overview of the quantitative research methods employed in this thesis. Quantitative research is characterised by its systematic and empirical investigation of phenomena through numerical data and statistical analyses.

At the outset of this study, data collection began just before the COVID-19 pandemic, with firms being contacted to invite their employees to participate in an online questionnaire. The onset of the pandemic necessitated a shift to a fully online approach, disrupting the original data collection plans. Additionally, many firms began conducting their own internal surveys to assess employee well-being and mental health, which made it more challenging to add this study's survey into their schedules. Of the four firms that were approached for the survey, two agreed to participate, necessitating the recruitment of additional participants by promoting the survey in professional closed groups on the social networking site LinkedIn.

Despite these challenges, the pandemic also presented a unique opportunity as it led to a significant increase in remote work. This shift enabled the collection of rich and deeper insights, as more employees had firsthand experience with remote working. Consequently, data was collected via two methods: quantitative surveys, which are discussed in this chapter, and in-depth interviews with employees, which are covered in the following chapter on qualitative research methods.

The following sections detail the research design, including the formulation of hypotheses, the selection of variables, and the methods employed for data collection. Additionally, it will outline the statistical techniques applied for data analysis, providing transparency regarding the tools used to draw meaningful conclusions from the gathered data.

4.2. Quantitative research methods

Quantitative research holds significant value in the social sciences for several reasons. It provides a systematic and rigorous approach to studying social phenomena (Laher, 2016; Maula & Stam, 2019), enabling researchers to collect and analyse numerical data, and meets the thesis research objectives as discussed in Chapter 1.

Quantitative research emphasises objectivity and strives to minimise researcher bias (Field & Derksen, 2020; Jindal et al., 2015). It employs standardised measures and statistical techniques, allowing for replication and generalisation of findings, which can enhance the reliability and validity of the research (Bryman, 2016).

Further, quantitative research is well-suited to testing and explaining relationships, which justifies using this approach to explore the influence of remote working on the amount of IWB displayed by Australian employees. We can use statistical analysis to uncover patterns, relationships, and trends in data to examine the magnitude and significance of associations between variables, aiding in identifying causal relationships (Gravetter & Forzano, 2018). Quantitative techniques are powerful at studying groups of people and making generalisations from the studied sample to broader groups beyond that sample (Creswell, 2018). Quantitative research generates empirical evidence that can inform evidence-based decision-making. Policymakers, practitioners, and organisations can utilise the quantitative research findings to develop effective interventions, policies, and strategies.

4.2.1. Sampling method and sample size

A sample refers to a subset of a larger population (Lohr, 2019), and the sampling strategy is the process that defines how the researcher will select a sample from which population to ensure that the sample will be representative of that particular population.

The study focused on Australian-based employees. Since the study targeted Australian employees, the questionnaire was adapted to the Australian context and reframed to align with the study's objectives. A random sampling technique was used to allow for the generalisation of results and to avoid any possible bias attributed to a specific group of workers. Creswell (2018, p. 250) defines random sampling as 'a procedure in quantitative research for selecting participants. It means that each individual has an equal probability of being selected from the population, ensuring that the sample will be representative of the population'. The questionnaire included respondents from all age groups (18 years and above), all genders, working within any industry in Australia.

The calculation of sample size ensures that the survey results are statistically significant and can be generalised to the entire population with the given confidence and precision (Del Águila & González-Ramírez, 2014; Singh & Masuku, 2014).

An online calculator (<u>www.qualtrics.com</u>) was used to identify a suitable sample size by entering the desired confidence level, margin of error, and population size. A confidence level of 95% and a 5% margin of error were chosen, as well as the ABS population of people working from home during 2021. For a population of 2,500,000, you would need to survey approximately 385 individuals (see Figure 4.1) to make inferences about the entire population within the specified confidence and precision parameters.

In total, 431 questionnaires were returned, an acceptable response rate higher than the sample size calculation. A total of 107 of these questionnaires were removed from the sample due to invalid responses. Invalid responses were those where the respondents answered 'never' to the question of how often they worked remotely, where respondents did not answer any of the IWB questions or where they did not answer all the demographic questions. They were not removed if they answered some but not all of the IWB questions. This resulted in 324 usable questionnaires.

Figure 4-1: Calculation of sample size

Sample size calculator
Confidence Level:
95%
2500000
Margin of Error:
5% -
Ideal Sample Size:
385

4.2.2. Participant recruitment

The recruitment of participants came from various sources. Four firms within existing professional networks were contacted to participate in the study. These organisations were selected because they are Australian-based and have employees who work remotely either some or all of the time. Collectively, these firms employ over 3,500 people. However, due to the challenges arising from the COVID-19 situation and the

significant number of surveys employees were being asked to complete during that time, only two firms agreed to participate.

The initial step in data collection involved distributing the Qualtrics questionnaire link to these two firms. Between September 2021 and February 2022, the survey link was sent to employees at these firms located in Melbourne and Perth. The link was forwarded to internal employees through designated contacts such as the Head of Human Resources. Approximately 200 and 65 employees are employed at each organisation, respectively. From these, 35 and 17 surveys were returned, resulting in response rates of 18% and 26%, respectively.

Since only two firms agreed to participate and the response rate was lower than expected, a new approach was necessary to gather sufficient data. A new questionnaire link was generated and shared across various professional LinkedIn groups specifically targeting Australian-based employees. This broader distribution aimed to increase participation and ensure a representative sample.

4.3. Data collection

An online questionnaire was chosen as the data collection method, and responses were collected through the use of online survey software Qualtrics. The questionnaire consisted of a set of questions designed to meet the aims and objectives of the study and expected to provide a useful dataset that addresses the research questions and tests the research hypotheses. This includes how often employees (self-rated) demonstrate IWB overall and during each of the four phases while working remotely.

4.3.1. Questionnaire design

The questionnaire was designed to achieve the following objectives: (a) measure the frequency of IWB overall and within each phase of the IWB model (De Jong & Den Hartog, 2010) among remote working employees; and (b) examine if demographic factors influence IWB overall and within each phase of the IWB model among employees who work remotely.

The questionnaire started with a screening question to be answered before commencing the main questionnaire to confirm that the participant was currently working in Australia. If respondents answered 'No' to the question about currently working in Australia, the survey would terminate, preventing them from proceeding. To ensure data integrity, the IP addresses of all returned responses were checked to confirm there were no duplicates from individuals who initially selected 'No' and then reopened the link to select 'Yes'.

In summary, the questionnaire contained 35 items within six sections. This is shown in Table 4.1 below.

Section	Heading	Question #	No. of items	Responses
1	Demographics	1–7	7	Multiple choice
2	Idea Exploration	8–17	10	Five-point Likert scale
3	Idea Generation	18–24	7	Five-point Likert scale
4	Idea Championing	25–29	5	Five-point Likert scale
5	Idea Implementation	30–34	5	Five-point Likert scale
6	Participating in interview	35	1	Yes/No

Table 4-1: Questionnaire sections and items

4.3.2. Likert scale

A five-point Likert scale provided scores for low or high values to represent the frequency of IWB activities undertaken by the respondents when working remotely. Table 4.2 below outlines the Likert scale used in this thesis to measure the employees' frequency of IWB when working remotely. To ensure the scales were clear in terms of frequency levels, scales 2 to 4 specified the assumed frequency in their description.

Likert scales are widely recognised and utilised in research and surveys. One primary advantage is their ability to measure attitudes, opinions, and perceptions with a high degree of reliability and validity, which allows the researcher to capture participants' responses in a measurable way (Norman, 2010; Sullivan & Artino, 2013). The scales

provide a structured format that allows respondents to express their level of agreement or disagreement with a series of statements, thus facilitating data collection and analysis. The scales also promote straightforward interpretation and data comparison, making them suitable for both quantitative and qualitative analyses (Jamieson, 2004).

Scale	Description
1	Never
2	Rarely (once a month)
3	Sometimes (2–3 times per month)
4	Often (4–8 times per month)
5	Always

Table 4-2: Five-point Likert scale

4.3.3. Section 1: Demographics

The questions in this section sought to obtain demographic information from respondents to describe the study population. In this section, the questionnaire items requested information on the following demographic variables: age, gender, employment status, tenure at the current organisation, employment level and frequency of working remotely.

4.3.4. Sections 2–5: Idea Exploration, Generation, Championing, and Implementation

In the following four sections of the instrument, the questionnaire asked respondents to indicate how often they undertook certain IWB activities while working remotely. The questions were divided into the four phases of the IWB model; idea exploration, idea generation, idea championing and idea implementation (De Jong & Den Hartog, 2010). Respondents were asked to select between a scale of 'Never' to 'Always'. There were 27 items to assess IWB, adapted from three IWB measurement scales: De Jong and Den Hartog (2010), Messmann and Mulder (2012) and Kleysen and Street (2001). Previous studies have found the survey scales to be reliable and valid. Questions included, 'When you are working away from the office, how often do you wonder how things can be improved?', 'When you are working away from the office, how often do

you try to generate solutions for problems?' and 'When you are working away from the office, how often do you promote new ideas to gain the support of my managers?' Appendix B identifies the source of study used for each question.

4.3.5. Section 6: Participation in the research interview

The final question asked respondents if they were willing to participate in an interview with the researcher. The question explained that the interview would take 30 minutes via video conferencing. The respondent could select 'yes' or 'no'. If they selected 'no', the questionnaire was completed. If they selected 'yes', a new window was brought up, and the respondent was asked to enter their email address so the researcher could make contact to arrange a time and date. The purpose of the new questionnaire window was to ensure that the respondent's email address could not be directly linked to their questionnaire response, thereby ensuring anonymity.

4.4. Quantitative data analysis method

Frequently, the first step in undertaking statistical analysis is examining the data one variable at a time, which is called univariate analysis (Gray, 2007). Univariate analysis explores each variable in a data set separately and seeks to describe each variable on its own. It looks at the range of values, frequency, and central tendency of the values and uses descriptive statistics to describe and summarise the data. This approach allows researchers to understand the characteristics of individual variables without considering the relationships between them.

Researchers typically follow the univariate analysis with bivariate statistical methods. The aim is to examine the association between the dependent variables of interest and various independent variables. Methods such as correlation coefficients and contingency tables are adopted. If the researcher is interested in understanding the impact of several independent variables on a dependent variable, the bivariate analysis is followed by multivariate analysis, which includes regression analysis methods.

This thesis utilised several statistical methods to achieve the study objectives, including descriptive statistics (univariate level), ANOVA (bivariate level), and logistic regression (multivariate). The software used for the statistical and hypothesis analyses was SPSS v. 29.0.

4.1.1. Univariate Analysis

This analysis involved the application of descriptive statistical techniques to systematically organise, analyse, and interpret the quantitative data gathered in the study. Descriptive analysis was employed to clearly understand the sample population and evaluate various aspects of the research. The distributions of variables were primarily examined and quantified using frequencies and percentages, offering valuable insights into the participants' responses. The computed descriptive statistics, including means, standard deviations, frequencies, and percentages, were tailored to each variable, and effectively visualised through frequency distributions and histograms.

4.1.2. Bivariate and Multivariate Analysis

This section discusses the data analysis methods used to answer sub-research questions 1 and 2 to determine the effect of working remotely on IWB and the effect of demographic variables on IWB when working remotely. The study objectives of these two research questions were achieved by examining a set of hypotheses using different data analysis techniques, both bivariate (ANOVA) and multivariate analysis (regression). ANOVA allows us to determine if the mean scores of different groups differ, while regression allows us to determine if and how variables are related (Rutherford, 2011).

4.1.3. Significance level

In statistical hypothesis testing, the significance level (often denoted as α) is the threshold used to determine whether to reject or fail to reject the null hypothesis. The conventional significance level commonly used in social sciences and other fields is 0.05 (or 5%). However, some researchers may choose to use a slightly higher or lower significance level based on factors such as the nature of the research question, the complexity of the data, or the specific field of study.

The determination of the significance level in hypothesis testing is contingent upon the specific context of the study (Johnson, 1999; Luo, 2018), and the use of a pvalue of 0.05 merely serves as a heuristic proposed by Ronald Fisher in his book *Statistical Methods for Research Workers* (1925), without intending to be construed as an unequivocal yes-or-no threshold (Meyer et al., 2017). While in medical studies for example, where there could be severe consequences if a wrong decision is made about a hypothesis, it would be more reasonable to use 0.05 or even 0.01 (Cramer & Howitt, 2004; Meyer et al., 2017), the 0.1 level of significance is considered reasonable in social sciences.

To test the hypotheses, the significance level was set at 0.1 level. For this thesis, 0.1 was considered reasonable for two reasons. Firstly, as the sample size is small, statistical power may be limited, making it more difficult to detect significant effects (Thiese et al., 2016). In such cases, researchers might opt for a slightly higher significance level to increase the chances of identifying meaningful relationships. Secondly, as this study is exploratory, a higher significance level is more likely to detect potential patterns or relationships that can guide further research (Bischof & Velden, 2019; Wicherts et al., 2016). A higher threshold allows for a broader exploration of the data, acknowledging the possibility of finding initial associations that could be investigated further, in this case, during the qualitative data collection phase.

4.1.4. Variables

In this thesis, a series of comprehensive tests were undertaken using six independent variables and five dependent variables. The classification of variables is shown in Table 4.3.

Variable	Label	Categories	
Dependent Innovative Work		'Never', 'Rarely', 'Sometimes', 'Often'	
	Behaviour	and 'Always'	
	Idea Exploration		
	Idea Generation		
	Idea Championing		
	Idea Implementation		
Independent	Remote Working	'1–3 times per month', '1 day per	
	Frequency	week', '2–4 days per week' and	
		'Always work remotely'	
Independent	Gender	'Female', 'Male', 'Non-binary', 'Prefer	
		not to say'	
Independent	Age	'18–30 years', '31–40 years', '41–50	
		years', '51 years +'	
		1	

Table 4-3: Variables

Independent	Employment Status	'Full time', 'Part-time' and 'Casual/Contract'
Independent	Tenure	'Less than 2 years', 'Between 2–5 years', and 'More than 6 Years'
Independent	Employment Level	'Administration', 'Non-manager', 'Manager', 'Senior Leadership'

4.2. Hypothesis

Table 4-4: Hypothesis table

Code	Description	Test
H1	RWF has a significant effect on IWB	ANOVA
H2	Age has a significant effect on IWB when working remotely	ANOVA
НЗ	Gender has a significant effect on IWB when working remotely	T-Test
H4	Tenure has a significant effect on IWB when working remotely	ANOVA
H5	Employment Status (ES) has a significant effect on IWB when working remotely	ANOVA
H6	Employment Level (EL) has a significant effect on IWB when working remotely	ANOVA
H7	RWF, Age, Gender, Tenure, ES, EL has a significant effect	Logistic
	on the odds of IWB occurring	Regression
H8	RWF, Age, Gender, Tenure, ES, EL has a significant effect	Logistic
	on the odds of Idea Exploration occurring	Regression
H9	RWF, Age, Gender, Tenure, ES, EL has a significant effect	Logistic
	on the odds of Idea Generation occurring	Regression
H10	RWF, Age, Gender, Tenure, ES, EL has a significant effect	Logistic
	on the odds of Idea Championing occurring	Regression
H11	RWF, Age, Gender, Tenure, ES, EL has a significant effect	Logistic
	on the odds of Idea Implementation occurring	Regression

4.3. Analysis – ANOVA

ANOVA (Analysis of Variance), also known as between-groups analysis of variance, is a widely used statistical technique in social science research and is considered one of the primary methods for analysing data (Hancock et al., 2010; Rutherford, 2011). Its primary objective is to determine whether an independent variable is responsible for variation in an outcome measure, which is the dependent variable. In the context of this thesis, ANOVA provides a method to examine significant differences in IWB among various groups, such as different employee levels, frequency of remote work, and tenure at the current organisation. Through ANOVA, researchers can gain valuable insights into the factors that may contribute to varying levels of innovation across different contexts. There is a considerable body of research utilising ANOVA to investigate IWB. For example, Sudeshna Basu and Anjali (2009) explored the relationship between the IWB of managers and the levels of stress arising out of stressful events in modernised organisations, Turgut and Sökmen (2018) investigated the role self-efficacy plays in organisational ethics and IWBs, while (Abun et al., 2023) examined the effect of an innovative work environment on the IWB of employees.

ANOVA is helpful in this thesis for several reasons:

- a) Social science research often compares means or group differences across multiple categories or conditions. ANOVA allows for the simultaneous comparison of two or more groups, making it suitable for examining the effects of different factors on dependent variables.
- b) ANOVA is robust to deviations from assumptions, such as normality and equal variances. This makes it applicable in a wide range of social science studies where data distribution may not strictly adhere to assumptions.
- c) ANOVA has a long history of application in social science research and is wellestablished in statistical literature. It is widely used in various fields, such as psychology, sociology, industry, commerce, education, and economics, making it a familiar and accepted tool within the social sciences and aligned with this study's research objectives (Rutherford, 2011).

In this thesis, ANOVA was used to analyse the means of each independent variable (remote working frequency and our demographic factors) on the dependent variable to understand whether there were any differences in means within groups. The results of the ANOVA analysis are found in Chapter 6.5.

For the gender variable, which consisted of only two groups, an independent ttest was conducted instead of ANOVA. The independent-samples t-test is used to determine if a difference exists between the means of two independent groups on a continuous dependent variable (Hassan et al., 2021; Lai et al., 2016). For this demographic variable, the independent-samples t-test was employed in lieu of ANOVA, as our dataset comprised only two categories (Gerald, 2018; Wilcox, 2017). ANOVA is typically utilised when analysing three or more categories (Donati et al., 2021; Wilcox, 2017).

4.3.1. Testing for normality

Researchers often conduct normality tests as a preliminary step before performing ANOVA to confirm whether the data conform to a normal distribution (Lantz, 2012). A normal distribution, also known as a Gaussian distribution or bell curve, is a symmetric probability distribution that is characterised by its bell-shaped curve when plotted on a graph (Thrane, 2023). In a normal distribution, the data is symmetrically distributed around the mean, with the majority of the data clustered around the mean and tapering off towards the tails. When the data adhere to a normal distribution, ANOVA can offer precise and robust estimates of group differences, facilitating the interpretation of the analysis (Bird, 2002), whereas deviations from normality can result in biased estimates and incorrect conclusions (Arias-Castro et al., 2011). The normal distribution is characterised by two parameters: the mean (μ), which represents the central tendency of the distribution, and the standard deviation (σ), which indicates the spread or dispersion of the data.

The Shapiro-Wilk test was used to assess whether a given sample of data follows a normal distribution, and is generally considered to be better at detecting departures from normality in smaller sample sizes than other normality tests such as Kolmogorov-Smirnow or Anderson-Darling tests (Jo et al., 2016; Mbah & Paothong, 2014; Wah & Sim, 2011).

However, at larger sample sizes the Shapiro-Wilk test will flag even minor deviations from normality as statistically significant (i.e. not normally distributed) (Bishara et al., 2021; Shatz, 2023). The implication is that the Shapiro-Wilk test might indicate that the data is not normally distributed due to minor deviations that have little practical impact on analyses that assume normality. The Normal Q-Q (Quantile-Quantile) plot is a good complementary visual approach to assess normality (Loy et al., 2016). It plots the quantiles of the sample data against the quantiles of a normal distribution. If the data is normally distributed, the points should fall approximately along a straight line. If the Q-Q plot shows data points roughly along a straight line, this suggests that the data is approximately normally distributed, and minor deviations detected by the Shapiro-Wilk test may be disregarded in the context of practical analysis (Korkmaz et al., 2014).

4.3.1.1. Kruskal-Wallis H Test

The Kruskal-Wallis H test is considered an alternative to the one-way ANOVA, and is useful when the assumptions of parametric tests, such as ANOVA, are violated, such as the requirements of normality or homogeneity of variance (Conover, 1999; Dinno, 2015; Sherwani et al., 2021; Sheskin, 2011).

In this study, in instances where it was found that the data was not normally distributed, a Kruskal-Wallis H Test was also conducted. The Kruskal-Wallis H test (sometimes also called the 'one-way ANOVA on ranks') is a non-parametric test that can be used to determine if there are statistically significant differences in median values between two or more groups of an independent variable. Unlike the Jonckheere-Terpstra test when you have a specific directional hypothesis about the order of the groups, expecting a trend (Hernández et al., 2021; Nielsen et al., 2017), the Kruskal-Wallis H test is used when you want to test for any differences among the groups without assuming any specific order or trend in the data.

Irrespective, the one-way ANOVA is considered 'robust' to violations of normality. This means that some violations of this assumption can be tolerated, and the test will still provide valid results. Furthermore, as sample size increases, the distribution can be very non-normal, and, thanks to the Central Limit Theorem, the one-way ANOVA can still provide valid results. Hence, both tests were reported in instances where the normality test was violated.

4.3.2. Homogeneity of variances

The assumption of homogeneity of variances states that the population variance of the dependent variable for each group of your independent variable is the same. The assumption of homogeneity of variances was tested using Levene's test of equality of variances (Gastwirth et al., 2009; Marozzi, 2011). It is employed as a preliminary analysis before conducting parametric tests such as ANOVA. Levene's test helps to determine whether the assumption of homogeneity of variances is met, which is a prerequisite for valid inference in ANOVA (Balnaves & Caputi, 2001; Mara & Cribbie, 2017; Wang et al., 2016).

When this assumption is not met, you cannot interpret the standard one-way ANOVA. In this thesis the Welch ANOVA was used. Welch's ANOVA does not assume equal variances and is more robust to heteroscedasticity (Dağ et al., 2018; Krishnamoorthy et al., 2007; Velina et al., 2016). It adjusts the degrees of freedom used in the F-ratio calculation to account for the inequality of variances, providing a more reliable test when the assumption of homogeneity of variances is violated (Ahad & Yahaya, 2014; Derrick et al., 2016).

4.3.3. Post-hoc testing

Post hoc testing was also conducted when the initial analysis indicated significant differences among group means, as evidenced by a statistically significant F-statistic. This significant F-test suggests that at least one group mean differs from the others, leading to the rejection of the null hypothesis that all group means are equal. Since ANOVA itself does not specify which specific group means differ, post hoc tests are necessary to identify these differences (Einot & Gabriel, 1975; Hsu, 1996; Lee & Lee, 2018).

This thesis used Tukey-Kramer post hoc analysis (Lee & Lee, 2018; Ramsey et al., 2011) to compare the means of all possible pairs of groups following a significant difference found in the overall ANOVA test. Where a significant different was found with the Kruskal-Wallis test, pairwise comparisons using Dunn's (1964) procedure with a Bonferroni adjustment were undertaken.

Tukey's test is ideal for parametric data as it assumes normality and homogeneity of variances, conditions that ANOVA also relies upon. This test is designed to compare all possible pairs of group means while controlling the family-wise error rate, making it efficient for identifying specific differences between group means when the ANOVA indicates significant results. The Tukey-Kramer test is an extension of Tukey's Honestly Significant Difference (HSD) test, specifically adapted for situations with unequal sample sizes across groups. This makes it highly suitable for post hoc analysis following ANOVA, as it maintains control over the family-wise error rate while accommodating the variability in sample sizes. This ensures that the comparisons of group means are robust and reliable, even when the sample sizes differ, thus enhancing the validity of the findings in parametric data contexts.

In contrast, the Kruskal-Wallis test is a non-parametric method that does not assume normality or homogeneity of variances, making it suitable for non-normally distributed data. When significant differences are found using Kruskal-Wallis, Dunn's (1964) procedure, used in conjunction with a Bonferroni adjustment, is ideal for post hoc comparisons following a Kruskal-Wallis test, which is a non-parametric method. The Bonferroni adjustment is conservative, reducing the risk of Type I errors (false positives), which is particularly important in non-parametric analyses where the data do not meet the stringent assumptions of parametric tests.

The use of Tukey's for ANOVA and Bonferroni for Kruskal-Wallis ensures appropriate, valid, and reliable post hoc analysis tailored to the nature of the data and the assumptions underlying each test.

4.4. Analysis – Logistic regression

Logistic regression is a statistical model extensively utilised in analysing binary outcomes, aiming to predict the probability of an event occurring or not. In the context of this thesis, logistic regression proves valuable in determining whether individuals exhibit IWB or not. The name 'logistic' regression originates from its utilisation of the logistic function, which effectively models the relationship between input variables and binary outputs. The application of logistic regression holds significant prominence in social science research and continues to gain traction in the field (King, 2008; Peng & So, 2002). Notably, numerous studies investigating innovation and innovative behaviour have

adopted logistic regression as their primary analytical method. For instance, Dwivedi and Pawsey (2023) delved into the drivers of marketing innovation in SMEs, Divisekera and Nguyen (2018) investigated determinants of innovation in the Australian tourism context, and Anand and Lokesh (2018) explored how the innovativeness in consumers accelerated innovative behaviour.

Logistic regression is useful in this study for several reasons:

- a) In social science research, many variables of interest are categorical, such as binary outcomes (e.g. yes/no, success/failure) or multinomial outcomes (e.g. low/medium/high). Logistic regression is specifically designed to model such categorical outcomes, making it well-suited for analysing data in social science studies.
- b) Logistic regression allows for estimating probabilities and odds ratios, which are valuable for understanding the relationships between predictor variables and the likelihood of specific outcomes occurring, such as whether remote employees demonstrate IWBs. This enables researchers to assess the impact of independent variables on the likelihood of an event or behaviour.
- c) Social science research frequently involves multiple predictor variables that may interact with each other or have independent effects on the outcome. Logistic regression facilitates the inclusion of multiple predictors in a single model, allowing for examining their independent and joint effects.
- d) Logistic regression provides interpretable coefficients and odds ratios, which can be easily understood and communicated. Researchers can assess the direction and magnitude of the relationship between predictors and the outcome, providing insights into the strength and significance of the associations.

In this thesis, logistic regression was used to analyse the effect of remote working frequency on the likelihood or probability of IWB occurring. The results of the logistic regression analysis are found in Chapter 6.6.

4.5. Chapter summary

This chapter provided an overview of the quantitative study method. It discussed the research design, data collection, and statistical analysis procedures of ANOVA and Logistic Regression. During the quantitative data collection, challenges were faced in getting firms onboarded to distribute the survey internally to their employees. To overcome this, the scope of the study had to be broadened and new ways to recruit participants were undertaken. However, the COVID-19 pandemic presented an opportunity as remote working became significantly more common and, for many employees, mandatory. This made the research topic highly relevant and timely.

The next chapter discusses the qualitative study approach and methods, while the quantitative study results are discussed in Chapter 6.

CHAPTER 5 - THE QUALITATIVE RESEARCH METHOD

5.1. Introduction

This chapter will discuss the research methods employed for the qualitative study. Selecting appropriate research methods is paramount in ensuring any academic investigation's rigour, validity, and reliability. In qualitative research, the emphasis lies on understanding and interpreting the complex phenomena under scrutiny, often within their natural settings, through a holistic and nuanced lens.

For the qualitative data collection, semi-structured interviews were the chosen approach. Collecting data for the qualitative side of this study was significantly challenged by the COVID-19 pandemic. Initially, not enough participants from the survey volunteered to take part in the interviews, necessitating a broadening of the recruitment scope. To address this, additional participants were sought through own professional networks and by promoting the study in professional closed groups on the social networking site LinkedIn.

Despite these challenges, the pandemic also provided a unique opportunity, as a greater number of people were currently working remotely, which enriched the pool of potential participants. Purposive sampling was critical in this data collection phase, as it allowed for the collection of detailed and relevant data from a targeted group of employees. By focusing on individuals with firsthand experience in remote working environments ensured the capture of rich, context-specific insights and the lived experiences of people, providing deep and nuanced insights into the impact of remote working on their creative and innovative behaviours.

The following sections provide an overview of the data collection methods employed in this study, the strategies and processes used for participant recruitment, and the steps involved in the data analysis process. The analysis of the qualitative study is discussed in Chapter 7.

5.2. Qualitative research methods

Qualitative research methods encompass techniques such as interviews, focus groups, ethnography, and case studies. Qualitative research holds significant value in the

social sciences due to its unique contributions to understanding complex social phenomena (Santana et al., 2021; Shelton et al., 2014). It enables a comprehensive approach to addressing the research objectives: to explore, understand and capture a range of perspectives, experiences, and themes.

Qualitative research provides rich and detailed insights into individuals' experiences, perceptions, and behaviours (Banyard & Miller, 1998; Creswell, 2007) and uncovering relationships. Qualitative research focuses on understanding the context and meaning of social phenomena, emphasising the importance of cultural, social, and historical contexts in shaping individuals' lives (Birkinshaw et al., 2011; Denzin & Lincoln, 2011) while being able to uncover new perspectives, concepts, and frameworks enhancing theoretical understanding in the social sciences (Charmaz, 2014).

For these reasons, qualitative methods were applied to understand from the employees perspective, the influence of remote working on their innovative behaviour.

5.2.1. Sampling method and sample size

The sampling method to select participants for the study was a non-probability sampling technique called 'purposeful' or 'purposive' sampling (Etikan et al., 2016). This is also known as judgmental, selective, or subjective sampling, where the units (in this case, employees) investigated are based on the researcher's judgment and is one of the most widely used sampling methods (Oppong, 2013; Palinkas et al., 2013; Robinson, 2014). This sampling method focuses on selecting individuals or cases with unique characteristics, experiences, or knowledge relevant to the research study (Creswell, 2007; Schindler, 2018), deliberately selecting participants based on their capacity to provide detailed insights (Patton, 2015; Robinson, 2014). This approach fits the objectives of the thesis by investigating employees with real experience and knowledge of utilising remote working practices.

The sample population for this study consisted of Australian-based employees who have experience working remotely who can provide insights into the challenges and benefits of remote work within the Australian context. Prior to conducting the interviews, it was confirmed via email that participants were currently working in Australia and had experience working away from the main office building. If they were not based in Australia or were working fully remotely, they were not interviewed for the study. Those Page 88
employees who worked fully remotely were not included as sub-research question 4 specifically investigates the lived experiences of employees comparing office-based to remote-based working.

A total of 18 interviews were conducted. The interviewing process continued until saturation was reached, where no new themes or ideas were emerging (Hennink & Kaiser, 2022). Reaching saturation refers to the point at which collecting additional data or sampling additional participants does not yield any new or meaningful information or insights related to the research topic (Mwita, 2022; L. Yang et al., 2022). It signifies that the researcher has gathered enough data to adequately understand and explore the phenomenon of interest and answer the research questions or achieve the research objectives. Further data collection is unlikely to contribute substantially to the overall understanding or analysis.

5.2.2. Participant recruitment

The recruitment of participants came from various sources. One source was the questionnaire during the quantitative data collection phase. At the end of the questionnaire, respondents were asked if they would be willing to participate in an interview. Those who selected 'yes' were directed to a new screen where they provided their email address. They received a confirmation email to reconfirm their willingness to participate. Five respondents were recruited from the questionnaires.

Since five interviews would be insufficient, additional participants were recruited. Invitations were posted in closed LinkedIn professional groups and distributed through the researcher's professional network to broaden participation. Refer to Table 5.1 for a summary of participants and the recruitment channel.

Prior to conducting the interviews, the researcher confirmed via email that participants were based in Australia. They were also emailed an information sheet about the study before the interview (refer to Appendix D). An electronic consent form was also required to be completed and signed before the interview (refer to Appendix E). If an employee agreed to participate, they were then emailed a Calendly.com scheduling link to schedule at an interview time that was convenient to them.

Each employee was assigned a code to ensure anonymity. The interviews were conducted between November 2021 and March 2023, using the Zoom communication platform, and lasted 40 minutes on average. All interviews were audio recorded, and auto transcription was used (with participant consent). The researcher also manually reviewed and cleaned each transcription.

Employee Code	Recruitment
EMP01	Questionnaire
EMP02	Questionnaire
EMP03	Questionnaire
EMP04	Questionnaire
EMP05	Questionnaire
EMP06	Network
EMP07	Network
EMP08	Professional group (LinkedIn)
EMP09	Network
EMP10	Professional group (LinkedIn)
EMP11	Network
EMP12	Professional group (LinkedIn)
EMP13	Professional group (LinkedIn)
EMP14	Professional group (LinkedIn)
EMP15	Professional group (LinkedIn)
EMP16	Professional group (LinkedIn)
EMP17	Professional group (LinkedIn)
EMP18	Professional group (LinkedIn)

 Table 5-1: Summary of participant recruitment

5.3. Data collection

'The interview is the primary data collection technique for gathering data in qualitative methodologies' (Schindler, 2018, p. 129).

In-depth semi-structured interviews were chosen as the data collection method as they enable 'a more accurate and clear picture of a respondents position or behaviour' (Ghauri, 2020, p. 117). Semi-structured interviews are an effective data collection instrument commonly used in qualitative research due to their flexibility, depth, and ability to capture rich and nuanced insights from participants (Adeoye - Olatunde & Olenik, 2021; Monteiro et al., 2016; Striukova & Rayna, 2015).

Semi-structured interviews allow researchers to have a predefined set of questions while also providing flexibility to explore unanticipated topics or delve deeper into participants' responses (Anderson & Holloway-Libell, 2014; Swain, 2018). This enables participants to clarify and elaborate their answers and allows the interviewer to ask probing questions or follow-up questions. By allowing participants to elaborate on their experiences, emotions, and thoughts, we can get a deeper understanding of the research topic (Brinkmann, 2013; Olson, 2016). Interviews also allow the researcher to gather information within the specific context of the participants in order to consider how this may influence their responses. Denzin and Lincoln (2011) explain how interviews can foster participant engagement and active involvement in the research process, as they can share their narratives, contribute insights, and shape the direction of the conversation.

5.3.1. Interview design

The interview protocol consisted of a set of questions designed to meet the aims and objectives of the study and expected to provide a useful dataset that addresses the research questions. Employees were asked questions to reflect on their experiences and behaviour when working remotely guided by the Ability, Motivation, Opportunity Framework (Appelbaum et al., 2000; Bailey, 1993) and the IWB four-phase model (De Jong & Den Hartog, 2010). Participants were asked to provide examples where possible to explain their response. The interviews were digitally recorded, and auto transcription of the audio was used to facilitate analysis. Each interview was transcribed verbatim, however any mentions of participant names or their organisation was later removed to ensure anonymity. Participants were assigned unique codes, from EMP1 to EMP18, for identification purposes. The interviews were designed to allow exploration of the following: (a) how employees perceive their IWB may be shaped by working remotely; (b) possible factors of remote working that foster and/or hinder their IWB; and their experience between remote and office-based working in terms of their IWB. The interview protocol consisted of 24 questions within three sections prepared in advance (see Appendix F), summarised in Table 5.2 below.

Table 5-2: Interview sec	tions and items
--------------------------	-----------------

Section	Heading	No. of	Responses
		items	
1	Participant background	2	Open-ended
2	Usage of remote working	3	Open-ended
3	Ability, Motivation, Opportunity factors	19	Open-ended

5.3.1.1. Section 1: Participant background

The questions in this section sought to obtain detailed information from employees about their work experience, including their background, current role, job level, and job responsibilities. The study specifically recruited participants with experience in remote working who were based in Australia. This purposive sampling approach aimed to gain a thorough understanding of the employees' professional backgrounds, providing context to their lived experiences of working both in the office and remotely. By collecting this background information, the study was able to contextualise the participants' insights and experiences, offering a view of how remote work may have impacted their creative and innovative behaviours. This enriched the data, allowing for a deeper analysis of the effects and implications of remote working practices.

5.3.1.2. Section 2: Usage of remote working

The questions in this section aimed to gain insights into the frequency with which employees worked remotely and their preferred locations for completing tasks outside the traditional office environment. Participants were asked to specify how often they engaged in remote work and to identify the locations they found most conducive to their productivity and creativity, such as home offices, coworking spaces, or other remote settings. Additionally, participants were requested to elaborate on the motivations driving their choice to work remotely. This approach sought to understand not only the logistical aspects of remote work but also the underlying reasons that influenced employees' decisions to opt for remote working arrangements.

5.3.1.3. Section 3: Ability, Motivation, Opportunity Factors

Section 3 of the interview protocol delved into the employee's perspectives, opinions, experiences, and perceptions regarding the impact of remote work on their innovative behaviour. The questions leveraged the phases of IWB (De Jong & Den Hartog, 2010) as a foundation to explore the typical activities associated with each phase. Moreover, the study sought to establish connections between these activities and the individual's ability, motivation, and opportunity to engage in such behaviours, using the AMO theoretical framework as a foundation (Appelbaum et al., 2000; Bailey, 1993). Table 5.3 provides the examples:

AMO framework	Example question(s)
Ability	When working remotely, how do you collaborate with your work colleagues? probe: how does the collaboration experience compare to when you are working in the office? What works/doesn't work?
Motivation	Where do you feel more 'creative'? E.g. In the office or working remotely? Why?
Opportunity	When you are not working in the office, are you hearing about new ideas from your workplace?probe: why/why not?Do you promote these new ideas to others or talk about them with others?Why/why not?

Table 5-3: Example questions from interview protocol

5.4. Qualitative data analysis method

A thematic analysis was used in this thesis to analyse the data collected. NVivo 14 software application was used to facilitate data analysis. Thematic analysis is 'a method for identifying, analysing and reporting patterns (themes) within data' (Braun & Clarke, 2012). It involves systematically coding and categorising qualitative data, such as interview transcripts, to uncover recurring themes that capture key aspects of the research topic. Thematic analysis aims to provide rich insights and understanding of phenomena by organising and interpreting qualitative data in a structured and meaningful way.

There are several benefits to using thematic analysis in this thesis. Firstly, thematic analysis offers flexibility in exploring and interpreting qualitative data. Unlike other analytic methods, thematic analysis is not bound to a specific theoretical or epistemological position (Maguire & Delahunt, 2017; Swain, 2018). This flexibility allows for in-depth exploration and interpretation of the data, capturing its complexity and diversity. For this study it enables the collection of rich and nuanced data about how remote working may shape employee experiences and innovative behaviour.

Secondly, thematic analysis is accessible to researchers across various disciplines and experience levels (Ozuem et al., 2022; Terry et al., 2017). It does not require specialised software or technical expertise, making it widely accessible and applicable in diverse research contexts. Importantly, it has been used in numerous studies exploring remote working (Deutrom et al., 2022; Henke et al., 2022; Newcomb & Venning, 2022) and innovation (Aldulaimi & Abdeldayem, 2020; Michalakopoulou et al., 2023; Santiago, 2023; Wipulanusat et al., 2019).

Thirdly, thematic analysis embraces an interpretive approach (Byrne, 2021; Peterson, 2017), focusing on the subjective meanings and perspectives of participants, a primary research objective of this thesis and alignment to the interpretivist research paradigm. It acknowledges the role of the researcher in constructing meaning from the data. This interpretive lens enables researchers to gain insights into participants' experiences, beliefs, and attitudes, enhancing the richness and depth of the analysis. Therefore, thematic analysis is not aimed at developing a theory but rather at

understanding a phenomenon by identifying, analysing, and interpreting themes found across the data.

Lastly, thematic analysis follows a systematic and rigorous process, ensuring the trustworthiness and credibility of the findings. Researchers engage in a careful and iterative process of coding, categorising, and analysing the data, facilitating a thorough dataset exploration (Naeem et al., 2023; Nowell et al., 2017).

5.4.1. Thematic analysis steps

Thematic analysis encompasses a set of steps designed to discern and delineate patterns of significance within qualitative datasets (Forbes, 2022; Nowell et al., 2017; Xu & Zammit, 2020). Figure 5.1 below outlines the key steps undertaken in this thesis to analyse the qualitative data obtained from the interviews (Braun & Clarke, 2006).





5.4.1.1. Data familiarisation

Familiarisation with the data involves immersing oneself deeply in the collected materials. This step included reading and re-reading transcripts to develop an understanding of the content, raw data and context. This part of the process is to identify patterns, key ideas, and initial impressions that may inform subsequent coding and thematic development.

5.4.1.2. Generate initial codes

The process of generating initial codes entails systematically labelling with short and meaningful labels segments of data that are relevant to the research objectives. Codes are typically descriptive, capturing specific concepts, actions, or phenomena present in the data. By coding the data, potential themes and patterns emerge from the raw data.

5.4.1.3. Search for themes

Searching for themes involves sorting and combining the initial codes to identify broader patterns, topics, relationships between different codes or recurring ideas within the dataset. Themes will emerge as clusters of codes are amalgamated into common concepts. This phase required a balance of deductive and inductive reasoning, as drawing on existing theoretical frameworks while remaining open to unexpected findings within the data. The goal was to uncover meaningful themes that encapsulate the dataset's central themes and provide insights into the research questions.

5.4.1.4. Review themes

Once potential themes have been identified, they were reviewed and refined through a thorough examination of the coded data and the dataset as a whole. Each theme was critically assessed to ensure it accurately reflected the dataset's content and captured relevant aspects of the research study.

5.4.1.5. *Define themes*

Defining and naming themes involves clearly articulating each identified theme in precise terms that capture its essence and distinguish it from other themes. This step involved revisiting the original transcripts and revising the thematic labels to accurately convey the underlying concepts. Clear definitions ensure the findings to be communicated effectively and facilitate comparisons with existing literature.

5.4.1.6. Write-up

The final step in thematic analysis involves synthesising the findings into a coherent narrative. The identified themes are supported by illustrative examples and quotations from the data. The findings are contextualised within the broader research context.

5.5. Chapter summary

This chapter provided an overview of the qualitative study method, including the interview design, data collection, and data analysis processes. Whilst recruiting participants for interviews posed challenges in the qualitative data collection phase; the COVID-19 pandemic inadvertently facilitated recruitment by increasing the number of employees with remote work experience. The next chapter discusses the results from the qualitative study, and Chapter 7 discusses the results from the qualitative study.

CHAPTER 6 - QUANTITATIVE DATA ANALYSIS

6.1. Introduction

The purpose of this chapter is to provide the results of the data analysis conducted in this study. It begins with an overview of the procedures and data preparation, which involves data coding and transformation, before providing the results of the descriptive statistics, ANOVA and logistic regression.

6.2. Data screening and transforming

A total of 431 questionnaires were thoroughly assessed for accuracy, missing values, and potential outliers to ensure the reliability of the data (Pallant, 2020). From the sample, questionnaires were omitted from analysis if respondents indicated 'never' in response to the inquiry concerning their frequency of remote work. This criterion was employed due to the study's focused examination of employee behaviour during remote work scenarios, necessitating active engagement in remote work to ensure coherence and relevance of the data under scrutiny.

Additionally, questionnaires were excluded from the dataset if they solely comprised demographic information without any accompanying responses to other questionnaire items (Allison, 2009; Newman, 2014). This practice of eliminating questionnaires with incomplete responses contributes to the refinement of data integrity (Jakobsen et al., 2017). Questionnaires that solely provide demographic information without engaging with the core research items fail to contribute meaningful data to address the study's objectives. Specifically, in the context of this study, the inclusion of responses related to IWB was imperative for conducting statistical analyses. Thus, questionnaires lacking such essential data were systematically omitted from further consideration (Mat Roni & Djajadikerta, 2021). However, questionnaires that contained some responses to the IWB questions were retained, as they could still provide valuable insights into the individual IWB phases.

Responses to the 'gender' question saw only eight respondents select 'Nonbinary' or 'Prefer not to say'. Removing data with very few responses to an item is essential for maintaining the statistical robustness and reliability of the analysis. When a dataset contains very small responses to a particular item, it can significantly skew the results and compromise the validity of any conclusions drawn from the analysis. This resulted in 324 usable responses.

Subsequently, the raw data underwent a process of transformation, resulting in the creation of additional variables while preserving the integrity of the original data. The analysis uses the transformed data rather than the untransformed because many statistical techniques such as regression and analysis of variance require that data follow a distribution of a particular kind (Bland & Altman, 1996; Fink, 2009; In & Lee, 2017).

The transformation process included:

- Recoding data entered as text (e.g. gender) to numeric data (e.g. 1, 2)
- Reducing the number of categories of a categorical variable (e.g. tenure) to decrease the number of cases where there was low frequency in distributions (Table 6.1)
- Adding up the scores from the items that made up the dependent variables (e.g. IWB) to give an overall continuous scale for IWB and each of the four phases (Table 6.2)
- Averaging the scores from the items that made up the dependent variables (e.g. IWB) to give an overall continuous scale for IWB and each of the four phases (Table 6.3)
- Using the average scores from the items that made up the dependent variables (e.g. IWB in Table 6.4) and assigning those cases where the average was equal to or greater than 4 as 'Yes' and assigning all other cases 'No' to create a binary dependent variable for IWB and each of the four phases (Table 6.4).

Table 6-1: Data categories following transformation

Variable	Categories before	Categories after
	transformation	transformation
RWF	'Once per month', 'Twice per	'1–3 times per month', '1 day
	month', '3 times per month', '1	per week', '2–4 days per week'
	day per week', '2–4 days per	and 'Always work remote'
	week' and 'Always work	
	remote'	
Gender	'Female', 'Male', 'Non-binary'	'Female', 'Male' and
	and 'Prefer not to say'	'Unknown'
Age	'18–30 years', '31–40 years',	'18-30 years', '31-40 years',
	'41–50 years', '51–60 years' and	'41–50 years' and '51+ years'
	'61+ years'	
Employment	'Full-time', 'Part-time',	'Full-time', 'Part-time' and
Status	'Casual/contract' and	'Casual/Contract'
	'Temporary/project-based'	
Tenure	'Less than 2 years', 'Between 2–	'Less than 2 years', 'Between 2–
	5 years', '6–10 years', and	5 years' and 'More than 6 years'
	'More than 10 years'	
Employment	'Administration', 'Non	'Administration', 'Non
Level	manager', 'Manager without	manager', 'Manager' and
	direct reports', 'Manager with	'Senior Leadership'
	direct reports', 'Senior	
	Executive/Senior Leadership',	
	and 'GM/CEO/Business owner'	

Table 6-2: Continuous dependent variables (sum)

Dependent variable	Question #	New continuous variable
	Σ	(sum)
IWB	8–34	IWB_scale
Idea Exploration	8–17	Explore_scale
Idea Generation	18–24	Gen_scale
Idea Championing	25–29	Champ_scale
Idea Implementation	30–34	Imp_scale

Table 6-3: Continuous dependent variables (avg)

Dependent variable	Question #	New continuous variable
	μ	(***'8/
IWB	8–34	IWB_avg
Idea Exploration	8–17	Explore_avg
Idea Generation	18–24	Gen_avg
Idea Championing	25–29	Champ_avg
Idea Implementation	30–34	Imp_avg

Table 6-4: Binary dependent variables

Dependent variable	Question #	New variable	New binary variable
variable	µ ≈		If ≤ 3.99 = No
	(rounded up to 0 decimal places)		If ≥ 4 = Yes
IWB	8–34	IWB_rounded	IWB_YN
Idea Exploration	8–17	Explore_rounded	Explore_YN
Idea Generation	18–24	Gen_rounded	Gen_YN
Idea Championing	25–29	Champ_rounded	Champ_YN
Idea Implementation	30–34	Imp_rounded	Imp_YN

6.3. Descriptive statistics

The respondent demographics are presented in Table 6.5. The analysis showed that almost half, 46.3% of respondents worked remotely 2–4 days per week, and 21.6% worked remotely all their working time. The majority of respondents were aged between 18 and 40 years (77.2%), and the majority worked full-time (63.9%). Of the 324 respondents, 47.8% identified as female and 52.2% identified as male. Regarding time at their current organisation, 40.7% indicated they had worked there for less than two years, 34.3% had from two to five years, and 25% had more than six years. In terms of employment level, the majority classified themselves as working in a non-manager role

(51.9%), while 27.2% classified themselves as managers. A further 12.7% worked in administrative roles while 8.3% were in senior leadership.

Profile	Employee details (n=324)	Ν	%
Gender	Female	155	47.8%
	Male	169	52.2%
Age	18–30 years	158	48.8%
	31–40 years	92	28.4%
	41–50 years	42	13.0%
	51+ years	32	9.9%
Employment status	Full-time	207	63.9%
	Part-time	61	18.8%
	Casual/Contract	56	17.3%
Tenure	Less than 2 years	132	40.7%
	2–5 years	111	34.3%
	More than 6 years	81	25.0%
Employment level	Administration	41	12.7%
	Not a manager	168	51.9%
	Manager	88	27.2%
	Senior Leadership	27	8.3%
Frequency of working	1–3 times per month	41	12.7%
Temotery	1 day per week	63	19.4%
	2–4 days per week	150	46.3%
	Always work remotely	70	21.6%

Table 6-5: Respondent demographics

6.3.1. Remote working frequency (RWF)

Of the total sample, almost half, 46.3% worked remotely 2–4 times per week. A further 21.6% always worked remotely (Table 6.6).

Frequency	Ν	%
1–3 times per month	41	12.7%
Once a week	63	19.4%
2–4 times per week	150	46.3%
Always	70	21.6%
	324	100.0%

Table 6-6: Overall frequency of working remotely

6.3.1.1. *RWF by age*

The category of 'working remotely 2–4 times per week' emerged as the most frequent selection across all age groups within the sample (Table 6.7). Individuals aged 41 years and above exhibited a higher inclination to 'always' work remotely compared to the other age groups. Similarly, half of those aged 51+ years reported working remotely 2–4 times per week.

Table 6-7: Cross tabulation of RWF by age

Age * How often working remotely Cross Tabulation

]	How often work	cing remotely		
		•	1–3 times per		2–4 times per		-
			month	Once a week	week	Always	Total
Age	18–30	Count	25	30	73	30	158
		% within Age	15.8%	19.0%	46.2%	19.0%	100.0%
	31-40	Count	10	19	44	19	92
		% within Age	10.9%	20.7%	47.8%	20.7%	100.0%
	41–50	Count	3	9	17	13	42
		% within Age	7.1%	21.4%	40.5%	31.0%	100.0%
	51+	Count	3	5	16	8	32
		% within Age	9.4%	15.6%	50.0%	25.0%	100.0%
Total		Count	41	63	150	70	324
		% within Age	12.7%	19.4%	46.3%	21.6%	100.0%

6.3.1.2. RWF remotely by gender

In the sample, both genders predominantly engaged in remote work '2–4 times per week' (Table 6.8). Females had a higher proportion of 'always' working remotely compared to males.

Table 6-8: Cross tabulation of RWF by gender

		_	How often working remotely				
			1–3 times per	Once a	2–4 times per		
			month	week	week	Always	Total
Gender	Female	Count	22	25	65	43	155
		% within Gender	14.2%	16.1%	41.9%	27.7%	100.0%
	Male	Count	19	38	85	27	169
		% within Gender	11.2%	22.5%	50.3%	16.0%	100.0%
Total		Count	41	63	150	70	324
		% within Gender	12.7%	19.4%	46.3%	21.6%	100.0%

Gender * How often working remotely Cross Tabulation

6.3.1.3. RWF by employment status

The majority of Contractors/Casual workers sampled are working remotely all of the time, compared to the other employment categories (Table 6.9). Among full-time workers, there is a greater tendency to engage in remote work '2–4 times per week', while part-time workers exhibit an equal likelihood of working remotely '2–4 times per week' and 'always'.

Table 6-9: Cross tabulation of RWF by employment status

]	How often working remotely				
			1–3 times per	•	2–4 times per		-	
			month	Once a week	week	Always	Total	
ES	Full-time	Count	22	41	119	25	207	
		% within ES	10.6%	19.8%	57.5%	12.1%	100.0%	
	Part-time	Count	10	15	18	18	61	
		% within ES	16.4%	24.6%	29.5%	29.5%	100.0%	
	Contract/Casua	l Count	9	7	13	27	56	
		% within ES	16.1%	12.5%	23.2%	48.2%	100.0%	
Total		Count	41	63	150	70	324	
		% within ES	12.7%	19.4%	46.3%	21.6%	100.0%	

ES * RWF Cross Tabulation

6.3.1.4. *RWF by tenure*

In the sample, all tenure groups predominantly engaged in remote work '2–4 times per week' (Table 6.10). Those with less than two years of tenure at their current organisations had the highest proportion of 'always' working remotely compared to other group proportions.

Table 6-10: Cross tabulation of RWF by tenure

Tenure * RWF Cross Tabulation

			Ho	w often wo	rking remotely			
		-	1–3 times per	Once a	2–4 times per		-	
			month	week	week	Always	Total	
Tenure	Less than 2	Count	18	28	52	34	132	
	years	% within Tenure	13.6%	21.2%	39.4%	25.8%	100.0%	
	2-5 years	Count	15	22	56	18	111	
		% within Tenure	13.5%	19.8%	50.5%	16.2%	100.0%	
	6+ years	Count	8	13	42	18	81	
		% within Tenure	9.9%	16.0%	51.9%	22.2%	100.0%	
Total		Count	41	63	150	70	324	
		% within Tenure	12.7%	19.4%	46.3%	21.6%	100.0%	

6.3.1.5. *RWF by employment level*

In the sample, both administration roles and senior level roles predominantly 'always' worked remotely, while managers and non-managers worked remotely '2–4 times per week' (Table 6.11).

Table 6-11: Cross tabulation of RWF by employment level

			How	often worl	king remotely		
			1-3 times per	Once a	2–4 times per		
			month	week	week	Always	Total
EL	Administration	Count	3	6	15	17	41
		% within EL	7.3%	14.6%	36.6%	41.5%	100.0%
	Not a manager	Count	23	30	78	37	168
		% within EL	13.7%	17.9%	46.4%	22.0%	100.0%
	Manager	Count	12	20	49	7	88
		% within EL	13.6%	22.7%	55.7%	8.0%	100.0%
	Senior	Count	3	7	8	9	27
	Leadership/CEO	% within EL	11.1%	25.9%	29.6%	33.3%	100.0%
Total		Count	41	63	150	70	324
		% within EL	12.7%	19.4%	46.3%	21.6%	100.0%

EL * RWF Cross Tabulation

6.3.1.6. Summary of RWF

Overall, the categories within the sample demonstrating the highest prevalence of remote work (i.e. always working remotely) were identified as follows: the 41–50 age group, contract/casual working females, with less than two years of tenure at their current organisation, working in administration positions.

The categories within the sample demonstrating a consistently high prevalence of remote work ('2–4 times per week'), were identified as follows: the 51+ age group, full-time working males, with 6+ years tenure, working in managerial positions.

6.3.2. Innovative Work Behaviour

As discussed in Chapter 3, the respondents were asked to answer how often they undertook a particular activity within each of the four phases of IWB (De Jong & Den Hartog, 2010), choosing from 1 = never, to 5 = always.

The standard deviation measures the size of the variation or spread around the mean of a continuous variable, indicating how much individual data points deviate from the mean value of the dataset. The larger the standard deviation, the more variation around the mean there is, the more spread out they are (Thrane, 2023). A low standard deviation indicates data observations very close to the mean (Albers, 2017). The standard deviation of 0.66 for the idea exploration phase indicates relatively low variability within the data set, the data points are relatively close to the mean. This suggests that most employees' frequency of undertaking idea exploration activities is very close to the average of idea exploration frequency. Conversely, the standard deviation for idea generation, championing and implementation suggests a moderate level of variability around the mean. While not extremely high, the values tend to spread out somewhat from the average value (Table 6.12).

The idea exploration and generation phases had higher means in comparison to the other phases and sit closer to the boundary between 'sometimes -2 to 3 times per month' and 'often -4 to 8 times per month'.

Variable	Mean	Standard deviation
IWB	3.30	0.704
Idea Exploration	3.39	0.657
Idea Generation	3.47	0.764
Idea Championing	3.01	0.904
Idea Implementation	3.25	0.889

Table 6-12: Means and standard deviation

Across all phases, with the exception of the idea generation phase, the majority of respondents provided responses categorised as 'sometimes'. However, in the idea

generation phase, there was an almost equal split, with 42.9% of respondents choosing 'sometimes' and 42.3% selecting 'often' (Table 6.13). The idea generation phase received the highest number of responses indicating 'always' (6.8%) compared to the other phases, with the idea implementation phase ranking second with 5.2% of respondents choosing 'always'. The idea generation phase had more responses of 'often' to 'always' than any other phase (49.1%).

	IWB	Idea Exploration	Idea Generation	Idea Championing	Idea Implementation
Never	0.6%	0.3%	0.9%	6.2%	3.7%
Rarely	9.9%	5.9%	7.1%	17.9%	13.6%
Sometimes	50.9%	50.9%	42.9%	48.1%	41.7%
Often	36.1%	40.1%	42.3%	24.1%	35.5%
Always	2.5%	2.8%	6.8%	3.7%	5.2%
missing data					0.3%

Table 6-13: IWB frequency table - percentages of rounded averages

Among all the phases, the idea championing phase had the highest percentage of respondents answering 'never' (6.2%) in comparison to the other phases, closely followed by the idea implementation phase (3.7%). The idea championing phase had more responses of 'never' to 'rarely' than any other phase (24.1%).

The findings regarding IWB indicate that approximately 51% of the respondents reported they sometimes undertake IWB activities. Moreover, 38.6% reported frequent to consistent engagement in such activities, reporting they often to always undertake IWB activities.

6.3.2.1. IWB by age

In all sampled age groups, with the exception of individuals aged 51+ years and above, the most frequently reported frequency for engaging in IWB was 'sometimes', followed by 'often' (Table 6.14). Those aged 51+ were equally as likely to report 'sometimes' and 'often', while 18–30-year-olds had the highest proportion of 'always' exhibiting IWB.

Table 6-14: Frequency of IWB by age

		_						
			Never	Rarely	Sometimes	Often	Always	Total
Age	18–30	Count	2	14	83	52	7	158
		% within Age	1.3%	8.9%	52.5%	32.9%	4.4%	100.0%
	31-40	Count	0	6	51	35	0	92
		% within Age	0.0%	6.5%	55.4%	38.0%	0.0%	100.0%
	41–50	Count	0	7	18	17	0	42
		% within Age	0.0%	16.7%	42.9%	40.5%	0.0%	100.0%
	51+	Count	0	5	13	13	1	32
		% within Age	0.0%	15.6%	40.6%	40.6%	3.1%	100.0%
Total		Count	2	32	165	117	8	324
		% within Age	0.6%	9.9%	50.9%	36.1%	2.5%	100.0%

Age * IWB round Cross Tabulation

Upon closer examination of the four distinct phases, some variations become apparent. In the first two phases, Idea Exploration and Idea Generation, employees aged 41–50 predominantly indicated 'often' as their response (Tables 6.15 and 6.16). In contrast, during the latter two phases, Idea championing and Idea Implementation phases, 41–50-year-olds predominantly indicated 'sometimes' (Tables 6.17 and 6.18).

Table 6-15:	Frequency	of Idea	exploration	by	age
-------------	-----------	---------	-------------	----	-----

			Never	Rarely	Sometimes	Often	Always	Total
Age	18–30	Count	1	9	85	56	7	158
		% within Age	0.6%	5.7%	53.8%	35.4%	4.4%	100.0%
	31-40	Count	0	3	48	40	1	92
		% within Age	0.0%	3.3%	52.2%	43.5%	1.1%	100.0%
	41–50	Count	0	5	17	20	0	42
		% within Age	0.0%	11.9%	40.5%	47.6%	0.0%	100.0%
	51+	Count	0	2	15	14	1	32
		% within Age	0.0%	6.3%	46.9%	43.8%	3.1%	100.0%
Total		Count	1	19	165	130	9	324
		% within Age	0.3%	5.9%	50.9%	40.1%	2.8%	100.0%

Age * Explore_round Cross Tabulation

Table 6-16: Frequency of Idea generation by age

			Never	Rarely	Sometimes	Often	Always	Total
Age	18–30	Count	2	8	70	63	15	158
		% within Age	1.3%	5.1%	44.3%	39.9%	9.5%	100.0%
	31-40	Count	0	5	43	41	3	92
		% within Age	0.0%	5.4%	46.7%	44.6%	3.3%	100.0%
	41–50	Count	0	6	14	19	3	42
		% within Age	0.0%	14.3%	33.3%	45.2%	7.1%	100.0%
	51+	Count	1	4	12	14	1	32
		% within Age	3.1%	12.5%	37.5%	43.8%	3.1%	100.0%
Total		Count	3	23	139	137	22	324
		% within Age	0.9%	7.1%	42.9%	42.3%	6.8%	100.0%

Age * Gen_rounded Cross Tabulation

Table 6-17: Frequency of Idea championing by age

Age * Cham	o rounded	Cross	Tabul	lation
------------	-----------	-------	-------	--------

			Champ_rounded					-
			Never	Rarely	Sometimes	Often	Always	Total
Age	18–30	Count	6	29	73	40	10	158
		% within Age	3.8%	18.4%	46.2%	25.3%	6.3%	100.0%
	31-40	Count	7	12	47	25	1	92
		% within Age	7.6%	13.0%	51.1%	27.2%	1.1%	100.0%
	41–50	Count	2	12	23	5	0	42
		% within Age	4.8%	28.6%	54.8%	11.9%	0.0%	100.0%
	51+	Count	5	5	13	8	1	32
		% within Age	15.6%	15.6%	40.6%	25.0%	3.1%	100.0%
Total		Count	20	58	156	78	12	324
		% within Age	6.2%	17.9%	48.1%	24.1%	3.7%	100.0%

Table 6-18: Frequency of Idea implementation by age

		-	Imp_rounded					_
			Never	Rarely	Sometimes	Often	Always	Total
Age	18–30	Count	4	26	63	54	11	158
		% within Age	2.5%	16.5%	39.9%	34.2%	7.0%	100.0%
	31-40	Count	4	7	38	39	4	92
		% within Age	4.3%	7.6%	41.3%	42.4%	4.3%	100.0%
	41–50	Count	1	7	22	10	1	41
		% within Age	2.4%	17.1%	53.7%	24.4%	2.4%	100.0%
	51+	Count	3	4	12	12	1	32
		% within Age	9.4%	12.5%	37.5%	37.5%	3.1%	100.0%
Total		Count	12	44	135	115	17	323
		% within Age	3.7%	13.6%	41.8%	35.6%	5.3%	100.0%

Age * Imp_rounded Cross Tabulation

6.3.2.2. IWB by gender

Across IWB overall, the most reported frequency was 'sometimes', followed by 'often' (Table 6.19). However, a slight contrast emerged between the genders, with males exhibiting a higher proportion of 'often' during the Idea Generation and Idea Implementation phases (Tables 6.21 and 6.23).

More males selected 'always' during the first two phases (Idea Exploration and Idea Generation), while more females selected 'always' during the latter two phases (Idea Championing and Idea Implementation).

Table 6-19: Frequency of IWB by gender

					IWB_round			
			Never	Rarely	Sometimes	Often	Always	Total
Gender	Female	Count	0	13	88	49	5	155
		% within	0.0%	8.4%	56.8%	31.6%	3.2%	100.0%
		Gender						
	Male	Count	2	19	77	68	3	169
		% within	1.2%	11.2%	45.6%	40.2%	1.8%	100.0%
		Gender						
Total		Count	2	32	165	117	8	324
		% within	0.6%	9.9%	50.9%	36.1%	2.5%	100.0%
		Gender						

Gender * IWB_round Cross Tabulation

Table 6-20: Rate of Idea exploration by gender

Gender * Explore_round Cross Tabulation

					Explore_round			
			Never	Rarely	Sometimes	Often	Always	Total
Gender	Female	Count	0	4	87	60	4	155
		% within	0.0%	2.6%	56.1%	38.7%	2.6%	100.0%
		Gender						
	Male	Count	1	15	78	70	5	169
		% within	0.6%	8.9%	46.2%	41.4%	3.0%	100.0%
		Gender						
Total		Count	1	19	165	130	9	324
		% within	0.3%	5.9%	50.9%	40.1%	2.8%	100.0%
		Gender						

Table 6-21: Frequency of Idea generation by gender

					Gen_rounded			_
			Never	Rarely	Sometimes	Often	Always	Total
Gender	Female	Count	1	9	72	64	9	155
		% within	0.6%	5.8%	46.5%	41.3%	5.8%	100.0%
		Gender						
	Male	Count	2	14	67	73	13	169
		% within	1.2%	8.3%	39.6%	43.2%	7.7%	100.0%
		Gender						
Total		Count	3	23	139	137	22	324
		% within	0.9%	7.1%	42.9%	42.3%	6.8%	100.0%
		Gender						

Gender * Gen_rounded Cross Tabulation

Table 6-22: Frequency of Idea championing by gender

Gender * Champ rou	ided Cross Tabulation
--------------------	-----------------------

				(Champ_rounde	d		
			Never	Rarely	Sometimes	Often	Always	Total
Gender	Female	Count	9	32	74	34	6	155
		% within	5.8%	20.6%	47.7%	21.9%	3.9%	100.0%
		Gender						
	Male	Count	11	26	82	44	6	169
		% within	6.5%	15.4%	48.5%	26.0%	3.6%	100.0%
		Gender						
Total		Count	20	58	156	78	12	324
		% within	6.2%	17.9%	48.1%	24.1%	3.7%	100.0%
		Gender						

Table 6-23: Frequency of Idea implementation by gender

					Imp_rounded			
			Never	Rarely	Sometimes	Often	Always	Total
Gender	Female	Count	5	22	73	44	10	154
		% within	3.2%	14.3%	47.4%	28.6%	6.5%	100.0%
		Gender						
	Male	Count	7	22	62	71	7	169
		% within	4.1%	13.0%	36.7%	42.0%	4.1%	100.0%
		Gender						
Total		Count	12	44	135	115	17	323
		% within	3.7%	13.6%	41.8%	35.6%	5.3%	100.0%
		Gender						

Gender * Imp rounded Cross Tabulation

6.3.2.3. IWB by employment status

Among the sample participants, more than half of both part-time employees and contract/casual workers selected 'sometimes' (Table 6.24). However full-time workers had the largest proportion with 'often' of all employment types.

Table 6-24: Rate of IWB by employment status

		-			IWB_round			
			Never	Rarely	Sometimes	Often	Always	Total
ES	Full-time	Count	0	20	93	89	5	207
		% within ES	0.0%	9.7%	44.9%	43.0%	2.4%	100.0%
	Part-time	Count	1	7	41	11	1	61
		% within ES	1.6%	11.5%	67.2%	18.0%	1.6%	100.0%
	Contract/	Count	1	5	31	17	2	56
	Casual	% within ES	1.8%	8.9%	55.4%	30.4%	3.6%	100.0%
Total		Count	2	32	165	117	8	324
		% within ES	0.6%	9.9%	50.9%	36.1%	2.5%	100.0%

ES * IWB round Cross Tabulation

Review of the four phases of IWB show that full time employees are more likely to 'often' undertake IWB during the Idea generation phase (Table 6.26) and equally as likely in the Idea implementation phase (Table 6.28).

Table 6-25: Rate of Idea exploration by employment status

					Explore_round	1		
			Never	Rarely	Sometimes	Often	Always	Total
ES	Full-time	Count	0	9	99	92	7	207
		% within ES	0.0%	4.3%	47.8%	44.4%	3.4%	100.0%
	Part-time	Count	0	6	38	16	1	61
		% within ES	0.0%	9.8%	62.3%	26.2%	1.6%	100.0%
	Contract/Casual	Count	1	4	28	22	1	56
		% within ES	1.8%	7.1%	50.0%	39.3%	1.8%	100.0%
Total		Count	1	19	165	130	9	324
		% within ES	0.3%	5.9%	50.9%	40.1%	2.8%	100.0%

ES * Explore_round Cross Tabulation

Table 6-26: Rate of Idea generation by employment status

ES * Gen_rounded Cross Tabulation

		_		Gen rounded						
			Never	Rarely	Sometimes	Often	Always	Total		
ES	Full-time	Count	0	14	83	94	16	207		
		% within ES	0.0%	6.8%	40.1%	45.4%	7.7%	100.0%		
	Part-time	Count	2	7	36	15	1	61		
		% within ES	3.3%	11.5%	59.0%	24.6%	1.6%	100.0%		
	Contract/Casual	Count	1	2	20	28	5	56		
		% within ES	1.8%	3.6%	35.7%	50.0%	8.9%	100.0%		
Total		Count	3	23	139	137	22	324		
		% within ES	0.9%	7.1%	42.9%	42.3%	6.8%	100.0%		

Table 6-27: Rate of Idea championing by employment status

		-		С	hamp_rounde	d		_
			Never	Rarely	Sometimes	Often	Always	Total
ES	Full-time	Count	10	30	97	62	8	207
		% within ES	4.8%	14.5%	46.9%	30.0%	3.9%	100.0%
	Part-time	Count	6	13	34	6	2	61
		% within ES	9.8%	21.3%	55.7%	9.8%	3.3%	100.0%
	Contract/Casual	Count	4	15	25	10	2	56
		% within ES	7.1%	26.8%	44.6%	17.9%	3.6%	100.0%
Total		Count	20	58	156	78	12	324
		% within ES	6.2%	17.9%	48.1%	24.1%	3.7%	100.0%

ES * Champ rounded Cross Tabulation

Table 6-28: Rate of Idea implementation by employment status

ES * Imp_rounded Cross Tabulation

		-		Imp rounded						
			Never	Rarely	Sometimes	Often	Always	Total		
ES	Full-time	Count	5	25	83	83	11	207		
		% within ES	2.4%	12.1%	40.1%	40.1%	5.3%	100.0%		
	Part-time	Count	4	12	27	15	2	60		
		% within ES	6.7%	20.0%	45.0%	25.0%	3.3%	100.0%		
	Contract/Casual	Count	3	7	25	17	4	56		
		% within ES	5.4%	12.5%	44.6%	30.4%	7.1%	100.0%		
Total		Count	12	44	135	115	17	323		
		% within ES	3.7%	13.6%	41.8%	35.6%	5.3%	100.0%		

6.3.2.4. IWB by tenure

Employees with six or more years of tenure at their current organisation 'often' exhibited IWB, while those with five or fewer years selected 'sometimes' (Table 6.29). This declined slightly during the idea championing phase, where most of those with six or more years tenure engaged in IWB activities 'sometimes' (Table 6.32).

Table 6-29: Rate of IWB by tenure

		-			IWB_round			
			Never	Rarely	Sometimes	Often	Always	Total
Tenure	Less than 2	Count	0	13	81	34	4	132
	years	% within Tenure	0.0%	9.8%	61.4%	25.8%	3.0%	100.0%
	2-5 years	Count	0	9	55	44	3	111
		% within Tenure	0.0%	8.1%	49.5%	39.6%	2.7%	100.0%
	6+ years	Count	2	10	29	39	1	81
		% within Tenure	2.5%	12.3%	35.8%	48.1%	1.2%	100.0%
Total		Count	2	32	165	117	8	324
		% within Tenure	0.6%	9.9%	50.9%	36.1%	2.5%	100.0%

Tenure * IWB_round Cross Tabulation

Table 6-30: Rate of Idea exploration by tenure

Tenure * Explore round Cross Tabulation

				Explore_round					
			Never	Rarely	Sometimes	Often	Always	Total	
Tenure	Less than 2	Count	0	9	76	43	4	132	
	years	% within Tenure	0.0%	6.8%	57.6%	32.6%	3.0%	100.0%	
	2–5 years	Count	0	3	57	48	3	111	
		% within Tenure	0.0%	2.7%	51.4%	43.2%	2.7%	100.0%	
	6+ years	Count	1	7	32	39	2	81	
		% within Tenure	1.2%	8.6%	39.5%	48.1%	2.5%	100.0%	
Total		Count	1	19	165	130	9	324	
		% within Tenure	0.3%	5.9%	50.9%	40.1%	2.8%	100.0%	

Table 6-31: Rate of Idea generation by tenure

				Gen_rounded					
			Never	Rarely	Sometimes	Often	Always		
Tenure	Less than 2	Count	1	8	63	51	9	132	
	years	% within Tenure	0.8%	6.1%	47.7%	38.6%	6.8%	100.0%	
	2-5 years	Count	0	7	51	44	9	111	
		% within Tenure	0.0%	6.3%	45.9%	39.6%	8.1%	100.0%	
	6+ years	Count	2	8	25	42	4	81	
		% within Tenure	2.5%	9.9%	30.9%	51.9%	4.9%	100.0%	
Total		Count	3	23	139	137	22	324	
		% within Tenure	0.9%	7.1%	42.9%	42.3%	6.8%	100.0%	

Tenure * Gen_rounded Cross Tabulation

Table 6-32: Rate of Idea championing by tenure

Tenure * Champ rounded Cross Tabulation

					Champ_round	led		
			Never	Rarely	Sometimes	Often	Always	Total
Tenure	Less than 2	Count	9	28	66	24	5	132
	years	% within Tenure	6.8%	21.2%	50.0%	18.2%	3.8%	100.0%
	2-5 years	Count	3	16	59	28	5	111
		% within Tenure	2.7%	14.4%	53.2%	25.2%	4.5%	100.0%
	6+ years	Count	8	14	31	26	2	81
		% within Tenure	9.9%	17.3%	38.3%	32.1%	2.5%	100.0%
Total		Count	20	58	156	78	12	324
		% within Tenure	6.2%	17.9%	48.1%	24.1%	3.7%	100.0%

Table 6-33: Rate of Idea implementation by tenure

				Imp_rounded					
			Never	Rarely	Sometimes	Often	Always	Total	
Tenure	Less than 2	Count	4	20	59	42	7	132	
	years	% within Tenure	3.0%	15.2%	44.7%	31.8%	5.3%	100.0%	
	2-5 years	Count	2	14	47	43	5	111	
		% within Tenure	1.8%	12.6%	42.3%	38.7%	4.5%	100.0%	
	6+ years	Count	6	10	29	30	5	80	
		% within Tenure	7.5%	12.5%	36.3%	37.5%	6.3%	100.0%	
Total		Count	12	44	135	115	17	323	
		% within Tenure	3.7%	13.6%	41.8%	35.6%	5.3%	100.0%	

Tenure * *Imp_rounded Cross Tabulation*

6.3.2.5. *IWB by employment level*

Across all employment levels except 'managers', the predominant frequency of IWB exhibited was 'sometimes' (Table 6.34). Managers had the highest proportion of 'always'. However, during Idea Generation, the majority of Senior Leader/CEOs reported they 'always' exhibit IWB (Table 6.36).

Table 6-34: Rate of IWB by employment level

					Total			
			Never	Rarely	Sometimes	Often	Always	
EL	Administration	Count	0	7	23	10	1	41
		% within EL	0.0%	17.1%	56.1%	24.4%	2.4%	100.0%
	Not a manager	Count	2	13	97	54	2	168
		% within EL	1.2%	7.7%	57.7%	32.1%	1.2%	100.0%
	Manager	Count	0	8	33	43	4	88
		% within EL	0.0%	9.1%	37.5%	48.9%	4.5%	100.0%
	Senior Leader/CEO	Count	0	4	12	10	1	27
		% within EL	0.0%	14.8%	44.4%	37.0%	3.7%	100.0%
Total		Count	2	32	165	117	8	324
		% within EL	0.6%	9.9%	50.9%	36.1%	2.5%	100.0%

EL * *IWB_round Cross Tabulation*

Table 6-35: Rate of Idea exploration by employment level

		-		Explore_round						
			Never	Rarely	Sometimes	Often	Always	Total		
EL	Administration	Count	0	2	25	13	1	41		
		% within EL	0.0%	4.9%	61.0%	31.7%	2.4%	100.0%		
	Not a manager	Count	1	10	86	69	2	168		
		% within EL	0.6%	6.0%	51.2%	41.1%	1.2%	100.0%		
	Manager	Count	0	4	42	39	3	88		
		% within EL	0.0%	4.5%	47.7%	44.3%	3.4%	100.0%		
	Senior Leader/CEO	Count	0	3	12	9	3	27		
		% within EL	0.0%	11.1%	44.4%	33.3%	11.1%	100.0%		
Total		Count	1	19	165	130	9	324		
		% within EL	0.3%	5.9%	50.9%	40.1%	2.8%	100.0%		

EL * *Explore_round Cross Tabulation*

Table 6-36: Rate of Idea generation by employment level

EL * Gen rounded C	ross Tabulation
--------------------	-----------------

				Gen_rounded						
			Never	Rarely	Sometimes	Often	Always	Total		
EL	Administration	Count	1	4	18	16	2	41		
		% within EL	2.4%	9.8%	43.9%	39.0%	4.9%	100.0%		
	Not a manager	Count	2	8	84	63	11	168		
		% within EL	1.2%	4.8%	50.0%	37.5%	6.5%	100.0%		
	Manager	Count	0	7	27	47	7	88		
		% within EL	0.0%	8.0%	30.7%	53.4%	8.0%	100.0%		
	Senior Leader/CEO	Count	0	4	10	11	2	27		
		% within EL	0.0%	14.8%	37.0%	40.7%	7.4%	100.0%		
Total		Count	3	23	139	137	22	324		
		% within EL	0.9%	7.1%	42.9%	42.3%	6.8%	100.0%		

Table 6-37: Rate of Idea championing by employment level

				Champ_rounded						
			Never	Rarely	Sometimes	Often	Always	Total		
EL	Administration	Count	3	12	17	9	0	41		
		% within EL	7.3%	29.3%	41.5%	22.0%	0.0%	100.0%		
	Not a manager	Count	10	28	93	32	5	168		
		% within EL	6.0%	16.7%	55.4%	19.0%	3.0%	100.0%		
	Manager	Count	2	14	34	33	5	88		
		% within EL	2.3%	15.9%	38.6%	37.5%	5.7%	100.0%		
	Senior Leader/CEO	Count	5	4	12	4	2	27		
		% within EL	18.5%	14.8%	44.4%	14.8%	7.4%	100.0%		
Total		Count	20	58	156	78	12	324		
		% within EL	6.2%	17.9%	48.1%	24.1%	3.7%	100.0%		

EL * Champ_rounded Cross Tabulation

Table 6-38: Rate of Idea implementation by employment level

				Imp_rounded						
			Never	Rarely	Sometimes	Often	Always			
EL	Administration	Count	1	13	13	13	1	41		
		% within EL	2.4%	31.7%	31.7%	31.7%	2.4%	100.0%		
	Not a manager	Count	6	21	84	51	6	168		
		% within EL	3.6%	12.5%	50.0%	30.4%	3.6%	100.0%		
	Manager	Count	2	9	30	40	6	87		
		% within EL	2.3%	10.3%	34.5%	46.0%	6.9%	100.0%		
	Senior Leader/CEO	Count	3	1	8	11	4	27		
		% within EL	11.1%	3.7%	29.6%	40.7%	14.8%	100.0%		
Total		Count	12	44	135	115	17	323		
		% within EL	3.7%	13.6%	41.8%	35.6%	5.3%	100.0%		

EL * Imp_rounded Cross Tabulation

6.3.2.6. IWB by *RWF*

More than half of the employees who 'always' worked remotely would 'sometimes' exhibit IWB. An equal number of employees working remotely '2–4 times per week' responded with 'sometimes' to 'always' (Table 6.39).

Table 6-39: Rate of IWB by RWF

					IWB_round			Total
			Never	Rarely	Sometimes	Often	Always	
RWF	1–3 times	Count	1	7	24	6	3	41
	per month	% within RWF	2.4%	17.1%	58.5%	14.6%	7.3%	100.0%
	Once a week	Count	0	9	31	21	2	63
		% within RWF	0.0%	14.3%	49.2%	33.3%	3.2%	100.0%
	2–4 times	Count	0	12	68	68	2	150
	per week	% within RWF	0.0%	8.0%	45.3%	45.3%	1.3%	100.0%
	Always	Count	1	4	42	22	1	70
		% within RWF	1.4%	5.7%	60.0%	31.4%	1.4%	100.0%
Total		Count	2	32	165	117	8	324
		% within RWF	0.6%	9.9%	50.9%	36.1%	2.5%	100.0%

RWF * IWB_round Cross Tabulation

However, during the first two phases (Idea Exploration and Idea Generation), the majority of employees working remotely '2–4 times per week' selected 'often' (Tables 6.40 and 6.41). Those working remotely '1–3 times per month' had the highest proportion who selected 'always' in each phase except for Idea Generation.

Table 6-40: Rate of Idea Exploration by RWF

					Explore_rour	ıd		Total
			Never	Rarely	Sometimes	Often	Always	
RWF	1–3 times	Count	1	5	22	11	2	41
	per month	% within RWF	2.4%	12.2%	53.7%	26.8%	4.9%	100.0%
	Once a	Count	0	6	32	23	2	63
	week	% within RWF	0.0%	9.5%	50.8%	36.5%	3.2%	100.0%
	2–4 times	Count	0	2	71	73	4	150
	per week	% within RWF	0.0%	1.3%	47.3%	48.7%	2.7%	100.0%
	Always	Count	0	6	40	23	1	70
		% within RWF	0.0%	8.6%	57.1%	32.9%	1.4%	100.0%
Total		Count	1	19	165	130	9	324
		% within RWF	0.3%	5.9%	50.9%	40.1%	2.8%	100.0%

RWF * *Explore_round Cross Tabulation*

Table 6-41: Rate of Idea Generation by RWF

			Gen_rounded					Total
			Never	Rarely	Sometimes	Often	Always	
RWF	1–3 times	Count	1	5	20	13	2	41
	per month	% within RWF	2.4%	12.2%	48.8%	31.7%	4.9%	100.0%
	Once a	Count	0	8	27	25	3	63
	week	% within RWF	0.0%	12.7%	42.9%	39.7%	4.8%	100.0%
	2–4 times	Count	1	9	57	69	14	150
	per week	% within RWF	0.7%	6.0%	38.0%	46.0%	9.3%	100.0%
	Always	Count	1	1	35	30	3	70
		% within RWF	1.4%	1.4%	50.0%	42.9%	4.3%	100.0%
Total		Count	3	23	139	137	22	324
		% within RWF	0.9%	7.1%	42.9%	42.3%	6.8%	100.0%

RWF * *Gen_rounded Cross Tabulation*

Table 6-42: Rate of Idea Championing by RWF

			Champ_rounded					Total
			Never	Rarely	Sometimes	Often	Always	
RWF	1–3 times	Count	4	11	16	8	2	41
	per month	% within RWF	9.8%	26.8%	39.0%	19.5%	4.9%	100.0%
	Once a	Count	5	9	33	14	2	63
	week	% within RWF	7.9%	14.3%	52.4%	22.2%	3.2%	100.0%
	2–4 times	Count	5	23	73	43	6	150
	per week	% within RWF	3.3%	15.3%	48.7%	28.7%	4.0%	100.0%
	Always	Count	6	15	34	13	2	70
		% within RWF	8.6%	21.4%	48.6%	18.6%	2.9%	100.0%
Total		Count	20	58	156	78	12	324
		% within RWF	6.2%	17.9%	48.1%	24.1%	3.7%	100.0%

RWF * Champ_rounded Cross Tabulation

			Imp_rounded					Total
			Never	Rarely	Sometimes	Often	Always	
RWF	1–3 times	Count	4	9	15	10	3	41
	per month	% within RWF	9.8%	22.0%	36.6%	24.4%	7.3%	100.0%
	Once a	Count	3	10	24	22	3	62
	week	% within RWF	4.8%	16.1%	38.7%	35.5%	4.8%	100.0%
	2–4 times	Count	4	15	64	60	7	150
	per week	% within RWF	2.7%	10.0%	42.7%	40.0%	4.7%	100.0%
	Always	Count	1	10	32	23	4	70
		% within RWF	1.4%	14.3%	45.7%	32.9%	5.7%	100.0%
Total		Count	12	44	135	115	17	323
		% within RWF	3.7%	13.6%	41.8%	35.6%	5.3%	100.0%

RWF * Imp rounded Cross Tabulation

6.3.2.7. Summary of Innovative Work Behaviour

Overall, the categories within the sample demonstrating the highest prevalence of IWB (i.e. 'Always') were identified as follows: the 18–30-year-old age group, contract/casual working females, with less than two years of tenure at their current organisation, working in managerial positions, working remotely 1–3 times per month.

The categories within the sample demonstrating a consistently high prevalence of IWB ('Often') were identified as follows: the 41–50-year-old age group, full-time working males, with 6+ years tenure, working in managerial positions, working remotely 2–4 times per week.

6.4. ANOVA and Logistic regression testing

6.4.1. Reliability analysis

Prior to undertaking the ANOVA and regression tests, reliability analysis using Cronbach's alpha was conducted to analyse the consistency of scales, and assess the degree to which the scale will produce stable and consistent results (McNeish, 2018; Sijtsma, 2008).
Cronbach's alpha is a statistical measure used to assess the internal consistency or reliability of a set of scale or questionnaire items (Cronbach, 1951). Cronbach's alpha evaluates the extent to which items within a questionnaire are interrelated or measure the same underlying construct. It assesses how closely related the responses to different items are within the same scale. A high Cronbach's alpha indicates that the items are highly correlated and consistently measure the same construct, while a low Cronbach's alpha suggests that the items may not be measuring the same underlying construct or that the scale may lack internal consistency. Cronbach's alpha typically ranges from 0 to 1, with values above 0.7 considered acceptable for research purposes, values above 0.8 as good reliability, and values above 0.9 as excellent reliability (Cronbach, 1951; Streiner, 2003).

The results shown in Table 6.44 indicated that the value of Cronbach's alpha of variables 'idea exploration' and 'idea generation' were 0.827 and 0.888 respectively, which were higher than 0.80, indicating good reliability. Similarly, the Cronbach's alpha of variables 'idea championing' and 'idea implementation' were 0.910 and 0.905 respectively, which were higher than 0.90, indicating excellent reliability. Hence, all variables were found to be reliable, so hypothesis testing could continue.

Variables	No. of Items	Cronbach's Alpha
Idea Exploration	10	0.827
Idea Generation	7	0.888
Idea Championing	5	0.910
Idea Implementation	5	0.905

Table 6-44: Reliability test

6.5. ANOVA analysis

6.5.1. IWB among RWF groups (H1)

H₀: There is no difference in innovative work behaviour between different RWF groups. H₁: There is a difference in innovative work behaviour between different RWF groups. One-way ANOVA was conducted to determine if IWB was different for each group of remote working frequency. Participants were classified into four groups: 1-3 times per month (n=41), Once a week (n=63), 2-4 times per week (n=150) and Always (n=70).

Data was normally distributed for each remote working frequency group, as assessed by Shapiro-Wilk's test (p > .05).

Variables	Kolmogorov-Smirnov		Shapiro-Wilk	
	Statistic	p-value	Statistic	p-value
1–3 times p/month	0.116	0.184	0.970	0.348
Once a week	0.109	0.059	0.971	0.140
2–4 times p/week	0.060	0.200	0.986	0.129
Always	0.061	0.200	0.977	0.227

Table 6-45: Normality test for IWB and RWF

There was homogeneity of variances (Table 6.46), as assessed by Levene's test for equality of variances (p = .402).

The differences between different remote working frequency groups was statistically significant (F (3, 320) = 3.673, p = 0.013, ω^2 = .024).

IWB increased as remote working frequency increased. From the 1–3 times per month (M=3.09, SD=.767), once a week (M=3.21, SD=.655), to 2–4 times per week (M=3.42, SD=.580). The always work from home group (M=3.25, SD=.562) declined slightly.

The group means were statistically significantly different (p < .1). Therefore, we can reject the null hypothesis and accept hypothesis H1.

Category – RWF	Mean (SD)	Levene's test –	One-way ANOVA
		Homogeneity of	F (p-value)
		Variances	
		F (p-value)	
1–3 times p/month	3.09 (0.767)	0.981 (0.402)	3.673 (0.013)
Once a week	3.24 (0.655)		
2–4 times p/week	3.42 (0.580)		
Always	3.25 (0.562)		

Table 6-46: One-way ANOVA – Comparison of IWB across RWF

Tukey post hoc analysis revealed that the mean increase from 1-3 times per month to 2–4 times per week (0.324, 95% CI [0.43, 0.61]) was statistically significant (p = .016), but no other group differences were statistically significant.

This indicates that individuals who work remotely 2–4 times per week exhibit a statistically significantly higher IWB compared to those who work remotely 1–3 times per month.

6.5.2. IWB among age groups (H2)

Ho: There is no difference in innovative work behaviour between different age groups.

H₁: There is a difference in innovative work behaviour between different age groups.

One-way ANOVA was conducted to determine if IWB was different for each age group. Participants were classified into four groups: 18-30 (n=158), 31-40 (n=92), 41-50 (n=42) and 51+ (n=32) years old.

Data was not normally distributed for all age groups, as assessed by Shapiro-Wilk's test (p < .05). However, as the sample sizes are greater than 50, the Normal Q-Q plot (Quantile-Quantile plot) was also considered to assess normality using graphical methods. The graph indicated data points appear as roughly a straight line. Hence, from the graphical representation of the variables, the variables were considered to be normal.

Table 6-47: Normality test for IWB and age

Variables	Kolmogorov-Smirnov		Shapiro-Wilk	
	Statistic	p-value	Statistic	p-value
18–30 years old	0.065	0.097	0.982	0.035
31-40 years old	0.090	0.061	0.952	0.002
41-50 years old	0.114	0.198	0.948	0.055
51+ years old	0.102	0.200	0.966	0.401

There was homogeneity of variances (Table 6.48), as assessed by Levene's test for equality of variances (p = .194).

IWB increased from the 41–50 years group (M=3.24, SD=.609), to 51+ years (M=3.26, SD=.733), 18-30 (M=3.32, SD=.652), to 31-40 (M=3.33, SD=.547) group, in that order.

However, the differences between these age groups was not statistically significant (F(3, 320) = 0.282, p = 0.839). Therefore, hypothesis H2 was rejected.

Category – Age	Mean (SD)	Levene's test –	One-way ANOVA
		Homogeneity of	F (p-value)
		Variances	
		F (p-value)	
18-30 years old	3.32 (0.652)	1.579 (0.194)	0.282 (0.839)
31-40 years old	3.33 (0.547)		
41-50 years old	3.24 (0.609)		
51+ years old	3.26 (0.733)		

Table 6-48: One-way ANOVA - Comparison of IWB across Age

A Kruskal-Wallis H test was also run to determine if there were differences in IWB between four age groups. Distributions of IWB scores were similar for all groups, as assessed by visual inspection of a boxplot. Median IWB scores increased from 18-30 years (3.30), 31-40 years (3.30), to 41-50 years (3.32), to 51+ years (3.41) age groups, but the differences were not statistically significant, $\chi^2(3) = 0.420$, p = .936.

6.5.3. IWB among gender groups (H3)

Ho: There is no difference in innovative work behaviour between different genders.

H₁: There is a difference in innovative work behaviour between different genders.

Independent-samples t-test was conducted to determine if there were differences in IWB between females and males. There were 155 female respondents and 169 male respondents.

Data was not normally distributed for males, as assessed by Shapiro-Wilk's test (p < .05). However, as the t-test has been found to be robust against violations of normality if sample sizes are equal or nearly equal , fairly large ($n \ge 30$) and you have equal variances (Poncet et al., 2016; Sawilowsky & Blair, 1992; Stonehouse & Forrester, 1998), the results were interpreted.

Table 6-49	: Normality	test for	IWB	and	gender
	r v				0

Variables	Kolmogorov-Smirnov		Shapiro-Wilk	
	Statistic	p-value	Statistic	p-value
Female	0.066	0.200	0.989	0.246
Male	0.085	0.004	0.972	0.002

There was homogeneity of variances (Table 6.50), as assessed by Levene's test for equality of variances (p = .287).

IWB was slightly higher for males (M=3.32, SD=0.66) than females (M=3.30, SD=0.58). However, the differences between genders was not statistically significant, t(322) = -0.246, p = 0.806. Therefore, hypothesis H3 was rejected.

Table 6-50: T-Test – Comparison of IWB between genders

Category – Gender	Mean (SD)	Levene's test –	T-Test
		Homogeneity of	F (p-value)
		Variances	
		F (p-value)	
Female	3.30 (0.582)	1.136 (0.287)	-0.246 (0.806)
Male	3.32 (0.664)		

6.5.4. IWB among tenure groups (H4)

Ho: There is no difference in innovative work behaviour between different tenure groups.

H₁: There is a difference in innovative work behaviour between different tenure groups.

One-way ANOVA was conducted to determine if IWB was different for the tenure group. Participants were classified into four groups: Less than 2 years (n=158), 2–5 years (n=92), and 6+ years (n=32).

Data was normally distributed for two tenure groups, as assessed by Shapiro-Wilk's test (p > .05), but not normally distributed for the 6+ year group (P < .05).

Variables	Kolmogorov-Smirnov		Shapiro-Wilk	
	Statistic	p-value	Statistic	p-value
Less than 2 years	0.076	0.056	0.986	0.191
2–5 years	0.075	0.162	0.980	0.104
6+ years	0.117	0.008	0.938	< 0.001

Fable 6-51: Normality	test for	IWB and	tenure
------------------------------	----------	---------	--------

The assumption of homogeneity of variances was violated (Table 6.52), as assessed by Levene's test for equality of variances (p = .003).

IWB increased from the less than 2 years tenure group (M=3.25, SD=.604), to the more than +6 years tenure group (M=3.30, SD=.609), to the 2–5 years group (M=3.38, SD=.553), in that order.

As we do not have homogeneity of variances, we cannot interpret the standard one-way ANOVA, but must use a modified version of the ANOVA; the Welch ANOVA was used (Delacre et al., 2018; Wilcox, 2017).

The differences between tenure groups was not statistically significant, Welch's F(2, 185.904) = 1.339, p = .265. Therefore, hypothesis H4 was rejected.

Category – Tenure	Mean (SD)	Levene's test –	Welch ANOVA
		Homogeneity of	F (p-value)
		Variances	
		F (p-value)	
Less than 2 years	3.25 (0.604)	3.420 (0.034)	1.339 (0.265)
2–5 years	3.38 (0.553)		
6+ years	3.30 (0.741)		

Table 6-52: Welch ANOVA – Comparison of IWB across tenure

A Kruskal-Wallis H test was also run to determine if there were differences in IWB scores between three tenure groups. Distributions of IWB scores were similar for all groups, as assessed by visual inspection of a boxplot. Median IWB scores increased from less than 2 years (3.19), to 2–5 years (3.33), to 6+ years (3.48), but the differences were not statistically significant, $\chi^2(2)=4.970$, p = .083.

6.5.5. IWB among employment status groups (H5)

H₀: There is no difference in innovative work behaviour between different employment status groups.

H₁: There is a difference in innovative work behaviour between different employment status groups.

One-way ANOVA was conducted to determine if IWB was different for each employment status group. Participants were classified into three groups: Full-time (n=207), Part-time (n=61), and Contract/Casual (n=56).

Data was normally distributed for two tenure groups, as assessed by Shapiro-Wilk's test (p > .05), but not normally distributed for part-time group (p < .05).

Variables	Kolmogorov-Smirnov		Shapiro-Wilk	
	Statistic	p-value	Statistic	p-value
Full-time	0.051	0.200	0.991	0.237
Part-time	0.139	0.005	0.945	0.008
Contract/Casual	0.091	0.200	0.962	0.074

Table 6-53: Normality test for IWB and employment status

There was homogeneity of variances (Table 6.54), as assessed by Levene's test for equality of variances (p = .850).

The differences between employment status groups was statistically significant (F (2, 321) = 4.696, p = 0.010, $\omega^2 = .022$).

IWB increased from the part-time group (M=3.12, SD=.630), to Contract/Casual (M=3.27, SD=.651) to the full-time group (M=3.38, SD=.605) in that order.

The group means were statistically significantly different (p < .1). Therefore, we can reject the null hypothesis and accept hypothesis H5.

Category – ES	Mean (SD)	Levene's test –	One-way ANOVA
		Homogeneity of	F (p-value)
		Variances	
		F (p-value)	
Full-time	3.38 (0.605)	0.199 (0.850)	4.696 (0.010)
Part-time	3.12 (0.630)		
Contract/Casual	3.27 (0.651)		

Table 6-54: One-way ANOVA - Comparison of IWB across employment status

Tukey post hoc analysis revealed that the mean increase from Part-time to Fulltime (0.272, 95% CI [0.60, 0.48]) was statistically significant (p = .008), but no other group differences were statistically significant.

This indicates that individuals who work full-time exhibit a statistically significantly higher IWB compared to those who work part-time.

A Kruskal-Wallis H test was also run to determine if there were differences in IWB scores between three employment status groups. Distributions of IWB scores were similar for all groups, as assessed by visual inspection of a boxplot. Median IWB scores increased from the Part-time (3.11), to Contract/Casual (3.30), to Full-time (3.37) employment status groups. Median IWB scores were statistically significantly different between groups, $\chi^2(2)=10.594$, p = .005.

Pairwise comparisons were performed using Dunn (1964) procedure with a Bonferroni correction for multiple comparisons. Adjusted p-values are presented. This post hoc analysis revealed statistically significant differences in IWB between the part-time (Mdn = 3.11) and full-time (Mdn = 3.37) (p = 0 .004) groups, but not between any other group combination.

6.5.6. IWB among employment level groups (H6)

Ho: There is no difference in innovative work behaviour between different employment level groups.

H₁: There is a difference in innovative work behaviour between different employment level groups.

One-way ANOVA was conducted to determine if IWB was different for employment level groups. Participants were classified into four groups: Administration (n=41), Not a manager (n=168), Manager (n=88) and Senior Leadership (n=27).

Data was normally distributed for three employment level groups, as assessed by Shapiro-Wilk's test (p > .05), but not normally distributed for the not a manager group (p < .05).

Variables	Kolmogoro	ov-Smirnov	Shapiro-Wilk		
	Statistic	p-value	Statistic	p-value	
Administration	0.071	0.200	0.990	0.978	
Not a manager	0.078	0.013	0.971	0.001	
Manager	0.066	0.200	0.983	0.318	
Senior Leadership	0.135	0.200	0.943	0.140	

Table 6-55: Normality test for IWB and employment level

There was homogeneity of variances (Table 6.56), as assessed by Levene's test for equality of variances (p = .172).

The differences in IWB between employment level groups was statistically significant ($F(3, 320) = 3.069, p = 0.028, \omega^2 = .019$).

IWB increased from the Administration group (M=3.17, SD=.634), to Not a manager (M=3.260, SD=.577), to Senior Leadership (M=3.290, SD=.793) to Managers (M=3.470, SD=.631) in that order.

The group means were statistically significantly different (p < .1). Therefore, we can reject the null hypothesis and accept hypothesis H6.

Category – EL	Mean (SD)	Levene's test –	One-way ANOVA
		Homogeneity of	F (p-value)
		Variances	
		F (p-value)	
Administration	3.168 (0.634)	1.678 (0.172)	3.069 (0.028)
Not a manager	3.260 (0.577)		
Manager	3.470 (0.631)		
Senior Leadership	3.290 (0.793)		

Table 6-56: One-way ANOVA - Comparison of IWB across employment level

Tukey post hoc analysis revealed that the mean increase from not a manager to manager (0.211, 95% CI [0.0005, 0.4213]) was statistically significant (p = .049), as well as from administration to manager (0.303, 95% CI [0.0002, 0.6049], p = .050), but no other group differences were statistically significant.

This indicates that managers exhibit a statistically significantly higher IWB compared to non-managers or administrators.

A Kruskal-Wallis H test was also run to determine if there were differences in IWB scores between four employment level groups of participants. Distributions of IWB scores were similar for all groups, as assessed by visual inspection of a boxplot. Median IWB scores increased from the Administration (3.22), to Not a manager (3.26), to Senior Leadership/CEO (3.33), to Manager (3.54) employment level groups. Median IWB scores were statistically significantly different, $\chi^2(3)=10.340$, p = .016.

Subsequently, pairwise comparisons were performed using Dunn (1964) procedure with a Bonferroni correction for multiple comparisons. Adjusted p-values are presented. This post hoc analysis revealed statistically significant differences in IWB between the Manager (Mdn = 3.54) and Administration (Mdn = 3.22) (p = 0.045) and

Manager and Non-manager (Mdn = 3.26) (p = 0.030) groups, but not between the Senior Leadership/CEO group (Mdn = 3.33) or any other group combination.

6.5.7. Summary of ANOVA tests

Table 6.57 provides a summary of the ANOVA tests relating to hypotheses H1 to H6.

Code	Description	Test	Outcome
H1	RWF has a significant effect on IWB	ANOVA	Reject the null hypothesis
H2	Age has a significant effect on IWB when working remotely	ANOVA	Fail to reject the null hypothesis
Н3	Gender has a significant effect on IWB when working remotely	T-Test	Fail to reject the null hypothesis
H4	Tenure has a significant effect on IWB when working remotely	ANOVA	Fail to reject the null hypothesis
Н5	Employment Status (ES) has a significant effect on IWB when working remotely	ANOVA	Reject the null hypothesis
H6	Employment Level (EL) has a significant effect on IWB when working remotely	ANOVA	Reject the null hypothesis

6.6. Logistic regression analysis

6.6.1. RWF, demographics and IWB (H7)

Ho: There is no significant relationship between RWF and demographics and the odds of innovative work behaviour.

H₁: There is a significant relationship between RWF and demographics and the odds of innovative work behaviour.

A binomial logistic regression was performed to ascertain the effects of RWF, age, gender, tenure at current organisation, employment status and employment level on the likelihood that participants will exert IWB overall.

The logistic regression model was statistically significant, $\chi^2(7) = 26.56$, p < .10

The model explained 10.7% (Nagelkerke R²) of the variance in IWB and correctly classified 65.1% of cases.

Sensitivity is the percentage of cases that had the observed characteristic (i.e. 'yes' for IWB) which were correctly predicted by the model. In this case, 35.2% of participants who exhibited IWB were also predicted by the model to have IWB.

Specificity is the percentage of case that did not have the observed characteristic (i.e. 'no' for IWB) and were also correctly predicted as not having the observed characteristic. In this case, 83.9% of participants who did not exert IWB were correctly predicted by the model.

Positive predictive value is the percentage of correctly predicted cases with the observed characteristic compared to the total number of cases predicted as having the characteristic. In this case, 57.9% of all cases predicted as having IWB were correctly predicted.

Negative predictive value is the percentage of correctly predicted cases without the observed characteristic compared to the total number of cases predicted as not having the characteristic. In this case, 67.3% of all cases predicted as not having IWB were correctly predicted.

Of the six predictor variables, three were statistically significant at 0.10: Employment Status, Tenure and Employment Level (Table 6.58). RWF was just outside of the statistically significant range. Table 6-58: Logistic Regression predicting likelihood of IWB based on RWF and demographics

								95% C.I.	for EXP(B)
		В	S.E.	Wald	df	Sig.	Exp(B)	Lower	Upper
Step 1 ^a	Age	041	.130	.097	1	.755	.960	.744	1.240
	Gender(1)	015	.250	.004	1	.952	.985	.604	1.607
	Employment Status			9.541	2	.008			
	Employment Status(1)	-1.124	.367	9.371	1	.002	.325	.158	.668
	Employment Status(2)	333	.333	1.000	1	.317	.717	.373	1.376
	Tenure at current organisation	.317	.171	3.455	1	.063	1.373	.983	1.919
	RWF	.218	.136	2.576	1	.108	1.243	.953	1.622
	Title	.277	.164	2.851	1	.091	1.319	.956	1.819
	Constant	-1.993	.606	10.805	1	.001	.136		

Variables in the Equation

a. Variable(s) entered on step 1: Age, Gender, Employment Status, Tenure at current organisation, How often working remotely, Title.

Being part-time was associated with a reduced likelihood of exhibiting IWB. This indicates that being in the part-time category is less likely to show IWB than being full-time. The odds ratio (OR) indicates that for every one-unit increment on the predictor (part-time employment), the odds of IWB increase by a factor of .325 (meaning the odds are decreasing).

Increased tenure and more senior roles are associated with an increased likelihood of exhibiting IWB. The OR for tenure indicating that for every one unit increase on this predictor, the odds of IWB change by a factor of 1.37 (meaning the odds are increasing, as tenure increases the odds of IWB also increase). The OR for employment level indicate that for every one unit increase on this predictor, the odds of IWB change by a factor of 1.32.

Although not statistically significant, the data suggests that RWF is positively associated with an increased likelihood of exhibiting IWB. The OR for RWF indicating that for every one unit increase on this predictor, the odds of IWB change by a factor of 1.24 (as RWF increases the odds of IWB also increase).

As the overall model tested statistically significant, it indicates that at least one of the independent variables has a significant impact on the dependent variable, we can reject the null hypothesis.

6.6.2. RWF, demographics and Idea Exploration phase (H8)

H₀: There is no significant relationship between RWF & demographics and the odds of idea exploration.

H₁: There is a significant relationship between RWF & demographics and the odds of idea exploration.

A binomial logistic regression was performed to ascertain the effects of RWF, age, gender, tenure at current organisation, employment status and employment level on the likelihood that participants will exert IWB during the idea exploration phase.

The logistic regression model was statistically significant, $\chi^2(7) = 12.53$, p < .10

The model explained 5.1% (Nagelkerke R²) of the variance in IWB and correctly classified 60.8% of cases.

Sensitivity was 33.1%, specificity was 81.6%, positive predictive value was 57.5% and negative predictive value was 61.9%.

Of the six predictor variables, one was statistically significant: Employment Status (Table 6.59). Being part-time was associated with a reduced likelihood of exhibiting Idea Exploration compared to being full-time. The OR indicating that for every one-unit increment on the predictor (part-time employment), the odds of Idea Exploration increase by a factor of .431 (meaning the odds are decreasing). Expressed inversely, being part-time had 2.32 times lower odds to exhibit idea exploration than being full-time.

Table 6-59: Logistic Regression predicting likelihood of Idea Exploration based on RWF and demographics

								95% EX	C.I.for XP(B)
		В	S.E.	Wald	df	Sig.	Exp(B)	Lower	Upper
Step 1 ^a	Age	.059	.125	.227	1	.634	1.061	.831	1.355
	Gender(1)	102	.241	.177	1	.674	.903	.563	1.450
	Employment Status			6.444	2	.040			
	Employment Status(1)	842	.332	6.442	1	.011	.431	.225	.826
	Employment Status(2)	192	.319	.363	1	.547	.825	.441	1.542
	Tenure at current organisation	.236	.165	2.032	1	.154	1.266	.915	1.750
	RWF	.072	.127	.321	1	.571	1.075	.837	1.380
	Title	.060	.158	.144	1	.705	1.062	.779	1.446
	Constant	938	.569	2.713	1	.100	.391		

Variables in the Equation

a. Variable(s) entered on step 1: Age, Gender, Employment Status, Tenure at current organisation, How often working remotely, Title.

As the overall model tested statistically significant, it indicates that at least one of the independent variables has a significant impact on the dependent variable, we can reject the null hypothesis.

6.6.3. RWF, demographics and Idea Generation Phase (H9)

Ho: There is no significant relationship between RWF and demographics and the odds of idea generation.

H₁: There is a significant relationship between RWF and demographics and the odds of idea generation.

A binomial logistic regression was performed to ascertain the effects of RWF, age, gender, tenure at current organisation, employment status and employment level on the likelihood that participants will exert IWB during the idea generation phase.

The logistic regression model was statistically significant, $\chi^2(7) = 22.90$, p < .10

The model explained 9.1% (Nagelkerke R^2) of the variance in IWB and correctly classified 61.1% of cases.

Sensitivity was 69.8%, specificity was 52.7%, positive predictive value was 58.7% and negative predictive value was 64.4%.

Of the six predictor variables, one was statistically significant: employment status (Table 6.60). Being part-time was associated with a reduced likelihood of exhibiting Idea Generation compared to being full-time. The OR indicating that for every one-unit increment on the predictor (part-time employment), the odds of Idea Exploration increase by a factor of .327 (meaning the odds are decreasing). Expressed inversely, being part-time had 3.06 lower odds to exhibit idea generation.

Table 6-60: Logistic Regression predicting likelihood of Idea Generation based on RWF and demographics

Variables in the Equation									
								95%	C.I.for
								EX	(B)
		В	S.E.	Wald	df	Sig.	Exp(B)	Lower	Upper
Step 1 ^a	Age	061	.127	.228	1	.633	.941	.734	1.207
	Gender(1)	126	.243	.271	1	.603	.881	.547	1.419
	Employment Status			14.229	2	<.001			
	Employment Status(1)	-1.117	.337	11.001	1	<.001	.327	.169	.633
	Employment Status(2)	.335	.320	1.096	1	.295	1.398	.747	2.617
	Tenure at current organisation	.153	.166	.842	1	.359	1.165	.841	1.615
	RWF	.175	.128	1.855	1	.173	1.191	.926	1.531
	Title	.251	.160	2.450	1	.118	1.285	.939	1.759
	Constant	-1.068	.575	3.449	1	.063	.344		

a. Variable(s) entered on step 1: Age, Gender, Employment Status, Tenure at current organisation, How often working remotely, Title.

As the overall model tested statistically significant, it indicates that at least one of the independent variables has a significant impact on the dependent variable, we can reject the null hypothesis.

6.6.4. RWF, demographics and Idea Championing Phase (H10)

Ho: There is no significant relationship between RWF and demographics and the odds of idea championing.

H₁: There is a significant relationship between RWF and demographics and the odds of idea championing.

A binomial logistic regression was performed to ascertain the effects of RWF, age, gender, tenure at current organisation, employment status and employment level on the likelihood that participants will exert IWB during the idea championing phase.

The logistic regression model was statistically significant, $\chi^2(7) = 21.61$, p < .10

The model explained 9.3% (Nagelkerke R^2) of the variance in IWB and correctly classified 71.3% of cases.

Sensitivity was 2.2%, specificity was 97.9%, positive predictive value was 28.6% and negative predictive value was 72.2%.

Of the six predictor variables, half were statistically significant: age, employment status and tenure (Table 6.61). Increasing age was associated with a reduction in the likelihood of idea championing. The OR indicating that for every one unit increase on this predictor, the odds of Idea championing change by a factor of .706. Expressed inversely, being older had 1.42 lower odds to exhibit idea championing.

Table 6-61: Logistic Regression predicting likelihood of Idea Championing based on RWF and demographics

Variable	Variables in the Equation									
								95% ΕΣ	C.I.for KP(B)	
		В	S.E.	Wald	df	Sig.	Exp(B)	Lower	Upper	
Step 1 ^a	Age	347	.153	5.165	1	.023	.706	.524	.953	
	Gender(1)	119	.269	.195	1	.659	.888	.524	1.504	
	Employment Status			7.459	2	.024				
	Employment Status(1)	-1.091	.420	6.746	1	.009	.336	.147	.765	
	Employment Status(2)	471	.373	1.593	1	.207	.624	.300	1.298	
	Tenure at current organisation	.344	.186	3.400	1	.065	1.410	.979	2.031	
	RWF	.050	.145	.122	1	.727	1.052	.792	1.396	
	Title	.233	.178	1.703	1	.192	1.262	.890	1.790	
	Constant	-1.367	.638	4.593	1	.032	.255			

a. Variable(s) entered on step 1: Age, Gender, Employment Status, Tenure at current organisation, How often working remotely, Title.

Being part-time was also associated with a reduced likelihood of exhibiting Idea Championing compared to being full-time. The OR indicating that for every one-unit increment on the predictor (part-time employment), the odds of Idea Championing increase by a factor of .336 (meaning the odds are decreasing). Expressed inversely, being part-time had 2.98 lower odds to exhibit idea championing.

Increased tenure was associated with an increased likelihood of exhibiting idea championing. The OR indicating that for every one unit increase on this predictor, the odds of Idea Championing change by a factor of 1.41.

As the overall model tested statistically significant, it indicates that at least one of the independent variables has a significant impact on the dependent variable, we can reject the null hypothesis.

6.6.5. RWF, demographics and Idea Implementation Phase (H11)

H₀: There is no significant relationship between RWF and demographics and the odds of idea implementation.

H₁: There is a significant relationship between RWF and demographics and the odds of idea implementation.

A binomial logistic regression was performed to ascertain the effects RWF, age, gender, tenure at current organisation, employment status and employment level on the likelihood that participants will exert IWB during the idea implementation phase.

The logistic regression model was statistically significant, $\chi^2(7) = 17.65$, p < .10

The model explained 7.2% (Nagelkerke R²) of the variance in IWB and correctly classified 60.1% of cases.

Sensitivity was 25%, specificity was 84.3%, positive predictive value was 52.4% and negative predictive value was 61.9%.

Of the six predictor variables, two were statistically significant: employment status and employment level (Table 6.62). Being part-time was associated with a reduced likelihood of exhibiting Idea Implementation compared to being full-time. The OR indicating that for every one-unit increment on the predictor (part-time employment), the odds of Idea Implementation increase by a factor of .568 (meaning the odds are decreasing). Expressed inversely, being part-time had 1.76 lower odds to exhibit idea championing.

 Table 6-62: Logistic Regression predicting likelihood of Idea Implementation based on RWF and demographics

								95% EX	C.I.for
		В	S.E.	Wald	df	Sig.	Exp(B)	Lower	Upper
Step 1 ^a	Age	152	.130	1.364	1	.243	.859	.665	1.109
-	Gender(1)	.285	.244	1.364	1	.243	1.330	.824	2.144
	Employment Status			2.890	2	.236			
	Employment Status(1)	566	.335	2.850	1	.091	.568	.294	1.095
	Employment Status(2)	184	.326	.318	1	.573	.832	.439	1.576
	Tenure at current organisation	010	.169	.003	1	.955	.991	.711	1.380
	RWF	.172	.131	1.723	1	.189	1.187	.919	1.535
	Title	.433	.163	7.003	1	.008	1.541	1.119	2.123
	Constant	-1.584	.588	7.271	1	.007	.205		

Variables in the Equation

a. Variable(s) entered on step 1: Age, Gender, Employment Status, Tenure at current organisation, How often working remotely, Title.

Increasing employment level was associated with an increased likelihood of exhibiting idea implementation. The OR indicating that for every one unit increase on this predictor, the odds of IWB change by a factor of 1.54.

As the overall model tested statistically significant, it indicates that at least one of the independent variables has a significant impact on the dependent variable, we can reject the null hypothesis.

6.6.6. Summary of logistic regression tests

Table 6.63 provides a summary of the logistic regression tests relating to hypotheses H7 to H11.

Code	Description	Predictor variables with significant effect at 0.1	Outcome
H7	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of IWB	Employment Status Tenure Employment Level	Reject the null hypothesis
H8	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Exploration behaviours	Employment Status	Reject the null hypothesis
H9	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Generation behaviours	Employment Status	Reject the null hypothesis
H10	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Championing behaviours	Age Employment Status Tenure	Reject the null hypothesis
H11	RWF, Age, Gender, Tenure, ES, EL has a significant effect on the odds of Idea Implementation behaviours	Employment Status Employment Level	Reject the null hypothesis

6.7. Summary of ANOVA and Logistic Regression tests

Table 6.64 presents a summary of the variables that exhibited a statistically significant relationship with IWB across both ANOVA and Logistic Regression testing. The '+' indicates increased IWB and '-' indicates that decresed IWB.

Table 6-64:	Summary of	ANOVA and	Logistic	Regression (tests

	ANOVA	VA Logistic Regression				
	IWB	IWB	IE	IG	IC	II
RWF	Increasing					
Аде	RWF (+)				Increasing	
					Age (-)	
Gender						
Tenure	Increasing	Increasing			Increasing	
	Tenure (+)	Tenure			Tenure	
		(+)			(+)	
Employment	Full-time	Full-time	Full-time	Full-time	Full-time	Full-time
Status	(+)	(+)	(+)	(+)	(+)	(+)
Employment	Manager	Increasing				Increasing
Level	(+)	EL (+)				EL (+)

6.8. Chapter summary

In summary, this chapter discussed the quantitative data analysis conducted through a rigorous application of two statistical methods, ANOVA and Logistic Regression, as well as descriptive statistics. The primary objective was to understand relationships within the dataset concerning IWB overall, and within each of the four phases of idea exploration, idea generation, idea championing and idea implementation.

The ANOVA tests found that RWF, employment status and employment level had a statistically significant effect on IWB when employees worked remotely. These results begin to shed light on the interplay between remote work arrangements, employment status, and hierarchical levels within organisational structures, and their influence on fostering IWBs. Conversely, the variables of age, gender, and tenure did not exhibit any statistically significant differences in group means concerning IWB. The logistic regression tests found that employment status, namely being full-time rather than part-time, had a statistically significant effect on the odds of IWB occurring, and within each of the four phases. Being contract/casual did not have any effect when compared to full-time.

Furthermore, the analysis delved deeper into the phases of IWB, shedding light on demographic differences within each phase. Longer tenure emerged as having a statistically significant effect on the odds of idea championing behaviours. Conversley, older was assocuated with a decrease in the likelihood of idea championing. This suggests that employees with extended organisational experience play a pivotal role in championing innovative ideas, whereas more mature workers are less likely to take on this role.

Additionally, the hierarchical dimension of employment (i.e. employment level) manifested as an important factor, with more senior employment levels exhibiting a statistically significant effect on the likelihood of idea implementation behaviours. This emphasises the vital role played by senior professionals in translating innovative concepts into innovation outcomes.

The findings of the quantitative analysis are discussed in Chapter 8. The next chapter will discuss the qualitative data analysis.

CHAPTER 7 - QUALITATIVE DATA ANALYSIS

7.1. Introduction

This chapter provides an overview of the key steps and concepts used for the qualitative data analysis process. The following sections discuss the data preparation, including the organisation and familiarisation with the data, transcription, coding, categorising, and interpretation.

The qualitative data analysis chapter delves into analysing and interpreting the data collected through 18 semi-structured interviews. This chapter aims to uncover the meaning, patterns, and themes embedded within the qualitative data, thereby addressing the research objectives and providing valuable insights into the research topic. The findings shed light on the employee experience, offering valuable insights into the employees' practices, attitudes, and behaviours.

7.2. Organisation of data

7.2.1. Data management

The initial step in the qualitative data analysis process involved carefully and accurately polishing the transcribed collected data. For each interview conducted, an automated transcription system was employed. Nonetheless, this system occasionally failed to capture the intended words accurately and incorporated numerous redundant verbal fillers such as 'umm' and 'like'. Part of the data cleaning process involved removing verbal fillers, removing introductions and farewells at the beginning and end of the interviews, and extraneous content. Only the questions and responses remained. Each interview file was renamed 'Fully Transcribed Interview_EMP#' to preserve anonymity. The researcher was labelled as 'Interviewer', and the employee was labelled as 'EMP#'. A process of familiarisation with the data was undertaken. This involved reading and re-reading transcripts to understand their contents fully.

7.2.2. Generate initial codes

Once the data was transcribed, the researcher embarked on the process of coding. Coding involves systematically identifying and labelling data segments relevant to the research objectives. These short, descriptive labels captured specific ideas and concepts within the data using an inductive process (they emerged directly from the data rather than being predefined). This process created a framework to organise and categorise the data.

7.2.3. Search for themes

As coding progressed, patterns and connections were formed, representing concepts, ideas and themes. Codes were grouped into broader categories based on the shared underlying concepts, which became the themes. Each theme represents a distinct aspect of the data that captures a central idea or concept. Each theme was defined and named. From this, a codebook was created, which included tables of themes. Further, to enhance clarity, examples of excerpts and quotes from the data illustrate each code and frequency within each theme. This demonstrates how the theme is manifested in the data.

7.3. Thematic analysis

This qualitative study aimed to understand the employee experiences with remote work and explore the factors influencing their innovative behaviour when working remotely. When understanding the factors, we want to understand those that support innovative behaviour and those that may hinder it. The AMO framework (Appelbaum et al., 2000) was used as the theoretical guide in understanding the lived experience of remote workers in Australia. This framework focuses on an individual's ability, motivation, and opportunity as key factors influencing behaviour and performance. It has been widely used in various domains including innovation research (Bos-Nehles et al., 2013; Rayner & Morgan, 2017; Shahzad et al., 2019) and has been instrumental in explaining the relationship between human resource management practices and organisational performance (Bos-Nehles et al., 2023; Doshi & Nigam, 2022; Marin-Garcia & Tomas, 2016).

The qualitative data analysis resulted in six themes, with 20 codes, presented in Table 7.1. The themes are discussed in turn in Section 7.3.1 onwards.

Table 7-1: Themes and codes

Theme	Code	Number of	Number of
		respondents	References
		mentioning	
		this code	
		(n=18)	
Resources and tools	Tangible resources for remote work	13	35
	Exploring and generating new ideas	10	36
Perspectives	Negative side of remote working	16	79
and feelings	Positive side of remote working	15	77
	Approach to risk	11	22
	Importance of remote working	7	15
Support and culture	Intangible support for remote work	15	29
	Company's support and encouragement of innovation	14	18
Work spaces	Work remotely	16	65
and tasks	Creative Work	14	53
	High Focus Work	14	32
	Work Environment	13	31
	Innovation while remote working	5	7
Communication	Interactions between coworkers	27	75
and interactions	Post-COVID work interactions	13	43
	Pre-COVID work interactions	12	27
	Impromptu connections	12	13
	In-person communications	6	7
Collaboration and	Collaborating in different contexts	14	33
relationships	Creativity and group bonding	12	21

7.3.1. Resources and tools

The thematic analysis explores the theme of 'Resources and tools', encapsulating the diverse array of tools and support the organisation offers to cultivate remote work and foster innovative thinking. Two prominent codes emerged: 'Tangible resources for remote work' and 'Exploring and generating new ideas when remote working'. The results are shown in Table 7.2.

Theme	Code	Theme	Examples
		description	
Resources	Tangible	Concrete	'So we got given the \$250 allowance
and tools	resources	support for	for office equipment.' EMP01
	for remote	remote working	
	work	success:	'We were all given money to go and
		Organisations	buy an extra screen or whatever it was
		provide tools	you needed to.' EMP02
		such as screens	
		and financial aid	'There was actually quite a few memos
			going out saying you are not to take
			any equipment home. Not even a
			mouse or a keyboard. You were not to
	D 1 ·	xx 11 .1	take anything. EMPTI
	Exploring	Unveiling the	So you have the tools now to kind of
	and	digital canvas:	set yourself two to three hours on "Do
	generating	Employees	not disturb or whatever, where you
	new ideas	share how	can just rip through the work
		remote working	distraction-free. EMP01
		and tools can	"Constant Chang Doint that's the main
		innovation and	So al our SharePoini, inal's life main
		ideo exchange	way inal we do inal. In terms of
		idea exchange	dona hyvidao agll hasiaglly on
			Microsoft Teams, ' FMP03
			'We had Zoom and we had an instant
			chat messenger messaging thing
			system that we used as well There'd
			be a lot of just quick messages you
			know, bouncing ideas.' EMP16

Table 7-2: Theme: Resources and tools

Employee responses highlighted diverse experiences regarding the provision of resources for remote work. Respondents described receiving financial allowances to acquire necessary equipment such as screens, chairs, and printers, reflecting organisational support for enhancing remote work efficiency. In contrast, some respondents expressed frustration over the lack of tools provided and the expectation that employees would self-fund their home office setups, indicating a disparity in resource allocation across organisations.

Furthermore, the analysis revealed variations in the types of support organisations offer to facilitate remote work. Some respondents mentioned a combination of monetary assistance and salary increments to cover expenses related to setting up a home office, demonstrating a multifaceted approach to resource provision. Other respondents highlighted the implementation of a working-from-home bursary for all employees, reflecting a concerted effort to mitigate financial burdens associated with remote work transitions. Respondents discussed a shift in organisational attitudes towards remote work support pre- and post-pandemic, with increased emphasis on equipping employees with necessary resources following the onset of the global health crisis.

Further insights emerge from employees' experiences regarding utilising resources and tools in remote work environments. Respondents emphasise the newfound ability to establish focused work periods through technology-enabled 'Do not disturb' settings, contrasting the potential distractions of office environments. This highlights the adaptability of remote work practices in enhancing productivity. Moreover, transitions to platforms like Microsoft Teams and Zoom facilitate seamless communication and collaboration, underscoring the organisational responsiveness to the demands of remote work.

Employees demonstrate innovative approaches to idea generation and collaboration in virtual settings. Respondents reflected on the transition from in-person creative processes to remote exchanges, indicating a shift towards leveraging digital platforms for idea dissemination. Employees utilise various strategies for maintaining effective communication with remote teams, emphasising the importance of structured meetings and asynchronous messaging channels like Slack. These testimonies underscore the significance of proactive communication strategies in sustaining team cohesion and fostering innovation in remote work settings.

However, challenges in virtual collaboration are also acknowledged. Concerns about the potential for negative interactions in persistent digital channels, highlighting the need for effective moderation and conflict resolution strategies. Collaborative tools like SharePoint and virtual meetings enable ongoing project alignment and progress tracking, indicating the potential for digital platforms to facilitate constructive collaboration despite physical distance. The narrative illuminates how the virtual realm sustains and augments collaborative potential, transcending physical barriers and geographical constraints. The findings emphasise the integral role played by these digital tools in creating an ecosystem conducive to cultivating and disseminating novel ideas.

The theme of 'Resources and tools' reveals a dynamic interplay between resource provision, technological adaptation, and innovative collaboration practices in remote work contexts. As organisations navigate the complexities of remote work, fostering a culture of adaptability and leveraging digital tools emerge as critical factors in ensuring productivity and team cohesion.

7.3.2. Perspectives and feelings

Within this thematic umbrella of 'Perspectives and feelings', participants candidly elucidated the dichotomy of the 'Negative side of remote working' juxtaposed with the 'Positive side of remote working'. They also discussed their approach to risk-taking activities and the importance of having the option to work remotely. The results are shown in Table 7.3.

Table 7-3: Theme: Perspectives and feelings

Theme	Code	Theme	Examples
		description	
Dorspostivos	Nagativa	Explaning the	'One thing that does happen I think
rerspectives	Negative side of	Exploring the	One thing that does happen I think
and reenings	side of	romoto work:	disconnect ' EMP01
	working	isolation	
	working	disrupted	'I cortainly don't think that I would
		work-life	want to work five days from home
		balance and	even day. I think that'd be
		prolonged task	isolating ' FMP03
		durations	
		revealing	'The negatives are the lack of
		nuanced	communication, the working too long
		challenges	an hour.' EMP07
	Positive	Illuminating	'Because it's a lot more private. And
	side of	the bright side	those interruptions. You get those
	remote	of remote	long breaks, of say four to five hours
	working	work:	uninterruptedSmash things out.
		enhanced	Strategy. We call it thinking time as
		focus, time	well.' EMP06
		flexibility, and	
		improved	'The positive, no travel and being
		work-life	close to home.' EMP09
		balance,	
		showcasing	'Positive says I'm far more
		diverse	productive. I'm far more creative
		advantages	and some more efficient. 'EMP11
	Approach	Navigating	'I'd say I have a healthy appetite for
	to risk	risk:	risk. ' EMP03
		Employees	
		share insights	'I take risk, if there's an
		into their risk	opportunity.' EMP06
		taking	
		tendencies	'I'm not a big risk taker.' EMP10

Importance	Perceptions	'I'd say it's quite important to
of remote	unfold:	meit's important to have the
working	Employees	flexibility.' EMP01
	reflect on	
	remote work's	'I know I want to keep working from
	impact on	home. I just enjoy, it gives you
	work dynamics	balance.' EMP03
		'I think it's very important. I think
		it's all about the hybrid balance and
		giving you that feeling of
		empowerment as well.' EMP06

The thematic analysis on the 'Negative side of remote working' unveils various challenges and concerns voiced by employees navigating remote work arrangements. Central to these concerns is the feeling of isolation. The absence of physical interaction with colleagues and the separation from the traditional office environment contribute to feelings of social disconnection and solitude, underscoring the importance of interpersonal dynamics in the workplace.

Remote work blurs the boundaries between professional and personal life, exacerbating challenges related to work-life balance and disengagement. Respondents highlight the difficulty in disconnecting from work, with the home environment facilitating prolonged working hours and impeding the delineation of leisure time. Respondents acknowledge the temptation to overwork, attributing it to the accessibility and omnipresence of work-related tasks in the home setting. This sense of home life conflict is compounded by practical constraints, such as inadequate workspace or noisy environments, further exacerbating feelings of isolation and marginalisation within the organisational context.

Additionally, communication gaps and feelings of invisibility within the organisation emerge as significant concerns for remote workers. Respondents lament the lack of informal interactions and the risk of being overlooked or forgotten by colleagues and supervisors.

Remote work poses challenges in maintaining social connections and fostering camaraderie among team members. Respondents reflect on the formalisation of communication channels and the absence of casual interactions that characterise office environments, highlighting the value of social interactions in nurturing a sense of belonging and cohesion within teams.

Conversely, the 'Positive side of remote working' reveals many benefits and advantages perceived by employees navigating remote work arrangements. Central to these positive experiences is the liberation from the daily commute. The elimination of travel time not only affords individuals additional leisure time but also reduces stress associated with commuting, thereby enhancing overall wellbeing. Moreover, remote work allows employees to structure their day according to personal preferences. This flexibility extends beyond work hours, allowing individuals to seamlessly integrate family responsibilities and personal pursuits into their daily routines, fostering a more balanced lifestyle.

Remote work environments offer a conducive space for focus and productivity. The absence of office distractions enables uninterrupted work periods, facilitating deep concentration and enhanced efficiency. This privacy and autonomy over one's workspace can be particularly beneficial for individuals with specific sensitivities or neurological conditions. Additionally, remote work empowers employees to approach tasks creatively and strategically, with the ability to carve out dedicated 'thinking time' for problemsolving and innovation.

Employees perceive the integration of work and personal life positively, as it allows for greater work-life balance and flexibility. Respondents appreciate the opportunity to manage multiple roles and responsibilities, such as caregiving and education, without the constraints of traditional office hours. This integration fosters a sense of empowerment and autonomy, enabling individuals to tailor their work schedules to accommodate diverse life priorities and pursuits.

The data on 'Approach to risk' uncovers a diverse spectrum of employee attitudes and behaviours when confronted with risk-taking opportunities. Some respondents identify themselves as moderately inclined towards risk-taking, adopting a pragmatic approach that balances the potential rewards with the associated uncertainties. This stance reflects a calculated willingness to embrace new ventures and concepts while exercising caution to mitigate potential downsides. They are willing to seize opportunities and take calculated risks, demonstrating a proactive attitude towards innovation and experimentation.

Conversely, other employees lean towards a more risk-averse disposition. These individuals prioritise careful planning and consideration before embarking on ventures, meticulously weighing the potential outcomes and risks. While not inherently opposed to taking risks, they prefer measured and calculated decision-making processes, minimising the likelihood of adverse consequences. Respondents emphasised the importance of evaluating the potential returns against the associated risks before committing to a course of action. This approach reflects a strategic mindset that prioritises long-term sustainability and risk management.

Other employees articulate a nuanced perspective on risk-taking, highlighting the necessity of adopting a managed approach, particularly in contexts involving financial investments or behaviour change initiatives. This stance underscores the importance of striking a delicate balance between innovation and prudence, recognising the inherent uncertainties while leveraging opportunities for growth and development.

The 'Importance of remote working' analysis uncovers contrasting attitudes towards the importance of flexibility in work arrangements among employees. While some individuals emphasise the significance of flexibility in achieving a balanced worklife integration and empowering employees to make choices that suit their preferences, others underscore the necessity of embracing flexibility as a strategic imperative for remaining competitive in the modern business landscape. These perspectives highlight the recognition of flexibility as a key factor in attracting and retaining talent, fostering employee satisfaction, and adapting to evolving work preferences and expectations.

Employees emphasise the tangible benefits of flexibility, such as avoiding lengthy commutes, maximising productivity, and allocating more time to personal pursuits and family commitments. For these individuals, flexibility represents a practical necessity and a source of fulfilment and happiness, enabling them to pursue career advancement while maintaining a fulfilling personal life. This perspective reflects a pragmatic approach that prioritises the tangible benefits of flexibility in enhancing overall wellbeing and job satisfaction.

Some respondents advocate for a hybrid approach that combines the advantages of remote and office-based work, emphasising the importance of striking a balance between flexibility and structure. Respondents emphasised the need for businesses to adapt to changing workforce preferences and market dynamics by offering greater flexibility in work arrangements. This perspective underscores the recognition of flexibility as a strategic imperative for remaining agile and responsive to employee needs and market demands, thereby ensuring organisational resilience and competitiveness.

The theme 'Perspectives and feelings' highlights the complex nature of employee experiences and attitudes towards remote work, risk-taking, and flexibility. It underlines the need for organisations to adopt holistic approaches that prioritise employee wellbeing, foster a culture of innovation, and adapt to evolving workforce preferences and expectations in the dynamic and ever-changing landscape of work.

7.3.3. Support and culture

The thematic analysis navigates the realm of 'Intangible support for remote working', wherein participants elaborate on measures addressing mental wellbeing and resources. Including wellbeing programs, psychological support, and initiatives aimed at enhancing mental health underscores a holistic approach to remote work support, acknowledging the intricacies of both the physical and mental dimensions of the remote work experience. Complementing this, the organisation's 'Support and encouragement of innovation' delves into the programs and organisational climate which fosters innovation. These codes encapsulate the diverse mechanisms through which organisations provide assistance and create conducive environments for employees, particularly in remote work facilitation and innovation promotion. The results are shown in Table 7.4. Table 7-4: Support and culture theme: Intangible support for remote work

Theme	Code	Theme	Examples
		description	
Support	Intangible	Nurturing	<i>`we've got mental health type</i>
and	support for	wellbeing	programs running. We try and
culture	remote work	remotely:	make sure that we do some team
		Insights into	building type stuff fairly regularly.'
		mental health	EMP02
		support,	' they've get a subscription to
		programs and	iney ve goi a subscription to
		programs, and	some employees service of
		resources for a	us It's called Re Well There's an
		balanced remote	actual company that they outsource
		work experience	it to.' EMP03
		1	
			'There was a lot of mental health
			support. Lot of mental health
			supporta lot of contact, a lot of
			checking in, all that kind of stuff.'.
			EMP06
	Company's	Fostering and	'So we definitely used to have a big
	support and	promoting	innovation program that was
	encouragement	innovation:	formally rewarded, but I don't think
	of innovation	proactive	that's necessarily happening
		encouragement,	anymore. EMP01
		shaping a	'We run an international run
		supportive ethos	innovation competition And some
		supportive etitos	guys out of our group went through
			the national final and they won a
			prize for that.' EMP02
			'I know we have an office
			awardAnd every quarter, there's
			one person from each service line
			that is selected as the winner of this
			award. And it's assessed on a
			submission basis.' EMP03

The data on 'Intangible support for remote work' elucidates the comprehensive efforts undertaken by organisations to prioritise employee wellbeing and mental health for employees working remotely. Employees across various contexts highlight the proliferation of mental health support initiatives, such as Employee Assistance Programs (EAPs), wellness programs, and resilience-building workshops, as integral components of organisational support frameworks. These initiatives underscore the proactive approach adopted by organisations in addressing the psychological and emotional challenges associated with remote work, fostering a culture of care and empathy within the workforce.

Respondents discuss how their organisations intensified their focus on promoting mental health awareness and resilience-building strategies. By equipping employees with tools and resources to manage stress, anxiety, and isolation effectively, organisations empower individuals to navigate the complexities of remote work with greater resilience and adaptability.

Crucially, the thematic analysis underscores the importance of fostering open communication channels and social connections to mitigate feelings of isolation and enhance social support networks among remote workers. Organisations prioritise regular check-ins, team-building activities, and virtual social gatherings to foster a sense of belonging and camaraderie among employees despite physical distancing measures. This emphasis on maintaining social connections underscores the recognition of social support as a vital determinant of employee wellbeing and job satisfaction in remote work settings.

The data on 'Company's support and encouragement of innovation' reveals a spectrum of approaches organisations adopt to foster a culture of innovation and recognise employees' contributions to creative problem-solving and enhancement. Some respondents highlight formalised recognition programs and awards to celebrate innovative ideas and initiatives within the organisation. These programs often entail a structured evaluation process, wherein individuals are encouraged to submit proposals or showcase their innovative contributions for assessment, thereby providing a platform for showcasing novel solutions and driving organisational change.
Conversely, other respondents underscore the informal avenues through which innovation is encouraged and supported within the organisation. These individuals emphasise the importance of autonomy and freedom in fostering innovation, wherein employees are empowered to explore new ideas, experiment with novel approaches, and drive change in their respective roles or departments. Such decentralised approaches to innovation underscore the organisation's commitment to nurturing a bottom-up culture of innovation, wherein creativity and ingenuity are valued and rewarded irrespective of formalised recognition programs.

The analysis highlights the role of organisational initiatives and platforms in promoting innovation and knowledge sharing among employees. For instance, one respondent mentions participation in international innovation competitions and conferences, wherein employees have the opportunity to showcase their innovative projects on a global stage and gain recognition for their contributions. Additionally, initiatives such as innovation newsletters and forums facilitate the dissemination of best practices, success stories, and innovative ideas, thereby fostering a collaborative and learning-oriented environment conducive to continuous improvement and innovation.

The theme 'Support and culture' emphasises an organisation's commitment to nurturing a dynamic and inclusive workplace culture that values employee wellbeing, fosters creativity and promotes organisational resilience in the face of working remotely.

7.3.4. Workspaces and tasks

In examining the dynamics of workspaces and tasks, the analysis delves into the multifaceted interplay between individuals' preferences for where work is conducted, and the nature of tasks being performed. This exploration entails dissecting various dimensions, from the rise of remote work to the nuances of creativity, focus, and innovation within different work environments. Five distinct codes emerge: 'Work remotely', reflecting the experience of decentralised work arrangements; 'Creative work', highlighting the symbiotic relationship between environment and imaginative output; 'High focus work', elucidating the significance of conducive settings for concentrated tasks; 'Work environment' and 'Innovation while working remotely', examining the potential for new ideas and problem-solving to emerge amidst remote work setups. The results are shown in Table 7.5.

Table 7-5: Theme: Workspaces and tasks

Theme	Code	Theme	Examples
		description	
Work spaces and tasks	Working remotely	Work transcends boundaries: Employees share experiences of remote work frequency	'I very rarely work remotely. I probably work probably a couple of days a month, if that.' EMP02 'I'd probably be fully remote. As in internal meetings I like, like I go to the strategy days, which are in person or you know, things like that. But outside of that, yeah, like I, I'd happily be fully remote.' EMP13 'Remote working covid has been a
			godsend for me. I'll never work full-time in an office again.' EMP14
	Creative work	Employees share strategies for private focus, autonomy, and unconventional	'I do my best thinking when I'm laying in bed. And quite often, and maybe that's because that's the only time of the day I actually manage to have peace and quiet and thinking time.' EMP02
		spaces fostering creative and problem- solving	'That 10 minutes in the shower with the hot water and that silence, just the water running, gives you that time to think.' EMP06
		thinking in a digital landscape	'Working remotely and in the park that's just around the corner sitting under a tree, like lying under a tree sometimes or walking around and just getting, getting a bit of fresh air and space.' EMP10
	High focus work	Employees share strategies and spaces for high concentration work	'I will a hundred percent do my deep thinking at home because there is no way on God's green earth it's happening in the office.' EMP11 'Definitely that sort of thing is better from home.' EMP16

		'I would always book a meeting roomI wanna be away from people away from the buzz of an open plan office.' EMP17
Work environment	Employees articulate preferences for work environments, highlighting the influence on productivity, energy, and interpersonal dynamics	 'There's still situations where especially the younger people really missed out by not having those sort of water cooler conversations you get in the office.' EMP02 'I think it's just easier at home. I've got a dual screen set up. Everything's set up, ready to go. It's quiet.' EMP07 'I think some things are a little bit easier to do when you're in the office. So sometimes, you know, you just need to check in a process with someone.' EMP17
Innovation while remote working	Employees delve into fostering innovation in the virtual workspace	'But the extent that you do identify a problem at home when you're at home, I think it's easier to find the time to get the head space to think about the solution than when you're in the office.' EMP02 '[I feel more creative] outside the office environment. It's the space. And because, well, my mind is at rest, I think.' EMP09 'If you're in meetings personally and you had, the detractors in the room, you would've been able to kind of move them and change 'em a little bit.' EMP12

'Remote working' has become increasingly prevalent, with many respondents noting a shift towards working from home or other remote locations. While some individuals had already established home offices prior to the COVID-19 pandemic, others adapted by setting up dedicated workspaces within their homes during lockdown periods. The flexibility afforded by remote work allowed individuals to customise their work environments to suit their needs, whether it involved creating a home office or occasionally working from cafes to break the routine.

Preferences regarding remote work varied among respondents, with some expressing a preference for a fully remote setup, while others preferred a hybrid model that balanced office and remote workdays. For many, the ability to work remotely offered a better work-life balance, particularly for those juggling childcare responsibilities or seeking to avoid lengthy commutes. This flexibility allowed individuals to allocate their time more efficiently and tailor their work arrangements to fit their lifestyle.

The hybrid model emerged as a popular choice, with many respondents indicating a desire to split their time between office and remote workdays. This approach offered the benefits of face-to-face interaction and collaboration in the office, combined with the flexibility and autonomy of remote work.

The data on 'creative work' highlighted various environments and activities that fostered creativity in respondents' work processes. Some emphasised the importance of having dedicated spaces for creative thinking, such as breakout rooms in the office or home offices free from distractions. Others found that unconventional settings, like lying in bed or sitting under a tree in the park, stimulated their creativity by providing mental space and tranquillity.

Routine activities such as showering or driving also emerged as conducive moments for generating creative ideas. The solitude and rhythmic nature of these activities allowed individuals to engage in reflective thinking and problem-solving. Similarly, physical activities such as riding a bike or going for a walk provided opportunities to clear the mind and stimulate fresh perspectives.

For some respondents, switching between focused work and breaks was essential for nurturing creativity. They described the need to step away from screens and engage in activities such as taking walks or conversing with colleagues to generate new ideas. Additionally, visits to cafes provided a change of scenery and ambience conducive to creative thinking, offering a break from the monotony of the office or home environment.

The data shows the importance of creating environments and routines that support creative thinking and problem-solving. Whether through intentional spaces, routine

activities, or changes in scenery, individuals sought ways to cultivate creativity in their work processes.

When undertaking 'high focus work' the respondents highlighted the importance of isolating themselves to engage in deep thinking and creative work. Many expressed a preference for working from home when they needed to apply intense focus to complex tasks. They emphasised the lack of distractions and interruptions in a home environment compared to the bustling atmosphere of an office. Some even mentioned setting up home offices with doors they could shut to minimise disruptions and maintain concentration.

For those who did work in office settings, they often sought out secluded spaces such as meeting rooms or quiet rooms away from the main workspace. These areas provided the solitude necessary for deep concentration and strategic thinking, especially when tackling high-level tasks like strategizing or planning long-term initiatives. The ability to escape the buzz of an open-plan office was crucial for maintaining productivity and creativity during these focused work sessions. Whether at home or in designated quiet spaces within the office, employees valued the opportunity to isolate themselves when engaging in tasks that required intense concentration and focus.

The responses regarding 'work environments' highlighted a variety of perspectives on the benefits and challenges of different settings. Many respondents expressed the value of face-to-face interactions in the office, emphasising the importance of water cooler conversations and the ability to build relationships with colleagues in person. They noted that starting a new job remotely could be particularly challenging due to the lack of opportunities to meet people in real life and get a feel for the workplace culture.

However, some respondents also acknowledged the convenience and efficiency of working from home, citing factors such as having a dual-screen setup, minimal distractions, and the ability to focus without interruptions. Others mentioned the preference for a dedicated workspace and the challenges of adapting to coworking spaces or hot-desking arrangements. Despite the advantages of remote work, some respondents highlighted the limitations of virtual meetings for high-level discussions and strategy sessions, noting that face-to-face interactions were often more effective in these contexts. There was recognition of the energy and collaboration that comes from working alongside colleagues in the office. Some respondents mentioned the difficulty of managing team performance remotely and the need to navigate permission for interruptions when working from home. For those in roles requiring people management, the office environment provided a source of energy and connection that was difficult to replicate remotely.

The responses underscored the complex interplay between physical work environments, interpersonal dynamics, and individual work preferences. While remote work offered advantages in terms of flexibility and focus, many respondents still saw value the social and collaborative aspects of working in an office setting.

The data on 'innovation while remote working' found the shift to remote work during the COVID-19 pandemic catalysed innovation in unexpected ways. Without the constraints of traditional office settings, employees found themselves empowered to streamline processes and implement new ideas with unprecedented speed. Freed from the distractions of office environments, individuals could focus more intently on problemsolving and creative thinking. This enhanced ability to concentrate has accelerated the development and approval of concepts that might otherwise have languished for years due to bureaucratic hurdles or organisational inertia. Employees report being able to generate solutions more quickly and efficiently, without the need to labour over problems late into the night.

However, the transition to remote work also presented challenges, particularly in terms of collaboration and communication. In the absence of in-person meetings, some employees found it harder to sway detractors or engage in dynamic brainstorming sessions. Virtual collaboration tools, while valuable, may not fully replicate the spontaneity and interactivity of face-to-face interactions. For some, the lack of physical whiteboards and collaborative spaces has been keenly felt, as these tools are seen as catalysts for creative thinking and problem-solving. Nevertheless, innovative solutions have emerged, such as scheduling virtual 'whiteboard sessions' to encourage creativity and idea-sharing among team members.

Despite these challenges, many employees find that working outside the traditional office environment enhances their creativity and problem-solving abilities.

The absence of workplace distractions allows individuals to achieve a state of mental clarity conducive to innovative thinking. Away from the hustle and bustle of office life, employees can tap into their creativity more freely, often finding inspiration in moments of rest or quiet reflection. Moreover, virtual collaboration platforms provide opportunities for asynchronous brainstorming and idea-sharing, enabling individuals to develop and refine their ideas collaboratively, even in remote settings.

The theme of 'Workspaces and tasks' provides comprehensive insights into the dynamics of remote work and its impact on creativity and innovation. The analysis reveals the complexity of remote work dynamics, encompassing individual preferences, work practices, organisational support, and technological considerations.

7.3.5. Communication and interactions

In the evolving landscape of modern workplaces, effective communication and interactions serve as fundamental pillars for fostering collaboration, productivity, and organisational cohesion. Through an examination of five distinct codes, 'Interactions between coworkers', 'Post-COVID work interactions', 'Pre-COVID work interactions', 'Impromptu connections', and 'In-person communications', participants discuss the multifaceted nature of communication and interactions within work environments, exploring workplace relationships, shedding light on the shifts, challenges, and opportunities that have emerged in the wake of remote work, virtual interactions, and hybrid working. The results are shown in Table 7.6.

Theme	Code	Theme	Examples
		description	
Communication	Interactions	Digital	'It is quite a lot different. So
and	between	dynamics in	if we're in the office, we
interactions	coworkers	focus:	might just all move our
		Employees	chairs over somewhere or
		explore	get a meeting room and kind
		interactions in	of do a similar thing. But I
		virtual spaces	think it's just a bit more
		and the use of	rigid and formal in the sense
		technology	that you have to find a time

 Table 7-6: Theme: Communication and interactions

		that lines up with everyone's
		calendars to schedule in a
		video callyou can just go
		and knock on the door. But
		now you have to find a time
		that works on both our
		calendars. It's a bit more
		rigid and formalised.'
		EMP01
		'So I sort of rely on being in
		a room in a meeting and
		sitting next to someone and
		striking a bit of a
		conversation before the
		meeting starts or after the
		meeting finishes. which you
		don't get to do on Zoom.'
		EMP02
		'If I was working from
		home. I would've just
		communicated as I would if
		I was in the office I'd still
		have that conversation It
		wouldn't be a barrier for
		m_{P} ' FMP07
		'We thought the board
		meeting was going to be
		live he was on his phone he
		didn't have the documents
		And so it made it a harder
		You couldn't see his face
		you couldn't read as much
		hody language into the
		meeting, 'EMP08
Post-COVID	Adapting to	'If there's anything we're
work	new normal.	not hearing it nrohably is if
interactions	Employees	I've out say innior staff that
moracions	discuss shifts	maybe don't communicate
	in work	unwards as much for
	interactions	whatever reasons That's
	meracions	whatever reasons. That s

post-COVID, navigating changing circumstances and hybridwhat I've found to be the main information gap." EMP01and hybrid modes of collaborationT think because we do them and I do make them turn their cameras on, it's like you're in the room with them if you 're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07If feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like pople's behaviour or how they 're receiving an idea.' EMP13So whether we 're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			
navigating changing circumstances and hybridmain information gap.' EMP01and hybrid modes of collaboration1 think because we do them and I do make them turn their cameras on, it's like you're in the room with them if you 're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07''I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the muances of like people's behaviour or how they're receiving an idea.' EMP13''So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people I don't see are those people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		post-COVID,	what I've found to be the
changing circumstances and hybrid modes of collaborationEMP01T think because we do them modes of collaborationT think because we do them and I do make them turn their cameras on, it's like you're in the room with them if you're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07T feel like I can't read the muances of what's going on in the roomCause I just like people's behaviour or how they're receiving an idea.' EMP13So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		navigating	main information gap. '
circumstances and hybrid modes of collaboration'I think because we do them and I do make them turn their cameras on, it's like you're in the room with them if you're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea. 'EMP13'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		changing	EMP01
and hybrid modes of collaborationThink because we do them and I do make them turn their cameras on, it's like you're in the room with them if you're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07"Ifeel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13"So whether we 're in the office or we 're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		circumstances	
modes of collaborationand I do make them turn their cameras on, it's like you 're in the room with them if you 're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13'So whether we 're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		and hybrid	'I think because we do them
collaborationtheir cameras on, it's like you're in the room with them if you're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07'I feel like I can't read the nuances of what's going on in the roomCause I just like people's behaviour or how they 're receiving an idea.' EMP13'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		modes of	and I do make them turn
you're in the room with them if you're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16		collaboration	their cameras on, it's like
them if you're having a conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea. 'EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			you're in the room with
conversation. So back in the days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff. I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we 're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			them if you're having a
days when it was only a phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07Ifeel like I can't read the nuances of what's going on in the roomCause I just like people's behaviour or how they're receiving an idea.' EMP13'So whether we 're in the office or we 're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.'			conversation. So back in the
phone call and you couldn't see them, but now that we've got Teams and got video conferencing and stuff, 1 think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			days when it was only a
see them, but now that we've got Teams and got video conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they 're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			phone call and you couldn't
got Teams and got video conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			see them, but now that we've
conferencing and stuff, I think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			got Teams and got video
think you can have a much more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			conferencing and stuff. I
more open and honest and transparent conversation and get someone's real thoughts around an idea.' EMP07'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people Who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			think vou can have a much
transparent conversation and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			more open and honest and
and get someone's real thoughts around an idea.' EMP07 'I feel like I can't read the muances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			transparent conversation
thoughts around an idea.' <i>EMP07</i> 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' <i>EMP13</i> 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' <i>EMP16</i>			and get someone's real
EMP07 'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.'			thoughts around an idea.'
¹ <i>I</i> feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 ⁵ So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			EMP07
'I feel like I can't read the nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.'			
nuances of what's going on in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see are those people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			'I feel like I can't read the
in the roomCause I just in the roomCause I just like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			nuances of what's going on
like can't see the nuances of like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people Who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			in the room Cause I just
like people's behaviour or how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			like can't see the nuances of
how they're receiving an idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			like people's behaviour or
idea.' EMP13 'So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			how they're receiving an
So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			idea.' EMP13
So whether we're in the office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			
office or we're online or you know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			'So whether we're in the
know, mixture of both, we meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			office or we're online or vou
meet every day and we have a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			know, mixture of both, we
a chat. But I guess what I don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			meet every day and we have
don't see or who the people I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			a chat. But I guess what I
I don't see are those people who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			don't see or who the people
who are not in my direct team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			I don't see are those neonle
team who I might have had a chat to in the kitchen or walking past their pod.' EMP16			who are not in my direct
a chat to in the kitchen or walking past their pod.' EMP16			team who I might have had
walking past their pod.' EMP16			a chat to in the kitchen or
EMP16			walking nast their nod '
			EMP16

work interactionsthe past: Employces reminisce about pre- COVID work interactionsoffice, you could just kind of eyeball everyone and say 'Hey, are you free? Do you about pre- Want to grab a room and we'll go through it together?' EMP01COVID work interactions''If I want to have a conversation with a client, I''Il pop into a meeting room to have conversation because I don't feel that it's appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problem''If I' m in the office I need to be at the computer to be considered working,' EMP15Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual dialogues: Employees share experiences of impromptu opportunities for sharing idcas in different work contexts''There is probably not as much (creativity) as if I had been in the office solving unfolds as well as casual casual dialogues: Employees share experiences of impromptu opportunities for sharing idcas in working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I'' working remotely]['I''' working remotely]['I''' working remotely]['I'''' working remotely]['I'''' working remotely]['I'''''''''' working remotely]['I'''''''''''''''''''''''''''''''''	Pre-COVID	Reflecting on	'Whereas if you're in the
interactionsEmployees reminisce about pre- COVID work interactionseyeball everyone and say 'Hey, are you free? Do you want to grab a room and COVID work interactionsCOVID work interactions'Hey, are you free? Do you want to grab a room and COVID work interactions'If J want to have a conversation with a client, I'll pop into a meeting room to have conversation because I don't feel that it's appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problem solving unfolds a swell as casual dialogues: Employces share experiences of impromptu opportunities for sharing ideas in different work contextsSpontaneous contextsImpromptu connectionsSpontaneous creativity or problem solving unfolds a swell as casualThere is probably not as much [creativity] as if I haa been in the office becauseit's far easier just to hop across to your collague's desk and say "What do you think about Employces this?" Rather than just the act of dialling someone is a filter.' EMP02Typically it's just an impromptu opportunities for sharing ideas in working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?"'EMP03	work	the past:	office, you could just kind of
reminisce about pre- COVID work interactions'Hey, are you free? Do you want to grab a room and We'll go through it together?' EMP01'If I want to have a conversation with a client, I'll pop into a meeting proof to have conversation because I don't feel that it's appropriate for people to bo listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontancous creativity or problem solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I haa been in the office been un the office been un the office solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I haa been in the office been un the office been un the officeEmployees share ideas in different work contexts'Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the ent "Oh gee, do you have two minutes for me to run something past you?''EMP03	interactions	Employees	eyeball everyone and say
about pre- COVID work interactionswant to grab a room and we 'll go through it together?' EMP01'If I want to have a conversation with a client, I'll pop into a meeting room to have conversation because I don't feel that it's appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as easual dialogues: Employees share easual impromptu contextsThere is probably not as much [creativity] as if I had been in the office to hop across to your casual colleague's desk and say "What do you think about this?" Rather than just the act of dialling someone is a experiences of filter.' EMP02Impromptu contexts'Typically it's just an impromptu opportunities for sharing ideas in different work contextsImpromptu oupportunities for sharing ideas in working remotely!if I're got a meeting with my manager, I might just tack on the ent "Ohge do you have two minutes for me to run something past you?''EMP03		reminisce	'Hey, are you free? Do you
COVID work interactionswe 'll go through it together?' EMP01'If I want to have a conversation with a client, I'll pop into a meeting room to have conversation because I don't feel that it's appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problem solving unfolds a swell as casual casual experiences of impromptu opportunitiesThere is probably not as much (creativity) as if I had been in the office solving unfolds as well as casual connectionsImpromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual casual connectionsImpromptu connectionsThere is probably not as much (creativity) as if I had been in the office solving unfolds as well as casual colleague's desk and say "What do you think about this?" Rather than just the act of dialling someone is a experiences of impromptu opportunities for sharing ideas in different work contextsTypically it's just an ingromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the ent ''O hype do you have two minutes for me to run something past you?''EMP03		about pre-	want to grab a room and
interactionstogether?' EMP01If I want to have a conversation with a client, I'll pop into a meeting room to have conversation because I don't feel that it's appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontancous creativity or problem as well as casual casual colleague's desk and say dialogues: Employces share experiences of filter.' EMP02Impromptu connectionsSpontancous creativity or problem as well as colleague's desk and say dialogues: Typically it's just an impromptu opportunities for sharing ideas in different work contextsTypically it's just an impromptu contextsTypically it's just an impromptu context and on the end "Oh gee, do you have two minutes for me to run something past you?''EMP03		COVID work	we'll go through it
Impromptu connectionsSpontaneous creativity or problem'If I want to have a conversation with a client, I'II pop into a meeting room to have conversation because I don't feel that it's appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual dialogues:'If ind if I'm in the office I need to be at the computer to be considered working.' EMP15Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual dialogues:'There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual to hop across to your casual idialogues:Employces share opportunities for sharing ideas in different work contexts'Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end ''Oh gee, do you have two minues for me to run something past you?'''EMP03		interactions	together?' EMP01
appropriate for people to be listening to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problemImpromptu connectionsSpontaneous creativity or problemImpromptu connectionsSpontaneous creativity or problemImpromptu connectionsSpontaneous creativity or problemImpromptu connectionsSpontaneous creativity or problemImpromptu connectionsSpontaneous creativity or problemImpromptu connectionsCreativity or problemImpromptu casual dialogues:There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual casual colleague's desk and say "What do you think about Employees this?" Rather than just the share act of dialling someone is a experiences of impromptu opportunities for sharing ideas in different work contextsImpromptu opportunities to sharing vorking remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?"'EMP03			'If I want to have a conversation with a client, I'll pop into a meeting room to have conversation because I don't feel that it's
Insteming to everything I say or me being disruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problemThere is probably not as much [creativity] as if I had been in the officeImpromptu connectionsSpontaneous creativity or problemThere is probably not as if a easier just to hop across to your 			appropriate for people to be
or me being alsruptive to them.' EMP02Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual dialogues:Employees share experiences of impromptu opportunities for sharing ideas in different work contextsThere is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual for dialling someone is a filter.' EMP02Typically it's just an impromptu opportunities for sharing ideas in different work contextsTypically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?''EMP03			listening to everything I say
Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual dialogues:Employees share experiences of impromptu opportunities for sharing ideas in different work contexts"There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual act of dialling someone is a filter.' EMP02"Typically it's just an impromptu opportunities for sharing ideas in different work contexts"Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?" 'EMP03			or me being alsruptive to
Impromptu connectionsSpontaneous creativity or problem'I find if I'm in the office I need to be at the computer to be considered working.' EMP15Impromptu connectionsSpontaneous creativity or problem'There is probably not as much [creativity] as if I had been in the office becauseit's far easier just as well as casual dialogues: Employees share experiences of impromptu opportunities'There is probably not as much [creativity] as if I had been in the office becauseit's far easier just as well as to hop across to your casual dialogues: Employees this?' Rather than just the share experiences of impromptu opportunities'Typically it's just an impromptu opportunities for sharing ideas in different work contexts'Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?'''EMP03			inem. EMP02
Impromptu connectionsSpontaneous creativity or problem <i>There is probably not as</i> much [creativity] as if I had been in the officeImpromptu connectionsSpontaneous creativity or problem <i>There is probably not as</i> much [creativity] as if I had been in the officesolving unfolds as well as casual dialogues: <i>There is probably not as</i> much [creativity] as if I had been in the officeSpontaneous problem solving unfolds as well as casual dialogues: <i>There is probably not as</i> much [creativity] as if I had been in the officeSpontaneous casual dialogues: <i>There is probably not as</i> much [creativity] as if I had been in the officeSpontaneous casual dialogues: <i>There is probably not as</i> much [creativity] as if I had been in the officeSpontaneous casual <i>Colleague's desk and say</i> this?" Rather than just the share experiences of impromptu opportunitiesSpontaneous for sharing ideas in different work contexts <i>Typically it's just an</i> impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?"'EMP03			<i>'I find if I'm in the office I</i>
Impromptu connectionsSpontaneous creativity or problem'There is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual dialogues:''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual dialogues:''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual dialogues:''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual dialogues:''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual''Uhere is probably not as much [creativity] as if I had been in the officesolving unfolds as well as casual''Uhere is probably not as motopacesit's far easier just to hop across to your casual to hop across to your casual this?'' Rather than just the share opportunitiesfor sharing ideas in different work contexts''Uhere manager, I might just tack on the end ''Oh gee, do you have two minutes for me to run something past you?'''EMP03			need to be at the computer
Impromptu connectionsSpontaneous creativity or problem solving unfolds as well as casual dialogues: experiences of impromptu opportunities'There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual colleague's desk and say dialogues: "What do you think about Employees share experiences of impromptu opportunities'There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your casual colleague's desk and say dialogues: "What do you think about Employees this?" Rather than just the share act of dialling someone is a filter.' EMP02impromptu opportunities ideas in different work contexts'Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?'''EMP03			to be considered working.'
Impromptu connectionsSpontaneous creativity or problemThere is probably not as much [creativity] as if I had been in the office solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I had been in the office solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I had been in the office solving unfolds as well as casual dialogues:There is probably not as much [creativity] as if I had been in the office solving unfolds as well as colleague's desk and say "What do you think about this?" Rather than just the share act of dialling someone is a filter.' EMP02Share opportunitiesact of dialling someone is a filter.' EMP02Typically it's just an ideas in different work contexts"Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?"'EMP03			EMP15
	Impromptu connections	Spontaneous creativity or problem solving unfolds as well as casual dialogues: Employees share experiences of impromptu opportunities for sharing ideas in different work contexts	'There is probably not as much [creativity] as if I had been in the office becauseit's far easier just to hop across to your colleague's desk and say "What do you think about this?" Rather than just the act of dialling someone is a filter.' EMP02 'Typically it's just an impromptu call [when working remotely]if I've got a meeting with my manager, I might just tack on the end "Oh gee, do you have two minutes for me to run something past you?"'EMP03

		'I love it when I go into the office because I get to see my team and that typical "Let's go for a coffee" and just having that informal, that chit chat where you get to connect on a more personal and a human level so that you're building up that trust and rapport.' EMP09
In-person communications	Employees reminisce on in-office interactions, exploring the nuances of face-to-face communication in traditional work settings	 'you could read a person's energy a lot easier face-to-face than you can on the camera.' EMP06 'I think we got a better outcome because it was face-to-face than online. You could deal with some of the emotion, some of the tensionsyou can see people's reactions.' EMP08 'So you can't really tell what people are thinking, but in a room I can tell what they're thinking without them having to say as much.' EMP17

The data on 'interactions between coworkers' reveal both challenges and opportunities in the transition to remote work. Respondents noted a shift in the dynamics of communication, with traditional office interactions being replaced by digital platforms such as Microsoft Teams and Zoom. While these tools facilitate communication with colleagues, when necessary, they also introduce new barriers to spontaneous interactions and informal conversations that occur naturally in the office environment.

Remote work has affected the learning opportunities for junior staff members who miss out on overhearing conversations and absorbing knowledge from more experienced colleagues. The absence of physical proximity has made it difficult for junior staff to stay engaged and learn from their peers, leading to a regression in their learning curve.

Additionally, remote work has changed the nature of collaboration, making it more formalised and structured. While digital platforms allow for virtual meetings and screen sharing, the spontaneity of in-person interactions is lost. Scheduling video calls becomes necessary, leading to a more rigid and formalised approach to collaboration.

Respondents also highlighted the challenges of maintaining connections with colleagues outside of formal work-related interactions. Without the opportunity for casual encounters in the office, building relationships and staying connected with colleagues requires more deliberate effort. Some respondents expressed reluctance to schedule informal catch-ups, preferring the spontaneous interactions that occur naturally in the office setting.

However, despite these challenges, respondents identified strategies to maintain connections and foster collaboration in a remote work environment. This includes regular video calls with specific team members, deliberate efforts to reach out and communicate, and creating structured opportunities for team dialogue and collaboration. By adapting to the new reality of remote work and implementing strategies to overcome its limitations, teams can continue to collaborate effectively and maintain strong working relationships, even in a virtual setting.

The data on 'post-COVID work interactions' shed light on the evolving nature of communication and collaboration in the modern workplace. Respondents highlighted both benefits and challenges associated with remote work and virtual interactions.

One recurring theme was the speed and accessibility of information in a remote work environment. With the advent of digital tools like Microsoft Teams and video conferencing platforms, information sharing has become more efficient and immediate. Younger employees, in particular, are adept at multitasking and utilising real-time resources during discussions, enhancing the speed of decision-making and problemsolving.

However, despite the efficiency of virtual interactions, there are concerns about the loss of nuanced communication cues present in face-to-face interactions. Respondents noted challenges in reading nonverbal cues and gauging reactions during virtual meetings, which can impact the depth of understanding and collaboration within teams.

Furthermore, the shift to remote work has changed the dynamics of information flow within organisations. While firm communication and developments are still disseminated effectively, there may be gaps in information for junior staff who do not communicate upward as frequently. The lack of casual encounters with colleagues outside one's direct team also contributes to a sense of isolation and reduced visibility across the organisation.

Respondents also highlighted the importance of creating opportunities for personal connection and collaboration in the virtual workspace. Efforts to turn on cameras during meetings and schedule regular check-ins help maintain a sense of camaraderie and transparency within teams. Additionally, organisations are reassessing the role of physical office spaces to better support collaborative sessions and team-building activities, recognising the need for a balance between remote and in-person interactions.

While remote work has enabled greater flexibility and efficiency in communication, it has also presented challenges in maintaining interpersonal connections and fostering a sense of belonging within organisations. By leveraging digital tools effectively and prioritising opportunities for personal connection, teams can navigate the complexities of remote work and cultivate a culture of collaboration and innovation in the post-COVID era.

The data on 'pre-COVID work interactions' provide a glimpse into the traditional office dynamics and the challenges associated with work before the pandemic. Respondents highlighted the ease of spontaneous interactions and collaboration in the office setting. Casual conversations and impromptu meetings were common, facilitating quick decision-making and problem-solving. The ability to physically see colleagues and gauge their availability allowed for fluid communication and efficient teamwork.

However, despite the benefits of in-person interactions, some respondents also expressed challenges associated with office culture, such as the pressure to appear busy and the stigma associated with remote work. There was a cultural expectation to be visibly present at one's desk, contributing to a sense of guilt or apprehension about working remotely. Additionally, there was a perception that productivity was tied to physical presence in the office, leading to a reliance on traditional work habits and environments.

Furthermore, being present at the client's premises was viewed as advantageous, as it allowed for greater interaction and collaboration with clients and colleagues. Faceto-face interactions fostered stronger relationships and accelerated the pace of work, facilitating the change process and driving innovation.

Pre-COVID work interactions were characterised by a blend of in-person collaboration and traditional office norms. While there were benefits to being physically present in the office, there were also challenges and expectations that influenced work behaviour and perceptions of productivity. The shift to remote work during the pandemic prompted a re-evaluation of these dynamics and highlighted the need for flexibility and adaptability in the modern workplace.

The data on 'impromptu connections' highlight the shift in communication dynamics between colleagues in the transition to remote work. Respondents noted a decrease in spontaneous interactions and casual conversations that typically occur in the office environment. The absence of physical proximity has made it more challenging to engage in ad hoc discussions or seek immediate assistance from colleagues. Instead, communication has become more formalised, with scheduled meetings and structured interactions replacing impromptu exchanges.

The loss of impromptu connections has impacted the breadth of communication within organisations. Respondents noted a narrowing of communication channels, with interactions primarily focused within their immediate teams or project groups. The lack of casual encounters with colleagues from other departments or teams has reduced opportunities for networking and cross-functional collaboration, limiting exposure to different perspectives and areas of expertise.

Additionally, the shift to remote work has altered the nature of relationshipbuilding and rapport-building among colleagues. Informal interactions, such as grabbing coffee or chatting in the hallway, have become less frequent, impacting the development of personal connections and camaraderie within teams. The absence of these spontaneous moments has made it challenging to foster trust and establish rapport, particularly with colleagues outside one's immediate circle. Despite these challenges, respondents expressed a desire to maintain impromptu connections and informal interactions, highlighting their value in building relationships and fostering a sense of community within organisations. The transition to remote work necessitated a re-evaluation of communication strategies and the adoption of new approaches to facilitate spontaneous interactions in a virtual environment.

'In-person communication' emerged as a vital component in facilitating effective collaboration and understanding within work environments. Respondents highlighted the unique aspects of face-to-face interactions that are difficult to replicate in virtual settings. One notable aspect was the ability to read non-verbal cues and energy, which can provide valuable insights into people's thoughts and reactions. This intuitive understanding often leads to smoother communication and quicker decision-making processes. Additionally, face-to-face interactions were perceived as conducive to brainstorming and innovation, as the spontaneity and fluidity of in-person discussions allow for the exchange of ideas and feedback in real-time.

Respondents emphasised the importance of body language in communication, noting that it adds depth and context to conversations. Being able to see facial expressions and gestures enhances understanding and fosters a more natural flow of communication. Face-to-face interactions were seen as essential for building strong working relationships and trust, particularly between managers and their teams. The ability to have informal side conversations and read subtle cues contributes to a deeper connection and a sense of camaraderie among colleagues.

Despite the increasing reliance on digital communication tools, the value of faceto-face communication remains important to employees. Many respondents expressed that certain aspects of collaboration and relationship-building are simply more effective when conducted in person. While remote work offers flexibility and convenience, it also presents challenges in capturing the nuances of human interaction. Therefore, maintaining a balance between virtual and in-person communication is essential for fostering effective teamwork, innovation, and interpersonal relationships within organisations.

The theme of 'Communication and interactions' highlights the multifaceted nature of workplace communication and the importance of adapting communication strategies to meet the evolving needs of remote and hybrid work environments. By leveraging digital platforms and creating opportunities for engagement, organisations can mitigate the loss of impromptu connections and cultivate a culture of collaboration and camaraderie, even in remote work settings.

7.3.6. Collaboration and relationships

Collaboration and relationships play pivotal roles in driving innovation, productivity, and overall workplace satisfaction. The final theme delves into the intricate dynamics of 'Collaboration and relationships' within diverse work environments. Two primary codes emerged from the data: 'Collaborating in different contexts' and 'Creativity and group bonding'. The former encapsulates participants' reflections on the effectiveness of working together across various contexts, including virtual spaces, meetings, and brainstorming sessions. Participants shared insights into the challenges and opportunities inherent in collaborative efforts, shedding light on the nuances of ideation and problem-solving in different work settings. Conversely, 'Creativity and group bonding' delves into maintaining and fostering connections and friendships in the digital realm. As remote and hybrid work models become increasingly prevalent, participants elucidated strategies for nurturing group cohesion, fostering creativity, and sustaining meaningful relationships in virtual environments. The results are shown in Table 7.7.

Table 7-7: Theme: Collaboration and relationships

Theme	Code	Theme	Examples
		description	
Collaboration and relationships	Collaborating in different contexts	Employees reveal effectiveness of working together, ideation and brainstorming sessions in different work contexts	['] If it's like sort of I guess more creative stuff where you're coming up with new ideas, I'd maybe stay in the officeI do benefit from bouncing ideas off people in those kind of situations rather than just coming up with something on my own' EMP01 ['] there's things that you miss out on and there's things that others miss out on because you are not in the office. The person who brings you the problem also joins in terms of trying to create the solution. So it sort of almost becomes a brainstorming session, just an informal brainstorming session rather than just sort of sitting in the corner quietly contemplating by yourself.' EMP02 ['] I guess my ideas tend to sort of pop up organically when I'm actually in a group environment, whether that's in the same room as people, or whether that's on a group call sort of thingSo need being in the same room at the office, or being on a group call together with the same people. But all kind of really make a difference for me. the content is the same. You can have the same discussion in a room, or on a call together.' EMP03 ['] It was amazing [working remotely]. And just the way people
			were more confident to share their

		ideas and yeah, it was interesting. I actually really enjoyed that dynamic where people that didn't speak up before suddenly did and had a lot to share because there were different ways for them to communicate.' EMP09 'But something felt different about being face-to-face and being able to have that rapid connection for that time. The engagement, the body language, I don't think as much as we try to adopt to the best kind of tech tools to support some of this planning, there's nothing better than being in a room with post-it notes and markers and being able to, you know, mix and communicate and sort of edit people's stuff on the fly' EMP12
Group bonding	Employees share how connections and friendships are maintained and fostered in the digital realm	'T'd be speaking with them throughout the office and that kind of thing. Whereas now I have a social call or two a day with some a couple of my close friends in the office. And then really only be speaking with the people that are on my team rather than a wider division. There'd be people in the audit team that if we were in the office, I'd be speaking to once a week that I probably haven't spoken to this month or even the last couple of months [because I am remote working].' EMP01 'I don't think it can replaceget everybody together in a room with a drink in their hand. I think that's a different experience.' EMP05 'we don't have a lot of day-to-day

	interactions. So we made time to
	FaceTime yesterday and, and we're
	gonna do that, you know, maybe
	once a month, just have a half hour
	chat on FaceTime.' EMP10
	'And so there's generally kind of
	there's a couple of key people that I
	kind of, hey, you know, thinking
	about this quite informal, it'll come
	through on a teams chat or I'll give
	them a call like, if, if you are green
	on teams, I'll call you instead of
	emailing you So if I'm working
	and I've got a bit of a free time, I'll
	have a look in my, my chat list on
	who's green and I'll just ring 'em.
	So a lot of people interact for
	purposeful reasons now. So
	because I need something from
	you.' EMP14
	'I had a weekly team meeting and
	so that was really good. We'd catch
	up on a Monday morning, talk
	about our weekends. So it was. vou
	know, ves we'd talk about work, but
	we'd also be trying to catch up a bit
	socially 'EMP16
	Socially. Linit 10

The data on 'collaborating in different work contexts' reveal a nuanced understanding of the benefits and drawbacks of remote versus in-person collaboration. Respondents emphasised the value of face-to-face interaction for certain types of work, particularly in creative endeavours where brainstorming and idea generation thrive. Being in the office facilitates spontaneous conversations and informal brainstorming sessions, allowing for the organic development of ideas through real-time collaboration. These dynamic fosters innovation by enabling individuals to bounce ideas off one another and leverage collective creativity. However, remote work also offers unique advantages, such as increased productivity for some individuals and the ability to engage introverted team members who may find it challenging to speak up in traditional meetings. The use of digital tools facilitates collaboration across distances, albeit with some limitations on the spontaneity and fluidity of in-person interactions. Despite these advantages, some respondents noted that remote work can hinder certain aspects of collaboration, such as workshopping ideas and reading nonverbal cues during discussions.

In contrast, face-to-face interactions are praised for their ability to enhance communication through body language and immediate feedback. The physical presence of colleagues fosters a sense of connection and urgency, leading to more spontaneous problem-solving and decision-making. Additionally, the tactile nature of in-person collaboration, such as whiteboard sessions and impromptu discussions, allows for a level of engagement and creativity that may be difficult to replicate remotely.

The data regarding 'creativity and group bonding' reveal the perceived importance of social interaction in fostering a sense of camaraderie and creativity within teams. Respondents expressed a sense of loss in spontaneous interactions and casual conversations that often occur in office settings. Remote work has led to a reduction in day-to-day interactions with colleagues outside one's immediate team, resulting in fewer opportunities for serendipitous conversations and cross-departmental collaboration. This lack of organic interaction can hinder the flow of ideas and limit the exchange of diverse perspectives, which are essential for fostering creativity and innovation.

Respondents highlighted the unique value of face-to-face gatherings for building rapport and strengthening team bonds. Social gatherings, such as informal meetups with drinks, provide opportunities for colleagues to connect on a personal level, share experiences, and build trust. These interactions not only contribute to a positive work culture but also facilitate the exchange of ideas and the exploration of new concepts in a relaxed setting. The absence of these in-person gatherings during remote work has left a void in team dynamics and camaraderie, impacting both social cohesion and creative collaboration.

However, despite the challenges of remote work, respondents also identified innovative ways to maintain connections and foster creativity virtually. Scheduled social calls, virtual coffee chats, and regular check-ins via video conferencing platforms help bridge the gap and facilitate meaningful interactions outside of formal meetings. These intentional efforts to stay connected, share ideas, and celebrate successes contribute to a sense of belonging and mutual support within the team, even in a remote environment.

The 'Collaboration and relationships' theme features the multifaceted nature of interpersonal dynamics and collaborative processes within the modern workplace. The analysis suggests that both remote and in-person collaboration have their merits, and the ideal approach may vary depending on the nature of the work and the preferences of individuals involved. The analysis also reveals that effective collaboration and meaningful relationships are essential drivers of exploring and generating ideas. Employees navigate diverse work contexts, from traditional office spaces to virtual environments, each offering unique opportunities and challenges for collaboration.

Furthermore, fostering strong relationships, whether through face-to-face interactions or digital channels, enhances team cohesion and nurtures employee relationships and teamwork. Remote work offers opportunities for teams to adapt and find new ways of connecting and collaborating. By leveraging digital tools and cultivating a culture of communication, organisations can mitigate the drawbacks of remote work and encourage innovative behaviours, regardless of physical distance.

7.4. Chapter summary

In summary, this chapter discussed the qualitative data analysis undertaken using a thematic analysis approach. The thematic analysis results provide a detailed understanding of the multifaceted experiences and perspectives of employees engaged in remote work. From the challenges and advantages of remote work to the evolving nature of work interactions and the dynamics of creativity and technology, the employee's narratives highlight the complexity of remote work experiences.

The next chapter discusses findings from both the quantitative and qualitative studies, addressing the research questions and objectives.

CHAPTER 8 - FINDINGS AND DISCUSSION

8.1. Introduction

This chapter aims to provide a comprehensive synthesis of the research findings, offering insights into their implications, significance, and potential contributions to the field. This chapter represents a critical juncture in the research process, as it allows for integrating the results with existing knowledge, fostering a deeper understanding of the research objectives and their broader implications.

At the beginning of the chapter, the research questions and objectives that guided this study are revisited. The research questions serve as the foundation for the investigation and provide a clear focus for the subsequent discussion. By restating the research questions, we establish a framework for evaluating the extent to which the research objectives have been achieved and for exploring the implications of the findings.

Sections 8.3 and 8.4 discuss the quantitative study findings and address the first two research sub-questions. Sections 8.6 and 8.7 discuss the qualitative study findings and address the latter two research sub-questions. Beyond presenting the findings, the discussion chapter delves into their interpretation and analysis, establishing connections with existing literature.

8.2. Revisiting the research questions

Although remote work is not a novel concept, the COVID-19 pandemic prompted a sudden shift for many Australian employees from office-based work to remote work in their homes (Beck & Hensher, 2022; Vij et al., 2023). The pandemic has indelibly altered the foundational tenets of the workplace, with employees indicating their desire to persist in remote work either part or all of their working time (Ng et al., 2022; Smite et al., 2023). Therefore, it is reasonable to infer that working remotely is a lasting aspect of the modern work landscape. The thesis aims to enhance our understanding of the dynamic interplay between remote work and IWB from the employee's perspective. The thesis explored the factors under which remote working facilitates or inhibits IWB and how remote work compares to office-bound working arrangements in terms of innovative behaviours. The central question guiding this research is: How do employees perceive the relationship between remote work and their innovative work behaviour (IWB)?

The research sub-questions shown in Table 8.1 were as follows:

	Research sub-questions	Research objectives	Type of study
SQ1	What is the extent of IWB among employees engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?	RO1: Understand the extent to which remote workers engage in innovative work behaviour.	Quantitative
SQ2	How do demographics influence the outcomes of employees' IWB engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?	RO2: Explore the impact of demographic variables of remote working employees to discern their influence on innovative work behaviours.	Quantitative
SQ3	What factors contribute to fostering or inhibiting innovative work behaviours in remote working environments?	RO3: Explore the factors of remote working that foster or hinder innovative work behaviours.	Qualitative
SQ4	What are the distinctions in employee experience regarding innovative work behaviour when comparing remote and office- based working environments?	RO4: Explore the employee lived experience between remote and office-based working in terms of innovative behaviour.	Qualitative

Table 8-1: Research questions and research objectives

8.3. SQ1: The extent of IWB among employees engaged in remote work

This research investigated the relationship between the frequency of remote work and the self-reported levels of IWB among Australian employees. Participants were asked to specify the frequency of their remote work engagements (1–3 times per month, once a week, 2–4 times per week, and always) and detail the frequency of their involvement in activities conducive to innovative behaviour while working remotely. The IWB data is collated as IWB overall and for each phase within the four-phase model: idea exploration, idea generation, idea championing and idea implementation (De Jong & Den Hartog, 2010).

8.3.1. Remote Working Frequency (RWF) and IWB

The findings of the study indicate a positive relationship between the frequency of remote work and the level of IWB. Specifically, IWB increased as employees worked remotely more often. However, those who worked remotely every day showed a slight decrease in IWB compared to employees who worked remotely 2-4 times per week. This suggests that for organisations wishing to increase IWB, employees should partake in regular remote work arrangements yet refrain from working solely remotely (Costa et al., 2021).

This finding is consistent with literature suggesting that remote working can be a double-edged sword for innovation. While it offers flexibility, autonomy, and improved work-life balance, which in turn can enhance innovation (Coun et al., 2021; Lorentzon et al., 2023; Rachman et al., 2022), exclusive remote work may impede innovation efforts due to social isolation and work-home conflicts (Beňo & Koreňová, 2021; Chafi et al., 2021; Henke et al., 2022) and reduced opportunities to connect, decreasing the chances of working together on innovative collaborations (Alvarez-Torres & Schiuma, 2022; Costa et al., 2022).

Autonomy and flexibility have been identified as core values that stimulate creativity and innovation within organisations (Akhtar & Ali, 2023; Alpkan et al., 2010; Amankwaa et al., 2019; Martins & Terblanche, 2003; Nguyen, 2021; Supriyanto et al., 2023). Remote work environments often afford individuals greater autonomy and flexibility in managing their work tasks and schedules, which can enhance innovativeness among employees. Increased flexibility in work location can lead to improved work-life balance, wellbeing, and effective communication, enhancing the conditions for innovative endeavours (Boccoli et al., 2022; Charalampous et al., 2018; Hoeven & Zoonen, 2015).

However, while remote work provides autonomy and flexibility, it may limit critical interaction opportunities necessary for collaborative innovation. The 'autonomy paradox' in remote work literature highlights the potential challenges of balancing autonomy with the need for social interaction and collaboration (Çobanoğlu, 2021; Fatima et al., 2022; Mazmanian et al., 2013).

It is essential to consider how autonomy impacts collaborative innovation efforts that rely on teamwork and face-to-face interactions, whereby organisations must also create opportunities for collaboration and teamwork (Mazmanian et al., 2013; Neidlinger et al., 2022). The digital connectivity and collaboration tools available in remote work environments can facilitate communication and knowledge sharing among geographically dispersed teams, enabling individuals to collaborate effectively and exchange ideas in real-time, enhancing the ideation and implementation process.

With reduced distractions and interruptions commonly found in traditional office settings, remote workers may have more focused and uninterrupted time to engage in creative thinking and problem-solving activities (Adekoya et al., 2022), essential components of innovative behaviours. However, too much remote work may lead to isolation (Golden et al., 2008; Hu & Subramony, 2022; Toscano & Zappala, 2020; W. Wang et al., 2020; Zoonen et al., 2021). While the isolation fostered by remote work initially appears conducive to individual creativity and tasks requiring deep focus (Michinov & Michinov, 2021; Takeuchi et al., 2021), it may not effectively support the collaborative activities essential for driving organisational change and development. This might further clarify the slight decrease found in this study in IWB for those employees who always worked remotely.

As the frequency of remote work rises, so does IWB, except where the employee always works remotely. This understanding is pivotal for organisations navigating evolving work landscapes. For managers and organisations, the study indicates that promoting remote work can be advantageous for nurturing innovation, challenging the traditional notion that physical co-location is necessary for creativity and innovation. However, the findings underscore the importance of a balanced approach in implementing remote work arrangements to enhance employee innovation. When examining the distinct phases of IWB as outlined by De Jong and Den Hartog (2010), the data findings, while lacking statistical significance, imply that with an increase in remote work frequency, there is a corresponding rise in self-reported IWB, notably observed during the phases of Idea Generation and Idea Implementation.

Idea Generation and Idea Implementation

Several factors may contribute to the observed increase in self-reported IWB during idea generation and implementation phases as remote working frequency increases. Firstly, during the idea-generation phase, employees need the time and space to engage in creative thinking (Agrawal et al., 2018; Mullens & Glorieux, 2023). Remote work provides individuals with greater autonomy and flexibility over their work schedules and environment (Akhtar & Ali, 2023; Wicaksono & Pusparini, 2022), which can facilitate creativity. Without the constraints of traditional office settings, remote employees have the freedom to allocate time to engage in deep thinking, reflection, and problem-solving activities conducive to generating new ideas. Additionally, remote work often involves fewer interruptions and distractions than office environments (Amir et al., 2021; Henke et al., 2022), which may facilitate deep reflection and ideation, enabling employees to generate innovative solutions more effectively.

Similarly, during the idea implementation phase, greater control over one's work environment and schedule, enabled by remote work, can enhance productivity and efficiency in executing implementation plans. Remote employees have the flexibility to organise their work in a manner that best suits their preferences and working styles, enabling them to focus their efforts on implementing innovative ideas with minimal disruptions. Moreover, using digital collaboration tools and communication platforms in remote work environments facilitates seamless coordination and collaboration among team members, streamlining the implementation process and enhancing the likelihood of successful outcomes.

Idea Exploration and Idea Championing

However, it is unsurprising that there was no observed relationship between RWF and the phases of idea exploration and idea championing. Increased remote working can hinder employees' ability to explore innovative approaches during idea exploration. Remote work diminishes the frequency of impromptu discussions and casual encounters that often spark creativity needed in the early phase of innovation. Without the spontaneous exchange of ideas that occurs naturally in office settings, employees may find it more challenging to generate innovative thoughts. Social engagement plays a crucial role in stimulating creativity. Remote work can lead to feelings of isolation and reduced social interaction, which may negatively impact employees' motivation and ability to think creatively.

During idea championing, increased remote working can pose challenges to championing new ideas and garnering support. In traditional office settings, spontaneous discussions, water cooler chats, and casual interactions often provide opportunities to pitch ideas and gather initial feedback. Remote working reduces these informal opportunities, making it harder to introduce new concepts casually.

Further, building trust and rapport with colleagues is essential for gaining support for new ideas. Remote working can hinder the development of these interpersonal relationships, making it harder to persuade others to back innovative initiatives. In a remote work environment, it may be more challenging for innovative efforts to gain visibility across the organisation. Without the physical presence in the office, it can be more challenging for new ideas to spread organically and attract the attention of critical stakeholders or decision-makers.

8.4. SQ2: Demographics and IWB

Demographics have long been recognised as influential determinants in workplace dynamics, impacting individual behaviours, perceptions, and interactions. This research explored the relationship between age, gender, tenure at current organisation, employment status and employment level and the self-reported levels of IWB among Australian employees while they are working remotely.

The results showed the highest prevalence of undertaking IWB activities (i.e. 'Always'), were categorised as follows: the 18–30-year-old age group, female, contract/casual, less than 2 years tenure, managers, working remotely 1–3 times per month.

The categories that showed a consistently high prevalence of undertaking IWB (i.e. 'Often'), were the 41–50-year-old age group, male, full-time, with 6+ years tenure, managers, working remotely 2–4 times per week.

8.4.1. Demographics and IWB overall

The findings suggest that age and gender had no significant relationship to IWB overall. However, full-time employees exhibited higher IWB than part-time and contract/casual employees. Additionally, IWB increased with higher employee levels, reaching its peak among 'Managers', and rose with longer tenure at the current organisation.

Employment Status

When analysing employee behaviour in the context of remote work, employment status must be considered due to its significant impact on various aspects of work dynamics, such as employee wellbeing, engagement, and productivity.

The research found that full-time employees had higher levels of IWB than their part-time counterparts. Full-time employees often experience enhanced job security, fostering a sense of stability and enduring commitment to their roles and the organisation (Feng et al., 2021; Kalmi & Kauhanen, 2008; Peng & Li, 2021). Job security contributes to employees' sense of meaningfulness, purpose, and connectedness, which are particularly important in remote work settings. The sense of increased stability and assurance can reduce uncertainty and anxiety, especially in remote work settings where the physical distance from the workplace can amplify feelings of isolation and detachment (Golden et al., 2008; Maqableh et al., 2023; Miyake et al., 2022).

Increased job security provides full-time employees with the assurance to explore innovative ideas, take calculated risks, and invest time in creative problem-solving endeavours (Kalmi & Kauhanen, 2008; Loi et al., 2011; Y. Wang et al., 2017). Employer support for creative work significantly correlates with enhanced job security and job satisfaction, ultimately improving the quality of employees' working lives. This support gives individuals the confidence to explore new ideas and invest time in creative endeavours (Raykov, 2014).

Conversely, part-time, temporary, or contract employees may experience heightened job insecurity, which can diminish their motivation for innovative activities (Montani et al., 2021; Probst et al., 2019; Spiegelaere et al., 2014). Such individuals might prioritise short-term tasks to safeguard their immediate position rather than invest in innovative projects beneficial for the organisation's future.

Furthermore, full-time employees typically have greater access to opportunities for skill enhancement and ongoing learning within the organisation, such as training and development (Jaworski et al., 2018; Kauhanen & Nätti, 2015; Lyonette, 2015), which augment their innovative capabilities (Chaubey & Sahoo, 2019; Miah & Hafit, 2020). Skill enhancement and training opportunities in remote environments can provide employees with the necessary tools and knowledge to adapt to new work structures and emerging technologies, fostering innovation (Ferguson et al., 2023; B. Wang et al., 2021). In contrast, temporary or part-time employees may face constraints in accessing training programs and resources thus impeding their capacity to contribute meaningfully to innovative initiatives. This limited availability also extends to having the time to work on organisational projects. Full-time employees may have more available work time to work on projects than part-time employees with limited availability.

Employment Level

The research found that managers and senior leaders had higher IWB than nonmanagers and administrators. Considering employment level is essential when analysing employee behaviour in remote work settings. Employment levels have been found to impact job satisfaction, organisational commitment, turnover intentions, communication patterns, collaboration styles, learning behaviours, and knowledge sharing (Ma et al., 2022; Naveed et al., 2022; H. Yu et al., 2021). The research found that a managerial or senior leadership position leads to increased IWB when working remotely compared to administrative staff and non-managers.

More junior employees, often starting their careers, frequently bring fresh perspectives and creative ideas driven by their enthusiasm and eagerness to explore novel concepts (Binnewies et al., 2007; Liu et al., 2016; Miron-Spektor et al., 2022). However, they may lack the experience and confidence needed to effectively pursue their ideas when compared to more senior employees. Creative self-confidence is particularly

important in stimulating innovative behaviour in the workplace (Dar et al., 2022). In remote settings, where physical distance and virtual communication can present challenges, confidently exploring new ideas and approaches is essential for driving innovation.

Managers and senior leaders typically have greater access to resources and information and possess greater decision-making authority within the organisation compared to lower-level employees (Niranga & Dharmadasa, 2021). This autonomy in decision-making also allows them to implement innovative ideas more readily without facing bureaucratic hurdles or needing approval from higher levels of management. In remote settings, decision-making authority allows employees to take ownership of their ideas and initiatives, leading to a sense of empowerment and autonomy that can drive innovative thinking (Hu et al., 2018). Moreover, decision-making authority enables employees to make timely and informed decisions, which is essential for adapting to the dynamic and fast-paced nature of remote work environments (Turner et al., 2017). Having the authority to make decisions empowers employees to experiment, take risks, and explore new ideas, all of which are vital components of the innovation process (Zameer et al., 2020). In remote environments, where communication and collaboration may be challenging, decision-making authority, particularly entrepreneurial decision-making, can streamline processes, enhance efficiency, and create a conducive environment for creativity and innovation (Blauth et al., 2014).

At the upper levels of the organisational hierarchy, leaders establish the groundwork for innovation by cultivating a culture of creativity, embracing experimentation, and tolerating failure. Their strategic vision and direction steer the organisation toward innovative initiatives aligned with its objectives. Managers and senior leaders are responsible for setting strategic direction and long-term goals for their teams or departments. Their strategic perspective enables them to identify opportunities for innovation that align with organisational objectives and drive sustainable growth. Junior employees, lacking exposure to such strategic endeavours, may not receive the requisite support and recognition to encourage them to engage in innovative behaviours (Shipton et al., 2006).

Managers and senior leaders frequently collaborate cross-functionally with colleagues from different departments or units. This collaboration exposes them to

diverse perspectives, expertise, and experiences, fostering creativity and innovation through exchanging ideas and knowledge (Melander, 2017).

While senior employees often exhibit a wealth of experience and a proven track record of innovation, it is crucial not to overlook the potential contributions of junior employees within an organisation. Employees in lower hierarchical positions may have greater motivation to validate their worth and effect a positive influence, motivating them to exert innovative behaviours (Dai et al., 2022). These employees, alongside administrators, usually assume operational responsibilities, engaging more intimately in day-to-day operations and possessing a profound comprehension of operational problems and customer requirements, thereby often being closer to the problems requiring innovative solutions. Junior employees frequently bring fresh perspectives, enthusiasm, and a willingness to explore new ideas that can inject vitality into the innovation process. Their motivation to prove themselves and make a positive impact can drive them to approach challenges with creativity and ingenuity. Moreover, their lack of entrenched habits or biases may enable them to see opportunities that more seasoned employees might overlook.

Tenure

The research found that longer tenure was associated with higher IWB. Tenure within an organisation can influence employee behaviour in remote work settings by affecting performance at both individual and team levels, as well as work attitudes and motivations.

The length of an employee's tenure within a specific organisation can significantly impact their IWB (Liu et al., 2016; Woods et al., 2018). Long-tenured employees often deeply understand the organisation's culture, goals, and existing processes (Ng & Feldman, 2010). This accumulated experience enables them to navigate organisational complexities more effectively and identify opportunities for innovation. With increasing tenure, employees accumulate many experiences, including successes and failures. Reflecting on past experiences allows employees to learn from mistakes, adapt strategies, and apply insights to future innovative endeavours. Remote workers who have been with a company longer tend to be more comfortable with the tools, systems, and processes in place, making them more adaptable to using those processes in different work contexts. This leaves more time to focus on innovative endeavours.

This familiarity with the organisation can also translate into increased innovation due to the networks and relationships they have cultivated over time (Ng & Feldman, 2010). Longer-tenured employees typically have extensive networks and occupy central positions in internal networks, enabling them to effectively disseminate and implement innovation within organisations and also facilitate collaboration and knowledge sharing while working remotely (Liu et al., 2016; Ng & Feldman, 2010).

Long-standing employees often enjoy higher levels of trust and credibility within the organisation. This trust enables them to propose and implement innovative ideas more effectively, as they are perceived as reliable and knowledgeable contributors. As remote workers gain trust and autonomy from their employers over time, they feel more empowered to explore innovative ideas without constant supervision. Longer-tenured remote workers may have earned this trust, allowing them more freedom to experiment and innovate. These employees also often attain higher levels of job mastery and proficiency in their roles. Greater job mastery gives individuals the confidence and autonomy to experiment with new approaches and solutions, increasing innovative behaviours.

However, long-tenured employees can become entrenched in established modes of thinking, thereby resisting change and impeding the adoption of innovative practices (Dam et al., 2008; Lauterbach & Kunze, 2023). Conversely, employees with shorter tenures bring fresh perspectives and a readiness to challenge existing norms (Liu et al., 2016; Woods et al., 2018). Their detachment from historical processes can propel them to propose innovative solutions and explore uncharted territories. Ng and Feldman (2010) discovered a positive relationship between tenure and job performance for employees with shorter average tenure, while for those with longer average tenure, the relationship turns negative. Liu et al. (2016) identified that factors like status hierarchy and position tenure influence the relationship between organisational tenure and innovative behaviour, being positive for individuals with low-status hierarchy and short position tenure but negative for those with longer tenure. Similarly, Yuan (2021) observed that as tenure increases, employees with higher levels of conscientiousness tend to display less innovative behaviour. Employees with longer tenure often possess a wealth of experience and a deep understanding of the organisation's inner workings, which can lead to heightened levels of innovation. However, this extensive tenure can also inadvertently result in employees becoming set in their ways and resistant to change. In contrast, with their fresh perspectives and eagerness to make an impact, newcomers to the organisation can bring innovative ideas to the table. Their lack of historical attachment to established processes enables them to approach challenges with a sense of curiosity and a willingness to try new approaches.

Gender

The findings around the relationship of gender and age to IWB suggest that they are not linearly related but rather influenced by several personal and contextual variables.

The findings reveal that gender does not significantly influence IWB among remote employees, suggesting that regardless of gender, individuals demonstrate similar levels of creativity and innovation in their work. This is supported by J. Wang et al. (2017), who suggested that employees of different genders are similarly willing to share knowledge and engage in innovation.

However, the relationship between gender and IWB in remote working contexts is complex. It can be influenced by various factors, thereby explaining why our findings did not show a statistical effect. Moreover, numerous studies IWB often adopt the employer perspective, potentially introducing gender biases and perceptions into the analysis of which gender shows more IWB (Carmeli & Spreitzer, 2009; Carpini et al., 2023; Cortland & Kinias, 2019; Heilman & Eagly, 2008; Luksyte et al., 2017; Martin & Barnard, 2013; Nusair et al., 2012; Reuvers et al., 2008). In contrast, this study adopts the employee's perspective, reducing the likelihood of bias against one's own gender, as individuals are less inclined to exhibit such biases when evaluating themselves. Other factors such as organisational cultures, access to resources, work-life balance initiatives, and technology utilisation are among the numerous variables that can shape this relationship. Therefore, while gender may not directly impact IWB in remote work settings, its influence can be mediated by these multifaceted elements.

The presence of work-life balance opportunities notably affects IWB, indicating that access to such opportunities can positively enhance an employee's capacity to engage

in IWB (Damayanti & Kurniawan, 2023; Li & Liu, 2023; Mishra et al., 2017). This can be achieved through opportunity-enhancing human resources (Bos-Nehles, Renkema, et al., 2017; Prieto & Pérez-Santana, 2014) that impact autonomy, task composition, feedback, job demands and time pressure (Berkel et al., 2021; Bos-Nehles, Bondarouk, et al., 2017; Singh et al., 2020). While increased flexibility aids in achieving work-life balance, there is a risk of overworking and difficulty disengaging from work while operating from home.

The heavy use of technology in remote work removes many traditional barriers that would otherwise have hindered women's participation in innovation. The digital workplace provides equal access to information and resources that are essential for innovation activities. Remote work is particularly beneficial for women, who often tend to have a 'second shift' at home afforded by the availability of technology, juggling work with a host of domestic obligations. This flexibility and control over where and when to do one's job help enhance job satisfaction and positively impact creativity.

While gender did not directly impact IWB, the literature suggests that diverse teams, including those with gender diversity, tend to foster a wider array of ideas and perspectives, consequently enhancing innovation (Dutcher & Rodet, 2018; H. Leroy et al., 2021; Li et al., 2018; Salazar et al., 2017). Diverse teams often outperform homogenous ones in creativity, problem-solving, and idea generation. Further, gender-diverse teams improve participative communication, coordination, cohesion, and mutual support within the team, thereby bolstering innovative activities (Dai et al., 2018; Rejeb et al., 2019).

Age

The findings reveal that age does not exert a significant influence on IWB among remote employees, suggesting that regardless of age, individuals can demonstrate similar levels of creativity and innovation when they work remotely (Parsons, 2015; Stoffers & Heijden, 2018; Tams & Dulipovici, 2019). Although age may have some influence on factors such as experience and perspective, it is not a determining factor in predicting IWB in remote work contexts. Other individual and contextual factors play a more significant role in fostering innovation in the workplace (Russo et al., 2020).

Traditional perspectives argue that younger employees are best positioned to adapt to the technological and uncertain nature of remote work, making them inherently better suited to innovative performance in such settings. Younger employees frequently offer novel viewpoints and possess an innate familiarity with technology, facilitating their rapid adaptation to new tools and platforms (Kim et al., 2011). In a remote environment, effectively using digital communication tools is vital. Often labelled as digital natives, these younger employees tend to demonstrate greater innovation levels than their older counterparts, often referred to as digital immigrants. The digital natives' ease with technology and adeptness in utilising digital tools significantly contribute to their heightened innovation behaviour (Buhl et al., 2016).

Younger employees' enthusiasm and willingness to take risks can drive creative thinking and experimentation, leading to innovative solutions (Kim et al., 2011). In contrast, older employee' preference for the security and more stable nature of traditional work settings may inhibit risk-taking behaviours and engagement in innovative activities when working remotely. This perspective is rooted in theories of adaptability and work motivation; as employees age, they might prefer work environment stability and routine, reducing their inclination for innovation in less structured environments.

So, while younger employees often exhibit a greater receptiveness to change and may challenge conventional approaches, thereby nurturing an atmosphere conducive to innovation, older employees, drawing from their extensive experience and profound industry knowledge, offer invaluable perspectives (Kim et al., 2011; Meyer, 2008). Moreover, older employees typically excel in problem-solving and decision-making, which are pivotal competencies in the innovation process (Guillén & Kunze, 2019; Meyer, 2008).

The findings around age suggest that work outcomes and behaviour are more likely a function of changing life circumstances than chronological age per se. Rather, other factors such as life stage, career phase, personal circumstances, and individual adaptability should be considered when considering employees in remote working contexts. As employees age, there are indeed changes in their behaviour and motivations (Klopotan et al., 2018). Older employees are less motivated by extrinsic factors but more by intrinsic rewards, indicating a change in motivational drivers with age (Inceoglu et al., 2011; Kooij et al., 2011). As individuals age, there is a transition in motivational priorities from knowledge acquisition and competition towards the pursuit of emotional satisfaction (Williams et al., 2006). This shift is attributed to an increasing awareness of a limited lifespan and changes in emotional stability over age.

8.4.2. Demographics and each phase of IWB

Exploring each phase of IWB separately, in addition to an overall assessment, is crucial for several reasons. It allows for a more nuanced understanding of how different aspects of innovation may unfold throughout the innovation process (Davids & Frenken, 2017; Koskull & Strandvik, 2014). Examining each phase individually allows researchers to identify whether demographic or contextual factors may exert varying degrees of influence across different stages of the innovation process. The first two phases of IWB are defined as 'creativity' phases, and the latter two phases are defined as 'innovation' phases. This distinction is supported by De Jong and Den Hartog (2007), who suggest that creativity is a fundamental component of innovative behaviour, particularly evident in the early stages of the innovation process, where ideas are generated in response to the need for innovation. This aligns with the notion that the early stages of IWB are characterised by creative thinking and idea generation. Costa et al. (2022) also emphasise the temporal separation of creativity from other stages of IWB, highlighting creativity as a precursor to subsequent phases of innovation. This temporal distinction underscores the idea that creativity is foundational to the innovation process and is particularly prominent in the initial phases of IWB.

The research findings underscore the significant influence of employment status on IWB when working remotely. Being full-time over part-time is associated with higher levels of innovative behaviour across all four phases of IWB.

Longer tenure was found to be correlated with increased innovative behaviour during the idea-championing phase. Conversely, advancing age was linked to lower levels of IWB during this same phase.

Additionally, increasing employment levels corresponded with increased innovative behaviour during the idea implementation phase.
8.4.2.1. Demographics and idea exploration

During the first phase of IWB, idea exploration, employees engage in activities to identify opportunities, explore, contemplate, and seek solutions to identified problems or challenges.

This phase is characterised by individual efforts to brainstorm, research, and conceptualise innovative concepts. This phase involves the process of investigating, researching, and understanding a problem or opportunity. The focus of idea exploration is to gather information, identify relevant insights, and explore different perspectives related to the subject matter.

Employees rely on their cognitive abilities (Bagheri & Akbari, 2017; Eisenbart et al., 2023; Shahid et al., 2022), knowledge, and creativity to generate novel ideas that have the potential to drive innovation within the organisation. This phase involves a process of deep reflection, analysis, and creativity as individuals delve into the problem space, seeking to understand its intricacies and potential avenues for resolution. Employees may research extensively and gather relevant information and data to gain insights into the underlying issues.

This phase focuses on exploration rather than immediate idea generation, with individuals actively probing and testing different hypotheses, concepts, and approaches to uncover innovative solutions. Ultimately, the idea exploration phase serves as a critical precursor to idea generation, laying the groundwork for developing novel and creative solutions to problems. The goal is to lay the groundwork for idea generation by creating a rich and diverse pool of information and insights to draw upon.

The research found that part-time employees are less likely to engage in idea exploration behaviours when working remotely compared with their full-time counterparts. There is a critical link between the nature of one's employment and the ability to engage in the early, creative stages of innovation in remote working settings. This may be due to factors such as time availability, continuity of focus, collaboration opportunities, autonomy, and flexibility.

Full-time remote employees typically have more dedicated time available for work compared to part-time employees (Adekoya et al., 2022). This extra time allows them to delve deeper into ideas, engage in brainstorming sessions, and explore new ideas without the constraints of limited work hours. They can also maintain continuity in their work and thought processes without the distractions and interruptions common in office settings (Ford et al., 2020; McGee et al., 2023), which is essential for effective idea exploration. Unlike part-time employees who may need to switch between work and other commitments, full-time remote employees have more work time which they can dedicate uninterrupted blocks of time to explore ideas thoroughly.

Full-time remote employees often have more opportunities, by virtue of increased time, for collaboration with colleagues, both within and outside their organisation. Virtual collaboration platforms, which are used heavily in remote working environments, facilitate idea exchange, feedback sessions, and collective brainstorming, enabling full-time remote workers to explore ideas collaboratively and benefit from diverse perspectives (Chantziaras et al., 2021; Kwon et al., 2019; Suortti & Sivunen, 2023).

Full-time remote workers typically enjoy greater autonomy and flexibility in managing their work schedules and priorities. Increased autonomy has been consistently linked to fostering creative and innovative behaviours among employees (Acar et al., 2018; Elsetouhi et al., 2022; Spiegelaere et al., 2014). Studies such as those by Kpinpuo et al. (2022), Peng et al. (2019), and Purc and Łaguna (2019) have highlighted the positive relationship between autonomy and innovation. These studies suggest that when employees have a higher level of autonomy in their work, they are more likely to engage in innovative behaviours due to increased intrinsic motivation and self-determination.

8.4.2.2. Demographics and idea generation

During the second phase of IWB, idea generation, employees actively generate new ideas, devise solutions to problems, and develop innovative approaches to address identified challenges.

A burst of creative activity characterises this phase as individuals harness their creativity and critical thinking skills to conceive novel concepts and possibilities. Employees may engage in brainstorming sessions, idea-generation workshops, or collaborative discussions to generate various potential solutions. These activities often involve free-flowing ideation, where individuals explore unconventional perspectives, challenge assumptions, and push the boundaries of conventional thinking.

Additionally, employees may draw inspiration from diverse sources, including previous experiences, industry trends, customer feedback, and emerging technologies, to inform their idea-generation process. The goal of idea generation is to produce a diverse set of potential solutions or concepts that can be further refined, evaluated, and developed into actionable plans or prototypes.

As with the first phase, the findings show that during this second phase, an employee's employment status effects their capacity to participate in the early, creative stages of innovation within remote work setups. Part-time employees are less likely to engage in idea-generation behaviours when working remotely compared to their full-time counterparts.

Part-time employees' reduced exposure to an organisation's day-to-day dynamics may be a critical challenge. Part-time employees may have limited engagement with the organisation's culture, values, and long-term goals due to their reduced hours and presence in the workplace (Joung et al., 2018; Wang, 2014). Such limited immersion can make it harder for them to understand the bigger goals of the organisation and how they fit in with innovation efforts (Cletus et al., 2018). Studies have shown that awareness of organisational goals increases organisational commitment, work attitudes, engagement, and performance outcomes, such as creativity (Ayers, 2015; Lü, 2023).

Research indicates that job security is associated with organisational commitment, job satisfaction and employee performance. The commitment and increased job security associated with full-time employment may foster a greater sense of ownership and investment in the organisation's goals and challenges, motivating employees to proactively engage in innovative solutions (Hur & Perry, 2019; Maqableh et al., 2023; Umrani et al., 2019).

Part-time employees often have limited access to resources that are available to full-time employees (Burgess, 2005; Conway & Briner, 2002; Saini & Jawahar, 2019; Sobaih et al., 2011). Accessible resources include physical and intangible resources such as training and development, mentoring in the creative process and participation in meetings, such as brainstorming meetings, where creativity is practised and developed. Organisations that provide adequate resources, such as infrastructure, intellectual capital development programs, incentives, and knowledge-sharing platforms, are more likely to

cultivate a creative environment that enhances innovation and problem-solving capabilities (Cai et al., 2020; Cirella, 2021; Sonenshein, 2014).

8.4.2.3. Demographics and idea championing

During the third phase of IWB, idea championing, employees undertake a series of strategic activities to advocate for and promote a new idea within the organisation. Once an idea is generated, it needs support and endorsement to gain traction and move towards implementation.

This phase involves actively championing the proposed innovation, gathering support from key stakeholders, overcoming resistance, and rallying backing for its implementation. Employees typically engage in persuasive communication efforts to articulate the benefits and potential impact of the idea, compelling others to align with its vision and objectives. This may involve delivering persuasive presentations, conducting stakeholder meetings, and leveraging various communication channels to disseminate information and garner support.

Additionally, employees may seek to identify and enlist influential backers and allies within the organisation who can champion the idea on their behalf and advocate for its adoption at higher levels. This phase focuses on building momentum and consensus around the idea, mobilising support from across different departments and hierarchical levels to ensure its successful implementation. Ultimately, the idea championing phase requires proactive leadership, effective communication skills, and political acumen to navigate organisational dynamics and secure buy-in for the proposed innovation.

As observed in the findings of preceding two phases, part-time employees are less inclined to exhibit idea-championing behaviours than their full-time counterparts when operating in a remote environment. Also, longer tenured employees exhibited higher levels of idea-championing behaviours, while older employees exhibited decreased ideachampioning behaviours.

Part-time employees may have varying levels of commitment to their roles and the organisation compared to full-time employees. Their roles often entail narrower job scopes or specific tasks, which may not always align with activities related to promoting new ideas and championing them within the organisation. Therefore, part-time employees might prioritise other commitments of their work and role responsibility, over their engagement in idea promotion and advocacy activities. Part-time employees may have also have other more critical roles outside of the organisation and are therefore only partially involved in their job (Wang, 2014).

Similar to the idea exploration and generation phases, the limited work hours of part-time employees may restrict their capacity to effectively promote new ideas and advocate for them. Further, factors such as commitment and job security can cultivate a stronger sense of ownership and dedication to the organisation's objectives and obstacles (Hur & Perry, 2019; Maqableh et al., 2023; Umrani et al., 2019). This heightened allegiance can fuel an employee's motivation to champion and advocate for novel ideas that will benefit their organisation

Unlike in the preceding two phases of IWB, longer tenure led to increased ideachampioning behaviours. The initial two phases predominantly depend on the individual and their personal skills and cognition. However, there is a heightened requirement for interacting with others in this phase, which can pose different challenges in remote work settings.

Employees with longer tenure typically have a deeper understanding of the organisation's culture, goals, and decision-making processes (Ng & Feldman, 2010; Pei et al., 2018). This familiarity allows them to navigate the organisational landscape more effectively and confidently advocate for new ideas. Employees with longer tenure might possess insights into the organisation's social fabric, aiding in pinpointing the optimal channels for introducing new ideas and navigating resistance to change (Ng & Feldman, 2013). Individuals with lengthier tenure within the organisation may exhibit greater confidence in the worth of their innovative concepts, thereby experiencing reduced apprehension in advocating for their new ideas (Clercq & Pereira, 2019). This experience and accumulated knowledge of the organisation also give them insights into past initiatives, successful strategies, and potential barriers to innovation, which they can leverage to promote and advocate for new ideas more effectively.

Longer-tenured employees have established organisational networks and relationships (Ng & Feldman, 2010). Employees with longer tenure may enjoy advantages in championing innovative concepts compared to those with shorter tenure. This is because long-tenured employees often occupy more central roles in the organisation's internal network, and possess broader external social networks for the organisation (Liu et al., 2016). In remote settings, these established connections are critical due to the reduced physical visibility to garner support for new ideas and champion them among key stakeholders (Björk & Magnusson, 2009). A deep understanding of organisational dynamics, coupled with networks and relationships, can be essential in navigating the politics of managing, selling, and advocating for new ideas, and how to frame and promote innovative ideas in a manner that resonates with key stakeholders. In remote work settings, idea champions need to be even more strategic and intentional in their communication, using synchronous and asynchronous tools to involve stakeholders who can help develop their own or others' new ideas.

Employees who have been with the organisation longer may feel a stronger sense of ownership and investment in its long-term success. As a result, they may be more motivated to promote and advocate for new ideas that they believe will contribute to the organisation's growth and longevity (Hur & Perry, 2019). Moreover, longer tenure often signifies a proven track record of performance and reliability, which enhances their credibility and trust among colleagues and superiors. Research supports the importance of trust in remote working environments for enhancing performance (Ferguson et al., 2023; Soomar, 2020; Stavrova et al., 2023; Zheng et al., 2023). Increased trust has also been consistently linked to increased creativity in various contexts. Studies have shown that trust plays a crucial role in fostering creativity at both individual and team levels (Kmieciak, 2020; Rodrigues & Veloso, 2013; C. Yu et al., 2021). C. Yu et al. (2021) highlight that trust can be particularly beneficial in enhancing creativity, especially in situations involving high uncertainty and risk, where creativity requires a departure from existing norms. Building trust in remote settings can present various challenges. Thus, longer-tenured employees who have established trust may enjoy certain advantages in navigating these obstacles.

Differing from the preceding two phases of IWB, advancing age resulted in diminished idea championing behaviours. The reduced likelihood of older employees engaging in the idea-championing phase challenges our expectation that with their years of experience and influence, they are more likely to advocate for new ideas. Years of accumulated knowledge and insights enable older employees to draw upon past experiences and lessons learned to craft persuasive arguments and compelling narratives in support of new ideas. However, it may suggest a reluctance on the part of older employees to take risks in advocating for change. Older individuals tend to report lower willingness to take risks compared to younger individuals (Albert & Duffy, 2012; Banks et al., 2020; Boyle et al., 2012). Younger employees, who may be earlier in their careers and less risk-averse, may be more willing to take the initiative and champion innovative ideas to distinguish themselves and advance their professional development. In contrast, older employees, who may be closer to retirement or have established career trajectories, may prioritise stability and risk avoidance over innovation and may be less inclined to invest time and energy in championing new ideas. Older employees may be more resistant to change, as they may be accustomed to traditional methods and processes and may feel uncomfortable or sceptical about embracing new ideas and innovations.

The reluctance on the part of older employees to engage in idea-championing behaviours may stem from technological proficiency and adaptability variations. Younger employees, who are often more familiar and comfortable with digital communication tools and social media platforms, may find it easier to navigate virtual networks and effectively promote their ideas in remote work environments (Buhl et al., 2016). In contrast, older employees may face challenges adapting to new communication technologies (Hecker et al., 2021; Raymundo & Castro, 2019; Soja & Soja, 2020) or may be less inclined to engage in online promotion and advocacy due to generational differences in communication preferences (Bencsik et al., 2016; Mehra & Nickerson, 2019; Stewart et al., 2017). Older employees who may also be less familiar with emerging technologies and trends than their younger counterparts and therefore find it challenging to understand and advocate for new ideas effectively.

8.4.2.4. Demographics and idea implementation

During the fourth phase of IWB, idea implementation, employees are actively involved in translating conceptualised ideas into tangible actions and outcomes within the organisation.

This phase encompasses a range of activities to execute the proposed innovation or new idea, turning it into a practical solution that addresses identified needs or challenges. Employees may collaborate with cross-functional teams to develop detailed implementation plans, outlining specific tasks, timelines, and resource allocations required to bring the innovation to fruition.

Additionally, individuals may take on leadership roles, overseeing the execution of various project components, monitoring progress, and addressing any obstacles or setbacks encountered along the way. Communication and coordination play a pivotal role during this phase, as employees work collaboratively to ensure alignment and synergy across different departments and stakeholders involved in the implementation process. Moreover, employees may engage in continuous monitoring and evaluation to assess the effectiveness and impact of the implemented innovation, making necessary adjustments and refinements as needed to optimise outcomes.

Comparable to the previous three phases, part-time employees are less inclined to participate in idea implementation activities when working remotely compared to their full-time counterparts. Also, employees holding increased seniority exhibited more idea implementation behaviours.

The findings also show that the idea implementation phase is a crucial stage in the innovation process, as it involves converting creative ideas into concrete outcomes or products. Innovation and creativity are often discussed in relation to each other, with a key distinction being that innovation involves the transformation of creative ideas into tangible outcomes. Creativity is typically seen as the generation of novel ideas, while innovation encompasses the implementation of these ideas to create new products, processes, or services (Anderson et al., 2014; Doran & Ryan, 2017; Panasiewicz, 2021).

An interesting facet of the idea implementation phase is its reliance on collaborative action (Nguyen & Hunter, 2021; Sørensen & Torfing, 2016), which has unique dynamics in remote settings (Arunprasad et al., 2022; Rudzin et al., 2022; Waizenegger et al., 2020). As with the idea championing phase, this phase necessitates increased interaction with others. It also requires considerable time and effort, as it involves participating in the implementation process (Ng & Feldman, 2012). Full-time employees often have more frequent interactions with colleagues and supervisors by virtue of their additional working hours, which enables better coordination and collaboration on idea implementation projects, which can be crucial in remote work

environments where physical proximity is lacking. These additional hours also allow them to allocate sufficient time and focus to implement ideas internally.

Additionally, full-time employees typically enjoy enhanced access to resources, support systems, and professional development opportunities within the organisation (Burgess, 2005; Conway & Briner, 2002; Saini & Jawahar, 2019; Sobaih et al., 2011). This access equips full-time employees with the necessary tools and knowledge to execute ideas effectively, as well as fostering a proactive attitude towards idea implementation initiatives, even in dynamic and challenging remote work environments. In contrast, part-time employees receive lesser investment from organisations and fewer job resources. As a result, they tend to exhibit lower levels of organisational identification compared to their full-time counterparts, which can result in a lack of commitment to the organisation (Mauno et al., 2015), thereby reducing their motivation to participate in idea implementation activities.

The idea implementation phase is the only IWB phase where the employment level of the employee exerts some influence. Managers and senior leaders are more likely to work on idea implementation activities when working remotely compared to junior employees.

Managers and senior leaders are more likely to have roles that include overseeing and facilitating the implementation of ideas within their organisations (Birken et al., 2012; Engle et al., 2017). Managers are therefore more likely to be accountable for the success of projects and initiatives within their purview. They may feel a greater sense of responsibility to ensure that idea implementation activities are carried out effectively, meeting organisational objectives and delivering desired outcomes.

Senior leaders are instrumental in providing support and resources to their employees needed for idea implementation. Supervisor support, can enhance employees' access to necessary resources needed for implementing ideas, as well as increasing responsiveness within organisations, which can foster innovation and change and make creative ideas more implementable (Bedenik et al., 2024; Škerlavaj et al., 2014). Their organisational authority allows them to mobilise and allocate resources, allocate budgets, and prioritise initiatives in alignment with strategic objectives, facilitating the implementation of innovative ideas. They may also have a greater insight into the organisation's strategic priorities and be better equipped to coordinate and execute tasks autonomously. In remote work environments, the freedom to carry out responsibilities autonomously enables employees to swiftly address emerging challenges or seize opportunities, free from the constraints of bureaucratic procedures or delays.

Autonomy in decision making also plays a role. Senior leaders and managers possess greater decision-making authority (Shaed et al., 2018), which enables them to drive the execution of innovative ideas with greater efficacy (Chenger & Woiceshyn, 2020; Hammedi et al., 2011). By having this authority to make decisions autonomously, their involvement can accelerate the execution of ideas and remove obstacles that may hinder the implementation process. In remote work settings, decision-making authority can increase efficiency when implementing new ideas. These managers and senior leaders can make decisions in real-time without needing to wait for approval from higher levels of management, streamlining the process and eliminating bottlenecks.

In contrast, administrative or non-managerial employees may have limited authority, autonomy, and influence to execute innovative ideas independently. While they may contribute to the implementation process as team members or collaborators, their role in driving and shaping the implementation strategy is often more circumscribed. The leadership and managerial roles held by senior employees afford them greater agency, resources, and influence in remote work settings.to actively lead and participate in the implementation of innovative ideas, without needing to be in the office environment.

8.5. Summary of quantitative study findings

The findings of this study reveal a nuanced relationship between remote work and IWB, where increased remote working correlates positively with greater levels of IWB up to a certain threshold. However, a noteworthy caveat emerges: employees who work remotely exclusively may experience diminishing returns regarding their IWB. Therefore, the key takeaway underscores the significance of maintaining a balance between remote and in-office work arrangements. While remote work offers opportunities for autonomy and flexibility, improved work-life balance, minimised distractions, and heightened concentration conducive to innovation, occasional in-office interactions are essential to sustain optimal levels of IWB. This counteracts obstacles such as social isolation, conflicts between work and home life, and limited collaborative

opportunities, which may hinder the innovation process. Achieving this balance ensures that individuals can leverage remote work's benefits while capitalising on the collaborative and social dynamics inherent in traditional office settings to effectively support innovation endeavours.

The findings also show that demographic factors' impact on IWB in remote work settings was a complex relationship that showed differences in the various phases of IWB.

Compared to their part-time counterparts, full-time employees demonstrate a higher propensity for IWB across all phases, due to enhanced job security, organisational commitment, increased access to resources and skill development opportunities, and greater time allocation towards innovation-related tasks.

Younger employees show increased levels of promoting and championing new ideas, demonstrating that younger employees find it easier to navigate virtual networks to promote ideas in remote work settings. It also suggests a difference in attitude towards change and risk tolerance.

Senior employees and those with longer tenures exhibit increased levels of IWB during the latter two phases, which require increased interaction with colleagues. This is attributed to their elevated decision-making authority, confidence, experience, established networks, historical insights, and job mastery.

In contrast, gender does not wield a notable influence on IWB among remote employees. Irrespective of gender, individuals exhibit comparable levels of creativity in their work, implying that IWB is not inherently linked to gender identity within the framework of remote work environments.

Companies aiming to encourage IWB within remote work settings should broaden their focus beyond mere work arrangements. It demonstrates the importance of recognising how various organisational roles and responsibilities impact such behaviour and what tools can be implemented to facilitate this process. This emphasises the necessity for inclusive strategies to enable all employees, irrespective of their age, tenure, position, or employment type, to actively participate in the innovation process.

8.6. SQ3: Factors of remote working that foster and hinder IWB

In the contemporary landscape of remote work, understanding the dynamics of innovative behaviour among employees is crucial for organisations striving to thrive in an increasingly competitive and dynamic environment. The Ability, Motivation, and Opportunity (AMO) framework offers a comprehensive lens (Cai et al., 2020) through which to examine the dynamics of innovative behaviour among remote employees and the factors that either nurture or impede IWB. Table 8.2 outlines the themes, sub-themes (codes) and AMO framework.

Theme	Ability,	Code
	Motivation,	
	Opportunity	
Resources and	А	Tangible resources for remote work
tools		Exploring and generating new ideas
Perspectives and	М	Negative side of remote working
feelings		Positive side of remote working
		Approach to risk
		Importance of remote working
Support and culture	0	Intangible support for remote work
		Company's support and encouragement of
		innovation
Work spaces and	А	Work remotely
tasks		Creative Work
		High Focus Work
		Work Environment
		Innovation while remote working
Communication	0	Interactions between coworkers
and interactions		Post-COVID work interactions
		Pre-COVID work interactions
		Impromptu connections
		In-person communications
Collaboration and	A	Collaborating in different contexts
relationships		Creativity and group bonding

Table 8-2: Themes, codes and the AMO Framework

8.6.1. Ability of remote working employees to demonstrate IWB

Within the AMO framework, 'ability' is a foundational pillar, encompassing the skills, knowledge, and cognitive capacities that individuals bring to their work (Bhatti et al., 2020; Obaid et al., 2022). For remote employees, the ability to engage in IWB hinges on their proficiency in leveraging digital tools and technologies, as well as their aptitude for independent problem-solving and creative thinking. Moreover, the rapidly evolving nature of remote work demands adaptability and continuous learning as employees navigate virtual collaboration platforms and adapt to new modes of communication.

8.6.1.1. Tangible resources for remote work

The tangible resources provided to employees for remote working can play a crucial role in enabling their ability to engage in innovative behaviours while working remotely. The availability of resources is strongly related to employees' innovative results (Choi et al., 2021; De Jong & Den Hartog, 2007; Simon et al., 2019; Yang & Wu, 2021). These resources, ranging from office equipment allowances to financial contributions towards home office setups, can significantly impact employees' comfort and, therefore, their productivity and creativity in their remote work environments (Ahmad et al., 2023; Russo et al., 2020). With the right tools, employees are better equipped to focus on creative problem-solving and explore new ideas without the limitations of an inadequate setup (Barrero et al., 2021; Lee, 2016).

'So we got given the \$250 allowance for office equipment, like if it was a chair or a screen, something like that, whatever you felt you needed. So I used the allowance on the screen, but I went out and bought a better chair...' EMP01

'We were all given money to go and buy an extra screen or whatever it was you needed to...It might have been a printer or something, whatever equipment you needed to be able to do your job efficiently from home.' EMP02

Providing financial contributions towards home office setups, such as subsidies for desks, monitors, or even electricity costs, demonstrates an organisation's commitment to facilitating the adaptation of office spaces in their homes. By investing in employees' remote work environments, organisations not only remove barriers to innovation but also signal their recognition of the importance of remote work in today's landscape (Baumann & Marcum, 2023; Chafi et al., 2021; Klerk et al., 2021; Toh et al., 2022). Employees who feel supported and valued are more likely to feel motivated to think creatively, experiment with new ideas, and drive innovation within their roles (Janssen, 2005; Jebali & Meschitti, 2020; Opoku et al., 2021).

"...But after the pandemic, everyone is given, well, it's \$750 to go to towards a work home, a home work setup." EMP09

`...they offered you know, give you monitors, you could get money back for buying a desk. And then they also paid a portion of, they increased our salary by \$40 a fortnight I think it was, or \$40 a month to contribute to the cost of electricity.' EMP14

8.6.1.2. Exploring and generating new ideas

Remote work transformed how employees explored and generated new ideas, leveraging various digital tools and experiences to facilitate collaboration, knowledge sharing, and innovation (Gudžinskienė & Mačiuikienė, 2022; Olokundun et al., 2022). With the shift to remote work, employees have adapted using platforms like Microsoft Teams, Zoom, and instant messaging systems to connect with colleagues, share ideas, and brainstorm solutions (Shockley et al., 2021). These tools facilitated seamless communication and provided a platform for spontaneous collaboration, enabling employees to engage in innovative behaviours regardless of their physical location (Kähkönen, 2023; Lee & Kim, 2022; Oyekan et al., 2017).

'We moved from Skype for Business to Microsoft Teams, which allowed us to video call a lot more easily. I assume it was part of the whole shift of working from home that led them to do it...So we have a kind of teams group chat. So if it's small things like minor processes that we need to improve on, I'll drop it in that.' EMP01

'It's a very simple matter of setting up a Team's call. And almost it's amazingly convenient to...And I find that people are more inclined to attend...' EMP02

"...if it's talking to the developers about, you know, this is an idea I had, can we implement it? This is what I've had a look at and, and scoped out. That's usually by Slack message. So the, the whole company uses Slack.' EMP10 "...we had Zoom and we had a an instant chat messenger messaging thing system that we used as well...So there'd be a lot of, and also some colleagues at, at level. There'd be a lot of just quick messages you know, bouncing ideas.' EMP16

Moreover, remote working has spurred the development of new processes and work activities that foster idea exchange and creativity (Costa et al., 2022; Coun et al., 2021; Rachman et al., 2022). Employees have embraced technologies like SharePoint for collaborative document editing and project management, creating a virtual space where ideas can be refined and developed collaboratively. Additionally, some respondents discussed the adoption of weekly or monthly team meetings which provided a structured forum for discussing priorities, sharing progress, and aligning on strategic objectives. Regular meetings are crucial for fostering innovation within organisations as they play a significant role in regulating and ordering interactions in the context of innovation development, serving as a valuable resource for driving innovative activities (Bürkland et al., 2019). Team meetings promote interaction, cooperation, and information sharing, positively influencing innovation activities (Prokop & Hájek, 2023). By leveraging these digital platforms and processes, employees can harness the collective intelligence of their teams and drive innovation forward in a remote work environment.

'It was amazing. And just the way people were more confident to share their ideas and yeah, it was interesting. I actually really enjoyed that dynamic where people that didn't speak up before suddenly did and had a lot to share because there were different ways for them to communicate, particularly if you're introverted, that you had different tools to use to speak up about things. Where in meetings, you have to be an extrovert to speak up...' EMP09

"...so I ramped up my comms with my team ...We have a weekly meeting at the beginning cuz I wanna create the space and understand the priorities and people hearing from each other what they're doing." EMP14

However, the transition to remote work has also presented challenges that can impact employees' ability to engage in innovative behaviours. In a virtual environment, where interactions are mediated through digital channels, some employees may find it challenging to generate ideas organically or to express themselves fully. The absence of face-to-face interactions can limit spontaneous brainstorming sessions and informal exchanges, which can hinder the flow of creative energy that often occurs in physical workspaces (Ferguson et al., 2023; Gajendran & Harrison, 2007; Gruber et al., 2022). Additionally, the asynchronous nature of some digital communication tools may lead to misinterpretation or misunderstandings, potentially stifling collaboration and innovation. The asynchronous nature of these tools can limit immediate feedback, visual cues, and the time available for conversations to develop, potentially leading to misinterpretations and communication breakdowns (Forbes & Gedera, 2019; Vukašinović et al., 2020).

"...if you're in meetings personally and you had, you know, the detractors in the room, you would've been able to kind of move them and change 'em a little bit. Whereas now, you know, if you've got a teams channel going that's 24-7, someone could write a really negative comment that's seen by everyone.' EMP12

Despite these challenges, remote working has created opportunities for employees to explore new ways of thinking and collaborating. By leveraging digital tools and experiences, employees can overcome geographical barriers, connect with colleagues across time zones, and access a wealth of resources and expertise. With the right tools and processes, employees can continue to engage in innovative behaviours, driving meaningful change and progress within their organisations, even in a remote work environment.

8.6.1.3. Creative work

Employees employ various strategies and unconventional spaces to foster creative thinking while working remotely, influencing their ability to engage in innovative behaviours.

Respondents found unconventional settings outside the traditional workspace more conducive to creative thinking. Whether lying in bed, sitting under a tree in the park, or taking a walk, these individuals seek out environments that provide peace, quiet, and a change of scenery to stimulate their creativity. The freedom to explore different spaces outside the confines of the office allows them to approach problems from new perspectives, opens up the free flow of ideas and stimulate creativity (Keinänen, 2015; Oppezzo & Schwartz, 2014). 'I do my best thinking when I'm laying in bed. And quite often, and maybe that's because that's the only time of the day I actually manage to have peace and quiet and thinking time. So I'm not sure. I've never stopped to analyse why I do it...but that's where I solve problems.' EMP02

'Working remotely and in the park that's just around the corner sitting under a tree, like lying under a tree sometimes or walking around and just getting, getting a bit of fresh air and space. For some reason that triggers my brain to kind of have a, a different kind of focus...If I'm in the office, I know that I don't, I don't have the space to actually allow creativity.' EMP10

Routine activities such as showering or driving also serve as opportunities for private focus and problem-solving. The solitude and repetitive nature of these activities create moments of mental clarity and reflection, enabling individuals to work through challenges and generate ideas. Solitude has been demonstrated to offer various benefits that can enhance reflection, mental clarity, and problem-solving abilities. Research indicates that spending time alone can promote concentration, insight and creativity (Lay et al., 2019; Pauly et al., 2018; Thomas et al., 2021).

'That 10 minutes in the shower with the hot water and that silence, just the water running, gives you that time to think. And you're in the robot mode anyway, taking care of business and stuff. But also then the drive.' EMP06

'If I have to really think a problem through, I'll, I'll have a, I'll have a shower.' EMP11

Furthermore, many remote workers highly value the ability to create a private, distraction-free environment. Whether turning off distractions in their home office or stepping away from the screen to think deeply, these individuals prioritise privacy and autonomy to engage in focused, uninterrupted work. Away from the hustle and bustle of the office, they find solace in their home environment, allowing them to delve into complex problems with fewer distractions, thereby fostering innovative behaviours (Adekoya et al., 2022; Amir et al., 2021; Henke et al., 2022).

"...if there's things that I really know I need to knuckle down without distractions, those are the days that over the last 12 months when the office was an option, I chose to work from home to kind of work through those with fewer distractions...I need time to work through it and time to myself to kind of get through the detail of the problem and work out what the resolution will be. I'd say I'd prefer to do that in private at home just because of fewer distractions.' EMP01

'I'm sort of that operator that is almost a little bit pensive. And I kind of brewed over things. And I find when I'm at home, I can just sort of sit there, literally I can turn away from my computer and just look at a blank wall and sort of let things tick over in my brain. You can't do that in the office.' EMP03

8.6.1.4. Innovation

Participants highlighted various strategies and experiences influencing their ability to engage in innovative behaviours while working remotely.

For some employees, the shift to remote work accelerated decision-making and concept approval processes, removing barriers and encouraging a proactive approach to problem-solving. The ability to focus without distractions and allocate dedicated brainstorming time allowed for quicker solutions to emerge.

'In sort of period of the month once COVID hit, we approved our concepts that would've otherwise taken a decade because there was all the naysayers who couldn't say nay because there was no option. You just had to do it...I've got the ability to focus and not be distracted as much as if I was sitting in the office, I can come up with a solution a lot quicker than...I don't have to lay awake at night coming up with it. I've got time during the day to actually turn my mind to it.' EMP02

'... But the extent that you do identify a problem at home when you're at home, I think it's easier to find the time to get the head space to think about the solution than when you're in the office...And plus, the person who brings you the problem also joins in terms of trying to create the solution. So it sort of almost becomes a brainstorming session, just an informal brainstorming session rather than just sort of sitting in the corner quietly contemplating by yourself. And that's got merit.' EMP02 However, virtual collaboration also presents challenges, such as the potential for miscommunication (Vukašinović et al., 2020) or the lack of spontaneous interactions that occur in traditional office settings. Despite technological advancements, some individuals need help with specific platforms or find virtual brainstorming less dynamic than inperson sessions (LaPensee et al., 2021), which can hinder the flow of ideas and impede the creative process. The dynamic nature of face-to-face interactions, where participants can engage in spontaneous discussions and build on each other's ideas, is often challenging to replicate in virtual brainstorming environments. Nonetheless, the motivation to innovate remains high, driven by the need to overcome inherent weaknesses and adapt to new ways of working.

'Or a new idea is really a solution to a problem, right? So if you don't have the problem. The problem is the thing that stimulates the new idea. You come across more problems in the office because other people are inclined to tell you their problems just by virtue of the fact that you're sitting there. So you do spend more time stimulated thinking about solutions when you're in the office than when you're at home, I think.' *EMP02*

'If you're in meetings personally and you had, the detractors in the room, you would've been able to kind of move them and change 'em a little bit. Whereas now, if you've got a teams channel going that's 24-7, someone could write a really negative comment that's seen by everyone for, you know, however long it takes you to be able to respond. But then you go back and like the opportunity for things to blow up can be quite strong...I just think for multiple barriers, even if you've got a great technology, not everyone has the same handle on it...' EMP12

'I'd say [motivation to be innovative] it's possibly higher more than when we are in person just because, like I said before, there's that kind of inherent weaknesses that you need to overcome...It's much more difficult for people these days to kind of work with their desk neighbour to solve an issue like they would've done in the past. You just ask the person next to you if they're on your team or not...' EMP01

Participants employ various techniques to foster innovation in the remote environment, such as engaging in creative thinking outside the office, leveraging alternative spaces like cafes or implementing virtual whiteboard sessions to encourage brainstorming and idea generation. Minimising external distractions fosters clearer thinking and problem-solving. Similarly, engaging in creativity exercises like virtual brainstorming, which mirrors office practices, enhances overall IWB.

'I do that at home, but then I've also, I'll also go to the cafe. So I might take a break and, go and get a coffee and sit down there and with my notebook and do some sort of creative thinking there...However, I do miss a whiteboard sesh, so I'm such a big fan of, let's jump in the room and chuck it on a whiteboard kind of thing. I did a whiteboard, I've got one here. That was one of the first, I bought that before a decent desk. Cause that was the first thing I noticed I missed working at home was, a whiteboard. Like just that creative piece. And so yeah, I still try and encourage that with the types of meetings with my team and my direct reports...let me know if you wanna just book a book the time and say whiteboard sesh. Yeah. And so I know that we're just gonna have a brainstorm creative time.' EMP14

'[feel more creative] outside the office environment. It's the space. And because, well, my mind is at rest, I think. So I'm not thinking about, but I have to do this and I have to do that at rest, but that particular big problem has been weighing on my mind all day. And so it's sitting there and then suddenly it comes to me. So I think it's just getting rid of the white noise.' EMP09

Furthermore, participants emphasise the importance of collaboration and dialogue in the virtual workspace in order to foster innovation. While remote communication may lack the spontaneity of face-to-face interactions (Ferguson et al., 2023), effective communication technology provides opportunities for sharing ideas, receiving feedback, and refining solutions collaboratively (Adibe et al., 2023; Batarseh et al., 2017). Regular virtual team meetings and virtual presentations for example enable individuals to share insights, trigger ideas, and explore new possibilities collectively, enabling innovative thinking despite physical distance (Bürkland et al., 2019; Prokop & Hájek, 2023).

'So I'm like, oh, I saw a great article. I'll just share it on the, on the messaging. Like, just have a read of this guys. This could be great for us doing blah, blah, blah. And see what the reaction is. And then we have a team meeting online. And the good thing about that is that I can do presentations, I can share my screen and I can show where I'm thinking with things...I need some time to think that through. And so yeah, I'd prefer to do that in sort of a quieter space, but where that might, I get some ideas triggered and I start thinking about things maybe after a conversation with people.' EMP17.

8.6.1.5. Connections in the digital realm

Maintaining connections and relationships while working remotely poses challenges (C.-L. Yang et al., 2022), but employees have devised strategies to nurture these bonds, recognising their importance in fostering innovation and collaboration (Charalampous et al., 2018; Cripe & Burleigh, 2022; Salaran & Maritz, 2013).

Some respondents found that remote working limits their day-to-day interactions, leading to a sense of disconnection from colleagues outside their immediate teams, which is supported by literature (Felstead & Henseke, 2017; Hafermalz & Riemer, 2020).

'I'd be speaking with them throughout the office and that kind of thing. Whereas now I have a social call or two a day with some a couple of my close friends in the office. And then really only be speaking with the people that are on my team rather than a wider division. There'd be people in the audit team that if we were in the office, I'd be speaking to once a week that I probably haven't spoken to this month or even the last couple of months [because I am remote working].' EMP01

'I don't think it can replace...get everybody together in a room with a drink in their hand. I think that's a different experience.' EMP05

To bridge this gap, they schedule regular social calls or virtual meetings to catch up with coworkers, fostering a sense of camaraderie and shared experiences. These informal gatherings provide opportunities to discuss work-related matters and personal experiences, providing a sense of belonging and connection (Flannery et al., 2022; Kuntz, 2020)

"...we don't have a lot of day-to-day interactions. So we made time to FaceTime yesterday and, and we're gonna do that, you know, maybe once a month, just have a half-hour chat on FaceTime...' EMP10. 'I had a weekly team meeting and so that was really good. We'd catch up on a Monday morning, talk about our weekends. So it was, you know, yes we'd talk about work, but we'd also be trying to catch up a bit socially.' EMP16

Additionally, employees emphasise the importance of celebrating successes and fostering a culture of gratitude in virtual environments. Research has shown that recognising and celebrating achievements can act as a catalyst for promoting creative problem-solving and thinking (Dai et al., 2019; Malik et al., 2023; Naushad, 2022). Weekly virtual team meetings serve as forums for recognising achievements and expressing appreciation, reinforcing a sense of belonging and motivation. By acknowledging and celebrating each other's contributions, teams cultivate a positive work environment that encourages creativity and innovation.

'And it also prompts other people I''ve noticed too, to say, okay, you know, this person's done something extraordinary, I'm gonna tell everybody about it. And so it just, it fosters that that kind of real vibe of celebrating everybody's successes, even if they're really tiny.' EMP10

'And then the Friday meeting was a, what have we done this week and kind of a congratulations and a kind of Oh, fantastic. Gratitude and recognising what had been done well that week.' EMP11

Furthermore, employees leverage technology to maintain ongoing communication and connection with colleagues. They make deliberate efforts to reach out to coworkers for casual conversations, ensuring that interactions are not solely task oriented. By incorporating spontaneous interactions into their daily routines, employees create opportunities for serendipitous connections and idea-sharing, mirroring the informal interactions that occur in physical office spaces.

'And then our meetings were turning into more about the health and wellbeing, "What are you doing, what ideas have you got to share with the team about managing your mental and physical health when we're in lockdown?" and we can't easily connect with our friends and we can't or our family.' EMP09 'And so there's generally kind of there's a couple of key people that I kind of, hey, you know, thinking about this quite informal, it'll come through on a teams chat or I'll give them a call like, if, if you are green on teams, I'll call you instead of emailing you. I think you know, it gives us more of that connection as well. And you get to speak to different people all day and it's not just an email inbox kind of thing...it's less about the in-person and virtual, it's more about the don't make every interaction only when you need something from them...I'll have a look in my, my chat list on who's green and I'll just ring 'em. So, a lot of people interact for purposeful reasons now. So, because I need something from you. Yeah. I think yeah, people need to be more mindful of those kind of, you know, pass by the office, create those moments.' EMP14

Several key insights emerge in analysing employees' 'ability' to engage in IWB while working remotely. Providing tangible resources for remote work, such as additional screens and communication platforms, enhances employees' technical capabilities. This facilitates innovative behaviours by providing the necessary infrastructure and tools for idea generation, collaboration, and problem-solving in virtual environments. Employees demonstrate their ability to adapt to remote work setups and utilise digital platforms effectively, contributing to the organisation's innovation efforts.

Ability also refers to individuals' skills, knowledge, and resources necessary to innovate effectively. In the remote work context, participants discussed their ability to adapt to digital tools and platforms for collaboration, with acknowledgment of technology's role in enabling virtual brainstorming sessions. Participants demonstrate cognitive skills such as problem-solving and critical thinking while working remotely, particularly in environments and situations that amplify these abilities, which are essential for IWB.

Employees demonstrate varying capabilities to innovate based on their access to resources, skills, and knowledge. For example, those who express a preference for face-to-face collaboration rely heavily on interpersonal communication skills and the ability to read nonverbal cues to foster creativity. Conversely, individuals who thrive in virtual environments possess strong digital literacy skills and are adept at leveraging technology to facilitate idea-sharing and collaboration. Thus, IWB is contingent upon employees' proficiency in utilising both traditional and digital mediums for collaboration and communication.

8.6.2. Motivation of remote working employees to demonstrate IWB

At the heart of the AMO framework is 'Motivation', which is pivotal in shaping individuals' willingness and eagerness to engage in IWB (Knies & Leisink, 2013; Obaid et al., 2022). For remote employees, motivation emerges as a key determinant, fuelled by factors such as autonomy, work-life balance, flexibility, and personalised work environment. Nevertheless, the virtual nature of remote work also brings forth challenges, including sentiments of isolation, disconnection, and tendencies towards overworking, which can impede employee motivation to undertake innovative activities.

8.6.2.1. Positive side to remote working

Remote working offers a plethora of benefits that can significantly enhance employees' motivation to exert innovative behaviours. One notable advantage highlighted by respondents is the elimination of the daily commute. For many, this means reclaiming valuable time that would otherwise be spent navigating traffic or crowded public transportation. This newfound time can be directed towards other endeavours, like creative thinking, brainstorming or ideation, ultimately increasing IWB (Agrawal et al., 2018; Mullens & Glorieux, 2023).

"... avoiding that commute just to be in a certain location for no particular reason." EMP10

`...just having that opportunity not to have to travel...You gain some more time, which is really nice. And you do get a little bit of thinking time.' EMP07

Moreover, remote work provides employees the freedom of creating their ideal physical work environment to suit their individual preferences and needs. Personal control over the workspace can reduce environmental distractions, enhance mood, improve environmental satisfaction, and increase productivity and creativity at work (Samani et al., 2015; Yu & Wu, 2021). This autonomy over one's surroundings can inspire a sense of ownership and empowerment, increasing motivation to engage in positive work outcomes such as exploring new ideas (Jebali & Meschitti, 2020).

'If I'm in the office, I know that I don't, I don't have the space to actually allow...having that time and the quiet and the space to, to just focus on something differently. To just take it a take a different angle in trying to find a solution.' EMP10

'So positives definitely that for me the ability to have my own space...I don't know if you've ever heard of misophonia, it's a neurological condition where certain noises can cause their triggers and they can cause a reaction. And even you know, some noises in the office. So, for me that's fantastic having my own space.' EMP16.

Additionally, the absence of office noise and distractions (Amir et al., 2021; Henke et al., 2022) can be advantageous for concentration and uninterrupted focus. The findings underscore the necessity of having the time, quiet, and space to focus on tasks without interruptions, a luxury often challenging to achieve in a traditional office setting. Without the interruptions and distractions commonly found in a bustling office environment, employees can delve deeply into complex problems, explore new ideas, and devise innovative solutions with greater clarity and efficiency (Eerde et al., 2022).

'So, you have the tools now to kind of set yourself two to three hours on 'Do not disturb' or whatever, where you can just rip through the work distraction-free. Whereas if you're in the office, someone might come up and ask you a question, and you're probably more easily going to get sidetracked....we've come up with a lot of things that help us work better remotely over the last two years.' EMP01

'Because it's a lot more private. And those interruptions. You get those long breaks, of say four to five hours uninterrupted...Smash things out. Strategy. We call it thinking time as well.' EMP06

"...And for somebody like me who is an introvert...Having the luxury of not having all that noise and distraction and activity around me.' EMP10

The flexibility to manage one's schedule enables individuals to pursue personal interests and empowers employees to work in a manner that best suits their individual preferences and rhythms (Lange & Kayser, 2022). Integrating work and personal life facilitated by remote work can contribute to a greater sense of work-life balance (Blumberga & Berga, 2022; Mattarelli et al., 2022; Neidlinger et al., 2022). Improved work-life balance can lead to higher levels of creativity, productivity, and job

performance (Haider et al., 2018). By eliminating the need for a physical separation between work and home, employees can structure their day to accommodate both professional responsibilities and personal commitments. This balance allows individuals to approach their work with a clear mind and a renewed sense of purpose, enabling them to tackle challenges with creativity and innovation.

`…having that time and the quiet and the space to, to just focus on something differently. To just take it a take a different angle in trying to find a solution.' EMP10

"... being able to balance my lifestyle around my work." EMP12

"... allows me to have a better work life balance and to do more things in my life which makes my weekend freer." EMP14

"...work-life balance and being able to manage being a working parent, manage being a carer. And I went back to study last year as well, so I was excellent too and trying to manage that and, could do all of that because I didn't have the commute and feel a bit more flexible in my own like work if I don't finish things when I need to, I can stagger my day a bit more.' EMP17

Furthermore, remote work offers flexibility for working parents and caregivers (Augustine et al., 2023; Koreshi & Alpass, 2023). The convenience of being close to home allows employees to attend to family matters or unexpected emergencies swiftly. This seamless integration of work and personal life provided by remote working enhances overall employee wellbeing and engagement (Costantini & Rubini, 2021; Klerk et al., 2021; Wong et al., 2020), which in turn can positively influence innovative behaviours (Honkaniemi et al., 2015; Smith et al., 2022; J. Wang et al., 2017).

'I would say the positives definitely like flexibility around family life. Like it just, it's nice to be able to do drop-offs pickups can ride the bike and pick my daughter up...I think structuring my day and being able to, you know, work how and when I, I like Yeah.' EMP13.

'The positives are definitely for me have been around work-life balance and being able to manage being a working parent, manage being a carer. And I went back to study last year as well, so I was excellent too and trying to manage that and, and could do all of that because I didn't have the commute and feel a bit more flexible in my own like work if I don't finish things when I need to, I can, I can stagger my day a bit more and yeah.' EMP17

8.6.2.2. Negative side to remote working

While remote working offers numerous advantages, such as flexibility and autonomy, it also presents various challenges that may hinder employees' motivation to display innovative behaviours. One of the most commonly cited drawbacks is the feeling of isolation (Saurombe et al., 2022; Toscano & Zappala, 2020). The lack of face-to-face interaction with colleagues can lead to feelings of isolation and disconnection (Juchnowicz & Kinowska, 2021; Noh & Lee, 2022) which hinders collaboration and idea exchange (Babin et al., 2021; Waizenegger et al., 2020), essential components of IWB.

'I certainly don't think that I would want to work five days from home every day. I think that'd be isolating.' EMP03

'I know some of my colleagues were finding it quite isolating. And we used to have a daily check-in and I was a bit like, can we not have a daily check-in? And they went, no, this might be the only time I see another person, so could we keep them on?' EMP17

Furthermore, remote work can blur the boundaries between work and personal life, making it difficult for individuals to disconnect (Andrade & Fernandes, 2021; Baumann & Sander, 2021; Palumbo et al., 2020). Work can tend to encroach on personal time, leading to longer work hours, potential conflict and burnout (Bakarich et al., 2022; Hayes et al., 2021). Individuals experiencing high levels of burnout tend to exhibit lower overall creativity (Alessandri et al., 2018; Şahin, 2023). Burnout has been identified as a moderating factor between intellectual capital and IWB, indicating a significant negative relationship between burnout and IWB (Narzary & Palo, 2021). The absence of clear delineation between work and leisure can diminish motivation and impede creativity, as employees struggle to find time for relaxation and rejuvenation (Yu & Wang, 2022).

'One thing that does happen I think is it makes it a little bit harder to disconnect. Even though you can set your own hours and set your own day, it's definitely easier for things to creep and for you to while I'm already at home, it's easier for me to do the extra hours and just work through to like say 10:00 pm or something...I think now I've noticed that there are things that take a lot longer than they used to because the face time made it easier to sort of teach people and also to interact with the clients. Working as a team is easier when you actually are kind of together as a team.' EMP01

"...the working too long an hour. So, for example, you jump on first thing in the morning, and then I'm still online at 7:30 at night, and I've barely left my seat. And on weekends, you'd walk past your desk, and your computer was on, so you would check your emails or...It's just you were never off. I found that challenging.' EMP07

'I found that I was working on Friday and then it would just go to 8:00 and then I'd find I was working on Saturday and Sunday as well, just because it was there. It was almost an addiction ... ' EMP09

Remote working employees may encounter challenges in proving their value and achievements to their peers and superiors, which can impact motivation. In addition to feelings of isolation and reduced social interactions, employees experience a lack of visibility, which may hinder employees in showcasing their contributions effectively (Chafi et al., 2021). Supervisors may also struggle to monitor and evaluate the performance of remote employees effectively due to reduced visibility (Gong et al., 2022; Maden-Eyiusta & Alparslan, 2022; Zoonen et al., 2021). Employees working remotely may need to adapt their communication strategies, seek opportunities to showcase their work through virtual platforms, and actively engage with supervisors to ensure that their value and achievements are recognised in a remote work environment. There is also the perceived additional effort required to demonstrate productivity in remote settings. This pressure to constantly showcase one's work can create a culture focused more on output quantity rather than quality, potentially stifling innovative thinking and experimentation (Lykourentzou et al., 2022).

'The negatives I think it is just a little bit of extra energy to prove your value or especially if you're a remote and then it's a KPI-driven job. Like you just need to make sure you're achieving and that's, you are sharing with people your achievements.'

EMP13

'Some of the negatives are that yes, you're a bit invisible in the organisation. In terms of there's a lot of that also out of sight, out of mind...' EMP17

Another expressed significant downside of remote work is that it can exacerbate communication challenges, particularly in collaborative settings. Without the face-to-face interactions and spontaneous, informal discussions that occur in an office environment, communication may become more formalised (Cao et al., 2021; Kershaw et al., 2021) and motivation to develop new ideas can decline. Remote work has been shown to lead to a decrease in informal communication and incidental exposure to other employees (Choudhary & Mishra, 2021; Ferguson et al., 2023; Viererbl et al., 2022). Coordinating meetings and ensuring everyone is on the same page can require additional time and effort, potentially slowing decision-making processes and hindering collaboration. This lack of informal communication can impede the flow of ideas and inhibit IWB.

'I think now I've noticed that there are things that take a lot longer than they used to because the face time made it easier to sort of teach people and also to interact with the clients. Working as a team is easier when you actually are kind of together as a team.' EMP01

'Now you got to set up a time and you got to see if you're both free. And then it's quite formal because it's a call. So, there's something to be said about being back in the

office...' EMP02

Furthermore, logistical issues such as inadequate workspace or distractions at home can impede concentration (Edilson et al., 2022; Pang et al., 2021) and hinder innovative behaviours. This can include noisy home environments or limited space for dedicated work areas. The lack of necessary equipment and adequate space can impede employees' creativity and innovation (Bergefurt et al., 2023).

"...if you've got kids at home or if you've got a noisy dog or if you don't have the space." EMP10 Employees in remote settings also lament the loss of the social aspect of work (Felstead & Henseke, 2017; Hajjami & Crocco, 2023; Pradhan & Hati, 2019). There is the loss of casual interactions (Gilmartin et al., 2020; C.-L. Yang et al., 2022) and camaraderie with colleagues, which can serve as sources of inspiration and support for innovative thinking. Casual interactions and social aspects have been shown to play a significant role in fostering creativity and innovation. Spontaneous communication, casualness, and interpersonal familiarity create an atmosphere conducive to innovation, idea generation, and creativity (Gerards et al., 2020; Lee et al., 2005; Tønnessen, 2022).

"...there's also the social aspect. People you'd like to have that, what did you do on the weekend? Or did you win on the races or whatever...' EMP02

8.6.2.3. Importance of remote working to employees

The importance of remote working to employees extends beyond mere convenience; it represents a fundamental shift in how individuals perceive and approach their work-life balance. For many, remote working is not just a preference but a necessity, offering them the flexibility and autonomy needed to thrive in both their professional and personal lives. This recognition of remote work's value is a powerful motivator for employees to engage in innovative behaviours, even when working remotely.

Employees who value remote working emphasise the importance of flexibility and empowerment. They view remote work as a means of reclaiming control over their time and environment (Adekoya et al., 2022; Felstead & Henseke, 2017; Koekemoer et al., 2021), enabling them to balance professional responsibilities and personal commitments (Brooks et al., 2022). Increased autonomy is linked to fostering creativity and innovation (Beugelsdijk, 2008; Lumpkin et al., 2009) and IWB (Criscuolo et al., 2014; Giebels et al., 2016; Sönmez & Yıldırım, 2019). This sense of autonomy empowers individuals to approach their work with renewed enthusiasm and creativity, knowing that they have the freedom to structure their day in a way that optimises their work pursuits.

'I'd say it quite important to me. I'm someone who wants to go back into the office at least to some extent. So, I'm pretty comfortable with saying it's important to have the flexibility.' EMP01

'I think it's very important. I think it's all about the hybrid balance and giving you that feeling of empowerment as well.' EMP06

Remote working is viewed as a competitive advantage in today's evolving work landscape. Employees recognise the need for organisations to adapt and embrace flexible work arrangements to remain competitive (Klindžić & Marić, 2019; Xiu et al., 2017), attract top talent (Onken-Menke et al., 2017; Schmoll & Süß, 2019) and retain valuable employees (Bainbridge & Townsend, 2020; Bilan et al., 2020; Haddad et al., 2023). By offering remote working options, businesses signal their commitment to supporting employee wellbeing leading to higher levels of organisational commitment and job satisfaction. This positive perception of the organisation can contribute to increased motivation to engage in innovative behaviours, knowing that their organisation values and prioritises flexibility and their wellbeing.

'I think we need to be more flexible...I think as a business, we need to offer some more flexibility for our staff. Because everybody's offering it. If we don't get on this bandwagon, we are going to get left behind.' EMP07

Many employees express a deep appreciation for the ability to integrate work and life seamlessly, allowing them to pursue their professional goals while also prioritising personal priorities, such as spending time with family. In this regard, remote work is closely linked to job satisfaction (Kähkönen, 2023; Mostafa, 2021). Higher levels of job satisfaction positively affect innovation performance (Banerjee & Bag, 2022; Hu & Zhao, 2016; Nasution et al., 2021). This alignment between work and personal values enhances employee motivation, creating an environment where individuals are more inclined to explore new ideas, and engage in innovative behaviours.

'I want to keep working from home. I just enjoy, it gives you balance...I think it really comes down to just the freedom to be able to sort of work how you want to work. And where you think you're going to be able to deliver the best results.' EMP03

"...the ability to integrate work and life, gotta get the boundaries right...the ability of work my life, work around my life so much more now is just, it's so great for me to be able to progress and work at a level in my career that I didn't think was possible...'

EMP14

Several key insights emerge from analysing the 'motivation' of employees to engage in IWB while working remotely. Employees prioritising flexibility often cite its positive impact on work-life balance, job satisfaction, and personal fulfilment. The ability to balance work and personal commitments is a crucial driver of job satisfaction and wellbeing. This balance between work and personal life can drive individuals to approach tasks and challenges with renewed energy, enthusiasm, and creativity, thereby fostering IWB.

However, while remote work offers flexibility and comfort, many employees yearn for the camaraderie of office life. The absence of daily interactions can lead to feelings of isolation and detachment. Recognising this, organisations must prioritise fostering social connections and providing opportunities for meaningful engagement to cultivate a thriving remote work culture.

8.6.3. Opportunity of remote working employees to demonstrate IWB

Within the AMO framework, 'opportunity' is a critical determinant, delineating the organisational and environmental conditions that facilitate or hinder innovation (Marín-García & Martinez-Tomas, 2016). For remote employees, opportunities for innovation manifest in various forms, ranging from access to intangible resources and technology to supportive organisational culture and encouragement of innovation. Remote work environments often offer unparalleled flexibility and autonomy, providing individuals with the freedom to explore new ideas and experiment with novel approaches. Additionally, virtual communication tools and digital collaboration platforms bridge geographical distances, fostering seamless interactions and idea exchange among remote teams. However, the virtual nature of remote work also presents challenges, such as potentially limited informal interactions, which can impact employees' opportunities for innovative collaboration.

8.6.3.1. Intangible support for remote working

Remote workers often receive intangible support from their organisations, ranging from mental health programs to team-building initiatives aimed at enhancing wellbeing and resilience. These support systems play a crucial role in ensuring that employees feel valued, connected, and equipped to navigate the challenges of remote work (Park et al., 2021; Tuzovic & Kabadayi, 2020), where increased wellbeing can enhance creativity (Smith et al., 2022; Tan et al., 2021; J. Wang et al., 2017).

...we've got mental health type programs running. We try and make sure that we do some team building type stuff fairly regularly...' EMP02

`...they've got a subscription to some employees' service or something or other, that they give us....It's called Be Well. There's an actual company that they outsource it to. So we've got that subscription...' EMP03

One aspect of this support is the emphasis on mental health and wellbeing, particularly during times of crisis such as the COVID-19 pandemic. Organisations may offer access to Employee Assistance Programs (EAPs) and mental health resources to help employees cope with stress and maintain their emotional wellbeing. By supporting employees in achieving a healthy balance between work and personal life, organisations create an environment where individuals feel more energised, motivated, and capable of generating innovative ideas.

'They enrolled us into I think it's called The Resilience Project. We got those presentations. I think that was mostly all linked to kind of improving mental health and that sort of thing for working from home...' EMP01

'There was a lot of mental health support. Lot of mental health support. The EAP program went through the roof. That was offered to...immediate family...But again, the same things, a lot of contact, a lot of checking in, all that kind of stuff,'. EMP06

"...we do get access to EAP through work, those sorts of things. We have health and wellbeing teams, so they do take that quite seriously...So they're constant, so that was long before the pandemic, but obviously I think a few more layers were added to that, a few more offerings." EMP09

In addition to mental health support, organisations may implement team-building activities and virtual initiatives to foster a sense of community and belonging among remote workers. Virtual social events, such as online gatherings and casual meetups, provide opportunities for employees to connect with their colleagues, share experiences, and build meaningful relationships outside of work tasks (Ghosh et al., 2023; Rathnaweera & Jayathilaka, 2021). Social interactions can stimulate creativity (Costa et al., 2022; Lebuda et al., 2016), and in remote settings it can reduce social isolation and contribute to a positive work culture, which are essential ingredients for IWB.

'We made a point of just having little check-in points with them, a message or a phone call.' EMP10

'There was certainly a focus on getting people together socially every now and then socially online.' EMP16

8.6.3.2. Organisational support for innovation

Companies actively support and encourage innovation among their employees through various initiatives and programs, recognising the value of fostering a culture of creativity and forward-thinking. One common approach is the implementation of formal innovation programs or competitions where employees are invited to submit their ideas for consideration. These programs often involve a structured process of evaluation and selection, with awards or recognition given to individuals or teams whose ideas demonstrate significant potential for driving positive change within the organisation. These programs serve not only to acknowledge innovative efforts but also to inspire and motivate others to think outside the box and explore new ideas, even when working remotely (Engelen et al., 2017; Malik et al., 2023; Shanker et al., 2017).

'So we definitely used to have a big innovation program that was formally rewarded, but I don't think that's necessarily happening anymore. Innovation is one thing that we look for as we kind of encourage people to develop when they're going for promotions.' EMP01

'We run an international run innovation competition. And some guys out of our group went through the national final...And they won a prize for that.' EMP02

'I know we have an office award,...And every quarter, there's one person from each service line that is selected as the winner of this award. And it's assessed on a submission basis.' EMP03 In addition to formal programs, companies may create opportunities for idea generation and collaboration through informal channels such as working groups or innovation hubs. Innovation oriented clusters and hubs play a crucial role in driving collaboration and idea generation (Haukipuro et al., 2023; Littlewood & Kiyumbu, 2018). By providing platforms for employees to share their ideas and collaborate with colleagues across different departments or service lines, organisations can tap into the diverse perspectives and expertise of their workforce, leading to more innovative solutions and approaches (Cui et al., 2022; Yoo et al., 2023). By leveraging digital platforms and communication tools, organisations can facilitate virtual brainstorming sessions, innovation workshops, and knowledge-sharing sessions, allowing employees to connect and collaborate regardless of their physical location.

'We used to have the opportunity hub...and the simplification program...I think the role of simplification, and we haven't met as a committee for some time now, but the opportunities of simplification program are still there.' EMP08

'I've always had a lot of freedom in the role because I manage all the operations, so I have quite a bit of freedom to do what I want...But yeah, always encouraged with innovation.' EMP09

'There's definitely recognition when people do come up with something that, that creates a change that's a really good enhancement. And there, I mean, everybody's got an opportunity to contribute ideas and I mean...we actually created a working group because quite often I was asking for a lot of changes. And I thought there must be other people in the business that, that have ideas as well. So we'd get people to bring together ideas.' EMP10

Company support and encouragement for innovation play a crucial role in shaping the opportunities for employees to engage in innovative behaviours (Khalili, 2016; Kock et al., 2014; Sharifirad & Ataei, 2012). Through formal programs, informal initiatives, recognition efforts, and professional development opportunities, organisations can create an environment where creativity, collaboration, and ideation are not only encouraged but also rewarded, ultimately driving innovative behaviours.

8.6.3.3. Impromptu connections

The shift to remote work has altered the dynamics of impromptu connections within workplaces, influencing innovative behaviours. Spontaneous interactions in corridors, coffee breaks, and desk-side chats are common in office environments (Maisha et al., 2023; Oshima & Asmuß, 2017; Seddigh et al., 2015). These unplanned interactions often sparked brainstorming sessions, facilitated idea exchanges, and nurtured the relationships crucial for fostering innovation. However, remote work has introduced a more formalised approach to communication (Ferguson et al., 2023; Viererbl et al., 2022), with impromptu interactions giving way to scheduled meetings and structured conversations (Cao et al., 2021; Kershaw et al., 2021), which can be more rigid. The informality and spontaneity of face-to-face exchanges have been replaced by deliberate efforts to initiate communication, often through digital platforms. Without the casual interactions that occur in traditional office settings, employees may they have less opportunity to spontaneously brainstorm or bounce ideas off one another, reducing IWB.

'It's a bit more difficult when you're working remotely, because yeah, like you say, it's more rigid. You schedule a time, so it seems more formal. It's not just

that sort of impromptu, "Hey, can I just ask you this? I'm having a bit of trouble with something." Yes, definitely a lot less. And it's a lot more narrow focused.' EMP01

'I mean, if I've got a meeting with my manager sort of coming up, I might just sort of tack on at the end, "Oh gee, do you have two minutes me to run something past you?"' *EMP03*

'Yes, that's right. So people had to deliberately contact me rather than stop me in the corridor.' EMP07

The absence of impromptu connections extends beyond mere convenience; it also impacts the depth and breadth of professional relationships. Respondents lament the loss of small talk and informal check-ins, recognising these interactions as crucial for building trust and rapport on a personal level. Small talk serves as a social ritual that is fundamental to nurturing relationships and building new relationships (Ferrante, 2021; Methot et al., 2021). Fostering both formal and informal relationships is key to innovation performance and desirable employee behaviour (Ahmad et al., 2022; Muñoz-Pascual & Galende,
2020). If employees lack relationships with their colleagues, it can reduce the opportunity for them to collaborate and generate new ideas together.

'I think the challenge was not being able to do those ad hoc, you know, just turn around and have a chat...I've already got established relationships with a lot of people where I am now, so that probably makes it a bit easier because I've got that establishment. But certainly you know, you're just not having those regular chats ' EMP016

'I love it when I go into the office because I get to see my team and that typical, "Let's go for a coffee," and just having that informal, that chit chat where you get to connect on a more personal and a human level so that you're building up that trust and rapport with someone.' EMP09

'Most of the time in the office I would be meeting with people, like chatting, you grab people in the hallways and that's, it's valuable stuff.'EMP17

The formality of remote communication presents challenges in maintaining the spontaneity and informal exchange of ideas that drive innovation. The act of initiating a conversation has become a deliberate decision, perceived as a barrier that can restrict the flow of ideas and inhibit spontaneous collaboration.

'And just you being able to sort of turn your head and ask the person next to you and answer to a question.' EMP02

'I do feel like there's a whole segment of the company that I, I feel like I don't know well at all. Because it's, you know, they're busy too. So suddenly popping up in their chat and saying, "Hey, can we have a chat and can we organise a meeting?" It's different to popping by their, their desk and having a two-minute chat with them and, you know, then popping back in a month later and saying, "How are you settling in? How are you finding things?" I guess cuz it's impromptu, whereas like this, it's more purposeful. You, you have to actually have it, you know, make a point of sending an email or picking up the phone.' EMP010

Several vital insights emerge in analysing the 'opportunity' of employees to engage in IWB while working remotely. Opportunity plays a pivotal role in enabling employees to translate their abilities and motivations into innovative actions. The findings highlight the importance of creating a conducive environment that fosters collaboration, interaction, relationship building and knowledge sharing. Platforms such as innovation forums, newsletters, and conferences provide employees with opportunities to exchange ideas, seek feedback, and collaborate on innovative projects. While digital platforms offer new avenues for collaboration and knowledge sharing, they may also impose barriers to spontaneous and informal communication. There are diminished opportunities for impromptu discussions, which foster collaborative brainstorming and idea generation. As organisations navigate remote work arrangements, addressing the need to foster impromptu connections become crucial for encouraging IWB.

Additionally, organisational support and resources, such as mental health programs and engaging social team initiatives, play a pivotal role in bolstering employee wellbeing by providing avenues for support and connection within the workplace. These initiatives not only contribute to a healthier and happier workforce but also cultivate a sense of belonging and camaraderie among employees. This conducive environment not only encourages collaboration but also sparks innovation as individuals feel empowered to share ideas and work together towards common goals, ultimately fostering IWB.

8.6.4. Summary of factors that foster and hinder IWB

Employee perceptions of remote work presented a myriad of factors that foster and hinder IWB. On the fostering end, providing tangible resources to employees, such as necessary equipment and materials, lays a solid foundation for creative endeavours. Additionally, financial contributions to enable remote working and ensuring access to the right tools and technology empower employees to explore new ideas and methods efficiently. Regular meetings and get-togethers, whether virtual or in-person, facilitate collaboration and idea exchange, while access to unconventional settings and moments of peace and quiet offer opportunities for uninterrupted creative thinking. Private spaces and fewer distractions contribute to an environment conducive to deep focus and innovation, as does dedicated time allocated explicitly for brainstorming and ideation. Social calls and informal gatherings, as well as celebrating successes, cultivate a sense of camaraderie and motivation, fostering a positive atmosphere for innovation. Autonomy, flexibility and a balanced work-life dynamic further empower individuals to experiment and innovate, while personalised work environments, mental health support, and wellbeing initiatives ensure that employees feel supported and motivated to bring their best ideas forward. Lastly, organisational encouragement to exchange and generate new ideas fosters a culture of innovation and continuous improvement within remote work settings.

Conversely, several factors can impede IWB in remote environments. The lack of face-to-face interactions diminishes spontaneous exchanges and can lead to misinterpretations or misunderstandings in digital communications, hindering the flow of ideas. Disparities in technology skills can create barriers to collaboration and idea sharing, while feelings of disconnection and isolation from colleagues can stifle creativity. Burnout and blurred boundaries between work and personal life can erode motivation and creative energy, as can the expectation of working longer hours and constant distractions at home. Inadequate home workspaces and a diminished social aspect further compound these challenges, making it harder for individuals to stay engaged and inspired. Formalised communication channels and added time spent coordinating meetings detract from time that could be allocated to innovative pursuits, while feelings of invisibility and difficulty proving one's worth can dampen enthusiasm for contributing new ideas. Additionally, a focus on outcomes rather than quality may prioritise quantity over innovation, inhibiting the exploration of novel approaches and solutions.

8.7. SQ4: The employee experience between remote and office-based working

In today's rapidly evolving work landscape, there is a question surrounding the optimal work environment for encouraging innovative behaviours, particularly in light of the contrasting experiences of employees in office-based versus remote settings. This research question seeks to compare the intricate dynamics of IWB within these divergent contexts. By delving into these findings, this section aims to offer a comprehensive understanding of how the working environment shapes the IWB of employees.

8.7.1.1. Frequency of working remotely

Employees' frequency with remote working varies widely, ranging from those who predominantly work from home to those who only occasionally opt for remote work. The frequency of remote work can significantly impact an individual's ability to engage in innovative behaviours as found in the quantitative study, influencing factors such as work environment, routine, and collaboration opportunities. For employees who predominantly work remotely, having a dedicated home office setup is crucial to create an environment conducive to innovation (Ahmad et al., 2023; Russo et al., 2020; Samani et al., 2015). These individuals have invested time and resources into creating a workspace tailored to their needs, allowing them to focus and be productive. However, there is still an appreciation for going into the office occasionally.

'But predominantly from home on like what was...I probably didn't even have a setup at the time. I would've just been sitting at a dining table. Whereas now when I work remotely, it's pretty much all from home. I've got a proper home office set up now that I work from.' EMP01

'So I set up a home office, so I was sort set up before the lockdown occurred. Well, truly, yeah. But I'd already organised and I've done it for some years since, well, since I first had children, so needing to juggle the childcare and those sorts of things.' EMP09

'I'd probably be fully remote. As in internal meetings I like, like I go to the strategy days, which are in person or you know, things like that. But outside of that, yeah, like I, I'd happily be fully remote.' EMP13

"...as a single working mom, that's so good for me...so I work at least one day in the office every week and I might work a second day...So the four days at home and the one day in the office, maybe sometimes the second day in the office.' EMP14

Hybrid models of remote and in-office work offer a balanced approach that combines the benefits of both settings (Baumann & Marcum, 2023; Chafi et al., 2021). This flexibility allows individuals to leverage the collaborative advantages of face-to-face interactions and social interaction while enjoying the focus, quiet and autonomy of remote work, enhancing their innovative contributions.

'I don't think I've spoken to a single person that is dead set on coming into the office five days, or zero days. I think easily 90% of the people that I speak to, within my team and my friend and family, just colleagues and peers more generally, but all very heavily weighted towards some sort of hybrid model. Whether that's one day in the office, two days, three or four. Very rarely do you get someone that feels five days, either way.'

EMP03

'And even so now, you look at the numbers we talked about when we first started this call, 50% of the office continuously working from home is...Or working remotely rather is...And seeing no tangible impact to the bottom line, I think that's the answer there.' EMP06

'Remote working covid has been a godsend for me. I'll never work full-time in an office again.' EMP14

'I'm contracted four days a week with them. Um, and I've generally been doing two in the office and two at home...' EMP15

'It's probably maximum two days in the office. On average it will probably change over coming months. But my preference is to stick to two days in the office, three days at home.' EMP16

The diverse preferences for remote working arrangements underscore the importance of flexibility in fostering innovative behaviours. By accommodating individual work preferences, organisations can create environments that empower employees to thrive creatively, whether they prefer remote, in-office, or hybrid work setups.

8.7.1.2. High focus work

Participants in remote work settings employ various strategies and seek out specific spaces conducive to high concentration and focus work, recognising the impact of their environment on their ability to engage in innovative behaviours. Many individuals prefer isolating themselves in quiet environments when tackling tasks that require deep thinking and problem-solving. For them, the tranquillity of their home environment provides the ideal setting for uninterrupted focus (Amir et al., 2021; Henke et al., 2022), allowing them to delve into complex tasks without distractions more often found in office settings (Adekoya et al., 2022; Eerde et al., 2022).

'I like isolating myself, when I really have to kind of apply my brain.' EMP03

'I think it's just easier at home. I've got a dual screen set up. Everything's set up, ready to go. It's quiet...I like my space...' EMP07

"...so yes, I will a hundred percent do my deep thinking work at home because there's no way on God's green earth it's happening in the office...The added energy to try and screen out the bright lights and the constant loud noise and the people wanting to chat about the football...when I go to office, I literally don't get any work done.'

EMP11

In contrast, the office environment, with its open-plan layout and constant foot traffic, can be challenging to maintain high concentration levels. Employees find themselves drawn to quieter spaces within the office, such as meeting rooms or designated quiet areas, to escape disruptions and immerse themselves in deep thinking or problemsolving work. However, even these measures may not always suffice, as the presence of colleagues and the buzz of the office can still prove distracting (Haynes et al., 2017; Methot et al., 2021; Öhrn et al., 2021).

'But then the flip side of that is, it's also quite distracting when I'm in the... because of all the people walking through the door and the constant disruption. So really, I think it's outside that office environment.' EMP09

'so most of the time I'd put headphones in, you know, occasionally block out a meeting room to try and get some space away...Definitely at home because I, I get very distracted with this like, open office space does not work for me well...At the hospital I have an office where I can like shut the door and Yeah. So I do, I will go there if I've got sort of things I wanna focus on. But yeah, home, home.' EMP12

'Yeah, so when I was working from an office, I would always book a meeting room somewhere a bit off. There used to be one that was downstairs and away from everyone or at one place I've worked in previously that had something called a quiet room that you could go to and not be disturbed...I wanna be away from people away from the buzz of an open plan office.' EMP17 Away from the office, individuals value the ability to create a private, distractionfree workspace where they can focus without interruptions. Whether it's having a dedicated home office or finding solace in a quiet room, individuals prioritise environments that allow them to concentrate deeply and engage in innovative thinking. The absence of external disturbances and the ability to step away from the computer when needed further enhance their capacity for high-focus work.

'I'll often just pack up and go for a walk depending on the weather. Yeah. and, and moving and thinking for me is, is kind of combined. So I, I feel like my brain shifts gear a little bit.' EMP09

`...there were times where I found it was good to be home to have kind of an uninterrupted thinking base. I think part of it is even being able to be away from the computer.' EMP15

'So yeah, definitely that sort of thing is, is better from home. And I'm really lucky, as I said, I've got, my office is set up here so I can shut the door.

There's no distractions, I don't have kids.' EMP16

Irrespective of work location, employees adopt various strategies and seek out specific environments conducive to deep concentration and focus, recognising the profound impact of their surroundings on their ability to engage in innovative behaviours.

8.7.1.3. Work environments

Employee preferences for work environments significantly influence their productivity, energy levels, and interpersonal dynamics, ultimately impacting their ability to engage in innovative behaviours. While remote work offers benefits such as quiet and less distractions, some individuals miss the spontaneous interactions and informal conversations that occur in traditional office settings. Water cooler conversations and face-to-face interactions provide opportunities for relationship-building and camaraderie, which can be challenging to replicate virtually. For certain tasks, being in the office facilitates quick access to colleagues for assistance or clarification, streamlining processes and decision-making.

'And just you being able to sort of turn your head and ask the person next to you and answer to a question.' EMP02

'I think some things are a little bit easier to do when you're in the office. So sometimes, you know, you just need to check in a process with someone. It's easier to just to go to their desk and, and say "How do I do this?" Or "What do I do for that?"' EMP17

Equally, the office serves as a vital source of energy and inspiration. The presence of colleagues fosters a dynamic atmosphere which can be conducive to collaboration and idea generation. Physical proximity can also increase collaboration (Miranda & Claudel, 2021; Wohlers & Hertel, 2016). The physical workspace also allows for socialising and networking, especially with new team members, while maintaining existing working relationships. These interactions not only facilitate relationship-building but also encourage the exchange of diverse perspectives (Chantziaras et al., 2021; Suortti & Sivunen, 2023), which is essential for IWB.

'There's still situations where especially the younger people really missed out by not having those sort of water cooler conversations you get in the office. I can only imagine how hard it would be starting a new job where you couldn't actually meet people. I mean, virtually meet them that's okay. But meet them in the flesh and just get a feel for the people.' EMP02

'I need people. I'm a HR person, and so managing people is what I do. And it's very difficult to do that, although it can be done from home. But I just prefer...I get the energy from the people around me. So even when we have a choice, my preference is to be here. Unless we're in lockdown and I have to, my preference is still to work from the office.' EMP07

'There's a lot of companies that are very successful in not having face-to-face time. But I think I just like to have that as the basis of, you know, the, the working relationship. It certainly, it still works. I still have a really good working relationship with anybody who's joined that I haven't met face-to-face, but it's just nicer.' EMP10 'That's been one of the benefits of going back into the office. For me there's lots of people who are new since I left the department and came back. So getting to know who they are...' EMP16

Moreover, the shift to remote work has prompted individuals to reconsider their management styles and communication methods. Managers must navigate the balance between respecting employees' boundaries and fostering open communication. While remote work offers autonomy, it may also lead to a sense of isolation or a need to seek permission before interrupting colleagues, potentially hindering the flow of ideas and collaboration.

'...it's a really interesting thing if I reflect on lockdown and what a lot of people worried about with their teams is, "Oh, how do I manage their performance?" And I just think, goodness sakes, then you're not a very good manager, are you? I don't need to have people with me or I don't need to be eyeballing someone...what's interesting with working from home is I feel I have to ask permission more for interrupting somebody. You feel that, "Oh, am I imposing on your time?" Whereas yeah, I have felt that when I'm in the office, we just go in and out of each other's offices and say, "Oh, have you got a quick minute? Or why don't we run for a coffee and have a chat about X, Y. Z?' EMP09

Employee preferences for work environments vary based on individual needs, tasks and job requirements. Striking a balance between these preferences is essential for organisations seeking to cultivate a culture of innovation and collaboration, regardless of the chosen work environment.

8.7.1.4. Collaborating in different contexts

Collaborating in different work environments, whether remote or in-office, presents unique dynamics that can influence an employee's ability to engage in innovative behaviours.

Some participants find that being in the office and having face-to-face interaction facilitates spontaneous idea generation and collaboration, as they benefit from bouncing ideas off colleagues and engaging in face-to-face discussions. Collaboration is a multifaceted process that involves individuals working together towards a common goal through direct interpersonal interactions. In office environments, where interactions occur in person, nonverbal cues and engaging in group brainstorming sessions can foster IWB (Singh et al., 2022; Yagolkovskiy, 2013).

'If it's like sort of I guess more creative stuff where you're coming up with new ideas, I'd maybe stay in the office...I do benefit from bouncing ideas off people in those kind of situations rather than just coming up with something on my own. So if it was something like that where the team had to work through a solution or where I had to come up with something as a member of a team and I wanted to bounce ideas around, I'd go in the office. There are other things that are easier when we're all in the same room because we can bounce ideas off each other more easily...' EMP01

"...but there's things that you miss out on and there's things that others miss out on because you are not in the office. The person who brings you the problem also joins in terms of trying to create the solution. So it sort of almost becomes a brainstorming session, just an informal brainstorming session rather than just sort of sitting in the corner quietly contemplating by yourself.' EMP02

'I think still in the office would've worked better for me. Yeah, I just work better with people around me. And normally, most of my ideas are because people have come up, "I've got this really good idea." That's where my creativity comes from. Or I see someone else doing something, "I could do that."' EMP07

In remote working environments, ensuring the correct tools for effective and seamless communication and information sharing can facilitate remote brainstorming sessions and idea exchanges (Oyekan et al., 2017; Shockley et al., 2021). Remote workers leverage virtual platforms for brainstorming and collaboration, utilising screen sharing and online meetings to facilitate idea generation and problem-solving.

'It was amazing [working remotely]. And just the way people were more confident to share their ideas and yeah, it was interesting. I actually really enjoyed that dynamic where people that didn't speak up before suddenly did and had a lot to share because there were different ways for them to communicate, particularly if you're introverted,

that you had different tools to use to speak up about things. Where in meetings, you have to be an extrovert to speak up...I think workshopping and getting people on board with your ideas [is harder when working remotely]. And it might be where I'm working, Page 242 I'm working in higher education in a university, people with old ways of thinking where they prefer that being in together and thrashing out ideas.' EMP09"

'Because we can actually share screens so effectively where it's actually better sometimes to, you know, if you, you are showing how it's all working. You can share the screen better. People can, you know, put it onto multi screens and work their way through it. If you're in a, an office environment, a meeting room for example, and you're putting it up on a screen, it's often harder. So that's, that's actually better being able to just let people look at it in their own way. Whether it's clients or colleagues, it's, it's, it's actually an advantage to have people just sitting at their own laptops and viewing it in their own way.' EMP10

'I send a quick message and say, do you have five minutes? And you know, we jump on and I share a screen and so it's not much different. I think the, in the office stuff is more off the cuff, so it's less about the deliberate and planned work, but it's more about, you know, hearing conversations, maybe listening into something so that you've got an idea of what's going on, that sort of thing.' EMP16

'So my boss and I might catch up on teams, what we do for an hour a week. We'll talk about some stuff and I'll start triggering some things. I'll do some research and then I'll come up with the ideas and then by the next week I can bounce that back with her and do this is what, where I'm going with it, this is what I'm thinking.' EMP17

However, some respondents note challenges in capturing the spontaneity and dynamics of in-person interactions, as well as immediacy of feedback and the ability to read nonverbal cues (Singh et al., 2022), which can impact the quality of collaboration and creativity.

'And of course we could have done that online and I have done it online, but something felt different about being face-to-face and being able to have that rapid connection for that time...the engagement, the body language, I don't think as much as we try to adopt to the best kind of tech tools to support some of this planning, there's nothing better than being in a room with post-it notes and markers and being able to, you know, mix and communicate and sort of edit people's stuff on the fly...' EMP12" 'So I definitely will arrange face-to-face stuff on campus with like the key stakeholders...I feel like I can't read the nuances of what's going on in the room...Cause I just like can't see the nuances of like people's behaviour or how they're receiving an idea [when remotely working].' EMP13

Effective collaboration transcends physical boundaries, as individuals find ways to connect, communicate, and collaborate regardless of their work environment. By leveraging the strengths of both remote and in-office settings, and ensuring effective tools and technology, employees can engage in idea generation and IWB.

8.7.1.5. In person communications

The perceived importance of in-person (face-to-face) communication in fostering innovative behaviours cannot be overstated, as highlighted by respondents' insights.

'I think now I've noticed that there are things that take a lot longer than they used to because the face time made it easier to sort of teach people and also to interact with the clients.' EMP01

'Yeah, that's pretty important. I had an advisory board meeting yesterday, I think we got a better outcome because it was face to face than online.' EMP08

'If you're in meetings personally and you had the detractors in the room, you would've been able to kind of move them and change 'em a little bit.' EMP12

The nuanced understanding gained from face-to-face interactions, where subtle cues and energies are more easily discerned compared to virtual encounters (Sidi et al., 2021; Thunberg & Arnell, 2021). This ability to read non-verbal cues contributes to a deeper understanding of colleagues' perspectives, which is needed for effective collaboration and problem solving. Nonverbal cues, such as facial expressions and body language which are considered to convey the majority of meaning when communicating feelings, intentions and attitudes, provide valuable information that is often lost in purely digital interactions or text based digital communication (Chretien & Kind, 2013; Erle et al., 2022). While virtual platforms offer opportunities for collaboration, face-to-face interactions are easier for gauging reactions during collaborative endeavours.

"...but you could read a person's energy a lot easier face-to-face than you can on the camera." EMP06

'So you can't really tell what people are thinking, but in a room I can tell what they're thinking without them having to say as much. And so yeah, it's, I don't know that any is better, but it's, it's good to have both options. It's good to be able to go, this is you know, explain. I think the online version helps me explain the concept more thoroughly and the in person is helping me gauge reaction better.' EMP17

All aspects I think that body language thing...body language thing that we talked about before is pretty important in the overall communication field. I've got expressive

eyes and I can't hide. With my speech I can sort of mask how I'm really feeling, but apparently the eyes give it away so that's not happening on Teams. You can't pick that up. So I think people are probably almost a bit more polite. And because when two people start talking at the same time, it becomes a bit awkward, whereas in a room they just, the body language and everything else.' EMP02

'You could then deal with some of the emotion, some of the tensions that the general manager was going through. And you could read that, you could experience it a little bit more. And then, the general manager could say something on the side, without other people hearing. And whereas with an online meeting, everyone's there, you've then got to, say, make the request, "Oh, Craig, can you stay on? I want to have another chat with you." Everyone knows you're having another chat, versus something just happens on the side...Because you can have more conversations, you can explain things, you can see people's reactions.' EMP08

Employees found the absence of face-to-face interactions in remote working settings diminishes the spontaneity of idea generation and reduces opportunities for inclusive discussions. The lack of casual exchanges and impromptu brainstorming sessions hinders the serendipitous flow of ideas that often occurs in physical work environments. 'Working as a team is easier when you actually are kind of together as a team. I think that's the main challenge. Obviously in audit, we work with a lot of young

staff that come in, graduate intakes that need a lot of training and need to ask a lot of questions. And I think the one thing that I've noticed as someone who's meant to manage these people is that it's harder to get that upwards feedback or for them to ask questions because they can't just tap you on the shoulder...there are other things

that are easier when we're all in the same room because we can bounce ideas off each other more easily.' EMP01

'There's still situations where especially the younger people really missed out by not having those sort of water cooler conversations you get in the office. I can only imagine how hard it would be starting a new job where you couldn't actually meet people. I mean, virtually meet them that's okay. But meet them in the flesh and just get a feel for the people.' EMP02

'Cause that was the first thing I noticed I missed working at home was, a whiteboard. Like just that creative piece. And so yeah, I still try and encourage that with the types of meetings with my team and my direct reports...let me know if you wanna just book a book the time and say whiteboard sesh.. And so I know that we're just gonna have a brainstorm creative time.' EMP14

8.7.2. Summary of employee experience between remote and office-based working

Different work environments, such as remote, hybrid, and in-office setups, can influence employees' ability to engage in IWB. Those predominantly working from home create dedicated workspaces conducive to thinking, focus and innovation. Employees adopt strategies, whether at home or in-office for deep concentration and problemsolving, preferring quiet environments for focused work.

However, while remote settings allow for uninterrupted focus, some miss spontaneous interactions of office settings. In-person interactions can facilitate spontaneous idea generation, as well as foster energy, collaboration, and relationshipbuilding. Digital tools available in remote working environments enable effective communication and brainstorming, however virtual interactions can present challenges in capturing the energy of in-person meetings and reading nonverbal cues. Nonverbal cues enhance communication and relationship-building, which may be lost in virtual interactions.

Hybrid models foster IWB by offering a balance between the benefits of office environments, such as collaboration and informal discussions, and the benefits of remote working, such as fewer disruptions and space to think. Effective collaboration to generate new ideas can transcend physical boundaries, and organisations can strike a balance between remote and in-person interactions to cultivate IWB.

8.8. Summary of qualitative study findings

The findings show that remote working has a multitude of factors that both encourage and inhibit IWB, contrasting with lived experiences of remote working compared to in-office working. On the positive side, providing tangible resources such as necessary equipment and materials lays a strong foundation for creativity. Financial support for remote work and access to appropriate tools empower employees to explore new ideas efficiently. Regular virtual or in-person meetings facilitate collaboration, while access to quiet spaces allows for uninterrupted creative thinking. Autonomy, flexibility, and a balanced work-life dynamic further empower individuals to innovate. Additionally, personalised work environments and mental health support ensure employees feel motivated. Organisational encouragement fosters a culture of innovation.

However, several challenges hinder IWB in remote settings. The lack of face-toface interactions can lead to misunderstandings, while disparities in technology skills create barriers to collaboration. Feelings of isolation and burnout can stifle creativity, as can blurred work-life boundaries and constant distractions at home. Inadequate home workspaces and a diminished social aspect make it harder for individuals to stay engaged. Formalised communication channels detract from time for innovative pursuits, while a focus on outcomes over quality inhibits the exploration of novel solutions. Different work environments, such as remote, hybrid, and in-office setups, influence employees' ability to engage in innovative thinking. The thesis advocates for the promotion of hybrid models as a strategic approach to fostering IWB by leveraging the advantages inherent in both traditional office settings and remote work environments (Baumann & Marcum, 2023; Chafi et al., 2021). By integrating elements of both work environments, hybrid approaches offer a unique opportunity to transcend physical limitations and optimise conditions for innovation. This approach aligns with the findings of both the quantitative and qualitative studies that found that increased, but not exclusive, remote working fosters IWB.

Hybrid models, by their very nature, enable organisations to harness the collaborative energy and spontaneous interactions often associated with in-person office environments while simultaneously providing the flexibility and autonomy characteristic of remote work setups. This blend of benefits not only acknowledges the diverse needs and preferences of employees but also maximises the potential for creative thinking and problem-solving.

However, the successful implementation of hybrid models requires a delicate balance. Organisations must navigate the intricacies of remote and in-person interactions, ensuring that each mode of engagement is effectively utilised to support and enhance IWB. This entails thoughtful consideration of factors such as communication channels, meeting structures, and collaboration tools to create an environment conducive to innovation regardless of physical location.

8.9. Summary of findings by research question

Four sub-research questions were developed to guide the study. A short summary of the findings for each research question is provided below.

SQ1: What is the extent of IWB among employees engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?

The key findings for this question are as follows. First, employees who work remotely two to four times per week exhibit the highest levels of IWB. Therefore, increased remote work leads to higher IWB, as long as it is not the sole mode of working. Employees who work remotely exclusively may experience diminishing returns in their IWB. Second, although the findings were not statistically significant, an increase in remote work frequency corresponded to a rise in IWB particularly during the idea generation and idea implementation phases. During the idea generation phase, employees require time and space to engage in creative thinking. Remote work offers employees the autonomy and flexibility to engage in creative thinking without the constraints of a traditional office setting. This environment allows for deep thinking, reflection, and problem-solving, fostering the creation of new ideas. Additionally, remote work often involves fewer interruptions and distractions, enabling employees to focus more effectively on generating innovative solutions.

During the idea implementation phase, remote work enables greater control over one's work environment and schedule, which can enhance productivity and efficiency in executing plans. Employees can organise their work to suit their preferences and working styles, minimising disruptions. The predominant use of digital collaboration tools and communication platforms in remote settings also facilitates seamless coordination and collaboration among team members, streamlining the implementation process.

SQ2: How do demographics influence the outcomes of employees' IWB engaged in remote work, both overall and across each phase (idea exploration, idea generation, idea championing, and idea implementation)?

The key findings for this question are as follows. In the context of demographic factors, several key variables emerged from the findings as influential in shaping IWB among remote workers. Full-time employment status, longer tenure within an organisation, and managers and senior leaders had increased levels of IWB overall. Conversely, gender showed no association with IWB, either overall or within any of the four phases.

The findings also revealed the following profile of employees who exhibited the most IWB when considering all demographic variables together: 'Always' undertaking IWB activities were the 18–30-year-old age group, female, contract/casual, less than 2 years tenure, managers, working remotely 1–3 times per month. The categories that showed a consistently high prevalence of undertaking IWB (i.e. 'Often'), were the 41–50-year-old age group, male, full-time, with 6+ years tenure, managers, working remotely 2–4 times per week.

When examining each phase individually, the findings showed that employment status was the only demographic variable which showed an influence across all four phases. Full-time employment status was associated with higher IWB across each phase. Longer tenure was found to increase IWB specifically during the idea championing phase, whereas higher age was associated with decreased IWB in this same phase. Additionally, higher levels of employment seniority were linked to increased IWB during the idea implementation phase.

Companies aiming to foster IWB in remote work settings should expand their focus beyond work arrangements. The findings highlight the importance of recognising how different organisational roles and responsibilities influence such behaviour. These findings underscore the need for inclusive strategies that enable all employees, regardless of employment type, tenure, age or position, to be able to actively engage in the innovation process.

SQ3: What factors contribute to fostering or inhibiting innovative work behaviours in remote working environments?

The key findings for this question are as follows. First, perceptions of employees engaged in remote work presented a number of key factors that foster their IWB. Providing essential resources like equipment and materials formed a strong foundation for creativity. Financial support for remote work, along with access to necessary tools and technology, empowers efficient exploration of new ideas. Regular meetings to facilitate collaboration and idea exchange, while access to quiet spaces supports their uninterrupted creative thinking. Private environments and minimised distractions enhances deep focus and innovation, supported by dedicated time for brainstorming. Social connections and celebrations cultivate camaraderie and motivation, fostering a positive atmosphere where ideas can be discussed and exchanged. Autonomy, flexibility, and work-life balance encourage experimentation and innovation. Personalised work environments and well-being initiatives ensured employee support and motivation. Further, organisational encouragement for idea exchange fosters a culture of continuous improvement in remote work settings.

Second, the findings showed that there were factors perceived by employees as impeding their IWB in remote environments. Reduced face-to-face interactions limited spontaneous exchanges and could lead to misunderstandings in digital communications, hindering idea flow. Technology skill disparities and feelings of isolation from colleagues can impede collaboration and creativity. Burnout, blurred work-life boundaries, and distractions at home diminished motivation and creative energy. Inadequate home workspaces and reduced social interaction further challenged engagement and inspiration. Formal communication channels and increased meeting coordination detract from time available for innovative pursuits. Feelings of invisibility and challenges in proving one's value can dampen enthusiasm for contributing ideas. Lastly, a focus on outcomes over quality was thought to prioritise quantity, limiting exploration of innovative solutions.

Therefore, while employees perceive that remote work supports IWB by providing autonomy, freedom, quiet, and time needed for ideation and innovation, the findings also highlight that isolation, overworking, reduced opportunities for spontaneous collaboration, reduced face-to-face interactions, and challenging interpersonal connections can hinder their IWB. Consequently, organisations need to ensure remote working employees sustain regular communication and interpersonal connectivity to maintain relationships and meaningful connections, despite the physical distance. The integration of technological infrastructure and collaborative tools assumes paramount importance to perpetuate cohesive teamwork and foster IWB. Implementing practices aimed at fostering interactivity and sharing of ideas, among all employees, irrespective of their work location is important.

SQ4: What are the distinctions in employee experience regarding innovative work behaviour when comparing remote and office-based working environments?

The key findings for this question are as follows. In today's rapidly evolving work landscape, there is a question surrounding the optimal work environment for encouraging innovative behaviours, particularly in light of the contrasting experiences of employees in office-based versus remote settings. The findings revealed that when employees work remotely, they cultivate dedicated spaces at home conducive to thinking and focus, while both remote and in-office employees prioritise quiet environments for deep concentration and problem-solving. However, the findings showed that remote settings, while offering uninterrupted focus, may lack the spontaneous interactions and energy of office environments that foster collaboration, and relationship-building. In-person interactions were valued for their ability to generate new ideas, facilitate spontaneous discussions and brainstorming, and enhance communication through nonverbal cues. These factors were identified by employees as crucial for building relationships. The findings found that when working remotely, employees use and valued digital tools to facilitate remote communication and brainstorming, but considered they do not fully capture the energy and nuances of face-to-face meetings.

8.10. Chapter summary

This chapter discussed the analysis, interpretation, and synthesis of the research findings to address the research objectives and provide valuable insights. This chapter represents the culmination of the study, integrating the results with existing knowledge and theories while highlighting their implications.

The chapter began with a restatement of the research questions and aims, serving as a guidepost for the subsequent discussion. It then presented a comprehensive overview of the key findings obtained from the data analysis from both of the studies conducted. The findings are systematically explored and contextualised within the broader academic discourse, allowing for a deep understanding of their implications. The next chapter provides a summary and conclusion of the thesis.

CHAPTER 9 - CONCLUSION

9.1. Introduction

The purpose of this chapter serves as the culmination of the research journey, bringing together the key findings, contributions to knowledge, limitations, and future areas of research. The previous chapter summarised the main research findings that emerged from the data analysis and interpretation. The conclusions drawn from these findings provide answers to the research questions and demonstrate the extent to which the research objectives have been achieved. The conclusion chapter provides an opportunity to reflect on how the study has expanded and enhanced existing knowledge and theoretical frameworks in the field.

The aim of this thesis was to examine the influence of remote working on employees' innovative work behaviour within the Australian context. The specific objectives of the thesis were to assess the extent to which remote workers engage in IWB, explore how demographic variables might influence IWB among remote employees, explore employee perceptions regarding the supportive elements and obstacles in engaging in IWB, and understand from the employees' viewpoint, the distinctions in their innovative behaviour when contrasting remote work with in-office work. This thesis draws several broad conclusions and implications, identifying key contributions.

9.2. Key contributions and implications

Extensive data was collected through questionnaires and in-depth interviews to explore the experiences and perceptions of employees, specifically focusing on how remote working influences their innovative behaviour. Based on the key findings, the thesis makes a significant contribution towards existing literature and knowledge in the field.

9.2.1. Remote working frequency and demographics influence on IWB

While there have been studies looking at IWB and remote working (Coun et al., 2021; Rachman et al., 2022), focusing on factors such as employee engagement and human resource management, there are no studies, to the best of the author's knowledge that explore specifically the impact of remote working frequency on IWB and the

influence of specific employee demographics. The findings contribute to the literature by identifying the optimal frequency of remote working that fosters IWB, and by revealing which demographic factors influence IWB and which do not. The findings revealed that frequent remote working leads to higher IWB. This is attributed to increased autonomy, flexibility, lack of disruptions, spaces for focus and thinking, all of which are conducive to fostering ideation (Theurer et al., 2018; Thi Minh Ly et al., 2023; Wicaksono & Pusparini, 2022; C. Yu et al., 2021). However, to foster IWB, employees should not work remotely exclusively. Working exclusively remotely can bring forth feelings of isolation, disconnection and communication challenges (Garlatti Costa et al., 2022; Golden et al., 2008; Oyekan et al., 2017; Singh et al., 2022) which need to be addressed in remote work environments. Further, when considering IWB overall, the findings revealed that employment status, tenure, and employment level significantly influenced IWB, whereas gender and age did not have a notable impact. This is significant because it allows for the exploration of tailored strategies and interventions. By considering diverse working preferences and demographics, organisations can develop initiatives that support all employees in exhibiting IWB.

9.2.2. Employee experience and perspective

The findings of the study extends the empirical research in the domain of employee IWB by focusing on the employee's experience and perspectives to understand how they engage in innovative activities outside traditional office settings. Employee driven innovation and the role of employees in the innovation process is underscored as a significant and often overlooked resource for driving innovation within organisations (De Jong & Den Hartog, 2007; Høyrup, 2010; Janssen, 2000) . By focusing on the employee perspective, the findings explore the subjective experiences and agency of employees in driving innovation (Weigt-Rohrbeck & Linneberg, 2019). It is the individual employee who will look for new ideas and determine how to implement them (Agarwal, 2014; Carmeli & Spreitzer, 2009; De Jong & Den Hartog, 2007; Kozlowski & Klein, 2000), as usually they are closer to the customer or the process than actual departments tasked with innovation such as research and development (R&D). Getz and Robinson (2003, p. 134) found that in practice '...80 per cent of improvement activities.

Existing literature on employee driven innovation discuss in-person collaboration, shared physical spaces, and face-to-face in person interactions as catalysts for creativity (Tønnessen, 2022; Yagolkovskiy, 2013; Yström & Agogué, 2020). However, the emergence of remote work challenges these assumptions and the findings of this study contributes to the literature by re-evaluating the factors that foster or inhibit innovation in all work settings.

Employees perceived that there are communication challenges when remote working which can impede their IWB. Lack of informal communication, rigidity of communication and employee capability with using various communication tools (Shockley et al., 2021; Sidi et al., 2021; Suortti & Sivunen, 2023) were expressed. Respondents also missing out on the spontaneous discussions and hallway conversations which can spark new ideas or collaboration (Oyekan et al., 2017; Singh et al., 2022; Waizenegger et al., 2020). Organisations need to provide regular opportunities for brainstorming and collaboration (Domingo et al., 2021), where employees from different departments and levels of the organisation can come together irrespective of where they are working, to generate new ideas and solutions, encouraging diverse perspectives and collaboration. Further consideration for the types and method of communication platforms. Remote working employees should have access to the necessary technology and resources to access information and perform their jobs effectively. This includes not only providing the digital communication tools, but providing support for technical issues and accommodating any accessibility needs.

Employees also perceived that when working from home they benefit from the quiet, solitude and lack of disruptions, which affords them the time and space to explore new ideas (McGee et al., 2023; Skountridaki, 2023; Thi Minh Ly et al., 2023). In the office, organisations should replicate the elements of the home work environment by providing spaces for deep thinking, reflection and seclusion.

The study also found that employees often felt disconnected from their colleagues and missed the informal social aspects of work, which can impact on their working relationships and reduce motivation for IWB (Viererbl et al., 2022). Creating a supportive environment where employees feel comfortable and connected leads to higher job satisfaction and employee engagement, which have been shown to positively influence IWB. A supportive workplace culture extends to work-life balance and autonomy. Autonomy as perceived by the employee was shown to be a key contributor to IWB. Employees expressed that working remotely often led to working longer hours and experienced a blur in boundaries between work and home. This can lead to burnout and less motivation to engage in creative pursuits. Allowing employees to choose when and how they work, and encouraging rest and clear delineation between work responsibilities and connectivity expectations, can mitigate some of these challenges and provide employees the freedom and time to explore new ideas.

9.2.3. Four phase model of IWB

The study's examination of each of the four phases of IWB – idea exploration, idea generation, idea championing, and idea implementation - individually represents a theoretical contribution to prevailing literature, which often argues that IWB is a unidimensional construct (Phil-Thingvad & Klausen, 2020; Rodrigues & Rebelo, 2022; Wicaksono & Pusparini, 2022). Scott and Bruce (1994) and Basu and Green (1997) state that employees could be engaged in various phases simultaneously or in any order and the process is not always sequential. Dorenbosch et al. (2005) pointed out that each phase builds on its predecessor and the phases are highly interrelated and connected by feedback loops. However, there are distinctions in each phase in terms of activities that are undertaken and behaviours that are demonstrated (De Jong & Den Hartog, 2010; Gkontelos et al., 2022; Kleysen & Street, 2001; Sari et al., 2021; Shahid et al., 2022; Soputan & Sumual, 2022), that warranted exploration particularly when looking to understand which demographic factors could have an influence. Shahid et al. (2022) emphasised the necessity of evaluating each phase of IWB, recognising each stage of innovation holds its own importance. By examining each phase of IWB separately, the findings of the study provide a more nuanced understanding of how different factors impact the innovation process at various stages, offering deeper insights that can inform both existing theory and practice.

The study found that employment status was the only demographic variable that influences IWB across all four phases. Full time employees typically benefit from increased job security, greater access to resources, increased working hours and autonomy which are factors which contribute to increased IWB (Kumar et al., 2023; Probst et al., 2019; Spiegelaere et al., 2014).

Increasing age was found to decrease IWB during the idea championing phase only. The decreased prospect of older employees participating in the idea championing phase contradicts an assumption that, given their extensive experience and influence, they would be more inclined to support new ideas. The wealth of accumulated knowledge and insights empowers older employees to leverage past experiences and lessons learned, enabling them to construct persuasive arguments and compelling narratives to promote new ideas. Nonetheless, this might indicate a hesitation among older employees to embrace risks associated with advocating for change (Albert & Duffy, 2012; Banks et al., 2020). Younger employees, typically earlier in their careers and less risk-averse, are often more proactive in championing innovative ideas to distinguish themselves and advance professionally. Conversely, older employees, nearing retirement or with established career paths, tend to prioritise stability over innovation, showing reluctance in championing new ideas, particularly in different work contexts such as working remotely. Moreover, technological proficiency differences contribute to this reluctance (Hecker et al., 2021; Raymundo & Castro, 2019). Younger employees, more adept with digital tools and social media, find it easier to promote new ideas when communicating virtually (Mehra & Nickerson, 2019; Stewart et al., 2017). In contrast, older employees may struggle with new communication technologies, hindering their ability to engage in virtual advocacy.

Longer tenure was found to increase IWB during the idea championing phase only. Employees with longer tenure tend to possess a profound understanding of the organisation's culture, goals, and decision-making processes, enabling them to advocate for new ideas confidently and navigate the organisation remotely (Ng & Feldman, 2010; Pei et al., 2018). Their familiarity with the organisational landscape aids in identifying optimal channels for introducing innovations and managing resistance to change in a virtual setting. Moreover, their experience equips them with insights into past initiatives and potential barriers to innovation, enhancing their effectiveness in promoting new ideas (Clercq & Pereira, 2019; Ng & Feldman, 2013). Longer-tenured employees also benefit from established organisational and external networks, which are crucial for championing innovations, especially in remote settings where physical visibility is limited (Björk & Magnusson, 2009; Liu et al., 2016; Ng & Feldman, 2010). Their deep understanding of organisational dynamics, coupled with strong networks, facilitates navigating the politics of promoting new ideas and garnering support from key stakeholders. Longer tenure can foster a stronger sense of ownership and investment in the organisation's success, motivating employees to advocate for ideas that contribute to its growth (Hur & Perry, 2019). Their proven track record of reliability and performance enhances their credibility and trustworthiness among colleagues and superiors, which is particularly vital in remote work environments where trust is essential for enhancing performance and fostering creativity (Rodrigues & Veloso, 2013; Taştan & Davoudi, 2015; C. Yu et al., 2021).

This finding was particularly interesting because, while increased age was associated with lower IWB, increased tenure correlated with higher IWB, which is unexpected given that we would typically assume increased tenure to coincide with increased age. This difference could be due to several factors. Older employees who are resistant to change may not leverage their knowledge of the organisation's culture, processes, and key stakeholders to champion new ideas as effectively as longer-tenured employees. Older employees might not utilise their internal networks to rally support for new ideas as enthusiastically due to a preference for maintaining existing relationships and routines. Championing ideas often involves taking risks and stepping outside comfort zones. Older employees might be more risk-averse, focusing on minimising disruptions, whereas tenured employees, regardless of age, may feel more secure in their positions and therefore more willing to take risks. Also, older employees might face implicit agerelated biases that question their adaptability and enthusiasm for new ideas.

Increased employment level showed an influence during the idea implementation phase only. Managers and senior leaders play pivotal roles in overseeing and facilitating the implementation of ideas within organisations (Birken et al., 2012; Engle et al., 2017). They are accountable for project success and ensuring effective idea implementation to meet organisational goals. Senior leaders provide support and resources necessary for idea implementation, enhancing employee access to resources and fostering innovation (Bedenik et al., 2024; Škerlavaj et al., 2014). Their authority allows them to mobilise resources, allocate budgets, and prioritise initiatives, facilitating idea implementation and aligning with strategic objectives. Moreover, their decision-making authority accelerates the execution of ideas and removes obstacles (Chenger & Woiceshyn, 2020; Shaed et al., 2018), particularly advantageous in remote work settings where swift and autonomous decision-making is crucial. In contrast, non-managerial employees may have limited autonomy and influence in executing innovative ideas independently. Leadership roles afford senior employees greater agency, resources, and influence in driving and shaping implementation strategies, even in remote work settings.

9.2.4. Ability, Motivation, Opportunity (AMO) framework

The study contributes to the Ability, Motivation, Opportunity (AMO) framework literature by exploring how employees' ability, motivation, and opportunity to participate in IWB activities are driven in remote settings as well as how it compares between remote and office settings. The AMO framework has played a crucial role in examining the factors that influence innovation within organisations. Researchers have gained valuable insights into how HR practices, high-performance work systems (HPWS), and organisational strategies can affect innovation performance (Alkhalaf & Al - Tabbaa, 2023; AlMunthiri et al., 2023; Bhatti et al., 2020; Shahzad et al., 2019). The theoretical contribution of extending the AMO literature through the lens of remote working lies in exploring how remote work environments influence these three dimensions. It adds to existing knowledge of how remote work impacts employees' abilities (skills, knowledge), motivations (intrinsic and extrinsic factors driving behaviour), and opportunities (environmental factors enabling or constraining behaviour). This perspective helps to refine AMO theory in the context of remote work, offering insights into how organisations can optimise these factors to enhance performance and innovation in dispersed work settings.

The analysis of employees' 'ability' to engage in IWB while working remotely revealed that providing tangible resources, such as additional screens and communication platforms, significantly enhanced employee's technical capabilities. These resources facilitated IWB by offering the necessary infrastructure for idea generation, collaboration, and problem-solving in virtual environments. Employees effectively adapted to remote work setups and digital platforms. Moreover, the ability to innovate is tied to individuals' skills, knowledge, and resources. Participants highlighted their proficiency with digital tools, cognitive skills like problem-solving and critical thinking, and the ability to leverage technology for virtual collaboration.

The analysis of employees' 'motivation' to engage in IWB while working remotely revealed that flexibility is a key factor, positively impacting work-life balance, job satisfaction, and personal fulfillment. This balance drives employees to tackle tasks with renewed energy, enthusiasm, and creativity, enhancing IWB. However, despite the benefits of remote work, many employees missed the camaraderie of office life, experiencing feelings of isolation and detachment. To address this, organisations should prioritise fostering social connections and providing opportunities for meaningful engagement.

The analysis of employees' 'opportunity' to engage in IWB while working remotely revealed that creating a conducive environment for collaboration, interaction, relationship-building, and knowledge sharing was crucial. Opportunities such as innovation forums, newsletters, and conferences enabled idea exchange and feedback. However, digital platforms can hinder spontaneous and informal communication, reducing opportunities for impromptu discussions that drive collaborative brainstorming. Organisations must address the need for impromptu connections to encourage IWB. Additionally, organisational support enhances employee well-being, fostering a sense of belonging and camaraderie. This supportive environment empowers employees to share ideas and work together towards common goals, ultimately promoting IWB.

9.3. Conclusions from research findings

The thesis explores the relationship between IWB and remote work among Australian employees, acknowledging innovation as a pivotal strategy for organisational competitiveness and recognising that innovative ideas often stem from employees. The research problem arises from the contemporary landscape, characterised by the rise of remote working, and accentuated by the COVID-19 pandemic, necessitating an exploration of how remote work may impact an employee's ability to engage in innovative practices.

The pandemic provided a unique opportunity to explore the current and pertinent issue of remote work, which is increasingly recognised as a lasting feature of evolving workplace dynamics (Costa et al., 2022; Smite et al., 2023; B. Wang et al., 2021). Further, given the pivotal role of creativity and innovation in the enduring viability of organisations, managers must navigate the balance among accommodating employees' aspirations for flexible working arrangements, incorporating remote work, and sustaining the momentum of innovation to uphold organisational competitiveness (Thevanes & Harikaran, 2020; Wong et al., 2020).

The findings highlight that remote working serves as a double-edged sword for IWB, where it can both foster and impede in various ways. The thesis suggests finding a balance by embracing hybrid work models, integrating remote working with periodic inperson interactions (Baker, 2020; Degerli, 2023; Rauv, 2022; Telstra, 2021). These faceto-face encounters are essential for IWB, as they allow for spontaneous interactions, collaboration and relationship building. In a 2022 study of Australian knowledge workers, 54% reported adopting a hybrid work model, with 56% expressing that this arrangement was ideal for them (Hopkins & Bardoel, 2023). Hybrid working has gained significant attention due to its potential benefits such as optimisation of labour costs, increased organisational commitment, and enhanced flexibility (Beno, 2021; Peprah, 2023). Furthermore, the positive influence of work flexibility, work-life balance, and team building has been highlighted as benefits of the hybrid work model (Sampat et al., 2022).

When accounting for demographic factors and IWB, it becomes imperative for organisations to actively foster inclusivity and equity. The literature supports the notion that employees, regardless of their position or education, possess the potential to contribute to innovative endeavours through their experience-based knowledge and active participation in innovation processes (Kristiansen & Bloch-Poulsen, 2010; Tan & Sim, 2022). However factors such as education level, experience, and managerial position can impact individuals' attitudes towards innovation and their subsequent adoption of innovative practices in the workplace (Farmer et al., 2003; Jafari & Zarghami, 2017; Wipulanusat et al., 2021). Additional factors such as technology, job security, networks, time, authority and access to resources can influence employees' potential to engage in IWB. The thesis suggests organisations actively pursue measures to ensure that every member of their workforce, regardless of their position, employment status, or tenure, has an equal opportunity, capability and motivation to demonstrate IWB.

9.4. Limitations of the study

Doctoral research projects, like all research endeavours, are subject to limitations (Brutus et al., 2013; Hwang et al., 2015; Marshall & Rossman, 1999; Price & Murnan, 2004).

The COVID-19 pandemic introduced significant challenges in data collection for the study. Initially designed with some in-person interactions, the scope had to be broadened and transitioned fully online, necessitating new methods for recruiting participants for both the questionnaire and interviews. Traditional recruitment channels were disrupted, requiring adaptation and flexibility to reach a diverse and representative sample. However, the pandemic also presented a unique and valuable opportunity. With more people working remotely, the study's relevance heightened as it addressed timely and critical issues for both employers and employees. This shift allowed the research to capture real-time data on remote work experiences, making the findings particularly pertinent and impactful in understanding the evolving dynamics of the workplace during a global crisis.

The sampling drew on the views of 18 Australian-based employees in the interviews and 324 Australian-based employees in the survey. Therefore, the results may not be generalisable beyond Australia. Replicating the study in other countries would improve the generalisability of the results (Cheng et al., 2016; Franklin & Thomas, 2022; Schmidt, 2009). By replicating studies in various countries, researchers can assess the extent to which the results hold true across different cultural, social, and economic contexts. Nevertheless, the findings could be generalised to other countries with similar labour force characteristics, such as New Zealand, Canada and the United Kingdom. Australia and New Zealand share historical, cultural, and economic ties, resulting in similar labour force characteristics. Likewise, Australia and the United Kingdom and Canada have comparable labour market structures, where each possess diverse industries, a skilled workforce, and high labour force participation rates.

The level of IWB reported in the questionnaire is based on the judgments of the employees themselves. Therefore, these self-reported levels may be subject to bias (Chan, 2010; Kimberlin & Winterstein, 2008). However the adage 'perception is reality' (Fernandez & Pitts, 2011, p. 211) supports the objective of this thesis to explore employees' perceptions and their lived experiences. Behaviour is a function of individuals' interpretation (perception) of reality rather than reality itself (Cantor et al., 2013). The research objectives were to gain insight from the employees' standpoint. Hence, relying on self-reported data suffices as it captures their subjective reality and perspectives that we aim to uncover.

9.5. Future areas of research

Several avenues for future research stem from this thesis. First, future research in IWB and remote working should focus on addressing the intricate dynamics between technology and organisational culture. As technology continues to advance, understanding how emerging tools and platforms influence IWB is essential (Parker et al., 2017; Wen et al., 2021). Research could explore how artificial intelligence, virtual reality, and collaborative software impact creativity, problem-solving, and knowledge sharing among remote employees. Additionally, investigating the role of organisational culture in fostering innovation within virtual teams is crucial. Organisations with a supportive culture that encourages experimentation, rewards risk-taking, and values diverse perspectives are more likely to foster IWB among remote employees. Identifying the specific elements that promote innovation in a remote context can provide valuable insights into employee innovation in distributed work environments.

Second, future research could delve into the psychological aspects of remote work and how they influence IWB. Remote working can lead to feelings of isolation, burnout, and reduced motivation (Bakarich et al., 2022; Hayes et al., 2021), all of which can negatively impact an employee's ability to engage in innovative behaviours. Understanding the psychological factors that contribute to or hinder innovation in remote settings is vital for designing interventions and support systems that nurture creativity and innovative thinking among employees. Exploring evidence-based strategies to maintain a sense of belonging and social connection within virtual teams, as well as investigating the impact of flexible work schedules on creative ideation and problem-solving, can provide valuable insights into optimising remote work environments for innovation.

Finally, it is recommended that future research be conducted on the ethical implications of remote work and IWB. As remote work and hybrid working becomes more prevalent, ethical considerations related to data privacy, surveillance, and intellectual property rights become increasingly significant. Researchers should examine how these ethical concerns intersect with IWB in remote settings. Understanding the ethical challenges faced by remote employees and organisations can inform the development of policies and guidelines that protect both employees' rights, privacy and the organisation's intellectual assets, ensuring a balance between innovation and ethical responsibility in the evolving landscape of work.

SUMMARY

In conclusion, the thesis has investigated and addressed key research questions in the field of innovative work behaviour and remote working, making a significant contribution to the existing body of knowledge and offering valuable insights. Through rigorous collection of both quantitative and qualitative data, analysis, and interpretation, this study has shed light on an important phenomenon and their implications. The findings and conclusions of this research have not only expanded IWB theory and understanding of the topic but also provided practical implications for various stakeholders. While this study has made significant contributions, it is important to acknowledge its limitations and recognise the potential for further research. Ultimately, it is the hope of the researcher that this thesis will inspire and inform further investigations, fostering progress and advancements in the understanding of employee innovative behaviours regardless of where, when, and how we work.

REFERENCES

- Abhari, K., Pesavento, M., & Williams, D. (2023). Supporting participatory innovation during the COVID-19 pandemic: a comparative study of enterprise social media use. *Information Technology & People*, 36(5), 2030-2062.
- ABS. (2020). *Household impacts of COVID-19 survey*. Australian Bureau of Statistics. <u>https://www.abs.gov.au/statistics/people/people-and-communities/household-impacts-covid-19-survey/sep-2020</u>.
- ABS. (2021). Characteristics of Employment, Australia. Australian Bureau of Statistics. Retrieved July 2022 from <u>https://www.abs.gov.au/statistics/labour/earnings-and-</u>working-conditions/characteristics-employment-australia/aug-2021
- ABS. (2022a). 2021 Census: 2.5 million people working from home on Census day https://www.abs.gov.au/media-centre/media-releases/2021-census-25-millionpeople-working-home-census-day
- ABS. (2022b). Characteristics of Employment, Australia methodology. Australian Bureau of Statistics. <u>https://www.abs.gov.au/methodologies/characteristics-</u> <u>employment-australia-methodology/aug-2022</u>
- Abstein, A., & Spieth, P. (2014). Exploring HRM Meta-Features that Foster Employees' Innovative Work Behaviour in Times of Increasing Work-Life Conflict [Article]. Creativity & Innovation Management, 23(2), 211-225. <u>https://doi.org/10.1111/caim.12053</u>
- Abun, D., Macaspact, L. G. R., Valdez, E. B., & Julian, F. P. (2023). The effect of innovative work environment on the innovative work behavior of employees. *International Journal of Research in Business & Complexing Social Science*, 12(3), 140-158. <u>https://doi.org/10.20525/ijrbs.v12i3.2467</u>
- Acar, O. A., Tarakci, M., & Knippenberg, D. v. (2018). Creativity and Innovation Under Constraints: A Cross-Disciplinary Integrative Review. *Journal of Management*. <u>https://doi.org/10.1177/0149206318805832</u>
- Adekoya, O. D., Adisa, T. A., & Aiyenitaju, O. (2022). Going Forward: Remote Working in the Post-Covid-19 Era. *Employee Relations*. <u>https://doi.org/10.1108/er-04-2021-0161</u>
- Adeoye Olatunde, O. A., & Olenik, N. L. (2021). Research and scholarly methods: Semi - structured interviews. *Journal of the american college of clinical pharmacy*, 4(10), 1358-1367.
- Adibe, P., Vesin, S., Almpanopoulou, A., & Blomqvist, K. (2023). A Threshold for Collaborative Innovation: Exploring the Dimensions of Liminality in a Data Economy Initiative. *R and D Management*. <u>https://doi.org/10.1111/radm.12585</u>

- Afsar, B., & Afsar, B. (2019). Transformational Leadership and Innovative Work Behavior. *European Journal of Innovation Management*. https://doi.org/10.1108/ejim-12-2018-0257
- Afsar, B., & Badir, Y. (2016). The mediating role of psychological empowerment on the relationship between person-organization fit and innovative work behaviour. *Journal of Chinese Human Resource Management*, 7(1), 5-26. <u>https://doi.org/10.1108/JCHRM-11-2015-0016</u>
- Afsar, B., & Badir, Y. F. (2017). Workplace Spirituality, Perceived Organizational Support and Innovative Work Behavior. *Journal of Workplace Learning*. <u>https://doi.org/10.1108/jwl-11-2015-0086</u>
- Afsar, B., Badir, Y. F., & Saeed, B. B. (2014). Transformational leadership and innovative work behavior [Article]. *Industrial Management & Data Systems*, 114(8), 1270-1300. <u>https://doi.org/10.1108/IMDS-05-2014-0152</u>
- Afsar, B., & Umrani, W. A. (2019). Transformational Leadership and Innovative Work Behavior. *European Journal of Innovation Management*. https://doi.org/10.1108/ejim-12-2018-0257
- Agarwal, U. A. (2014). Linking justice, trust and innovative work behaviour to work engagement. *Personnel Review*, 43(1), 41-73. <u>https://doi.org/10.1108/PR-02-2012-0019</u>
- Agarwal, U. A., Datta, S., Blake-Beard, S., & Bhargava, S. (2012). Linking LMX, Innovative Work Behaviour and Turnover Intentions. *Career Development International*. <u>https://doi.org/10.1108/13620431211241063</u>
- Agrawal, A., Catalini, C., Goldfarb, A., & Luo, H. (2018). Slack time and innovation. *Organization Science*, 29(6), 1056-1073.
- Ahad, N. A., & Yahaya, S. S. S. (2014). Sensitivity analysis of Welch'st-test. AIP Conference proceedings,
- Ahlstrom, D. (2010). Innovation and growth: How business contributes to society. *Academy of Management Perspectives, 24*(3), 11-24.
- Ahmad, A., Kee, D. M. H., & Daghriri, Y. Q. (2022). Conflict Side of Creativity: Role of Supervisory Support and Team Affective Tone in Facilitating Creative Idea Validation. *Frontiers in Public Health*. <u>https://doi.org/10.3389/fpubh.2022.965278</u>
- Ahmad, A. L., Batool, S., & Madni, Z.-u.-A. (2023). Intrinsic Motivation and Creativity: The Role of Digital Technology and Knowledge Integration Ability in Facilitating Creativity. *International Journal of Management Studies*. <u>https://doi.org/10.32890/ijms2023.30.1.1</u>
- Akbari, M., Bagheri, A., Imani, S., & asadnezhad, m. (2020). Does Entrepreneurial Leadership Encourage Innovation Work Behavior? The Mediating Role of

Creative Self-Efficacy and Support for Innovation. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/ejim-10-2019-0283</u>

- Akhtar, M. S., & Ali, H. (2023). Innovative Work Behavior: An Association of Leadership Styles, Job Autonomy, and Employee Proactive Behavior. *Journal of Social Sciences Review*. <u>https://doi.org/10.54183/jssr.v3i1.128</u>
- Akhtar, M. W., Syed, F., Husnain, M., & Naseer, S. (2019). Person-Organization Fit and Innovative Work Behavior: The Mediating Role of Perceived Organizational Support, Affective Commitment and Trust [Article]. *Pakistan Journal of Commerce & Social Sciences, 13*(2), 311-333. <u>http://wallaby.vu.edu.au:2048/login?url=https://search.ebscohost.com/login.aspx ?direct=true&db=bth&AN=138266291&site=eds-live</u>
- Akram, T., Lei, S., Haider, M. J., & Hussain, S. T. (2020). The impact of organizational justice on employee innovative work behavior: Mediating role of knowledge sharing. *Journal of Innovation & Knowledge*, 5(2), 117-129.
- Al-Madadha, A., Khasawneh, M. H. A., Haddid, O. A., & Al-Adwan, A. S. (2022). Adoption of Telecommuting in the Banking Industry: A Technology Acceptance Model Approach. *Interdisciplinary Journal of Information Knowledge and Management*. <u>https://doi.org/10.28945/5023</u>
- Albers, M. J. (2017). Introduction to quantitative data analysis in the behavioral and social sciences / Michael J. Albers. Wiley.
- Albert, S., & Duffy, J. (2012). Differences in Risk Aversion Between Young and Older Adults. *Neuroscience and Neuroeconomics*. <u>https://doi.org/10.2147/nan.s27184</u>
- Aldulaimi, S. H., & Abdeldayem, M. M. (2020). A thematic analysis of leadership behaviours and change management in higher education to boost sustainability. *International Journal of Higher Education and Sustainability*, 3(1), 34-51.
- Alessandri, G., Perinelli, E., Longis, E. D., Schaufeli, W. B., Theodorou, A., Борг огни, Л., Caprara, G. V., & Cinque, L. (2018). Job Burnout: The Contribution of Emotional Stability and Emotional Self - efficacy Beliefs. *Journal of Occupational and Organizational Psychology*. <u>https://doi.org/10.1111/joop.12225</u>
- Ali, I., Hong, L., & Chen, J. (2022). Remote Cataloging Productivity: An exploratory Study In a national Library. *Library Management*. <u>https://doi.org/10.1108/lm-12-2021-0109</u>
- Ali, M. (2021). Imitation or innovation: To what extent do exploitative learning and exploratory learning foster imitation strategy and innovation strategy for sustained competitive advantage? A. *Technological Forecasting & amp; Social Change, 165.*

- Alkhalaf, T., & Al Tabbaa, O. (2023). The Effect of Ability, Motivation and Opportunity (AMO) on SMEs' Innovation Performance. *Creativity and innovation management*, 33(1), 21-38. <u>https://doi.org/10.1111/caim.12578</u>
- Allen, T. D., Golden, T. D., & Shockley, K. M. (2015). How effective is telecommuting? Assessing the status of our scientific findings. *Psychological Science in the Public Interest*, 16(2), 40-68.
- Allison, P. D. (2009). Missing data. In *The SAGE handbook of quantitative methods in psychology* (2nd ed., pp. 72-89). Emerald Group Publishing Limited.
- AlMunthiri, O., Bani-Melhem, S., Mohd-Shamsudin, F., & Raziq, M. M. (2023). Fostering Innovative Behaviours Of public Sector Employees: The Potency of Innovation-Based HR Practices, Risk Propensity and Error Tolerance. *International Journal of Public Sector Management*, 37(2), 159-182. <u>https://doi.org/10.1108/ijpsm-08-2023-0242</u>
- Alpkan, L., Bulut, Ç., Günday, G., Ulusoy, G., & Kılıç, K. (2010). Organizational Support for Intrapreneurship and Its Interaction With Human Capital to Enhance Innovative Performance. *Management Decision*. https://doi.org/10.1108/02517471080000697
- Alvarez-Torres, F. J., & Schiuma, G. (2022). Measuring the Impact of Remote Working Adaptation on Employees' Well-Being During COVID-19: Insights for Innovation Management Environments. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/ejim-05-2022-0244</u>
- Amankwaa, A., Gyensare, M. A., & Susomrith, P. (2019). Transformational Leadership With Innovative Behaviour. *Leadership & Organization Development Journal*. <u>https://doi.org/10.1108/lodj-10-2018-0358</u>
- Amir, N., Varma, S. B., Razi, S. F. M., Rahmat, N. H., Sukimin, I. S., & Anyau, E. (2021). Working Amidst Covid-19: A Study of Growth, Relatedness, and Existence Among Remote Workers. *International Journal of Academic Research in Business and Social Sciences*. <u>https://doi.org/10.6007/ijarbss/v11i7/10501</u>
- Anand, T., & Lokesh, J. (2018). A Logit Model to Predict Innovativeness among Mobile Telecom Service Users. *Global Business Review*, 19, 54.
- Anderson, D., & Kelliher, C. (2020). Enforced Remote Working and the Work-Life Interface During Lockdown. *Gender in Management an International Journal*. <u>https://doi.org/10.1108/gm-07-2020-0224</u>
- Anderson, K. T., & Holloway-Libell, J. (2014). A Review of "Interviewing as Qualitative Research: A Guide for Researchers in Education and the Social Sciences" Seidman, I.(2013).. New York, NY: Teachers College Press. 178 pp. In. Taylor & Francis.
- Anderson, N., De Dreu, C. K., & Nijstad, B. A. (2004). The routinization of innovation research: A constructively critical review of the state - of - the - science. *Journal of organizational behavior*, 25(2), 147-173.
- Anderson, N., Potočnik, K., & Zhou, J. (2014). Innovation and creativity in organizations: A state-of-the-science review, prospective commentary, and guiding framework. *Journal of Management*, 40(5), 1297-1333.
- Andrade, C., & Fernandes, J. L. (2021). Role Boundary Management During Covid-19 Pandemic: A Qualitative Analysis of Focus Group Data With Working-Student Mothers. *Psicologia*. <u>https://doi.org/10.17575/psicologia.v35i1.1694</u>
- Anindita, R. (2023). The Role of Employee Engagement Towards Innovative Work Behavior Through the Mediation Role of Creativity on Millennial IT Worker. International Conference on Education, Humanities, Social Science (ICEHoS 2022),
- Appelbaum, E., Bailey, T., Berg, P., & Kalleberg, A. (2000). Manufacturing Advantage: Why High-Performance Work Systems Pay Off (Vol. 26). <u>https://doi.org/10.2307/259189</u>
- Arias-Castro, E., Candès, E. J., & Plan, Y. (2011). Global Testing Under Sparse Alternatives: ANOVA, Multiple Comparisons and the Higher Criticism. *The Annals of Statistics*, 39(5). <u>https://doi.org/10.1214/11-aos910</u>
- Arunprasad, P., Dey, C., Jebli, F., Manimuthu, A., & Hathat, Z. E. (2022). Exploring the Remote Work Challenges in the Era of COVID-19 Pandemic: Review and Application Model. *Benchmarking an International Journal*. <u>https://doi.org/10.1108/bij-07-2021-0421</u>
- Ashlan, S., & Bahri, S. (2023). Effect of Work Environment and Innovative Behavior on Teachers Work for Motivation State High School in Banda Aceh City. *Edunesia Jurnal Ilmiah Pendidikan*. <u>https://doi.org/10.51276/edu.v4i1.323</u>
- Assarlind, M., Eriksson, H., Gremyr, I., & Jakobsson, T. (2013). Adopting new ways of working in small and medium-sized enterprises: findings from interventions in 12 European companies [Article]. *Total Quality Management & Business Excellence*, 24(7/8), 945-958. <u>https://doi.org/10.1080/14783363.2013.791108</u>
- Augustine, J. M., Kim, J., & Lee, M. (2023). Parents' Access to Flexible Work Arrangements and Time in Active Caregiving Activities. *Journal of Family Issues*. <u>https://doi.org/10.1177/0192513x231169653</u>
- Awan, M. A. A., & Jehanzeb, K. (2022). How CEO Transformational Leadership Impacts Organizational and Individual Innovative Behavior: Collaborative HRM as Mediator. *Leadership & Organization Development Journal*. <u>https://doi.org/10.1108/lodj-05-2021-0197</u>

- Axtell, C., Holman, D., Unsworth, K., Wall, T., Waterson, P., & Harrington, E. (2000). Shopfloor innovation: Facilitating the suggestion and implementation of ideas. *Journal of Occupational and Organizational Psychology*, 73(3), 265-285.
- Ayers, R. S. (2015). Aligning Individual and Organizational Performance. *Public Personnel Management*. <u>https://doi.org/10.1177/0091026015575178</u>
- Aziz-Ur-Rehman, M., & Siddiqui, D. A. (2019). Relationship Between Flexible Working Arrangements and Job Satisfaction Mediated by Work-Life Balance: Evidence From Public Sector Universities' Employees of Pakistan. *International Journal of Human Resource Studies*. <u>https://doi.org/10.5296/ijhrs.v10i1.15875</u>
- Babin, J. J., Foray, M., & Hussey, A. (2021). Shelter-in-Place Orders, Loneliness, and Collaborative Behavior. *Economics & Human Biology*. https://doi.org/10.1016/j.ehb.2021.101056
- Baer, M., & Frese, M. (2002). Innovation Is Not Enough: Climates for Initiative and Psychological Safety, Process Innovations, and Firm Performance. *Journal of* organizational behavior. <u>https://doi.org/10.1002/job.179</u>
- Baety, N., & Rojuaniah, R. (2022). Servant Leadership, Organizational Culture and Organizational Citizenship Behavior on Innovative Work Behavior. *Management Analysis Journal*. <u>https://doi.org/10.15294/maj.v11i1.54626</u>
- Bagheri, A., & Akbari, M. (2017). The Impact of Entrepreneurial Leadership on Nurses' Innovation Behavior. *Journal of Nursing Scholarship*. <u>https://doi.org/10.1111/jnu.12354</u>
- Bailey, D., E., & Kurland, N., B. (2002). A Review of Telework Research: Findings, New Directions, and Lessons for the Study of Modern Work [research-article]. *Journal of organizational behavior*, 23(4), 383. <u>http://wallaby.vu.edu.au:2048/login?url=https://search.ebscohost.com/login.aspx</u> <u>?direct=true&db=edsjsr&AN=edsjsr.4093813&site=eds-live</u>
- Bailey, T. (1993). Discretionary Effort and the Organization of Work: Employee Participation and Work Reform Since Hawthorne. Teachers College and Conservation of Human Resources, Columbia University. <u>https://books.google.es/books?id=KCoiMwEACAAJ</u>
- Bainbridge, H. T. J., & Townsend, K. (2020). The Effects of Offering Flexible Work Practices to Employees With Unpaid Caregiving Responsibilities for Elderly or Disabled Family Members. *Human Resource Management*. <u>https://doi.org/10.1002/hrm.22007</u>
- Bakarich, K. M., Marcy, A. S., & O'Brien, P. (2022). Has the Fever Left a Burn?A Study of the Impact of COVID-19 Remote Working Arrangements on Public Accountants' Burnout. Accounting Research Journal. <u>https://doi.org/10.1108/arj-09-2021-0249</u>

- Baker, M. (2020). Now Organizations Must Manage a More Complex, Hybrid Workforce <u>https://www.gartner.com/en/newsroom/press-releases/2020-07-14-gartner-survey-reveals-82-percent-of-company-leaders-plan-to-allow-employees-to-work-remotely-some-of-the-time</u>
- Bakker, A. B., & Demerouti, E. (2008). Towards a Model of Work Engagement. *Career Development International*. <u>https://doi.org/10.1108/13620430810870476</u>
- Balnaves, M., & Caputi, P. (2001). Introduction to Quantitative Research Methods : An Investigative Approach. SAGE Publications, Limited. <u>http://ebookcentral.proquest.com/lib/vu/detail.action?docID=370510</u>
- Banerjee, B., & Bag, S. (2022). Relationship Between Creativity, Adaptive Selling, and Sales Performance: The Moderating Role of Job Satisfaction. Asian Academy of Management Journal. <u>https://doi.org/10.21315/aamj2022.27.2.5</u>
- Banks, J., Bassoli, E., & Mammi, I. (2020). Changing Attitudes to Risk at Older Ages: The Role of Health and Other Life Events. *Journal of Economic Psychology*. <u>https://doi.org/10.1016/j.joep.2019.102208</u>
- Banyard, V. L., & Miller, K. E. (1998). The Powerful Potential of Qualitative Research for Community Psychology. *American Journal of Community Psychology*, 26(4), 485-505. <u>https://doi.org/10.1023/a:1022136821013</u>
- Baregheh, A., Rowley, J., & Sambrook, S. (2009). Towards a multidisciplinary definition of innovation. *Management Decision*, 47(8), 1323-1339.
- Barrero, J. M., Bloom, N., & Davis, S. J. (2021). Why working from home will stick.
- Basadur, M. (2004). Leading others to think innovatively together: Creative leadership. *The leadership quarterly*, 15(1), 103-121.
- Basu, R., & Green, S. G. (1997). Leader member exchange and transformational leadership: an empirical examination of innovative behaviors in leader member dyads. *Journal of applied social psychology*, 27(6), 477-499.
- Batarseh, F., Usher, J. M., & Daspit, J. J. (2017). Collaboration Capability in Virtual Teams: Examining the Influence on Diversity and Innovation. *International Journal of Innovation Management*. https://doi.org/10.1142/s1363919617500347
- Battistelli, A., Montani, F., & Odoardi, C. (2013, 02 / 01 /). The impact of feedback from job and task autonomy in the relationship between dispositional resistance to change and innovative work behaviour [Article]. *European Journal of Work* and Organizational Psychology, 22(1), 26-41. https://doi.org/10.1080/1359432X.2011.616653
- Baumann, H. M., & Marcum, T. M. (2023). Human Capital and Legal Perspectives on Remote Work: Recommendations for Organizations. *Management Research Review*. <u>https://doi.org/10.1108/mrr-06-2022-0412</u>

- Baumann, O., & Sander, E. J. (2021). Psychological Impacts of Remote Working Under Social Distancing Restrictions. <u>https://doi.org/10.4018/978-1-7998-6754-</u> <u>8.ch001</u>
- Baumol, W. J., & Strom, R. J. (2007). Entrepreneurship and economic growth. *Strategic entrepreneurship journal*, 1(3 4), 233-237.
- Baxter;, J., & Warren, D. (2021). Towards COVID Normal: Employment and workfamily balance in 2020 (The Families in Australia Survey, Issue. <u>https://aifs.gov.au/sites/default/files/publication-</u> documents/2105_2_fias_employment_and_work_family_balance_in_2020.pdf
- Bayo-Moriones, A., & Galdon-Sanchez, J. E. (2010, 2010/06/01). Multinational companies and high-performance work practices in the Spanish manufacturing industry. *The International Journal of Human Resource Management, 21*(8), 1248-1271. <u>https://doi.org/10.1080/09585192.2010.483848</u>
- Beaty, R. E., Kenett, Y. N., Christensen, A. P., Rosenberg, M. D., Benedek, M., Chen, Q., Fink, A., Qiu, J., Kwapil, T. R., Kane, M. J., & Silvia, P. J. (2018). Robust Prediction of Individual Creative Ability From Brain Functional Connectivity. *Proceedings of the National Academy of Sciences*. https://doi.org/10.1073/pnas.1713532115
- Beck, M. J., & Hensher, D. A. (2022). Australia 6 months after COVID-19 restrictions part 2: The impact of working from home. *Transport Policy*, *128*, 274-285. <u>https://doi.org/10.1016/j.tranpol.2021.06.005</u>
- Beck, M. J., Hensher, D. A., & Wei, E. T. (2020). Slowly Coming Out of COVID-19 Restrictions in Australia: Implications for Working From Home and Commuting Trips by Car and Public Transport. *Journal of Transport Geography*. <u>https://doi.org/10.1016/j.jtrangeo.2020.102846</u>
- Becker, W. J., Belkin, L. Y., Tuskey, S. E., & Conroy, S. A. (2022). Surviving remotely: How job control and loneliness during a forced shift to remote work impacted employee work behaviors and well - being. *Human Resource Management*, 61(4), 449-464.
- Bedenik, T., Kearney, C., & Ní Shé, É. (2024). Trust in Embedding Co-Design For innovation and Change: Considering the Role of Senior Leaders and Managers. Journal of Health Organization and Management. <u>https://doi.org/10.1108/jhom-07-2023-0207</u>
- Bencsik, A., Horváth-Csikós, G., & Juhász, T. (2016). Y and Z Generations at Workplaces. *Journal of competitiveness*, 8(3).
- Beno, M. (2021). On-Site and Hybrid Workplace Culture of Positivity and Effectiveness: Case Study From Austria. Academic Journal of Interdisciplinary Studies, 10(5), 331. <u>https://doi.org/10.36941/ajis-2021-0142</u>

- Beňo, M., & Koreňová, L. (2021). Data on an Austrian Company's Productivity in the Pre-Covid-19 Era, During the Lockdown and After Its Easing: To Work Remotely or Not? Frontiers in Communication. https://doi.org/10.3389/fcomm.2021.641199
- Bergefurt, L., Appel Meulenbroek, R., & Arentze, T. (2023). How Physical Home Workspace Characteristics Affect Mental Health: A Systematic Scoping Review. Work. <u>https://doi.org/10.3233/wor-220505</u>
- Berkel, R. v., Vries, J. P. d., & Knies, E. (2021). Managing Street-Level Bureaucrats' Performance by Promoting Professional Behavior Through HRM. *Public Personnel Management*. <u>https://doi.org/10.1177/00910260211046554</u>
- Beugelsdijk, S. (2008, 06 / 01 /). Strategic human resource practices and product innovation [Article]. *Organization Studies*, 29(6), 821-847. https://doi.org/10.1177/0170840608090530
- Bhatti, S. H., Zakariya, R., Vrontis, D., Santoro, G., & Christofi, M. (2020). High-Performance Work Systems, Innovation and Knowledge Sharing. *Employee Relations*, 43(2), 438-458. <u>https://doi.org/10.1108/er-10-2019-0403</u>
- Bick, A., Blandin, A., & Mertens, K. (2023). Work from home before and after the COVID-19 outbreak. *American Economic Journal: Macroeconomics*, 15(4), 1-39.
- Bierly, P. E., Stark, E. M., & Kessler, E. H. (2009). The Moderating Effects of Virtuality on the Antecedents and Outcome of NPD Team Trust. *Journal of Product Innovation Management*. <u>https://doi.org/10.1111/j.1540-5885.2009.00680.x</u>
- Bilan, Y., Mishchuk, H., Roshchyk, I., & Joshi, O. (2020). Hiring and Retaining Skilled Employees in Smes: Problems in Human Resource Practices and Links With Organizational Success. Verslas Teorija Ir Praktika. https://doi.org/10.3846/btp.2020.12750
- Bilbao Osorio, B., & Rodríguez Pose, A. (2004). From R&D to innovation and economic growth in the EU. *Growth and Change*, *35*(4), 434-455.
- Binnewies, C., Ohly, S., & Sonnentag, S. (2007). Taking Personal Initiative and Communicating About Ideas: What Is Important for the Creative Process and for Idea Creativity? *European Journal of Work and Organizational Psychology*. <u>https://doi.org/10.1080/13594320701514728</u>
- Bird, K. D. (2002). Confidence Intervals for Effect Sizes in Analysis of Variance. *Educational and Psychological Measurement*, 62(2), 197-226. <u>https://doi.org/10.1177/0013164402062002001</u>
- Birdi, K., Leach, D., & Magadley, W. (2014). The Relationship of Individual Capabilities and Environmental Support With Different Facets of Designers'

Innovative Behavior. *Journal of Product Innovation Management*. https://doi.org/10.1111/jpim.12250

- Birken, S. A., Lee, S. Y. D., & Weiner, B. J. (2012). Uncovering Middle Managers' Role in Healthcare Innovation Implementation. *Implementation Science*. <u>https://doi.org/10.1186/1748-5908-7-28</u>
- Birkinshaw, J., Brannen, M. Y., & Tung, R. L. (2011). From a Distance and Generalizable to Up Close and Grounded: Reclaiming a Place for Qualitative Methods in International Business Research. *Journal of International Business Studies*, 42(5), 573-581. <u>https://doi.org/10.1057/jibs.2011.19</u>
- Bischof, D., & Velden, M. v. d. (2019). The Use and Usefulness of P Values in Political Science: Introduction. *Swiss Political Science Review*, 25(3), 276-280. <u>https://doi.org/10.1111/spsr.12376</u>
- Bishara, A. J., Li, J., & Conley, C. (2021). Informal Versus Formal Judgment of Statistical Models: The Case of Normality Assumptions. *Psychonomic Bulletin* & *Review*, 28(4), 1164-1182. <u>https://doi.org/10.3758/s13423-021-01879-z</u>
- Björk, J., & Magnusson, M. (2009). Where Do Good Innovation Ideas Come From? Exploring the Influence of Network Connectivity on Innovation Idea Quality. *Journal of Product Innovation Management*. <u>https://doi.org/10.1111/j.1540-5885.2009.00691.x</u>
- Bland, J. M., & Altman, D. G. (1996). Statistics notes: Transforming data. *Bmj*, *312*(7033), 770.
- Blauth, M., Mauer, R., & Brettel, M. (2014). Fostering Creativity in New Product Development Through Entrepreneurial Decision Making. *Creativity and innovation management*. <u>https://doi.org/10.1111/caim.12094</u>
- Bleiker, J., Morgan Trimmer, S., Knapp, K., & Hopkins, S. (2019). Navigating the Maze: Qualitative Research Methodologies and Their Philosophical Foundations. *Radiography*, 25, S4-S8. <u>https://doi.org/10.1016/j.radi.2019.06.008</u>
- Blumberg, B., Cooper, D., & Schindler, P. (2008). *Business Research Methods* (2nd edition ed.). McGraw Hill.
- Blumberga, S., & Berga, S. (2022). Personnel Loyalty, Work-Life Balance During Remote Work. Proceedings of Cbu in Social Sciences. <u>https://doi.org/10.12955/pss.v3.300</u>
- Boccoli, G., Sestino, A., Gastaldi, L., & Corso, M. (2022). The Impact of Autonomy and Temporal Flexibility on Individuals' Psychological Wellbeing in Remote Settings. *Sinergie Italian Journal of Management*. https://doi.org/10.7433/s118.2022.15
- Bos-Nehles, A., Bondarouk, T., & Nijenhuis, K. (2017). Innovative work behaviour in knowledge-intensive public sector organizations: the case of supervisors in the

Netherlands fire services [Article]. *International Journal of Human Resource Management*, 28(2), 379-398. <u>https://doi.org/10.1080/09585192.2016.1244894</u>

- Bos-Nehles, A., Renkema, M., & Janssen, M. (2017, 11/15/). HRM and innovative work behaviour: a systematic literature review [Article]. *Personnel Review*, 46(7), 1228-1253. <u>https://doi.org/10.1108/PR-09-2016-0257</u>
- Bos-Nehles, A. C., Riemsdijk, M. v., & Looise, J. C. (2013). Employee Perceptions of Line Management Performance: Applying the AMO Theory to Explain the Effectiveness of Line Managers' HRM Implementation. *Human Resource Management*. <u>https://doi.org/10.1002/hrm.21578</u>
- Bos-Nehles, A. C., Townsend, K., Cafferkey, K., & Trullén, J. (2023). Examining the Ability, Motivation and Opportunity (AMO) Framework in HRM Research: Conceptualization, Measurement and Interactions. *International journal of management reviews*. <u>https://doi.org/10.1111/ijmr.12332</u>
- Boskovic, A. (2021). Employee Autonomy and Engagement in the Digital Age: The Moderating Role of Remote Working. *Ekonomski Horizonti*. <u>https://doi.org/10.5937/ekonhor2103241b</u>
- Boxall, P., & Macky, K. (2009). Research and theory on high-performance work systems: progressing the high-involvement stream. *Human Resource Management Journal*, 19(1), 3-23. <u>https://doi.org/https://doi.org/10.1111/j.1748-8583.2008.00082.x</u>
- Boxall, P., & Purcell, J. (2003, 10/01). Strategy and Human Resource Management. Industrial & Labor Relations Review, 57. https://doi.org/10.1108/00251740310479368
- Boyle, P. A., Yu, L., Buchman, A. S., & Bennett, D. A. (2012). Risk Aversion Is Associated With Decision Making Among Community-Based Older Persons. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2012.00205</u>
- Bracio, K., & Szarucki, M. (2020). Mixed Methods Utilisation in Innovation Management Research: A Systematic Literature Review and Meta-Summary. *Journal of Risk and Financial Management*, 13(11), 252. <u>https://doi.org/10.3390/jrfm13110252</u>
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative* research in psychology, 3(2), 77-101.
- Braun, V., & Clarke, V. (2012). Thematic Analysis. In *APA handbook of research methods in psychology, Vol 2: Research designs: Quantitative, qualitative, neuropsychological, and biologica* (pp. 57-71).
- Breuer, C., Hüffmeier, J., & Hertel, G. (2016). Does Trust Matter More in Virtual Teams? A Meta-Analysis of Trust and Team Effectiveness Considering Virtuality and Documentation as Moderators. *Journal of Applied psychology*. <u>https://doi.org/10.1037/apl0000113</u>

Brinkmann, S. (2013). Qualitative interviewing. Oxford University Press.

- Brooks, S. K., Hall, C. L., Patel, D., & Greenberg, N. (2022). "In the Office Nine to Five, Five Days a Week... Those Days Are Gone": Qualitative Exploration of Diplomatic Personnel's Experiences of Remote Working During the COVID-19 Pandemic. BMC Psychology. <u>https://doi.org/10.1186/s40359-022-00970-x</u>
- Brown, J., & Tousey, C. (2023). The Shifting Expectations for Work From Home. *The Federal Reserve Bank of Kansas City Economic Review*. <u>https://doi.org/10.18651/er/v108n2browntousey</u>
- Brunetto, Y., Xerri, M., & Farr Wharton, B. (2020). Comparing the role of personal and organisational support on the innovative behaviour of frontline healthcare workers in Australia and the United States. *Australian Journal of Public Administration, 79*(3), 279-297. <u>https://doi.org/10.1111/1467-8500.12414</u>
- Brutus, S., Aguinis, H., & Wassmer, U. (2013). Self-reported limitations and future directions in scholarly reports: Analysis and recommendations. *Journal of Management*, 39(1), 48-75.
- Bryman, A. (2016). Social research methods. Oxford university press.
- Buhl, A., Blazejewski, S., & Dittmer, F. (2016). The More, the Merrier: Why and How Employee-Driven Eco-Innovation Enhances Environmental and Competitive Advantage. *Sustainability*. <u>https://doi.org/10.3390/su8090946</u>
- Bulińska-Stangrecka, H., & Bagieńska, A. (2019). HR Practices for Supporting Interpersonal Trust and Its Consequences for Team Collaboration and Innovation. *Sustainability*. <u>https://doi.org/10.3390/su11164423</u>
- Burgess, J. (2005). Exploring Job Quality and Part-Time Work in Australia. Labour & Industry a Journal of the Social and Economic Relations of Work. https://doi.org/10.1080/10301763.2005.10669316
- Bürkland, S., Zachariassen, F., & Oliveira, J. (2019). Meeting Up for Management Control: Bracketing Interaction in Innovation Development. *Qualitative Research in Accounting & Management*. <u>https://doi.org/10.1108/qram-06-2017-0060</u>
- Buvik, M. P., & Tvedt, S. D. (2016). The Impact of Commitment and Climate Strength on the Relationship Between Trust and Performance in Cross-Functional Project Teams. *Team Performance Management*. <u>https://doi.org/10.1108/tpm-02-2015-0011</u>
- Bygballe, L. E., & Ingemansson, M. (2014). The Logic of Innovation in Construction. Industrial Marketing Management. https://doi.org/10.1016/j.indmarman.2013.12.019

- Byrne, D. (2021). A Worked Example of Braun and Clarke's Approach to Reflexive Thematic Analysis. *Quality & Quantity*, 56(3), 1391-1412. https://doi.org/10.1007/s11135-021-01182-y
- Cai, W., Khapova, S. N., Bossink, B., Lysova, E. I., & Yuan, J. (2020). Optimizing Employee Creativity in the Digital Era: Uncovering the Interactional Effects of Abilities, Motivations, and Opportunities. *International Journal of Environmental Research and Public Health*. https://doi.org/10.3390/ijerph17031038
- Camisón-Zornoza, C., Lapiedra-Alcamí, R., Segarra-Ciprés, M., & Boronat-Navarro, M. (2004). A meta-analysis of innovation and organizational size. *Organization Studies*, 25(3), 331-361.
- Cangialosi, N., Odoardi, C., Battistelli, A., & Baldaccini, A. (2021). The Social Side of Innovation: When and Why Advice Network Centrality Promotes Innovative Work Behaviours. *Creativity and innovation management*. <u>https://doi.org/10.1111/caim.12434</u>
- Cantor, D. E., Morrow, P. C., McElroy, J. C., & Montabon, F. (2013). The Role of Individual and Organizational Factors in Promoting Firm Environmental Practices. *International Journal of Physical Distribution & Logistics Management*. <u>https://doi.org/10.1108/ijpdlm-03-2012-0071</u>
- Cao, H., Lee, C.-J., Iqbal, S. T., Czerwinski, M., Wong, P. N. Y., Rintel, S., Hecht, B., Teevan, J., & Yang, L. (2021). Large Scale Analysis of Multitasking Behavior During Remote Meetings. <u>https://doi.org/10.1145/3411764.3445243</u>
- Carmeli, A., & Spreitzer, G. M. (2009). Trust, connectivity, and thriving: Implications for innovative behaviors at work. *The Journal of Creative Behavior*, 43(3), 169-191.
- Carpini, J. A., Luksyte, A., Parker, S. K., & Collins, C. G. (2023). Can a Familiar Gender Stereotype Create a Not - so - familiar Benefit for Women? Evidence of Gendered Differences in Ascribed Stereotypes and Effects of Team Member Adaptivity on Performance Evaluations. *Journal of organizational behavior*. <u>https://doi.org/10.1002/job.2702</u>
- Carroll, N., & Conboy, K. (2020, 2020/12/01/). Normalising the "new normal": Changing tech-driven work practices under pandemic time pressure. *International Journal of Information Management*, 55, 102186. <u>https://doi.org/https://doi.org/10.1016/j.ijinfomgt.2020.102186</u>
- Cavana, R. Y., Delahaye, B. L., & Sekaran, U. (2001). *Applied business research: Qualitative and quantitative methods.* John Wiley & Sons Inc.
- Ceci, F., Lazoi, M., Lezzi, M., & Mohammad, H. (2021). Fostering Knowledge Sharing in the Innovation Process: Information and Communication Technology<scp> -</Scp>based Versus Face - to - face Relationships. *Knowledge and Process Management*. <u>https://doi.org/10.1002/kpm.1668</u>

- Černe, M., Hernaus, T., Dysvik, A., & Škerlavaj, M. (2017). The Role of Multilevel Synergistic Interplay Among Team Mastery Climate, Knowledge Hiding, and Job Characteristics in Stimulating Innovative Work Behavior. *Human Resource Management Journal*. <u>https://doi.org/10.1111/1748-8583.12132</u>
- Chafi, M. B., Hultberg, A., & Yams, N. B. (2021). Post-Pandemic Office Work: Perceived Challenges and Opportunities for a Sustainable Work Environment. Sustainability. <u>https://doi.org/10.3390/su14010294</u>
- Chan, D. (2010). So why ask me? Are self-report data really that bad? In *Statistical and methodological myths and urban legends* (pp. 329-356). Routledge.
- Chan, X. W., Shang, S., Brough, P., Wilkinson, A., & Lu, C. q. (2023). Work, life and COVID - 19: a rapid review and practical recommendations for the post pandemic workplace. *Asia Pacific Journal of Human Resources*, 61(2), 257-276. <u>https://doi.org/10.1111/1744-7941.12355</u>
- Chantziaras, G., Triantafyllidis, A., Papaprodromou, A., Chatzikonstantinou, I., Giakoumis, D., Tsakiris, A., Votis, K., & Tzovaras, D. (2021). An Augmented Reality-Based Remote Collaboration Platform for Worker Assistance. <u>https://doi.org/10.1007/978-3-030-68787-8_30</u>
- Charalampous, M., Grant, C. S., Tramontano, C., & Michailidis, E. (2018). Systematically Reviewing Remote E-Workers' Well-Being at Work: A Multidimensional Approach. *European Journal of Work and Organizational Psychology*. <u>https://doi.org/10.1080/1359432x.2018.1541886</u>
- Charmaz, K. (2014). Grounded theory in global perspective: Reviews by international researchers. *Qualitative Inquiry*, 20(9), 1074-1084.
- Chatzoglou, P. D., & Chatzoudes, D. (2018). The Role of Innovation in Building Competitive Advantages: An Empirical Investigation. *European Journal of Innovation Management*. https://doi.org/10.1108/ejim-02-2017-0015
- Chaubey, A., & Sahoo, C. K. (2019). Enhancing Organizational Innovation in Indian Automobile Industry. *International Journal of Innovation Science*. https://doi.org/10.1108/ijis-02-2018-0022
- Chen, P., Zhang, J., & Qi, Y. (2020). Distant Temporal Distance and Creative Thinking: The Mediating Role of Promotion Motivation. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2020.576835</u>
- Chen, Y., Shek, D. T. L., & Bu, F.-F. (2011). Applications of Interpretive and Constructionist Research Methods in Adolescent Research: Philosophy, Principles and Examples. *International Journal of Adolescent Medicine and Health*, 23(2). <u>https://doi.org/10.1515/ijamh.2011.022</u>

- Chénéchal, M. L., Duval, T., Gouranton, V., Royan, J., & Arnaldi, B. (2019). Help! I Need a Remote Guide in My Mixed Reality Collaborative Environment. *Frontiers in Robotics and Ai*. https://doi.org/10.3389/frobt.2019.00106
- Cheng, Z., Dimoka, A., & Pavlou, P. A. (2016). Context may be King, but generalizability is the Emperor! *Journal of Information Technology*, 31(3), 257-264.
- Cheng, Z., Liu, W., Zhou, K., Che, Y.-J., & Han, Y. (2021). Promoting Employees' Pro - environmental Behaviour Through Empowering Leadership: The Roles of Psychological Ownership, Empowerment Role Identity, and Environmental Self - identity. *Business Ethics the Environment & Responsibility*. <u>https://doi.org/10.1111/beer.12366</u>
- Chenger, D., & Woiceshyn, J. (2020). Executives' Decision Processes at the Front End of Major Projects: The Role of Context and Experience in Value Creation. *Project Management Journal*. <u>https://doi.org/10.1177/8756972820977225</u>
- Cheung, S. Y., Gong, Y., Wang, M., Zhou, L., & Shi, J. (2016). When and How Does Functional Diversity Influence Team Innovation? The Mediating Role of Knowledge Sharing and the Moderation Role of Affect-Based Trust in a Team. *Human Relations*. <u>https://doi.org/10.1177/0018726715615684</u>
- Chiu, H.-C., & Chiang, P.-H. (2019). A Trickle-Down Effect of Subordinates' Felt Trust. *Personnel Review*. <u>https://doi.org/10.1108/pr-01-2018-0036</u>
- Choi, W., Kang, S. W., & Choi, S. B. (2021). Innovative Behavior in the Workplace: An Empirical Study of Moderated Mediation Model of Self-Efficacy, Perceived Organizational Support, and Leader–Member Exchange. *Behavioral Sciences*. <u>https://doi.org/10.3390/bs11120182</u>
- Choudhary, S., & Mishra, K. (2021). Understanding Knowledge Hiding in the Context of Virtual Workplaces. *Vine Journal of Information and Knowledge Management Systems*. <u>https://doi.org/10.1108/vjikms-10-2020-0185</u>
- Chowdhury, M. F. (2014). Interpretivism in Aiding Our Understanding of the Contemporary Social World. *Open Journal of Philosophy*, 04(03), 432-438. <u>https://doi.org/10.4236/ojpp.2014.43047</u>
- Chretien, K. C., & Kind, T. (2013). Social Media and Clinical Care. *Circulation*. <u>https://doi.org/10.1161/circulationaha.112.128017</u>
- Cirella, S. (2021). Managing Collective Creativity: Organizational Variables to Support Creative Teamwork. *European Management Review*. <u>https://doi.org/10.1111/emre.12475</u>
- Clark, V. L. P., & Creswell, J. W. (2008). *The Mixed Methods Reader* Sage Publications.

- Clegg, C., Unsworth, K., Epitropaki, O., & Parker, G. (2002). Implicating trust in the innovation process. *Journal of Occupational and Organizational Psychology*, 75(4), 409-422.
- Clercq, D. D., & Pereira, R. (2019). Resilient Employees Are Creative Employees, When the Workplace Forces Them to Be. *Creativity and innovation management*. <u>https://doi.org/10.1111/caim.12328</u>
- Çobanoğlu, N. (2021). The Relationship Between Shared Leadership, Employee Empowerment and Innovativeness in Primary Schools: A Structural Equation Modeling. *European Journal of Educational Research*. <u>https://doi.org/10.12973/eu-jer.10.1.327</u>
- Coenen, M., & Kok, R. (2014, 8/1/August 2014). Workplace flexibility and new product development performance: The role of telework and flexible work schedules [Article]. *European Management Journal, 32*, 564-576. https://doi.org/10.1016/j.emj.2013.12.003
- Conway, N., & Briner, R. B. (2002). Full-Time Versus Part-Time Employees: Understanding the Links Between Work Status, the Psychological Contract, and Attitudes. *Journal of Vocational Behavior*. <u>https://doi.org/10.1006/jvbe.2001.1857</u>
- Cortland, C., & Kinias, Z. (2019). Stereotype Threat and Women's Work Satisfaction: The Importance of Role Models. *Archives of Scientific Psychology*. <u>https://doi.org/10.1037/arc0000056</u>
- Costa, G., Bortoluzzi, G., & Černe, M. (2022). Can Innovative Work Behaviour Spur Creativity While Working Remotely? The Role of Work-home Conflict and Social Isolation. *Management Research Review*. <u>https://doi.org/10.1108/mrr-03-2022-0204</u>
- Costa, G. G., Aleksić, D., & Bortoluzzi, G. (2021). The power of balance: Interplay effects of exploitative leadership style, work–family balance and family-friendly workplace practices on innovation implementation. *European Journal of Innovation Management*, 25(5), 1266-1287.
- Costantini, A., & Rubini, S. (2021). Workplace Innovation Through Proactivity. Job Crafting and Work Engagement Dynamics Among Remote Workers in Italy. <u>https://doi.org/10.31234/osf.io/5g26e</u>
- Coun, M. J. H., Edelbroek, R., Peters, P., & Blomme, R. J. (2021). Leading Innovative Work-Behavior in Times of COVID-19: Relationship Between Leadership Style, Innovative Work-Behavior, Work-Related Flow, and IT-Enabled Presence Awareness During the First and Second Wave of the COVID-19 Pandemic. *Frontiers in Psychology, 12*. <u>https://doi.org/10.3389/fpsyg.2021.717345</u>
- Cramer, D., & Howitt, D. L. (2004). *The Sage dictionary of statistics: a practical resource for students in the social sciences*. Sage.

- Creswell, J. W. (2007). *Qualitative inquiry and research design: Choosing among five approaches*. Sage publications.
- Creswell, J. W. (2018). *Research design : qualitative, quantitative, and mixed methods approaches* (Fifth edition. ed.). SAGE Publications, Inc.
- Cripe, K. M., & Burleigh, C. (2022). Examining Leadership Skills, Behaviors, and Effective Communication for Virtual IT Project Managers. *Team Performance Management*. <u>https://doi.org/10.1108/tpm-11-2021-0085</u>
- Criscuolo, P., Salter, A., & Wal, A. L. J. T. (2014). Going Underground: Bootlegging and Individual Innovative Performance. *Organization Science*. <u>https://doi.org/10.1287/orsc.2013.0856</u>
- Cronbach, L. J. (1951). Coefficient Alpha and the Internal Structure of Tests. *Psychometrika*, 16(3), 297-334. <u>https://doi.org/10.1007/bf02310555</u>
- Crowley, S., Gonnerman, C., & O'Rourke, M. (2016). Cross-Disciplinary Research as a Platform for Philosophical Research. *Journal of the American Philosophical Association*, 2(2), 344-363. <u>https://doi.org/10.1017/apa.2016.16</u>
- Cuel, R., Cacciatore, E., Ravarini, A., & Varriale, L. (2022). Do Organizations Need a Head of Remote Work? <u>https://doi.org/10.1007/978-3-031-10902-7_19</u>
- Cui, G., Wang, F., & Zhang, Y. (2022). Buffer or Boost? The Role of Openness to Experience and Knowledge Sharing in the Relationship Between Team Cognitive Diversity and Members' Innovative Work Behavior. Current Psychology. <u>https://doi.org/10.1007/s12144-022-03633-7</u>
- Cui, Y., & Gui-lan, Y. U. (2021). A Cross-Level Examination of Team-Directed Empowering Leadership and Subordinates' Innovative Performance: An AMO Theory Perspective. *International Journal of Manpower*. <u>https://doi.org/10.1108/ijm-03-2020-0099</u>
- Dağ, O., Dolgun, A., & Konar, N. M. (2018). Onewaytests: An R Package for One-Way Tests in Independent Groups Designs. *The R Journal*, 10(1), 175. <u>https://doi.org/10.32614/rj-2018-022</u>
- Dai, D. Y., Cheng, H. T., & Yang, P. (2019). QEOSA: A Pedagogical Model That Harnesses Cultural Resources to Foster Creative Problem-Solving. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2019.00833</u>
- Dai, Q., Dai, Y., Zhang, C., Meng, Z., Chen, Z., & Hu, S. (2022). The Influence of Personal Motivation and Innovative Climate on Innovative Behavior: Evidence From University Students in China. *Psychology Research and Behavior Management*. <u>https://doi.org/10.2147/prbm.s381494</u>
- Dai, Y., Byun, G., & Ding, F. (2018). The Direct and Indirect Impact of Gender Diversity in New Venture Teams on Innovation Performance. *Entrepreneurship Theory and Practice*. <u>https://doi.org/10.1177/1042258718807696</u>

- Dam, K. v., Oreg, S., & Schyns, B. (2008). Daily Work Contexts and Resistance to Organisational Change: The Role of Leader–Member Exchange, Development Climate, and Change Process Characteristics. *Applied Psychology*. <u>https://doi.org/10.1111/j.1464-0597.2007.00311.x</u>
- Damanpour, F., & Evan, W. M. (1984). Organizational innovation and performance: the problem of organizational lag. *Administrative science quarterly*, 392-409.
- Damayanti, D. W. P., & Kurniawan, J. E. (2023). Innovative Work Behaviour in Employed Homemakers: The Role of Work-Family Balance and Resilience. *Journal of Eastern European and Central Asian Research (Jeecar)*. <u>https://doi.org/10.15549/jeecar.v10i2.1045</u>
- Dar, N., Sudiro, A., & Rahman, W. (2022). How and When Overqualification Improves Innovative Work Behaviour: The Roles of Creative Self-Confidence and Psychological Safety. *Personnel Review*. <u>https://doi.org/10.1108/pr-06-2020-0429</u>
- Davids, M. F., & Frenken, K. (2017). Proximity, Knowledge Base and the Innovation Process: Towards an Integrated Framework. *Regional Studies*. <u>https://doi.org/10.1080/00343404.2017.1287349</u>
- Dawadi, S., Shrestha, S., & Giri, R. A. (2021). Mixed-Methods Research: A Discussion on Its Types, Challenges, and Criticisms. *Journal of Practical Studies in Education*. <u>https://doi.org/10.46809/jpse.v2i2.20</u>
- De Jong, J., & Den Hartog, D. (2007, 01 / 01 /). How leaders influence employees' innovative behaviour [Article]. *European Journal of Innovation Management*, 10(1), 41-64. <u>https://doi.org/10.1108/14601060710720546</u>
- De Jong, J., & Den Hartog, D. (2010). Measuring Innovative Work Behaviour [Article]. *Creativity & Innovation Management, 19*(1), 23-36. <u>https://doi.org/10.1111/j.1467-8691.2010.00547.x</u>
- De Leede, J., & Nijland, J. (2016). Understanding teamwork behaviors in the use of new ways of working (Vol. 16) [Chapter]. Emerald Group Publishing Ltd. https://doi.org/10.1108/S1877-636120160000016005
- Degerli, M. (2023). New Normal for Gold and White-Collar Workers: The Hybrid Way. Business and Management Studies an International Journal. https://doi.org/10.15295/bmij.v11i1.2194
- Del Águila, M. R., & González-Ramírez, A. (2014). Sample size calculation. *Allergologia et immunopathologia, 42*(5), 485-492.
- Delacre, M., Lakens, D., Mora, Y., & Leys, C. (2018). Taking Parametric Assumptions Seriously Arguments for the Use of Welch's F-Test Instead of the Classical F-Test in One-Way ANOVA (In Press for the International Review of Social Psychology). <u>https://doi.org/10.31234/osf.io/wnezg</u>

- Delanoeije, J., Verbruggen, M., & Germeys, L. (2019). Boundary Role Transitions: A Day-to-Day Approach to Explain the Effects of Home-Based Telework on Work-to-Home Conflict and Home-to-Work Conflict. *Human Relations*. <u>https://doi.org/10.1177/0018726718823071</u>
- Denzin, N. K., & Lincoln, Y. S. (2011). *The Sage handbook of qualitative research*. Sage.
- Denzin, N. K., & Lincoln, Y. S. (2017). *The Sage handbook of qualitative research* (Fifth edition. ed.). Sage.
- Déprez, G. R. M., Battistelli, A., & Vandenberghe, C. (2021). Linking Proactive Behavior and Constructive Deviance to Affective Commitment and Turnover Intention: The Mediating Role of Idea Championing. *Journal of Management & Organization*. <u>https://doi.org/10.1017/jmo.2021.54</u>
- Derrick, B., Toher, D., & White, P. (2016). Why Welch's test is Type I error robust. *The quantitative methods for Psychology, 12*(1), 30-38.
- Deutrom, J., Katos, V., & Ali, R. (2022). Loneliness, life satisfaction, problematic internet use and security behaviours: re-examining the relationships when working from home during COVID-19. *Behaviour & amp; Information Technology*, 41(14), 3161-3175. https://doi.org/10.1080/0144929X.2021.1973107
- Divisekera, S., & Nguyen, V. (2018). Determinants of innovation in tourism evidence from Australia. <u>https://doi.org/10.1016/j.tourman.2018.01.010</u>
- Domingo, L., Gutzeit, M., Leifer, L., & Auernhammer, J. (2021). Remote Brainstorming: Methodological Interventions in Designing From a Distance. *Proceedings of the Design Society*. <u>https://doi.org/10.1017/pds.2021.515</u>
- Donati, S., Viola, G., Toscano, F., & Zappalà, S. (2021). Not All Remote Workers Are Similar: Technology Acceptance, Remote Work Beliefs, and Wellbeing of Remote Workers During the Second Wave of the COVID-19 Pandemic. *International Journal of Environmental Research and Public Health*, 18(22), 12095. <u>https://doi.org/10.3390/ijerph182212095</u>
- Dorairaj, S., Noble, J., & Malik, P. (2012). Understanding Lack of Trust in Distributed Agile Teams: A Grounded Theory Study. <u>https://doi.org/10.1049/ic.2012.0011</u>
- Doran, J., & Ryan, G. (2017). The Role of Stimulating Employees' Creativity and Idea Generation in Encouraging Innovation Behaviour in Irish Firms. *The Irish Journal of Management*. <u>https://doi.org/10.1515/ijm-2017-0005</u>
- Dorenbosch, L., Engen, M., & Verhagen, M. (2005). On the job innovation: The impact of job design and human resource management through production ownership. *Creativity and innovation management*, 14(2), 129-141.

- Doshi, P., & Nigam, P. (2022). Analysing the Effect of AMO Framework on the Employee's Voice Behaviour. *Journal of Organizational Effectiveness People* and Performance. <u>https://doi.org/10.1108/joepp-05-2022-0117</u>
- Driskell, J. E., Radtke, P. H., & Salas, E. (2003). Virtual Teams: Effects of Technological Mediation on Team Performance. *Group Dynamics Theory Research and Practice*. <u>https://doi.org/10.1037/1089-2699.7.4.297</u>
- Drucker, P. F. (2015). Innovation and entrepreneurship: practice and principles. Routledge. <u>http://wallaby.vu.edu.au:2048/login?url=https://search.ebscohost.com/login.aspx</u> <u>?direct=true&db=cat06414a&AN=vic.b3271976&site=eds-live</u> http://wallaby.vu.edu.au:2048/login?url=https://www.taylorfrancis.com/books/9781315
- Dul, J., & Ceylan, C. (2014). The Impact of a Creativity supporting Work Environment on a Firm's Product Innovation Performance. *Journal of Product Innovation Management*. https://doi.org/10.1111/jpim.12149

747453

- Dulai, R., Shunmugam, S. R., Veasey, R. A., Patel, N., Sugihara, C., & Furniss, S. S. (2020). An Economic Evaluation of an Advanced Video Conferencing System for Cardiac Multidisciplinary Team Meetings. *International Journal of Clinical Practice*. <u>https://doi.org/10.1111/ijcp.13562</u>
- Dunn, O. J. (1964, 1964/08/01). Multiple Comparisons Using Rank Sums. *Technometrics*, 6(3), 241-252. <u>https://doi.org/10.1080/00401706.1964.10490181</u>
- Duradoni, M., & Fabio, A. D. (2019). Intrapreneurial Self-Capital and Sustainable Innovative Behavior Within Organizations. *Sustainability*. <u>https://doi.org/10.3390/su11020322</u>
- Dutcher, E. G., & Rodet, C. S. (2018). Which Two Heads Are Better Than One? Uncovering the Positive Effects of Diversity in Creative Teams. *SSRN Electronic Journal*. <u>https://doi.org/10.2139/ssrn.3287255</u>
- Dutta, S., Lanvin, B., Leon, L. R., & Wunsch-Vincent, S. (2022). Global Innovation Index 2022: What is the Future of Innovation-driven Growth? W. I. P. Organization. <u>https://www.wipo.int/edocs/pubdocs/en/wipo-pub-2000-2022-en-main-report-global-innovation-index-2022-15th-edition.pdf</u>
- Dwivedi, A., & Pawsey, N. (2023). Examining the drivers of marketing innovation in SMEs. *Journal of Business Research*, 155. https://doi.org/10.1016/j.jbusres.2022.113409
- Dzimbiri, G. L., & Molefi, M. A. (2022). Model of Talent Management for Registered Nurses in Malawian Public Hospitals. *SA Journal of Industrial Psychology*. <u>https://doi.org/10.4102/sajip.v48i0.1953</u>
- Eddleston, K. A., & Mulki, J. P. (2015). Toward Understanding Remote Workers' Management of Work–Family Boundaries: The Complexity of Workplace

Embeddedness. *Group & Organization Management*. https://doi.org/10.1177/1059601115619548

- Edgar, F., Blaker, N. M., & Everett, A. M. (2020). Gender and Job Performance: Linking the High Performance Work System With the Ability-motivationopportunity Framework. *Personnel Review*. <u>https://doi.org/10.1108/pr-10-2019-0577</u>
- Edilson, A. R., Izabela, S. R., Milena, P. S., Filho, W. L., & Anholon, R. (2022). Productivity Analysis in Work From Home Modality: An Exploratory Study Considering an Emerging Country Scenario in the COVID-19 Context. *Work*. <u>https://doi.org/10.3233/wor-211212</u>
- Edison, H., Jabangwe, R., & Abrahamsson, P. (2018). Innovation Initiatives in Large Software Companies: A Systematic Mapping Study. *Information and Software Technology*. <u>https://doi.org/10.1016/j.infsof.2017.12.007</u>
- Edmondson, A. C., & Lei, Z. (2014). Psychological Safety: The History, Renaissance, and Future of an Interpersonal Construct. *Annual Review of Organizational Psychology and Organizational Behavior*. <u>https://doi.org/10.1146/annurevorgpsych-031413-091305</u>
- Eerde, W. v., Beeftink, F. F., & Rutte, C. G. (2022). Planning New Ideas: Does Time Management Tendency Benefit Daily Creativity? *Psychology of Aesthetics Creativity and the Arts*. <u>https://doi.org/10.1037/aca0000517</u>
- Ehrnrooth, M., & Björkman, I. (2012). An Integrative HRM Process Theorization: Beyond Signalling Effects and Mutual Gains. *Journal of Management Studies*, 49(6), 1109-1135. <u>https://doi.org/https://doi.org/10.1111/j.1467-6486.2012.01055.x</u>
- Eisenbart, B., Lovallo, D., Garbuio, M., Cristofaro, M., & Dong, A. (2023). Future thinking and managers' innovative behavior: an experimental study. *Journal of Knowledge Management, 27*, 1660-1679. <u>https://doi.org/10.1108/JKM-02-2022-0102</u>
- Elsetouhi, A., Elbaz, A. M., & Soliman, M. (2022). Participative Leadership and Its Impact on Employee Innovative Behavior Through Employee Voice in Tourism SMEs: The Moderating Role of Job Autonomy. *Tourism and Hospitality Research*. <u>https://doi.org/10.1177/14673584221119371</u>
- Engelen, A., Weinekötter, L., Saeed, S., & Enke, S. (2017). The Effect of Corporate Support Programs on Employees' Innovative Behavior: A Cross - Cultural Study. *Journal of Product Innovation Management*. <u>https://doi.org/10.1111/jpim.12386</u>

- Eraso, A. G. B., & Erro-Garcés, A. (2020). Teleworking in the Context of the Covid-19 Crisis. *Sustainability*. <u>https://doi.org/10.3390/su12093662</u>
- Erle, T. M., Schmid, K., Goslar, S., & Martin, J. (2022). Emojis as Social Information in Digital Communication. *Emotion*. <u>https://doi.org/10.1037/emo0000992</u>
- Etikan, I., Musa, S. A., & Alkassim, R. S. (2016). Comparison of convenience sampling and purposive sampling. *American journal of theoretical and applied statistics*, 5(1), 1-4.
- Evans, N. (2022). Policing the Pandemic in Australia and New Zealand: Lessons for Trust and Legitimacy. *Journal of Criminological Research Policy and Practice*. <u>https://doi.org/10.1108/jcrpp-10-2022-0050</u>
- Fadhil, A. M., Kumarasamy, M. M., Jamaluddin, Z., & Mubarak, S. M. (2022). The Impacts of Leadership Style on Innovative Work Behaviour During Covid-19 Pandemic in Oman. *International Journal of Human Resource Studies*. <u>https://doi.org/10.5296/ijhrs.v12i3.20069</u>
- Fagerberg, J., & Verspagen, B. (2009). Innovation studies—The emerging structure of a new scientific field. *Research policy*, 38(2), 218-233.
- Farmer, S. M., Tierney, P., & Kung Mcintyre, K. (2003). Employee Creativity in Taiwan: An Application of Role Identity Theory. Academy of Management Journal, 46(5), 618-630. <u>https://doi.org/10.2307/30040653</u>
- Fasbender, U., Gerpott, F. H., & Unger, D. (2021). Give and Take? Knowledge Exchange Between Older and Younger Employees as a Function of Generativity and Development Striving. *Journal of Knowledge Management*. <u>https://doi.org/10.1108/jkm-11-2020-0856</u>
- Fatima, F., Oba, P., & Sony, M. (2022). Exploring Employee Well-Being During the COVID-19 Remote Work: Evidence From South Africa. *European Journal of Training and Development*. <u>https://doi.org/10.1108/ejtd-06-2022-0061</u>
- Fatonah, S., & Haryanto, A. (2022). Exploring market orientation, product innovation and competitive advantage to enhance the performance of SMEs under uncertain evens. *Uncertain Supply Chain Management, 10*(1), 161-168.
- Fawcett, B., & Hearn, J. (2004). Researching Others: Epistemology, Experience, Standpoints and Participation. *International Journal of Social Research Methodology*, 7(3), 201-218. <u>https://doi.org/10.1080/13645570210163989</u>
- Felstead, A., & Henseke, G. (2017). Assessing the Growth of Remote Working and Its Consequences for Effort, Well - being and Work - life Balance. New Technology Work and Employment. <u>https://doi.org/10.1111/ntwe.12097</u>

- Feng, C., Cooper, B., & Zhu, C. J. (2021). How and When Job Security Reduces Resistance to Change in the Context of Organizational Change. *The Journal of Applied Behavioral Science*. <u>https://doi.org/10.1177/00218863211040613</u>
- Ferguson, S., van Velzen, E., & Olechowski, A. (2023). Team and Communication Impacts of Remote Work for Complex Aerospace System Development. Systems Engineering. <u>https://doi.org/10.1002/sys.21716</u>
- Fernandez, S., & Pitts, D. W. (2011). Understanding Employee Motivation to Innovate: Evidence from Front Line Employees in United States Federal Agencies. *Australian Journal of Public Administration*, 70(2), 202-222. <u>https://doi.org/https://doi.org/10.1111/j.1467-8500.2011.00726.x</u>
- Ferrante, L. D. (2021). Transitioning Between Small Talk and Work Talk Through Discourse Markers : Evidence From a Workplace Spoken Corpus. Brno Studies in English. <u>https://doi.org/10.5817/bse2021-2-2</u>
- Ferrarini, F., & Curzi, Y. (2022). AMO-enhancing Practices, Open Innovation and Organizations' Innovation in the European Context: Testing a Mediation Model. *European Journal of Innovation Management*, 26(6). <u>https://doi.org/10.1108/ejim-01-2022-0005</u>
- Ferreira, A. I., Mach, M., Martinez, L. F., & Miraglia, M. (2022). Sickness Presenteeism in the Aftermath of COVID-19: Is Presenteeism Remote-Work Behavior the New (Ab)normal? *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.748053</u>
- Ferreira, J., Coelho, A., & Moutinho, L. (2020). Dynamic capabilities, creativity and innovation capability and their impact on competitive advantage and firm performance: The moderating role of entrepreneurial orientation. *Technovation*, 92, 102061.
- Field, S., & Derksen, M. (2020). Experimenter as Automaton; Experimenter as Human: Exploring the Position of the Researcher in Scientific Research. *European Journal for Philosophy of Science*, 11(1). <u>https://doi.org/10.1007/s13194-020-00324-7</u>
- Fink, A., Grabner, R. H., Benedek, M., Reishofer, G., Hauswirth, V., Fally, M., Neuper, C., Ebner, F., & Neubauer, A. C. (2008). The Creative Brain: Investigation of Brain Activity During Creative Problem Solving by Means of EEG and FMRI. *Human Brain Mapping*. https://doi.org/10.1002/hbm.20538
- Fink, E. L. (2009). The FAQs on Data Transformation. *Communication Monographs*, 76(4), 379-397. <u>https://doi.org/10.1080/03637750903310352</u>
- Fisher, R. A. (1925). *Statistical methods for research workers* (1st ed.). Oliver and Boyd.
- Flannery, H., McParland, J., Baker, H., Hitchcock, M., & Portnoy, S. (2022). Morning Connections: How Do You Support Hospital Staff Working Remotely During a

Global Pandemic Without Providing 'Staff Support'? *Clinical Child Psychology* and *Psychiatry*. <u>https://doi.org/10.1177/13591045221119003</u>

- Forbes, D., & Gedera, D. S. P. (2019). From Confounded to Common Ground: Misunderstandings Between Tertiary Teachers and Students in Online Discussions. Australasian Journal of Educational Technology. <u>https://doi.org/10.14742/ajet.3595</u>
- Forbes, M. (2022). Thematic analysis: A practical guide. *Evaluation Journal of Australasia*, 22, 132-135. https://doi.org/https://doi.org/10.1177/1035719X211058251
- Ford, D., Storey, M.-A., Zimmermann, T., Bird, C., Jaffe, S., Maddila, C., Butler, J. L., Houck, B., & Nagappan, N. (2020). A Tale of Two Cities: Software Developers Working From Home During the COVID-19 Pandemic. <u>https://doi.org/10.48550/arxiv.2008.11147</u>
- Foss, N. J., Minbaeva, D., Pedersen, T., & Reinholt, M. (2009). Encouraging Knowledge Sharing Among Employees: How Job Design Matters. *Human Resource Management*. <u>https://doi.org/10.1002/hrm.20320</u>
- Franklin, B. D., & Thomas, E. J. (2022). Replicating and Publishing Research in Different Countries and Different Settings: Advice for Authors. *BMJ Quality & Safety*. <u>https://doi.org/10.1136/bmjqs-2021-014431</u>
- Fu, N., Flood, P. C., Bosak, J., Morris, T., & O'Regan, P. (2015). How Do High Performance Work Systems Influence Organizational Innovation in Professional Service Firms? *Employee Relations*. <u>https://doi.org/10.1108/er-10-2013-0155</u>
- Fukumura, Y. E., Schott, J. M., Lucas, G. M., Becerik-Gerber, B., & Roll, S. C. (2021). Negotiating time and space when working from home: experiences during COVID-19. OTJR: Occupation, Participation and Health, 41(4), 223-231.
- Fussell, S. R., Setlock, L. D., Yang, J., Ou, J., Mauer, E., & Kramer, A. D. I. (2004). Gestures Over Video Streams to Support Remote Collaboration on Physical Tasks. *Human-Computer Interaction*. https://doi.org/10.1207/s15327051hci1903_3
- Gajendran, R. S., & Harrison, D. A. (2007). The good, the bad, and the unknown about telecommuting: Meta-analysis of psychological mediators and individual consequences. *Journal of Applied psychology*, *92*(6), 1524.
- Galanti, T., Guidetti, G., Mazzei, E., Zappala, S., & Toscano, F. (2021). Work From Home During the COVID-19 Outbreak. *Journal of Occupational and Environmental Medicine*. <u>https://doi.org/10.1097/jom.00000000002236</u>
- Ganguly, A., Talukdar, A., & Chatterjee, D. (2019). Evaluating the Role of Social Capital, Tacit Knowledge Sharing, Knowledge Quality and Reciprocity in Determining Innovation Capability of an Organization. *Journal of Knowledge Management*. <u>https://doi.org/10.1108/jkm-03-2018-0190</u>

- Garlatti Costa, G., Bortoluzzi, G., & Černe, M. (2022). Can innovative work behaviour spur creativity while working remotely? The role of work-home conflict and social isolation. *Management Research Review*.
- Gastwirth, J. L., Gel, Y. R., & Miao, W. (2009). The Impact of Levene's Test of Equality of Variances on Statistical Theory and Practice. *Statistical Science*, 24(3). <u>https://doi.org/10.1214/09-sts301</u>
- George, G., Lakhani, K. R., & Puranam, P. (2020). What has changed? The Impact of Covid Pandemic on the Technology and Innovation Management Research Agenda. *Journal of Management Studies*, 57(8), 1754-1758. <u>https://doi.org/https://doi.org/10.1111/joms.12634</u>
- Gerald, B. (2018). A Brief Review of Independent, Dependent and One Sample T-Test. International Journal of Applied Mathematics and Theoretical Physics, 4(2), 50. https://doi.org/10.11648/j.ijamtp.20180402.13
- Gerards, R., Wetten, S. v., & Sambeek, C. v. (2020). New Ways of Working and Intrapreneurial Behaviour: The Mediating Role of Transformational Leadership and Social Interaction. *Review of Managerial Science*. <u>https://doi.org/10.1007/s11846-020-00412-1</u>
- Getz, I., & Robinson, A. G. (2003, 01 / 01 /). Innovate or die: Is that a fact? [Article]. *Creativity and innovation management*, 12(3), 130-136. <u>https://doi.org/10.1111/1467-8691.00276</u>
- Ghasemi, M., Nejad, M. G., & Aghaei, I. (2020). Knowledge Management Orientation and Operational Performance Relationship in Medical Tourism (Overview of the Model Performance in the COVID-19 Pandemic and Post-Pandemic Era). *Health Services Management Research*. <u>https://doi.org/10.1177/0951484820971438</u>
- Ghauri, P. N. (2020). *Research methods in business studies / Pervez Ghauri, Kjell Grønhaug, Roger Strange* (Fifth edition. ed.). Cambridge University Press.
- Ghosh, K., Sinha, S., & Sharma, D. (2023). Virtual Fun in a Virtual Workplace: Employee Socialization for "Work From Home". *Benchmarking an International Journal*. <u>https://doi.org/10.1108/bij-11-2021-0637</u>
- Giannetti, C., & Madia, M. (2013, 03 / 01 /). Work arrangements and firm innovation: Is there any relationship? [Article]. *Cambridge Journal of Economics*, 37(2), 273-297. <u>https://doi.org/10.1093/cje/bes067</u>
- Gidel, T., Tucker, A., Fujita, S., Moulin, C., Sugawara, K., Suganuma, T., Kaeri, Y., & Shiratori, N. (2020). Interaction Model and Respect of Rules to Enhance Collaborative Brainstorming Results. *Advances in Science Technology and Engineering Systems Journal*. <u>https://doi.org/10.25046/aj050262</u>

- Giebels, E., Reuver, R. S. M. d., Rispens, S., & Ufkes, E. G. (2016). The Critical Roles of Task Conflict and Job Autonomy in the Relationship Between Proactive Personalities and Innovative Employee Behavior. *The Journal of Applied Behavioral Science*. <u>https://doi.org/10.1177/0021886316648774</u>
- Gill, P., Stewart, K., Treasure, E., & Chadwick, B. (2008, 03/22/online). Methods of data collection in qualitative research: interviews and focus groups. *Bdj*, 204, 291. <u>https://doi.org/10.1038/bdj.2008.192</u>
- Gillet, N., Huyghebaert-Zouaghi, T., Austin, S., Fernet, C., & Morin, A. J. S. (2021). Remote Working: A Double-Edged Sword for Workers' Personal and Professional Well-Being. *Journal of Management & Organization*. <u>https://doi.org/10.1017/jmo.2021.71</u>
- Gilmartin, H., Connelly, B., Hebbe, A., Battaglia, C., & Kwan, B. M. (2020). Assessing the Impact of Remote Work During COVID-19 on Clinical and Translational Scientists and Staff in Colorado. *Journal of Clinical and Translational Science*. <u>https://doi.org/10.1017/cts.2020.570</u>
- Gilson, L. L., Maynard, M. T., Jones Young, N. C., Vartiainen, M., & Hakonen, M. (2015). Virtual teams research: 10 years, 10 themes, and 10 opportunities. *Journal of Management*, 41(5), 1313-1337.
- Gkontelos, A., Vaiopoulou, J., & Stamovlasis, D. (2022). Teachers' Innovative Work Behavior Scale: Psychometric Properties of the Greek Version and Measurement Invariance Across Genders. *Social Sciences*, 11(7), 306. <u>https://doi.org/10.3390/socsci11070306</u>
- Godart, O. N., Görg, H., & Hanley, A. (2017). Trust-Based Work Time and Innovation: Evidence from Firm-Level Data [Article]. *Industrial and Labor Relations Review*, 70(4), 894-918. <u>https://doi.org/10.1177/0019793916676259</u>
- Golden, T. D., Veiga, J. F., & Dino, R. N. (2008). The impact of professional isolation on teleworker job performance and turnover intentions: does time spent teleworking, interacting face-to-face, or having access to communicationenhancing technology matter? *Journal of Applied psychology*, 93(6), 1412.
- Golden, T. D., Veiga, J. F., & Simsek, Z. (2006). Telecommuting's differential impact on work-family conflict: Is there no place like home? *Journal of Applied psychology*, 91(6), 1340.
- Gong, B., Tobias, P., & Young-Bristol, J. (2022). Leveraging Resources to Improve Supervisors' Vision in the Remote Workplace. *Management Research Review*. <u>https://doi.org/10.1108/mrr-12-2021-0916</u>
- Gorokhova, A. E., Eskerkhanova, L., & Shishkin, A. (2021). The Effect of Remote Work on the Successful Development of Engineering Company. Nexo Revista Científica. <u>https://doi.org/10.5377/nexo.v34i06.13135</u>

- Grant, A. M., Fried, Y., & Juillerat, T. (2011). Work matters: Job design in classic and contemporary perspectives.
- Grant, C., Wallace, L., & Spurgeon, P. (2013). An Exploration of the Psychological Factors Affecting Remote E - worker's Job Effectiveness, Well - being and Work - life Balance. *Employee Relations*. <u>https://doi.org/10.1108/er-08-2012-0059</u>
- Grant, C. A., & Osanloo, A. F. (2014). Understanding, Selecting, and Integrating a Theoretical Framework in Dissertation Research: Creating the Blueprint for Your "House". Administrative Issues Journal Education Practice and Research. <u>https://doi.org/10.5929/2014.4.2.9</u>
- Gravetter, F. J., & Forzano, L.-A. B. (2018). *Research methods for the behavioral sciences*. Cengage learning.
- Gray, P. S. (2007). The research imagination : an introduction to qualitative and quantitative methods / Paul S. Gray [and others]. Cambridge University Press.
- Green, C. A., Duan, N., Gibbons, R. D., Hoagwood, K., Palinkas, L. A., & Wisdom, J. P. (2014). Approaches to Mixed Methods Dissemination and Implementation Research: Methods, Strengths, Caveats, and Opportunities. *Administration and Policy in Mental Health and Mental Health Services Research*. https://doi.org/10.1007/s10488-014-0552-6
- Gruber, J., Hargittai, E., & Nguyen, M. H. (2022). The Value of Face-to-Face Communication in the Digital World: What People Miss About in-Person Interactions When Those Are Limited. *Studies in Communication Sciences*. <u>https://doi.org/10.24434/j.scoms.2022.03.3340</u>
- Guaralda, M., Hearn, G., Foth, M., Yigitcanlar, T., Mayere, S., & Law, L. (2020). Towards Australian Regional Turnaround: Insights Into Sustainably Accommodating Post-Pandemic Urban Growth in Regional Towns and Cities. Sustainability. <u>https://doi.org/10.3390/su122410492</u>
- Gudžinskienė, V., & Mačiuikienė, D. (2022). Using Distance Communication and Collaboration Tools for General Social Services. *Society Integration Education Proceedings of the International Scientific Conference*. https://doi.org/10.17770/sie2022vol1.6819
- Guillén, L., & Kunze, F. (2019). When Age Does Not Harm Innovative Behavior and Perceptions of Competence: Testing Interdepartmental Collaboration as a Social Buffer. *Human Resource Management*. <u>https://doi.org/10.1002/hrm.21953</u>
- Gumusluoğlu, L., & İlsev, A. (2009). Transformational Leadership, Creativity, and Organizational Innovation. *Journal of Business Research*, 62(4), 461-473. <u>https://doi.org/10.1016/j.jbusres.2007.07.032</u>
- Gundry, L. K., Muñoz-Fernández, Á., Ofstein, L. F., & Ortega-Egea, T. (2015). Innovating in Organizations: A Model of Climate Components Facilitating the

Creation of New Value. *Creativity and innovation management*. https://doi.org/10.1111/caim.12116

- Gurran, N., Forsyth, A., Darcy, M. D., Searle, G., Buckle, C., & Zou, S. (2021). Population Growth, Regional Connectivity, and City Planning — International Lessons for Australian Practice. *Ahuri Final Report*. <u>https://doi.org/10.18408/ahuri7322301</u>
- Haddad, H., Khatib, N. E., & Ashaal, A. (2023). The Nexus Between Extrinsic Motivation and Employees' Retention; Do Compensation Packages and Flexible Working Hours Matter? *Open Journal of Business and Management*. <u>https://doi.org/10.4236/ojbm.2023.112028</u>
- Hafermalz, E., & Riemer, K. (2020). Productive and Connected While Working From Home: What Client-Facing Remote Workers Can Learn From Telenurses About 'Belonging Through Technology'. *European Journal of Information Systems*. <u>https://doi.org/10.1080/0960085x.2020.1841572</u>
- Haider, S., Jabeen, S., & Ahmad, J. (2018). Moderated Mediation Between Work Life Balance and Employee Job Performance: The Role of Psychological Wellbeing and Satisfaction With Coworkers. *Journal of Work and Organizational Psychology*. <u>https://doi.org/10.5093/jwop2018a4</u>
- Hajjami, O., & Crocco, O. S. (2023). Evolving Approaches to Employee Engagement: Comparing Antecedents in Remote Work and Traditional Workplaces. *European Journal of Training and Development*. <u>https://doi.org/10.1108/ejtd-10-2022-0103</u>
- Hallebone, E., & Priest, J. (2008). Business and management research: paradigms and practices. Macmillan International Higher Education.
- Hammedi, W., Riel, A. C. R. v., & Sasovova, Z. (2011). Antecedents and Consequences of Reflexivity in New Product Idea Screening^{*}. Journal of Product Innovation Management. <u>https://doi.org/10.1111/j.1540-5885.2011.00831.x</u>
- Hammond, M., & Wellington, J. (2021). *Research methods : the key concepts* (2nd edition. ed.). Routledge.
- Hammond, M. M., Neff, N. L., Farr, J. L., Schwall, A. R., & Zhao, X. (2011). Predictors of individual-level innovation at work: A meta-analysis. *Psychology* of Aesthetics, Creativity, and the Arts, 5(1), 90.
- Hancock, G. R., Mueller, R. O., & Stapleton, L. M. (2010). *The reviewer's guide to quantitative methods in the social sciences*. Routledge.
- Handke, L., Costa, P., Klonek, F. E., O'Neill, T. A., & Parker, S. K. (2020). Team Perceived Virtuality: An Emergent State Perspective. *European Journal of Work* and Organizational Psychology. <u>https://doi.org/10.1080/1359432x.2020.1806921</u>

- Harun, M. A. B., Yunhendri, R. B., Shaharudin, M. R. B., & Kanapathipillai, K. (2022). The Impact of Self-Efficacy and Creative Process Engagement on Innovative Work Behaviour in the Telecommunication Industry in Malaysia. *European Journal of Economic and Financial Research*. <u>https://doi.org/10.46827/ejefr.v6i2.1273</u>
- Hasan, I., & Tucci, C. L. (2010). The innovation–economic growth nexus: Global evidence. *Research policy*, 39(10), 1264-1276.
- Hashim, A., Othman, B., Mohammad, A. J., & Agala, S. R. (2022). Investigating the Role of Leadership and Organisational Culture in Fostering Innovation. *International Journal of Health Sciences*. <u>https://doi.org/10.53730/ijhs.v6ns5.9145</u>
- Hassan, M. U., Nawaz, H., & Faiz, A. u. H. (2021). Revealing the Effects of Teachers Stimulating Technological Inclusion on University Students Educational Gains: A Comparative Study. *Humanities & Social Sciences Reviews*, 9(3), 986-999. <u>https://doi.org/10.18510/hssr.2021.9397</u>
- Haukipuro, L., Väinämö, S., Virta, V., & Perälä-Heape, M. (2023). Key Aspects of Establishing Research, Knowledge, and Innovation - based Hubs as Part of the Local Innovation Ecosystem. *R and D Management*. <u>https://doi.org/10.1111/radm.12584</u>
- Hayes, S. W., Priestley, J. L., Moore, B. A., & Ray, H. E. (2021). Perceived Stress, Work-Related Burnout, and Working From Home Before and During COVID-19: An Examination of Workers in the United States. SAGE Open. <u>https://doi.org/10.1177/21582440211058193</u>
- Haynes, B., Suckley, L., & Nunnington, N. (2017). Workplace Productivity and Office Type. Journal of Corporate Real Estate. <u>https://doi.org/10.1108/jcre-11-2016-0037</u>
- He, P., Wu, T.-J., Zhao, H., & Yang, Y. (2019). How to Motivate Employees for Sustained Innovation Behavior in Job Stressors? A Cross-Level Analysis of Organizational Innovation Climate. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph16234608</u>
- Hecker, I., Spaulding, S., & Kuehn, D. (2021). Digital Skills and Older Workers: Supporting Success in Training and Employment in a Digital World. *Urban Institute*.
- Heilman, M. E., & Eagly, A. H. (2008). Gender Stereotypes Are Alive, Well, and Busy Producing Workplace Discrimination. *Industrial and Organizational Psychology*. <u>https://doi.org/10.1111/j.1754-9434.2008.00072.x</u>
- Helmy, I., Azizah, S. N., Purnomo, J., & Fitriani, N. (2023). The Effect of Transformational Leadership on Innovative Work Behavior in SMEs: Test of a

Mediating Model. *Journal of International Conference Proceedings*. https://doi.org/10.32535/jicp.v6i1.2245

- Henke, J. B., Jones, S. K., & O'Neill, T. A. (2022). Skills and Abilities to Thrive in Remote Work: What Have We Learned. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2022.893895</u>
- Hennink, M., & Kaiser, B. N. (2022). Sample sizes for saturation in qualitative research: A systematic review of empirical tests. *Social science & medicine*, 292, 114523.
- Hertel, G., Geister, S., & Konradt, U. (2005). Managing virtual teams: A review of current empirical research. *Human resource management review*, 15(1), 69-95.
- Hinds, P., & Bailey, D. (2003). Out of Sight, Out of Sync: Understanding Conflict in Distributed Teams. Organization Science. <u>https://doi.org/10.1287/orsc.14.6.615.24872</u>
- Hirst, G., Van Knippenberg, D., & Zhou, J. (2009). A cross-level perspective on employee creativity: Goal orientation, team learning behavior, and individual creativity. *Academy of Management Journal*, 52(2), 280-293.
- Hitt, M. A., Hoskisson, R. E., & Kim, H. (1997). International Diversification: Effects on Innovation and Firm Performance in Product-Diversified Firms [Article]. *Academy of Management Journal*, 40(4), 767-798. <u>https://doi.org/10.2307/256948</u>
- Hoeven, C. L. t., & Zoonen, W. v. (2015). Flexible Work Designs and Employee
 Well being: Examining the Effects of Resources and Demands. New
 Technology Work and Employment. <u>https://doi.org/10.1111/ntwe.12052</u>
- Honkaniemi, L., Lehtonen, M. J., & Hasu, M. (2015). Well-Being and Innovativeness: Motivational Trigger Points for Mutual Enhancement. *European Journal of Training and Development*. <u>https://doi.org/10.1108/ejtd-11-2014-0078</u>
- Hopkins, J. D., & McKay, J. (2019). Investigating 'Anywhere Working' as a Mechanism for Alleviating Traffic Congestion in Smart Cities. *Technological Forecasting and Social Change*. <u>https://doi.org/10.1016/j.techfore.2018.07.032</u>
- Hopkins, J. L., & Bardoel, A. (2023). The Future Is Hybrid: How Organisations Are Designing and Supporting Sustainable Hybrid Work Models in Post-Pandemic Australia. *Sustainability*, 15(4), 3086. <u>https://doi.org/10.3390/su15043086</u>
- Houghton, K., Foth, M., & Hearn, G. (2018). Working From the Other Office: Trialling Co-Working Spaces for Public Servants. *Australian Journal of Public Administration*. <u>https://doi.org/10.1111/1467-8500.12317</u>
- Howell, J. M., Shea, C. M., & Higgins, C. A. (2005). Champions of product innovations: defining, developing, and validating a measure of champion behavior. *Journal of business venturing*, 20(5), 641-661.

- Howes, L. M. (2015). Developing the Methodology for an Applied, Interdisciplinary Research Project: Documenting the Journey Toward Philosophical Clarity. *Journal of mixed methods research*, 11(4), 450-468. <u>https://doi.org/10.1177/1558689815622018</u>
- Høyrup, S. (2010). Employee-Driven Innovation and Workplace Learning: Basic Concepts, Approaches and Themes. *Transfer European Review of Labour and Research*, 16(2), 143-154. <u>https://doi.org/10.1177/1024258910364102</u>
- Hu, B., & Zhao, Y. (2016). Creative Self-Efficacy Mediates the Relationship Between Knowledge Sharing and Employee Innovation. Social Behavior and Personality an International Journal. <u>https://doi.org/10.2224/sbp.2016.44.5.815</u>
- Hu, J., Erdoğan, B., Jiang, K., Bauer, T. N., & Liu, S. (2018). Leader Humility and Team Creativity: The Role of Team Information Sharing, Psychological Safety, and Power Distance. *Journal of Applied psychology*. <u>https://doi.org/10.1037/ap10000277</u>
- Hu, X., & Subramony, M. (2022). Understanding the impact of COVID 19 pandemic on teleworkers' experiences of perceived threat and professional isolation: The moderating role of friendship. *Stress & amp; Health: Journal of the International Society for the Investigation of Stress, 38*(5), 927-939. <u>https://doi.org/10.1002/smi.3146</u>
- Hudson, S., Bryson, D., & Michelotti, M. (2017, Spring2017). Individuals' Assessment of Corporate Social Performance, Person-Organization Values and Goals Fit, Job Satisfaction and Turnover Intentions [Article]. *Industrial Relations*, 72(2), 322-344. <u>https://doi.org/10.7202/1040403ar</u>
- Hülsheger, U. R., Anderson, N., & Salgado, J. F. (2009). Team-level predictors of innovation at work: a comprehensive meta-analysis spanning three decades of research. *Journal of Applied psychology*, 94(5), 1128.
- Huo, D., & Photchanachan, S. (2021). Influence of Intergenerational and Peer-to-Peer Knowledge Sharing on Academics' Creativity. *International Business Research*. <u>https://doi.org/10.5539/ibr.v14n6p91</u>
- Hur, H., & Perry, J. L. (2019). Job Security Rule Changes and Employee Organizational Commitment. *Review of Public Personnel Administration*. <u>https://doi.org/10.1177/0734371x19842622</u>
- Hurps, D., Hurps, M., Spencer, A., Jafer, M., & Conyngham, P. (2018). 2018 Startup Muster Annual Report. <u>https://www.startupmuster.com/3d0391a0-3f9f-4738-903e-083bf2c7071d</u>
- Hwang, E., Smith, R. N., Byers, V. T., Dickerson, S., McAlister-Shields, L.,Onwuegbuzie, A. J., & Benge, C. L. (2015). Doctoral Students' PerceivedBarriers That Slow the Progress Toward Completing a Doctoral Dissertation: A

Mixed Analysis. *Journal of Educational Issues*. https://doi.org/10.5296/jei.v1i1.7703

- Imran, R., Saeed, T., Anis-Ul-Haq, M., & Fatima, A. (2010). Organizational climate as a predictor of innovative work behavior. *African Journal of Business Management*, 4(15), 3337.
- In, J., & Lee, S. (2017, Jun). Statistical data presentation. *Korean J Anesthesiol*, 70(3), 267-276. <u>https://doi.org/10.4097/kjae.2017.70.3.267</u>
- Inceoglu, I., Segers, J., & Bartram, D. (2011). Age related Differences in Work Motivation. *Journal of Occupational and Organizational Psychology*. <u>https://doi.org/10.1111/j.2044-8325.2011.02035.x</u>

IPAustralia. (2022). 2022 Australian Intellectual Property Report.

- Irawanto, D. W., Novianti, K. R., & Roz, K. (2021). Work From Home: Measuring Satisfaction Between Work–Life Balance and Work Stress During the COVID-19 Pandemic in Indonesia. *Economies*. <u>https://doi.org/10.3390/economies9030096</u>
- Jafari, M., & Zarghami, H. R. (2017). Effect of TRIZ on Enhancing Employees' Creativity and Innovation. *Aircraft Engineering and Aerospace Technology*, 89(6), 853-861. <u>https://doi.org/10.1108/aeat-01-2016-0004</u>
- Jakobsen, J. C., Gluud, C., Wetterslev, J., & Winkel, P. (2017). When and How Should Multiple Imputation Be Used for Handling Missing Data in Randomised Clinical Trials – A Practical Guide With Flowcharts. *BMC Medical Research Methodology*, 17(1). <u>https://doi.org/10.1186/s12874-017-0442-1</u>
- Jamal, M. T., Anwar, I., Khan, N. M., & Singh, M. (2023). An Empirical Analysis of Telecommuters: Their Level of Satisfaction, Commitment and Loyalty. *Management and Labour Studies*. <u>https://doi.org/10.1177/0258042x221148073</u>
- Jamieson, S. (2004). Likert scales: How to (ab) use them? *Medical education, 38*(12), 1217-1218.
- Janssen, O. (2000). Job demands, perceptions of effort--reward fairness and innovative work behaviour [Article]. *Journal of Occupational & Organizational Psychology*, 73(3), 287-302. <u>http://wallaby.vu.edu.au:2048/login?url=http://search.ebscohost.com/login.aspx?</u> <u>direct=true&db=bth&AN=3595672&site=eds-live</u>
- Janssen, O. (2005). The joint impact of perceived influence and supervisor supportiveness on employee innovative behaviour. *Journal of Occupational and Organizational Psychology*, 78(4), 573-579.
- Jarvenpaa, S. L., & Leidner, D. E. (2006). Communication and Trust in Global Virtual Teams. *Journal of Computer-Mediated Communication*. https://doi.org/10.1111/j.1083-6101.1998.tb00080.x

- Javed, B., Abdullah, I., Zaffar, M. A., Haque, A. u., & Rubab, U. (2018). Inclusive Leadership and Innovative Work Behavior: The Role of Psychological Empowerment. *Journal of Management & Organization*. <u>https://doi.org/10.1017/jmo.2018.50</u>
- Javed, B., Y. A. Rawwas, M., Khandai, S., Shahid, K., & Tayyeb, H. H. (2018). Ethical Leadership, Trust in Leader and Creativity: The Mediated Mechanism and an Interacting Effect. *Journal of Management & Organization*. <u>https://doi.org/10.1017/jmo.2017.56</u>
- Jaworski, C., Ravichandran, S., Karpinski, A. C., & Singh, S. (2018). The effects of training satisfaction, employee benefits, and incentives on part-time employees' commitment. *International Journal of Hospitality Management*, 74, 1-12.
- Jebali, D., & Meschitti, V. (2020). HRM as a Catalyst for Innovation in Start ups. Employee Relations. <u>https://doi.org/10.1108/er-03-2020-0140</u>
- Jensen, N. D., Lyons, E., Chebelyon, E., Bras, R. L., & Gomes, C. P. (2020). Conspicuous Monitoring and Remote Work. *Journal of Economic Behavior & Organization*. <u>https://doi.org/10.1016/j.jebo.2020.05.010</u>
- Jetha, A., Shamaee, A., Bonaccio, S., Gignac, M. A. M., Tucker, L. B., Tompa, E., Bültmann, U., Norman, C. D., Banks, C. G., & Smith, P. (2021). Fragmentation in the Future of Work: A Horizon Scan Examining the Impact of the Changing Nature of Work on Workers Experiencing Vulnerability. *American Journal of Industrial Medicine*. <u>https://doi.org/10.1002/ajim.23262</u>
- Jimenez-Jimenez, D., & Sanz-Valle, R. (2008). Could HRM support organizational innovation? [Article]. *International Journal of Human Resource Management*, 19(7), 1208-1221. <u>https://doi.org/10.1080/09585190802109952</u>
- Jindal, A. K., Singh, G., & Pandya, K. (2015). Qualitative Research in Medicine An Art to Be Nurtured. *Medical Journal Armed Forces India*, 71(4), 369-372. https://doi.org/10.1016/j.mjafi.2014.11.006
- Jo, A., Bm, G. K., & George, F. (2016). Performances of Several Univariate Tests of Normality: An Empirical Study. *Journal of Biometrics & Biostatistics*, 07(04). <u>https://doi.org/10.4172/2155-6180.1000322</u>
- Johari, A. B., Abdul Wahat, N. W., & Zaremohzzabieh, Z. (2021). Innovative Work Behavior among Teachers in Malaysia: The Effects of Teamwork, Principal Support, and Humor. *Asian Journal of University Education*, 17(2), 72-84.
- Johari, A. B., Wahat, N. W. A., & Idris, K. (2021). Creating Fun Time in Teaching-Learning Experiences Through Teachers' Teamwork, Humor, and Innovative Behavior. *International Journal of Human Resource Studies*. <u>https://doi.org/10.5296/ijhrs.v11i4s.19229</u>

- Johnson, D. H. (1999). The insignificance of statistical significance testing. *The journal* of wildlife management, 763-772.
- Johnson, H.-A. M., Worthington, R., Gredecki, N., & Wilks-riley, F. R. (2016). The Impact of Forensic Work on Home Life: The Role of Emotional Labour, Segmentation/Integration and Social Support. *Journal of Forensic Practice*. <u>https://doi.org/10.1108/jfp-05-2015-0029</u>
- Johnson, R. B., & Onwuegbuzie, A. J. (2004). Mixed Methods Research: A Research Paradigm Whose Time Has Come. *Educational Researcher*, *33*(7), 14-26. <u>https://doi.org/10.3102/0013189x033007014</u>
- Johnson, R. B., Onwuegbuzie, A. J., & Turner, L. A. (2007). Toward a definition of mixed methods research. *Journal of mixed methods research*, 1(2), 112-133.
- Jonker, J., & Pennink, B. (2010). *The essence of research methodology: A concise guide for master and PhD students in management science*. Springer Science & Business Media.
- Joung, H.-W., Choi, E.-K., & Taylor, J. (2018). Investigating Differences in Job-Related Attitudes Between Full-Time and Part-Time Employees in the Foodservice Industry. *International Journal of Contemporary Hospitality Management*. <u>https://doi.org/10.1108/ijchm-03-2016-0129</u>
- Juan, A. M.-G., & Juan Martinez, T. (2016). Deconstructing AMO framework: a systematic review [article]. *Intangible Capital*(4), 1040. <u>https://doi.org/10.3926/ic.838</u>
- Juchnowicz, M., & Kinowska, H. (2021). Employee Well-Being and Digital Work During the COVID-19 Pandemic. *Information*. <u>https://doi.org/10.3390/info12080293</u>
- Kähkönen, T. (2023). Remote Work During the COVID-19 Pandemic: Identification Of working Life Impacts, Employees' Data Protection Abilities and Trust Outcomes. *Journal of Organizational Change Management*. <u>https://doi.org/10.1108/jocm-06-2022-0179</u>
- Kalmi, P., & Kauhanen, A. (2008). Workplace Innovations and Employee Outcomes: Evidence From Finland. *Industrial Relations a Journal of Economy and Society*. <u>https://doi.org/10.1111/j.1468-232x.2008.00528.x</u>
- Kanter, R. M. (1988). When a Thousand Flowers Bloom: Structural, Collective, and Social Conditions for Innovation in Organizations [Article]. *Research in Organizational Behavior, 10*, 169. <u>http://wallaby.vu.edu.au:2048/login?url=http://search.ebscohost.com/login.aspx?</u> <u>direct=true&db=bth&AN=6815426&site=eds-live</u>
- Katrahmani, A., Ansah, A., Ch, N., Burmeister, J., Kun, A. L., Mills, C., Shaer, O., & Lee, J. D. (2022). The Instability of Creativity Over Weeks Through the Lens of

Distributional Models. *Proceedings of the Human Factors and Ergonomics* Society Annual Meeting. <u>https://doi.org/10.1177/1071181322661553</u>

- Kauhanen, M., & Nätti, J. (2015). Involuntary temporary and part-time work, job quality and well-being at work. *Social Indicators Research*, 120, 783-799.
- Keinänen, M. (2015). Taking Your Mind for a Walk: A Qualitative Investigation of Walking and Thinking Among Nine Norwegian Academics. *Higher Education*. <u>https://doi.org/10.1007/s10734-015-9926-2</u>
- Kershaw, M., Lupien, S. P., & Scheid, J. L. (2021). Impact of Web-Based Meeting Platform Usage on Overall Well-Being Among Higher Education Employees. *European Journal of Investigation in Health Psychology and Education*. <u>https://doi.org/10.3390/ejihpe11020028</u>
- Khalili, A. (2016). Linking Transformational Leadership, Creativity, Innovation, and Innovation-Supportive Climate. *Management Decision*. <u>https://doi.org/10.1108/md-03-2016-0196</u>
- Khan, M. A., Ismail, F. B., Hussain, A., & Alghazali, B. (2020, 01 / 01 /). The Interplay of Leadership Styles, Innovative Work Behavior, Organizational Culture, and Organizational Citizenship Behavior [Article]. SAGE Open, 10(1). <u>https://doi.org/10.1177/2158244019898264</u>
- Khan, M. S., Mubarik, M. S., Islam, T., Rehman, A., Ahmed, S. H., Khan, E. A., & Sohail, F. (2021). How Servant Leadership Triggers Innovative Work Behavior: Exploring the Sequential Mediating Role of Psychological Empowerment and Job Crafting. *European Journal of Innovation Management*. https://doi.org/10.1108/ejim-09-2020-0367
- Kim, J.-E., Choi, D. S., Sung, C.-S., & Park, J. H. (2018). The Role of Problem Solving Ability on Innovative Behavior and Opportunity Recognition in University Students. *Journal of Open Innovation Technology Market and Complexity*. <u>https://doi.org/10.1186/s40852-018-0085-4</u>
- Kim, J., Erdem, M., Byun, J.-W., & Jeong, H.-Y. (2011). Training Soft Skills via E learning: International Chain Hotels. *International Journal of Contemporary Hospitality Management*. <u>https://doi.org/10.1108/09596111111153457</u>
- Kim, K. W., & Cho, Y.-H. (2020). The Moderating Effect of Managerial Roles on Job Stress and Satisfaction by Employees' Employment Type. *International Journal* of Environmental Research and Public Health. https://doi.org/10.3390/ijerph17218259
- Kimberlin, C. L., & Winterstein, A. G. (2008). Validity and Reliability of Measurement Instruments Used in Research. *American Journal of Health-System Pharmacy*. <u>https://doi.org/10.2146/ajhp070364</u>
- King, J. E. (2008). Binary logistic regression. In J. W. Osborne (Ed.), *Best practices in quantitative methods* (pp. 358-384). Sage Publications.

- Kingma, S. (2019). New ways of working (NWW): work space and cultural change in virtualizing organizations. *Culture and Organization*, 25(5), 383-406.
- Klaeijsen, A., Vermeulen, M., & Martens, R. (2017). Teachers' Innovative Behaviour: The Importance of Basic Psychological Need Satisfaction, Intrinsic Motivation, and Occupational Self-Efficacy. *Scandinavian Journal of Educational Research*. <u>https://doi.org/10.1080/00313831.2017.1306803</u>
- Klerk, J. J. d., Joubert, M., & Mosca, H. F. (2021). Is Working From Home the New Workplace Panacea? Lessons From the COVID-19 Pandemic for the Future World of Work. SA Journal of Industrial Psychology. <u>https://doi.org/10.4102/sajip.v47i0.1883</u>
- Kleysen, R. F., & Street, C. T. (2001, 09 / 01 /). Toward a multi-dimensional measure of individual innovative behavior [Article]. *Journal of Intellectual Capital*, 2(3), 284-296. <u>https://doi.org/10.1108/EUM000000005660</u>
- Klindžić, M., & Marić, M. (2019). Flexible Work Arrangements and Organizational Performance – The Difference Between Employee and Employer-Driven Practices. Drustvena Istrazivanja. <u>https://doi.org/10.5559/di.28.1.05</u>
- Kline, S. J., & Rosenberg, N. (2010). An overview of innovation. In Studies On Science And The Innovation Process: Selected Works of Nathan Rosenberg (pp. 173-203). World Scientific.
- Klopotan, I., Mjeda, T., & Kurečić, P. (2018). Exploring the Motivation of Employees in a Firm: A Case-Study. Business Systems Research Journal. <u>https://doi.org/10.2478/bsrj-2018-0012</u>
- Kmieciak, R. (2020). Trust, Knowledge Sharing, and Innovative Work Behavior: Empirical Evidence From Poland. *European Journal of Innovation* Management. <u>https://doi.org/10.1108/ejim-04-2020-0134</u>
- Knies, E., & Leisink, P. (2013). Linking People Management and Extra-Role Behaviour: Results of a Longitudinal Study. *Human Resource Management Journal*. <u>https://doi.org/10.1111/1748-8583.12023</u>
- Kock, A., Heising, W., & Gemünden, H. G. (2014). How Ideation Portfolio Management Influences Front - End Success. *Journal of Product Innovation Management*. <u>https://doi.org/10.1111/jpim.12217</u>
- Koekemoer, L., Beer, L. T. d., Govender, K., & Brouwers, M. (2021). Leadership Behaviour, Team Effectiveness, Technological Flexibility, Work Engagement and Performance During COVID-19 Lockdown: An Exploratory Study. SA Journal of Industrial Psychology. <u>https://doi.org/10.4102/sajip.v47i0.1829</u>
- Kohler, T., Fueller, J., Matzler, K., Stieger, D., & Füller, J. (2011). Co-creation in virtual worlds: The design of the user experience. *MIS quarterly*, 773-788.

- Kooij, B. J. G. V. d. (2017). Search for a Common Ground in the Fogs of Innovation Definitions. *SSRN Electronic Journal*. <u>https://doi.org/10.2139/ssrn.3139486</u>
- Kooij, D., Lange, A. H. d., Jansen, P., Kanfer, R., & Dikkers, J. (2011). Age and Work - related Motives: Results of a Meta - analysis. *Journal of organizational behavior*. <u>https://doi.org/10.1002/job.665</u>
- Koreshi, S. Y., & Alpass, F. (2023). Understanding the Use of Flexible Work Arrangements Among Older New Zealand Caregivers. *Journal of Applied Gerontology*. <u>https://doi.org/10.1177/07334648231152153</u>
- Korkmaz, S., Göksülük, D., & Zararsız, G. (2014). MVN: An R Package for Assessing Multivariate Normality. *The R Journal*, 6(2), 151. <u>https://doi.org/10.32614/rj-2014-031</u>
- Koskull, C. v., & Strandvik, T. (2014). Discovering the Unfolding of Service Innovations. *Journal of Business and Industrial Marketing*. <u>https://doi.org/10.1108/jbim-05-2012-0084</u>
- Kowalski, G., & Ślebarska, K. (2022). Remote Working and Work Effectiveness: A Leader Perspective. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph192215326</u>
- Kozlowski, S. W. J., & Klein, K. J. (2000). A multilevel approach to theory and research in organizations: Contextual, temporal, and emergent processes. In *Multilevel theory, research, and methods in organizations: Foundations, extensions, and new directions.* (pp. 3-90). Jossey-Bass.
- Kpinpuo, S. D., Akolgo, I. G., & Naimi, L. (2022). Leading Change and Innovation in Ghana's Banking Sector: The Mediating Role of Work Autonomy. *Industrial* and Commercial Training. <u>https://doi.org/10.1108/ict-12-2021-0086</u>
- Krause, D. E. (2004). Influence-based leadership as a determinant of the inclination to innovate and of innovation-related behaviors: An empirical investigation. *The leadership quarterly*, *15*(1), 79-102.
- Krishnamoorthy, K., Lu, F., & Mathew, T. (2007). A parametric bootstrap approach for ANOVA with unequal variances: Fixed and random models. *Computational Statistics & Data Analysis*, *51*(12), 5731-5742.
- Kristiansen, M., & Bloch-Poulsen, J. (2010). Employee driven innovation in team (EDIT)–Innovative potential, dialogue, and dissensus. *International Journal of Action Research*, 6(2-3), 155-195.
- Krzyżowska, E. (2022). Relationship Between Self-Efficacy, Trust, and Knowledge Sharing Among IT Industry Employees Working Remotely. *European Conference on Knowledge Management*. <u>https://doi.org/10.34190/eckm.23.1.598</u>

- Kumar, N., Alok, S., & Banerjee, S. (2023). Personal attributes and job resources as determinants of amount of work done under work-from-home: empirical study of Indian white-collar employees. *International Journal of Manpower*, 44(1), 113-132. <u>https://doi.org/10.1108/IJM-08-2021-0466</u>
- Kuntz, J. (2020). Resilience in Times of Global Pandemic: Steering Recovery and Thriving Trajectories. *Applied Psychology*. <u>https://doi.org/10.1111/apps.12296</u>
- Kwon, J. U., Hwang, J.-I., Park, J., & Ahn, S. C. (2019). Fully Asymmetric Remote Collaboration System. *Ieee Access*. <u>https://doi.org/10.1109/access.2019.2912397</u>
- Laher, S. (2016). Ostinato Rigore: Establishing Methodological Rigour in Quantitative Research. *South African Journal of Psychology*, *46*(3), 316-327. <u>https://doi.org/10.1177/0081246316649121</u>
- Lai, K. S., Yusof, N. A., & Kamal, E. M. (2016). Innovation Orientation in Architectural Firms. *Construction Innovation*, 16(4), 425-445. <u>https://doi.org/10.1108/ci-05-2015-0030</u>
- Lange, M., & Kayser, I. (2022). The Role of Self-Efficacy, Work-Related Autonomy and Work-Family Conflict on Employee's Stress Level During Home-Based Remote Work in Germany. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph19094955</u>
- Lantz, B. (2012). The Impact of Sample Non normality on ANOVA and Alternative Methods. *British Journal of Mathematical and Statistical Psychology*, *66*(2), 224-244. <u>https://doi.org/10.1111/j.2044-8317.2012.02047.x</u>
- LaPensee, E. W., Doshi, A., Salem, B., Jazdzyk, D., Steen, K., Cantrell, M. A., & Somers, E. C. (2021). Mobilizing Cross-Disciplinary Teams to Advance Translational Research Using Design Thinking Methods. *Journal of Clinical* and Translational Science. <u>https://doi.org/10.1017/cts.2021.823</u>
- Laursen, K., & Foss, N. J. (2003). New human resource management practices, complementarities and the impact on innovation performance. *Cambridge Journal of Economics*, 27(2), 243-263. <u>https://doi.org/10.1093/cje/27.2.243</u>
- Lauterbach, A., & Kunze, F. (2023). A Quasi-Experimental Exploration of Activity-Based Flexible Office Design and Demographic Differences in Employee Absenteeism. *Environment and Behavior*. <u>https://doi.org/10.1177/00139165231163549</u>
- Lay, J. C., Pauly, T., Graf, P., Biesanz, J. C., & Hoppmann, C. A. (2019). By Myself and Liking It? Predictors of Distinct Types of Solitude Experiences in Daily Life. *Journal of Personality*. <u>https://doi.org/10.1111/jopy.12421</u>
- LeBel, P. (2008). The role of creative innovation in economic growth: Some international comparisons. *Journal of Asian Economics*, 19(4), 334-347.

- Lebuda, I., Galewska-Kustra, M., & Glăveanu, V. P. (2016). Creativity and Social Interactions. *Creativity Theories – Research – Applications*. <u>https://doi.org/10.1515/ctra-2016-0012</u>
- Lee, C.-c., Lin, Y.-h., Huang, H.-c., Huang, W., & Teng, H.-h. (2015). The Effects of Task Interdependence, Team Cooperation, and Team Conflict on Job Performance. Social Behavior and Personality an International Journal. <u>https://doi.org/10.2224/sbp.2015.43.4.529</u>
- Lee, S. H., Wong, P. K., & Chong, C.-L. (2005). Human and Social Capital Explanations for R&D Outcomes. *IEEE Transactions on Engineering Management*. <u>https://doi.org/10.1109/tem.2004.839955</u>
- Lee, S. Y., & Brand, J. (2010). Can Personal Control Over the Physical Environment Ease Distractions in Office Workplaces? *Ergonomics*. <u>https://doi.org/10.1080/00140130903389019</u>
- Lee, Y., & Kim, J. (2022). How Family-Supportive Leadership Communication Enhances the Creativity of Work-From-Home Employees During the COVID-19 Pandemic. *Management Communication Quarterly*, 37(3), 599-628. <u>https://doi.org/10.1177/08933189221144997</u>
- Lee, Y. S. (2016). Creative Workplace Characteristics and Innovative Start-Up Companies. *Facilities*. <u>https://doi.org/10.1108/f-06-2014-0054</u>
- Leroy, H., Buengeler, C., Veestraeten, M., Shemla, M., & Hoever, I. J. (2021). Fostering Team Creativity Through Team-Focused Inclusion: The Role of Leader Harvesting the Benefits of Diversity and Cultivating Value-in-Diversity Beliefs. *Group & Organization Management*. https://doi.org/10.1177/10596011211009683
- Leroy, S., Schmidt, A. M., & Madjar, N. (2021). Working from home during COVID-19: A study of the interruption landscape. *Journal of Applied psychology*, 106(10), 1448-1465.
- Li, C., Makhdoom, H. Q., & Asim, S. (2020). Impact of Entrepreneurial Leadership on Innovative Work Behavior: Examining Mediation and Moderation Mechanisms. *Psychology Research and Behavior Management*. <u>https://doi.org/10.2147/prbm.s236876</u>
- Li, G., Li, L., Xie, L., & López, O. S. (2023). The Effects of Ethical Leadership on Creativity: A Conservation of Resources Perspective. *Current Psychology*. <u>https://doi.org/10.1007/s12144-023-04703-0</u>
- Li, P., Han, S. S., & Wu, H. (2023). "Home Away From Home" in Pandemic Times: How Has COVID-19 Changed the Airbnb Market in Melbourne? *International Journal of Housing Markets and Analysis*. <u>https://doi.org/10.1108/ijhma-08-2022-0123</u>

- Li, Q., & Liu, M. (2023). The Effect of Family Supportive Supervisor Behavior on Teachers' Innovative Behavior and Thriving at Work: A Moderated Mediation Model. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2023.1129486</u>
- Li, Q., She, Z., & Yang, B. (2018). Promoting Innovative Performance in Multidisciplinary Teams: The Roles of Paradoxical Leadership and Team Perspective Taking. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2018.01083</u>
- Lim, W. M. (2023). Philosophy of Science and Research Paradigm for Business Research In the Transformative Age Of automation, Digitalization, Hyperconnectivity, Obligations, Globalization and Sustainability. *Journal of Trade Science*, 11(2/3), 3-30. <u>https://doi.org/10.1108/jts-07-2023-0015</u>
- Lin, H.-C., & Lee, Y.-D. (2017). A Study of the Influence of Organizational Learning on Employees' Innovative Behavior and Work Engagement by a Cross-Level Examination. *Eurasia Journal of Mathematics Science and Technology Education*. <u>https://doi.org/10.12973/eurasia.2017.00738a</u>
- Litchfield, R., Ford, C. M., & Gentry, R. J. (2014). Linking Individual Creativity to Organizational Innovation. *The Journal of Creative Behavior*. <u>https://doi.org/10.1002/jocb.65</u>
- Litchfield, R. C., Karakitapoğlu Aygün, Z., Gumusluoğlu, L., Carter, M. R., & Hirst, G. (2017). When Team Identity Helps Innovation and When It Hurts: Team Identity and Its Relationship to Team and Cross - Team Innovative Behavior. *Journal of Product Innovation Management*. <u>https://doi.org/10.1111/jpim.12410</u>
- Littlewood, D., & Kiyumbu, W. L. (2018). "Hub" Organisations in Kenya: What Are They? What Do They Do? And What Is Their Potential? *Technological Forecasting and Social Change*. <u>https://doi.org/10.1016/j.techfore.2017.09.031</u>
- Liu, D., Chen, X., & Yao, X. (2011). From Autonomy to Creativity: A Multilevel Investigation of the Mediating Role of Harmonious Passion. *Journal of Applied psychology*. <u>https://doi.org/10.1037/a0021294</u>
- Liu, Z., Ge, L., & Peng, W. (2016). How Organizational Tenure Affects Innovative Behavior? *Nankai Business Review International*. <u>https://doi.org/10.1108/nbri-01-2016-0001</u>
- Lohr, S. L. (2019). Sampling : design and analysis (Second edition. ed.). CRC Press.
- Loi, R., Ngo, H. Y., Zhang, L., & Lau, C. B. S. (2011). The Interaction Between Leader–member Exchange and Perceived Job Security in Predicting Employee Altruism and Work Performance. *Journal of Occupational and Organizational Psychology*. <u>https://doi.org/10.1348/096317910x510468</u>
- Loignon, A. C., Johnson, M. A., Veestraeten, M., & Boyd, T. L. (2022). A Tale of Two Offices: The Socioeconomic Environment's Effect on Job Performance While
Working From Home. *Group & Organization Management*. https://doi.org/10.1177/10596011221117724

- Lopes, J. M., Oliveira, M., Silveira, P., Farinha, L. s., & Oliveira, J. (2021). Business Dynamism and Innovation Capacity, an Entrepreneurship Worldwide Perspective. *Journal of Open Innovation Technology Market and Complexity*. <u>https://doi.org/10.3390/joitmc7010094</u>
- Lorentzon, J. I., Fotoh, L. E., & Mugwira, T. (2023). Remote Auditing and Its Impacts on Auditors' Work and Work-Life Balance: Auditors' Perceptions and Implications. Accounting Research Journal. <u>https://doi.org/10.1108/arj-06-2023-0158</u>
- Loy, A., Follett, L., & Hofmann, H. (2016). Variations of Q–Q Plots: The Power of Our Eyes! *The American Statistician*, 70(2), 202-214. https://doi.org/10.1080/00031305.2015.1077728
- Lü, Y. (2023). A Review of the Influence of Organizational Identity on Employee Performance. BCP Business & Management. https://doi.org/10.54691/bcpbm.v39i.4205
- Lukić, J., & Vračar, M. M. (2018). Building and Nurturing Trust Among Members in Virtual Project Teams. *Strategic Management*. https://doi.org/10.5937/straman1803010l
- Luksyte, A., Unsworth, K., & Avery, D. R. (2017). Innovative Work Behavior and Sex-Based Stereotypes: Examining Sex Differences in Perceptions and Evaluations of Innovative Work Behavior. *Journal of organizational behavior*. <u>https://doi.org/10.1002/job.2219</u>
- Lumpkin, G. T., Cogliser, C. C., & Schneider, D. R. (2009). Understanding and Measuring Autonomy: An Entrepreneurial Orientation Perspective. *Entrepreneurship Theory and Practice*. <u>https://doi.org/10.1111/j.1540-6520.2008.00280.x</u>
- Luo, X. (2018). Theoretical Study on the Hypothesis Testing P Value. 2018 2nd International Conference on Education Science and Economic Management (ICESEM 2018),
- Lykourentzou, I., Vinella, F. L., Ahmed, F., Papastathis, C., Papangelis, K., Khan, V.-J., & Masthoff, J. (2022). Self-Organization in Online Collaborative Work Settings. *Collective Intelligence*. <u>https://doi.org/10.1177/26339137221078005</u>
- Lyonette, C. (2015, 2015/07/03). Part-time work, work–life balance and gender equality. *Journal of Social Welfare and Family Law, 37*(3), 321-333. https://doi.org/10.1080/09649069.2015.1081225
- Ma, X., Wu, D., & Hou, X. (2022). Positive Affect and Job Performance in Psychiatric Nurses: A Moderated Mediation Analysis. Nursing Open. <u>https://doi.org/10.1002/nop2.1553</u>

- Mackenzie, N., & Knipe, S. (2006). Research dilemmas: Paradigms, methods and methodology. *Issues in educational research*, *16*(2), 193-205.
- Maden-Eyiusta, C., & Alparslan, S. E. (2022). Captain or Deckhand? The Impact of Self-Leadership on Employees' Work Role Performance Under Remote Work. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2022.988105</u>
- Maghlaperidze, E., Kharadze, N., & Kuspliak, H. (2021). Development of Remote Jobs as a Factor to Increase Labor Efficiency. *Journal of Eastern European and Central Asian Research (Jeecar)*. <u>https://doi.org/10.15549/jeecar.v8i3.669</u>
- Maguire, M., & Delahunt, B. (2017). Doing a thematic analysis: A practical, step-bystep guide for learning and teaching scholars. *All Ireland journal of higher education*, 9(3).
- Maisha, K., Frei, M., Quintana, M., Chua, Y. X., Jain, R., & Miller, C. (2023). Utilizing Wearable Technology to Characterize and Facilitate Occupant Collaborations in Flexible Workspaces. *Journal of Physics Conference Series*. <u>https://doi.org/10.1088/1742-6596/2600/14/142009</u>
- Maitlo, Q., Wang, X., Yan, J., Lashari, I. A., Faraz, N. A., & Hajaro, N. H. (2022). Exploring Green Creativity: The Effects of Green Transformational Leadership, Green Innovation Climate, and Green Autonomy. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2022.686373</u>
- Majchrzak, A., More, P. H., & Faraj, S. (2012). Transcending knowledge differences in cross-functional teams. *Organization Science*, 23(4), 951-970.
- Malhotra, N., Smets, M., & Morris, T. (2016). Career pathing and innovation in professional service firms [Article]. Academy of Management Perspectives, 30(4), 369-383. <u>https://doi.org/10.5465/amp.2012.0108</u>
- Malik, M. A. R., Butt, A. N., & Zahid, F. (2023). Drivers of Radical and Incremental Creativity—A Moderated Mediation Framework. *Canadian Journal of Administrative Sciences / Revue Canadienne Des Sciences De L Administration*. <u>https://doi.org/10.1002/cjas.1713</u>
- Mancini, S., & González, J. M. S. (2021). Role of Technology Transfer, Innovation Strategy and Network: A Conceptual Model of Innovation Network to Facilitate the Internationalization Process of SMEs. *Technology and Investment*. <u>https://doi.org/10.4236/ti.2021.122006</u>
- Mansor, F. A., Jusoh, Y. H. M., Said, N. S. M., Ali, S. R. O., & Mat, K. A. (2022). Correlation of Employees' Satisfaction, Employees' Wellbeing and Workload Between Work Life Balance. *International Journal of Academic Research in Business and Social Sciences*. <u>https://doi.org/10.6007/ijarbss/v12-i9/14654</u>
- Maqableh, O., Helalat, A., & Nor, C. S. M. (2023). Exploring the Mediating Influence of Job Satisfaction on the Relationship Between Job Security and Turnover

Intention: A Case Study of the Hospitality Industry of Jordan. *Problems and Perspectives in Management*. <u>https://doi.org/10.21511/ppm.21(1).2023.33</u>

- Mara, C. A., & Cribbie, R. A. (2017). Equivalence of Population Variances: Synchronizing the Objective and Analysis. *The Journal of Experimental Education*, 86(3), 442-457. <u>https://doi.org/10.1080/00220973.2017.1301356</u>
- Marín-García, J. A., & Martinez-Tomas, J. (2016). Deconstructing AMO Framework: A Systematic Review. *Intangible Capital*. <u>https://doi.org/10.3926/ic.838</u>
- Marin-Garcia, J. A., & Tomas, J. M. (2016). Deconstructing AMO framework: A systematic review. *Intangible Capital*, *12*(4), 1040-1087.
- Marozzi, M. (2011). Levene Type Tests for the Ratio of Two Scales. *Journal of Statistical Computation and Simulation*, 81(7), 815-826. <u>https://doi.org/10.1080/00949650903499321</u>
- Marshall, C., & Rossman, G. B. (1999). *Designing qualitative research, 3rd ed.* Sage Publications, Inc.
- Martin, P., & Barnard, A. (2013). The Experience of Women in Male-Dominated Occupations: A Constructivist Grounded Theory Inquiry. SA Journal of Industrial Psychology. <u>https://doi.org/10.4102/sajip.v39i2.1099</u>
- Martínez-Sánchez, A., Vela-Jiménez, M. J., Pérez-Pérez, M., & de-Luis-Carnicer, P. (2011, 06 / 01 /). The Dynamics of Labour Flexibility: Relationships between Employment Type and Innovativeness [Article]. *Journal of Management Studies, 48*(4), 715-736. <u>https://doi.org/10.1111/j.1467-6486.2010.00935.x</u>
- Martins, E. C., & Terblanche, F. (2003). Building Organisational Culture That Stimulates Creativity and Innovation. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/14601060310456337</u>
- Mat Roni, S., & Djajadikerta, H. G. (2021). Preliminary Data Analysis: An Analysis Before the Analysis. In *Data Analysis with SPSS for Survey-based Research* (pp. 15-54). Springer Singapore. <u>https://doi.org/10.1007/978-981-16-0193-4_3</u>
- Mattar, J., & Ramos, D. K. (2022). Paradigms and Approaches in Educational Research. International Journal for Innovation Education and Research, 10(4), 250-256. https://doi.org/10.31686/ijier.vol10.iss4.3380
- Mattarelli, E., Cochis, C., Bertolotti, F., & Ungureanu, P. (2022). How Designed Work Environment and Enacted Work Interactions Impact Creativity and Work–life Balance. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/ejim-01-2022-0028</u>
- Maula, M., & Stam, W. (2019). Enhancing Rigor in Quantitative Entrepreneurship Research. *Entrepreneurship Theory and Practice*, 44(6), 1059-1090. <u>https://doi.org/10.1177/1042258719891388</u>

- Mauno, S., Cuyper, N. D., Kinnunen, U., Ruokolainen, M., Rantanen, J., & Mäkikangas, A. (2015). The Prospective Effects of Work–family Conflict and Enrichment on Job Exhaustion and Turnover Intentions: Comparing Long-Term Temporary vs. Permanent Workers Across Three Waves. Work & Stress. <u>https://doi.org/10.1080/02678373.2014.1003997</u>
- Mayangsari, F. R. (2020). Australia Government Response to COVID-19: Coordination and the Effectivity of Policy. *Jurnal Global & Strategis*. https://doi.org/10.20473/jgs.14.2.2020.279-296
- Mazmanian, M., Orlikowski, W. J., & Yates, J. (2013). The Autonomy Paradox: The Implications of Mobile Email Devices for Knowledge Professionals. *Organization Science*. <u>https://doi.org/10.1287/orsc.1120.0806</u>
- Mbah, A. K., & Paothong, A. (2014). Shapiro–Francia test compared to other normality test using expected p-value. *Journal of Statistical Computation and Simulation*, 85(15), 3002-3016. <u>https://doi.org/10.1080/00949655.2014.947986</u>
- McGee, B., Couillou, R. J., & Maalt, K. (2023). Work From Home: Lessons Learned and Implications for Post-Pandemic Workspaces. *Interiority*. <u>https://doi.org/10.7454/in.v6i1.259</u>
- McNeish, D. (2018). Thanks Coefficient Alpha, We'll Take It From Here. *Psychological Methods*, 23(3), 412-433. <u>https://doi.org/10.1037/met0000144</u>
- Mehmood, K., Jabeen, F., Iftikhar, Y., Yan, M., Khan, A. R., AlNahyan, M. T., Alkindi, H., & Alhammadi, B. (2022). Elucidating the Effects of Organisational Practices on Innovative Work Behavior in UAE Public Sector Organisations: The Mediating Role of Employees' Wellbeing. *Applied Psychology Health and Well-Being*. <u>https://doi.org/10.1111/aphw.12343</u>
- Mehra, P., & Nickerson, C. (2019). Does Technology Divide or Unite Generations? International Journal of Organizational Analysis. <u>https://doi.org/10.1108/ijoa-10-2018-1576</u>
- Meiryani, M., Nelviana, Koh, Y., Soepriyanto, G., Aljuaid, M., & Hasan, F. (2022). The Effect of Transformational Leadership and Remote Working on Employee Performance During COVID-19 Pandemic. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2022.919631</u>
- Melander, L. (2017). Achieving Sustainable Development by Collaborating in Green Product Innovation. *Business Strategy and the Environment*. <u>https://doi.org/10.1002/bse.1970</u>
- Meng, L., Jiang, S., & Chen, H. (2022). Authentic Leadership and Innovation Behaviour Among Nurses in China: A Mediation Model of Work Engagement. *Journal of Nursing Management*. <u>https://doi.org/10.1111/jonm.13669</u>

- Merone, L., & Whitehead, O. (2021). COVID-19 and working within health care systems: The future is flexible. *Asia Pacific Journal of Health Management*, *16*(1), 28-32.
- Messmann, G., & Mulder, R. H. (2012). Development of a measurement instrument for innovative work behaviour as a dynamic and context-bound construct [Article]. *Human Resource Development International*, 15(1), 43-59. <u>https://doi.org/10.1080/13678868.2011.646894</u>
- Messmann, G., & Mulder, R. H. (2020). A short measure of innovative work behaviour as a dynamic, context-bound construct. *International Journal of Manpower*, 41(8), 1251-1267. <u>https://doi.org/10.1108/IJM-01-2019-0029</u>
- Methot, J. R., Rosado-Solomon, E., Downes, P. E., & Gabriel, A. S. (2021). Office Chitchat as a Social Ritual: The Uplifting Yet Distracting Effects of Daily Small Talk at Work. Academy of Management Journal. https://doi.org/10.5465/amj.2018.1474
- Meyer, J. (2008). The Adoption of New Technologies and the Age Structure of the Workforce. *SSRN Electronic Journal*. <u>https://doi.org/10.2139/ssrn.1210162</u>
- Meyer, K. E., van Witteloostuijn, A., & Beugelsdijk, S. (2017). What's in a p? Reassessing best practices for conducting and reporting hypothesis-testing research. *Journal of International Business Studies*, 48(5), 535-551. <u>https://doi.org/10.1057/s41267-017-0078-8</u>
- Miah, M. M., & Hafit, D. N. I. A. (2020). Enhancing Employee's Creative Performance by Providing Training Among Employees of Small and Medium-Sized Enterprises (SMEs) in Malaysia. *International Journal of Business and Social Science*. <u>https://doi.org/10.30845/ijbss.v11n6a7</u>
- Michalakopoulou, K., Bamford, D., Reid, I., & Nikitas, A. (2023). Barriers and opportunities to innovation for legal service firms: a thematic analysis-based contextualization. *Production Planning & amp; Control, 34*(7), 604-622. <u>https://doi.org/10.1080/09537287.2021.1946329</u>
- Michinov, E., & Michinov, N. (2021). Stay at Home! When Personality Profiles Influence Mental Health and Creativity During the COVID-19 Lockdown. *Current Psychology*. <u>https://doi.org/10.1007/s12144-021-01885-3</u>
- Miranda, A. S., & Claudel, M. (2021). Spatial Proximity Matters: A Study on Collaboration. *Plos One*. <u>https://doi.org/10.1371/journal.pone.0259965</u>
- Miron-Spektor, E., Vashdi, D. R., & Gopher, H. (2022). Bright Sparks and Enquiring Minds: Differential Effects of Goal Orientation on the Creativity Trajectory. *Journal of Applied psychology*. <u>https://doi.org/10.1037/apl0000888</u>
- Mishra, P., Bhatnagar, J., Gupta, R. K., & Wadsworth, S. M. (2017). How Work-family Enrichment Influence Innovative Work Behavior: Role of Psychological Capital

and Supervisory Support. *Journal of Management & Organization*. <u>https://doi.org/10.1017/jmo.2017.23</u>

- Mitchell, V. (2020). *Report: Most Australian employees to work from home*. CMO. Retrieved July 2022 from <u>https://www.cmo.com.au/article/672072/report-most-australian-employees-work-from-home/</u>
- Miyake, F., Chimed Ochir, O., Hino, A., Ikegami, K., Nagata, T., Tateishi, S., Tsuji, M., Matsuda, S., & Ishimaru, T. (2022). Job Stress and Loneliness Among Desk Workers During the COVID-19 Pandemic in Japan: Focus on Remote Working. *Environmental Health and Preventive Medicine*. <u>https://doi.org/10.1265/ehpm.22-00107</u>
- Moll, F., & de Leede, J. (2016). Fostering Innovation: The Influence of New Ways of Working on Innovative Work Behavior. In New Ways of Working Practices: Antecedents and Outcomes (Vol. 16, pp. 95-143). Emerald Group Publishing Limited. <u>https://doi.org/10.1108/S1877-636120160000016006</u>
- Montani, F., Courcy, F., Battistelli, A., & Witte, H. D. (2021). Job Insecurity and Innovative Work Behaviour: A Moderated Mediation Model of Intrinsic Motivation and Trait Mindfulness. *Stress and Health*. https://doi.org/10.1002/smi.3034
- Monteiro, C. V., da Silva, F. Q., & Capretz, L. F. (2016). The innovative behaviour of software engineers: Findings from a pilot case study. Proceedings of the 10th ACM/IEEE International Symposium on Empirical Software Engineering and Measurement,
- Mostafa, B. A. (2021). The Effect of Remote Working on Employees Wellbeing and Work-Life Integration During Pandemic in Egypt. *International Business Research*. <u>https://doi.org/10.5539/ibr.v14n3p41</u>
- Mullens, F., & Glorieux, I. (2023). Reducing Weekly Working Hours: Temporal Strategies and Changes in the Organization and Experiences of Work-Results From a Qualitative Study of a 30-Hour Workweek Experiment. *Time & Society*. <u>https://doi.org/10.1177/0961463x231156948</u>
- Muñoz-Pascual, L., & Galende, J. (2020). Ambidextrous Relationships and Social Capability as Employee Well-Being: The Secret Sauce for Research and Development and Sustainable Innovation Performance. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph17093072</u>
- Mustafa, M., Coetzer, A., Ramos, H. M., & Fuhrer, J. (2021). Exploring the Effects of Small- And Medium-Sized Enterprise Employees' Job Satisfaction on Their Innovative Work Behaviours: The Moderating Effects of Personality. *Journal of Organizational Effectiveness People and Performance*. <u>https://doi.org/10.1108/joepp-07-2020-0133</u>

- Mwita, K. (2022). Factors influencing data saturation in qualitative studies. International Journal of Research in Business and Social Science (2147-4478), 11(4), 414-420.
- Naeem, M., Ozuem, W., Howell, K., & Ranfagni, S. (2023). A Step-by-Step Process of Thematic Analysis to Develop a Conceptual Model in Qualitative Research. *International Journal of Qualitative Methods*. <u>https://doi.org/10.1177/16094069231205789</u>
- Nahapiet, J., & Ghoshal, S. (1998). Social Capital, Intellectual Capital, and the Organizational Advantage. *Academy of management Review*. <u>https://doi.org/10.5465/amr.1998.533225</u>
- Nahrgang, J. D., Morgeson, F. P., & Hofmann, D. A. (2011). Safety at Work: A Meta-Analytic Investigation of the Link Between Job Demands, Job Resources, Burnout, Engagement, and Safety Outcomes. *Journal of Applied psychology*. <u>https://doi.org/10.1037/a0021484</u>
- Nangoy, R., Mursitama, T., Setiadi, N., & Pradipto, Y. (2020). Creating sustainable performance in the fourth industrial revolution era: The effect of employee's work well-being on job performance. *Management Science Letters*, *10*(5), 1037-1042.
- Narzary, G., & Palo, S. (2021). Testing Interaction Effects of Intellectual Capital and Burnout on Innovative Work Behaviour of Professional Nurses. *Journal of Intellectual Capital*. <u>https://doi.org/10.1108/jic-01-2019-0017</u>
- Nasution, N. R., Siregar, Z. M. E., & Pristiyono, P. (2021). The Effect of Job Autonomy on Employee Innovative Behavior: the Role of Job Satisfaction as Intervening Variable. Budapest International Research and Critics Institute (Birci-Journal) Humanities and Social Sciences. <u>https://doi.org/10.33258/birci.v4i2.1994</u>
- Naushad, M. (2022). SME Innovation and Employee Creativity: The Role of Entrepreneurial Leadership. *Problems and Perspectives in Management*. <u>https://doi.org/10.21511/ppm.20(4).2022.42</u>
- Naveed, M., Zia, M. Q., & Cangialosi, N. (2022). The Nexus of Job Resources and Turnover Intentions With the Mediating Role of Employees' Work Engagement in the Hospitality Industry. *Consumer Behavior in Tourism and Hospitality*. <u>https://doi.org/10.1108/cbth-09-2021-0217</u>
- Nawab, S., Nazir, T., Zahid, M., & Fawad, S. M. (2015). Knowledge Management, Innovation and Organizational Performance. *International Journal of Knowledge Engineering*. <u>https://doi.org/10.7763/ijke.2015.v1.7</u>
- Neidlinger, S. M., Felfe, J., & Schübbe, K. (2022). Should I Stay or Should I Go (To the Office)?—Effects of Working From Home, Autonomy, and Core Self– Evaluations on Leader Health and Work–Life Balance. *International Journal of Environmental Research and Public Health*. https://doi.org/10.3390/ijerph20010006

- Neumann, M., & Bogdanov, Y. (2022). The Impact of Covid 19 on Agile Software Development: A Systematic Literature Review. <u>https://doi.org/10.24251/hicss.2022.882</u>
- Newcomb, M., & Venning, A. (2022). One Size Fits All: The Role of Technology in Practitioner's Experience of Working from Home during COVID19. *Journal of Technology in Human Services*, 40(4), 319-338. https://doi.org/10.1080/15228835.2022.2087202
- Newman, A., Neesham, C., Manville, G., & Tse, H. (2017). Examining the Influence of Servant and Entrepreneurial Leadership on the Work Outcomes of Employees in Social Enterprises. *The International Journal of Human Resource Management*. <u>https://doi.org/10.1080/09585192.2017.1359792</u>
- Newman, D. A. (2014). Missing data: Five practical guidelines. *Organizational Research Methods*, 17(4), 372-411.
- Ng, P., Lit, K. K., & Cheung, C. (2022, Aug). Remote work as a new normal? The technology-organization-environment (TOE) context. *Technol Soc*, 70, 102022. <u>https://doi.org/10.1016/j.techsoc.2022.102022</u>
- Ng, T. W. H., & Feldman, D. C. (2010). Organizational Tenure and Job Performance. Journal of Management. <u>https://doi.org/10.1177/0149206309359809</u>
- Ng, T. W. H., & Feldman, D. C. (2012). Age and Innovation related Behavior: The Joint Moderating Effects of Supervisor Undermining and Proactive Personality. *Journal of organizational behavior*. <u>https://doi.org/10.1002/job.1802</u>
- Ng, T. W. H., & Feldman, D. C. (2013). A Meta analysis of the Relationships of Age and Tenure With Innovation - related Behaviour. *Journal of Occupational and Organizational Psychology*. <u>https://doi.org/10.1111/joop.12031</u>
- Nguyen, H. V., Nguyen, H. L., Dao, A. T. M., Nguyen, T. V., Le, P. M., Vu, K. D., Tran, A. T., Dao, P. K., Nguyen, C. T., & Debattista, J. (2021). The COVID -19 Pandemic in Australia: Public Health Responses, Opportunities and Challenges. *The International Journal of Health Planning and Management*. <u>https://doi.org/10.1002/hpm.3326</u>
- Nguyen, N. T. (2021). The Fearless Organization: Creating Psychological Safety in the Workplace for Learning, Innovation, and Growth. *The Learning Organization*. https://doi.org/10.1108/tlo-04-2021-266
- Nguyen, T.-M., Tran, N., & Nguyen, H. G. (2020). Knowledge Sharing and Innovation Capability at Both Individual and Organizational Levels: An Empirical Study From Vietnam's Telecommunication Companies. *Management & Marketing*. <u>https://doi.org/10.2478/mmcks-2020-0017</u>

- Nguyen, T. L., & Hunter, S. T. (2021). Not Worth My Time: Applying a Value Based Framework of Creative Idea Appraisals to Predict Investments of Time Toward Implementing Others' Ideas. *The Journal of Creative Behavior*. <u>https://doi.org/10.1002/jocb.529</u>
- Nikpour, A. (2018). Psychological Empowerment and Organizational Innovation: Mediating Role of Job Satisfaction and Organizational Commitment. *International Journal of Organizational Leadership*. <u>https://doi.org/10.33844/ijol.2018.60421</u>
- Nilles, J. M. (1994). *Making telecommuting happen: A guide for telemanagers and telecommuters.*
- Niranga, W. A. M., & Dharmadasa, M. P. P. (2021). Support for Innovations and Innovative Behavior: A Moderating Effect of Power Distance in Innovative Behaviour of Middle and Lower Level Employees in Sri Lanka. *Proceedings of International Conference on Business Management*. <u>https://doi.org/10.31357/icbm.v17.5213</u>
- Niu, H.-J. (2014). Is Innovation Behavior Congenital? Enhancing Job Satisfaction as a Moderator. *Personnel Review*. <u>https://doi.org/10.1108/pr-12-2012-0200</u>
- Nizar, S. F. b., Zain, R. b. S., Razali, M., Ibrahim, I. I., & Abdullah, N. A. b. N. (2021). The Association Between the Impacts of Telecommuting Work Arrangement and Employee Performance During Covid- 19 Pandemic: A Case Study in Private Sector. *International Journal of Academic Research in Business and Social Sciences*. https://doi.org/10.6007/ijarbss/v11-i7/10161
- Noh, E., & Lee, K.-H. (2022). The Empirical Research on Professional Isolation During Remote Work in Context of COVID-19 and Organizational Citizenship Behavior. Korean Journal of Industrial and Organizational Psychology. <u>https://doi.org/10.24230/kjiop.v35i3.513-558</u>
- Norman, G. (2010). Likert Scales, Levels of Measurement and the "Laws" of Statistics. *Advances in Health Sciences Education*. <u>https://doi.org/10.1007/s10459-010-</u> <u>9222-y</u>
- Nowell, L., Norris, J. M., White, D., & Moules, N. J. (2017). Thematic Analysis. International Journal of Qualitative Methods. https://doi.org/10.1177/1609406917733847
- Nsenduluka, E., & Shee, H. (2012). Organizational and Group Antecedents of Work Group Innovativeness in the Service Sector.
- Nusair, N., Ababneh, R. I., & Bae, Y. K. (2012). The Impact of Transformational Leadership Style on Innovation as Perceived by Public Employees in Jordan. *International Journal of Commerce and Management*. <u>https://doi.org/10.1108/10569211211260283</u>

- Obaid, S., Ahmad, S. F., & Mumtaz, F. (2022). Ability-Motivation-Opportunity Framework. *International Journal of Asian Business and Information Management*. <u>https://doi.org/10.4018/ijabim.309105</u>
- Odoardi, C., Montani, F., Boudrias, J. S., & Battistelli, A. (2015, 07 / 06 /). Linking managerial practices and leadership style to innovative work behavior: The role of group and psychological processes [Article]. *Leadership and Organization Development Journal*, 36(5), 545-569. <u>https://doi.org/10.1108/LODJ-10-2013-0131</u>
- OECD/Eurostat. (2018). Oslo Manual 2018: Guidelines for Collecting, Reporting and Using Data on Innovation (The Measurement of Scientific, Technological and Innovation Activities, Issue. <u>https://www.oecd-</u> ilibrary.org/content/publication/9789264304604-en
- Öhrn, M., Wahlström, V., Harder, M., Pettersson-Strömbäck, A., Danielsson, C. B., Olsson, D., Andersson, M., & Järvholm, L. S. (2021). Productivity, Satisfaction, Work Environment and Health After Relocation to an Activity-Based Flex Office—The Active Office Design Study. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph18147640</u>
- Okpara, L. O., Werner, C., Murray, A., & Damian, D. (2022). A Case Study of Building Shared Understanding of Non-Functional Requirements in a Remote Software Organization. <u>https://doi.org/10.48550/arxiv.2205.09220</u>
- Oldham, G. R., & Cummings, A. (1996). Employee Creativity: Personal and Contextual Factors at Work [research-article]. *The Academy of Management Journal, 39*(3), 607. <u>https://search.ebscohost.com/login.aspx?direct=true&AuthType=shib&db=edsjs</u> <u>r&AN=edsjsr.256657&site=eds-live&custid=s1145751</u>
- Olokundun, M., Ogbari, M. E., Falola, H. O., & Ibidunni, A. S. (2022). Leveraging 5G Network for Digital Innovation in Small and Medium Enterprises: A Conceptual Review. Journal of Innovation and Entrepreneurship. https://doi.org/10.1186/s13731-021-00181-5
- Olson, K. (2016). *Essentials of qualitative interviewing*. Routledge, Taylor & Francis Group.
- Onken-Menke, G., Nüesch, S., & Kröll, C. (2017). Are You Attracted? Do You Remain? Meta-Analytic Evidence on Flexible Work Practices. *Bur - Business Research*. <u>https://doi.org/10.1007/s40685-017-0059-6</u>
- Opoku, S. T., Apenteng, B. A., & Boakye, K. G. (2021). Rewards and Employee Creativity Among Rural Healthcare Employees: The Mediating Role of Organizational Support for Innovation and the Moderating Impact of Supervisory Support. *International Journal of Quality and Service Sciences*. <u>https://doi.org/10.1108/ijqss-11-2020-0187</u>

- Oppezzo, M., & Schwartz, D. L. (2014). Give Your Ideas Some Legs: The Positive Effect of Walking on Creative Thinking. *Journal of Experimental Psychology Learning Memory and Cognition*. <u>https://doi.org/10.1037/a0036577</u>
- Oppong, S. H. (2013). The problem of sampling in qualitative research. Asian journal of management sciences and education, 2(2), 202-210.
- Osayande, F., & Okolie, U. C. (2019). Person-Job Fit and Employee Engagement in Edo State Secretariat, Benin City. *Annals of Spiru Haret University Economic Series*. <u>https://doi.org/10.26458/1933</u>
- Oshima, S., & Asmuß, B. (2017). Mediated Business: Living the Organizational Surroundings – Introduction. *Culture and Organization*. <u>https://doi.org/10.1080/14759551.2017.1387965</u>
- Otonkorpi-Lehtoranta, K., Salin, M., Hakovirta, M., & Kaittila, A. (2021). Gendering Boundary Work: Experiences of Work–family Practices Among Finnish Working Parents During COVID - 19 Lockdown. *Gender Work and Organization*. <u>https://doi.org/10.1111/gwao.12773</u>
- Oyekan, J., Prabhu, V. A., Tiwari, A., Baskaran, V., Burgess, M., & McNally, R. (2017). Remote Real-Time Collaboration Through Synchronous Exchange of Digitised Human–workpiece Interactions. *Future Generation Computer Systems*. <u>https://doi.org/10.1016/j.future.2016.08.012</u>
- Öztürk, P., Avci, C. B., & Kaya, C. (2021). The Effect of Remote Collaborative Work on Design Processes During the Pandemic. *Strategic Design Research Journal*. <u>https://doi.org/10.4013/sdrj.2021.141.10</u>
- Ozuem, W., Willis, M., & Howell, K. E. (2022). Thematic Analysis Without Paradox: Sensemaking and Context. *Qualitative Market Research an International Journal*, 25(1), 143-157. <u>https://doi.org/10.1108/qmr-07-2021-0092</u>
- Pak, K., Kooij, D. T., De Lange, A. H., & Van Veldhoven, M. J. (2019). Human Resource Management and the ability, motivation and opportunity to continue working: A review of quantitative studies. *Human resource management review*, 29(3), 336-352.
- Palinkas, L. A., Horwitz, S. M., Green, C. A., Wisdom, J. P., Duan, N., & Hoagwood, K. (2013). Purposeful Sampling for Qualitative Data Collection and Analysis in Mixed Method Implementation Research. *Administration and Policy in Mental Health and Mental Health Services Research*, 42(5), 533-544. <u>https://doi.org/10.1007/s10488-013-0528-y</u>

Pallant. (2020). SPSS SURVIVAL MANUAL [electronic resource]. OPEN UNIV Press.

Palumbo, R., Manna, R., & Cavallone, M. (2020). Beware of Side Effects on Quality! Investigating the Implications of Home Working on Work-Life Balance in Educational Services. *The TQM Journal*. <u>https://doi.org/10.1108/tqm-05-2020-0120</u>

- Panasiewicz, L. (2021). Informative and Affective Determinants of Pro-Innovative Behaviours. *European Research Studies Journal*. <u>https://doi.org/10.35808/ersj/2310</u>
- Pang, Z., Becerik-Gerber, B., Hoque, S. M. A., O'Neill, Z., Pedrielli, G., Wen, J., & Wu, T. (2021). How Work From Home Has Affected the Occupant's Well-Being in the Residential Built Environment: An International Survey Amid the Covid-19 Pandemic. Asme Journal of Engineering for Sustainable Buildings and Cities. <u>https://doi.org/10.1115/1.4052640</u>
- Panuwatwanich, K., Stewart, R. A., & Mohamed, S. A. M. (2008). The Role of Climate for Innovation in Enhancing Business Performance. *Engineering Construction* & Architectural Management. <u>https://doi.org/10.1108/09699980810902712</u>
- Park, S., Jeong, S., & Chai, D. S. (2021). Remote E-Workers' Psychological Well-Being and Career Development in the Era of COVID-19: Challenges, Success Factors, and the Roles of HRD Professionals. *Advances in Developing Human Resources*. <u>https://doi.org/10.1177/15234223211017849</u>
- Park, S., Tseng, Y., & Kim, S. C. (2015). The Impact of Innovation on Job Satisfaction: Evidence From U.S. Federal Agencies. *Asian Social Science*. <u>https://doi.org/10.5539/ass.v12n1p274</u>
- Park, Y. K., Song, J. H., Yoon, S. W., & Kim, J. (2014). Learning organization and innovative behavior: The mediating effect of work engagement. *European Journal of Training and Development*, 38(1/2), 75-94.
- Parker, G. G., Alstyne, M. V., & Jiang, X. (2017). Platform Ecosystems: How Developers Invert the Firm. *MIS quarterly*. https://doi.org/10.25300/misq/2017/41.1.13
- Parker, S. K., Knight, C., & Keller, A. (2020). *Remote Managers Are Having Trust Issues*. <u>https://hbr.org/2020/07/remote-managers-are-having-trust-issues</u>
- Parker, S. K., Williams, H. M., & Turner, N. (2006). Modeling the Antecedents of Proactive Behavior at Work. *Journal of Applied psychology*. <u>https://doi.org/10.1037/0021-9010.91.3.636</u>
- Parris, J. (2017). Remote work or telecommute: What's the difference. *Retrieved May*, 14, 2018.
- Parsons, R. (2015). The Impact of Age on Innovation. *Management Research Review*. <u>https://doi.org/10.1108/mrr-10-2013-0241</u>
- Parter, C., & Wilson, S. (2021). My Research Is My Story: A Methodological Framework of Inquiry Told Through Storytelling by a Doctor of Philosophy Student. *Qualitative Inquiry*, 27(8-9), 1084-1094. <u>https://doi.org/10.1177/1077800420978759</u>

Pathak, K. P., & Thapaliya, S. (2022). Some Philosophical Paradigms and Their Implications in Health Research: A Critical Analysis. *International Research Journal of MMC*, 3(3), 9-17. <u>https://doi.org/10.3126/irjmmc.v3i3.48627</u>

Patton, M. Q. (2015). Qualitative research & evaluation methods : integrating theory and practice (Fourth edition. ed.) [Bibliographies Non-fiction]. SAGE.
<u>https://search.ebscohost.com/login.aspx?direct=true&AuthType=shib&db=cat06</u> 414a&AN=vic.b4051555&site=eds-live&custid=s1145751

- Paulus, P. B., & Brown, V. R. (2007). Toward More Creative and Innovative Group Idea Generation: A Cognitive-Social-Motivational Perspective of Brainstorming. Social and Personality Psychology Compass. <u>https://doi.org/10.1111/j.1751-9004.2007.00006.x</u>
- Pauly, T., Lay, J. C., Scott, S. B., & Hoppmann, C. A. (2018). Social Relationship Quality Buffers Negative Affective Correlates of Everyday Solitude in an Adult Lifespan and an Older Adult Sample. *Psychology and Aging*. <u>https://doi.org/10.1037/pag0000278</u>
- Pei, Z., Pan, Y., Skitmore, M., & Feng, T. (2018). Leader-member Exchange and Prohibitive Voice in Nonprofit Organizations in China: The Moderating Roles of Superior-subordinate Tenure Matching. *Nonprofit Management and Leadership*. <u>https://doi.org/10.1002/nml.21330</u>
- Peng, C.-Y. J., & So, T.-S. H. (2002). Logistic Regression Analysis and Reporting: A Primer. UNDERSTANDING STATISTICS, 1(1), 31-70.
- Peng, L., & Li, Z. (2021). Psychological Contract, Organizational Commitment, and Knowledge Innovation: A Perspective of Open Innovation. *Problems and Perspectives in Management*. <u>https://doi.org/10.21511/ppm.19(2).2021.33</u>
- Peng, Y., Zhang, W., Xu, X., Matthews, R. A., & Jex, S. M. (2019). When Do Work Stressors Lead to Innovative Performance? An Examination of the Moderating Effects of Learning Goal Orientation and Job Autonomy. *International Journal* of Stress Management. <u>https://doi.org/10.1037/str0000109</u>
- Peprah, E. O. (2023). Hybrid Workplace: Current Status, Positives, Negatives, Challenges, and Team Learning. *The Learning Organization*, 31(1), 88-103. https://doi.org/10.1108/tlo-11-2022-0150
- Perry-Smith, J. E., & Mannucci, P. V. (2017). From Creativity to Innovation: The Social Network Drivers of the Four Phases of the Idea Journey. Academy of management Review. <u>https://doi.org/10.5465/amr.2014.0462</u>
- Peterson, B. L. (2017). Thematic Analysis/Interpretive Thematic Analysis. 1-9. https://doi.org/10.1002/9781118901731.iecrm0249

- Pfund, C., Branchaw, J., McDaniels, M., Byars Winston, A., & Birren, B. W. (2020). Reassess - Realign - Reimagine: A Guide for Mentors Pivoting to Remote Research Mentoring. <u>https://doi.org/10.31235/osf.io/efgh8</u>
- Phil-Thingvad, S., & Klausen, K. K. (2020). Managing the implementation of innovation strategies in public service organisation—How managers may support employees innovative work behaviour. *International Journal of Innovation Management*, 24(04), 2050074.
- Phillips, S. (2020). Working through the pandemic: Accelerating the transition to remote working. *Business Information Review*, 37(3), 129-134. <u>https://doi.org/10.1177/0266382120953087</u>
- Pluye, P., & Hong, Q. N. (2014). Combining the Power of Stories and the Power of Numbers: Mixed Methods Research and Mixed Studies Reviews. *Annual Review of Public Health*, 35(1), 29-45. <u>https://doi.org/10.1146/annurevpublhealth-032013-182440</u>
- Poncet, A., Courvoisier, D. S., Combescure, C., & Perneger, T. V. (2016). Normality and sample size do not matter for the selection of an appropriate statistical test for two-group comparisons. *Methodology*.
- Pradhan, R. K., & Hati, L. (2019). The Measurement of Employee Well-Being: Development and Validation of a Scale. *Global Business Review*. <u>https://doi.org/10.1177/0972150919859101</u>
- Price, D. J., Shearer, F. M., Meehan, M. T., McBryde, E. S., Moss, R., Golding, N., Conway, E. J., Dawson, P., Cromer, D., Wood, J. G., Abbott, S., McVernon, J., & McCaw, J. M. (2020). Early Analysis of the Australian COVID-19 Epidemic. *Elife*. <u>https://doi.org/10.7554/elife.58785</u>
- Price, J. H., & Murnan, J. (2004). Research limitations and the necessity of reporting them. *35*(2), 66-67.
- Prieto, I. M., & Pérez-Santana, M. P. (2014). Managing innovative work behavior: the role of human resource practices. *Personnel Review*, 43(2), 184-208. <u>https://doi.org/10.1108/pr-11-2012-0199</u>
- Probst, T. M., Chizh, A., Hu, S., Jiang, L., & Austin, C. (2019). Explaining the Relationship Between Job Insecurity and Creativity. *Career Development International*. <u>https://doi.org/10.1108/cdi-04-2018-0118</u>
- Prokop, V., & Hájek, P. (2023). The Role of Top Management Meetings in Enhancing Product Innovation and Performance. *R and D Management*. <u>https://doi.org/10.1111/radm.12593</u>
- Purc, E., & Łaguna, M. (2019). Personal Values and Innovative Behavior of Employees. Frontiers in Psychology. <u>https://doi.org/10.3389/fpsyg.2019.00865</u>

- PwC. (2022). *Balancing Act The New Equation in hybrid working* (Future of Work, Issue. <u>https://www.pwc.com.au/important-problems/future-of-work/future-of-work/future-of-work-report-2022.pdf</u>
- Quigley, D., Poole, C., Whiting, S., O'Connor, E., Gleeson, C., & Alpine, L. (2022). University Student Experiences of Work-Based Placements During COVID-19 Pandemic: An Inter-Disciplinary Survey of Allied Health and Social Work Students. *Higher Education Skills and Work-Based Learning*, 13(1), 198-215. https://doi.org/10.1108/heswbl-11-2021-0218
- Quinlan, C., Babin, B., Carr, J., & Griffin, M. (2019). *Business research methods* (2nd Edition ed.). South Western Cengage.
- Quintane, E., Casselman, R. M., Reiche, B. S., & Nylund, P. A. (2011). Innovation as a Knowledge based Outcome. *Journal of Knowledge Management*. https://doi.org/10.1108/13673271111179299
- Rachman, T., Anindita, R., Izmi, N. N., Ewaldo, C., Rezki, F., & Andini, E. T. (2022). Fostering Innovative Work Behavior of Employees in the Tourism Industry: Strategies for Successful Remote Tourism During the COVID-19 Pandemic. *Jurnal Manajemen Dan Pemasaran Jasa*, 15(2), 227-240. <u>https://doi.org/10.25105/jmpj.v15i2.13573</u>
- Radaelli, G., Lettieri, E., Mura, M., & Spiller, N. (2014). Knowledge Sharing and Innovative Work Behaviour in Healthcare: A Micro-Level Investigation of Direct and Indirect Effects [Article]. *Creativity & Innovation Management*, 23(4), 400-414. <u>https://doi.org/10.1111/caim.12084</u>
- Rahman, M., Wahab, S. A., & Latiff, A. S. A. (2023). The Underlying Theories of Organizational Sustainability: The Motivation Perspective. *Journal of Business* and Management Studies. <u>https://doi.org/10.32996/jbms.2023.5.1.18</u>
- Ramamoorthy, N., Flood, P. C., Slattery, T., & Sardessai, R. (2005). Determinants of Innovative Work Behaviour: Development and Test of an Integrated Model [Article]. *Creativity & Innovation Management*, 14(2), 142-150. <u>https://doi.org/10.1111/j.1467-8691.2005.00334.x</u>
- Rashid, Y., Rashid, A., Warraich, M. A., Sabir, S., & Waseem, A. (2019). Case Study Method: A Step-by-Step Guide for Business Researchers. *International Journal* of Qualitative Methods, 18, 160940691986242. https://doi.org/10.1177/1609406919862424
- Rathnaweera, D., & Jayathilaka, R. (2021). In Employees' Favour or Not?—The Impact of Virtual Office Platform on the Work-Life Balances. *Plos One*. <u>https://doi.org/10.1371/journal.pone.0260220</u>
- Rauch, M., & Ansari, S. (2022). Waging War From Remote Cubicles: How Workers Cope With Technologies That Disrupt the Meaning and Morality of Their Work. *Organization Science*. <u>https://doi.org/10.1287/orsc.2021.1555</u>

- Rauv, S. (2022). Working from Home In 2023: The Hybrid Workforce Is Here to Stay. https://www.elcom.com.au/resources/blog/working-from-home-the-hybridworkforce-is-here-to-stay
- Raykov, M. (2014). Employer Support for Innovative Work and Employees' Job Satisfaction and Job - related Stress. *Journal of Occupational Health*. <u>https://doi.org/10.1539/joh.12-0192-oa</u>
- Raymundo, T. M., & Castro, C. d. S. S. (2019). Analysis of a training program for older workers in the use of technology: difficulties and satisfaction. *Revista Brasileira de Geriatria e Gerontologia*, 22(05), e190039.
- Rayner, J., & Morgan, D. (2017). An Empirical Study of 'Green' Workplace Behaviours: Ability, Motivation and Opportunity. Asia Pacific Journal of Human Resources. <u>https://doi.org/10.1111/1744-7941.12151</u>
- Rejeb, W. B., Berraies, S., & Talbi, D. (2019). The Contribution of Board of Directors' Roles to Ambidextrous Innovation. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/ejim-06-2018-0110</u>
- Reuvers, M., Van Engen, M. L., Vinkenburg, C. J., & Wilson Evered, E. (2008). Transformational leadership and innovative work behaviour: Exploring the relevance of gender differences. *Creativity and innovation management*, 17(3), 227-244.
- Risi, E., & Pronzato, R. (2021). Remote Work Between Narratives of Indipendence and Fractured Experiences. A Qualitative Research During the Pandemic Crisis. *Aoir Selected Papers of Internet Research*. https://doi.org/10.5210/spir.v2021i0.12234
- Robinson, R. S. (2014). Purposive Sampling. 5243-5245. <u>https://doi.org/10.1007/978-94-007-0753-5_2337</u>
- Rodrigues, A. F., & Veloso, A. (2013). Organizational Trust, Risk and Creativity. *Review of Business Management*. <u>https://doi.org/10.7819/rbgn.v15i49.1334</u>
- Rodrigues, N., & Rebelo, T. (2022). Portuguese Version of Parker's (1998) Role Breadth Self-Efficacy Scale: Examining Its Factor Structure, Reliability and Validity to Predict Innovative Work Behaviour. *Psychological Reports*, 125(6), 3263-3288. <u>https://doi.org/10.1177/00332941211037609</u>
- Rudzin, J. E., Soule, D. C., Whitaker, J. M., M, N. B. H., Clayton, S., & Fogaren, K. E. (2022). Catalyzing Remote Collaboration During the COVID-19 Pandemic and Beyond: Early Career Oceanographers Adopt Hybrid Open Science Framework. *Frontiers in Marine Science*. <u>https://doi.org/10.3389/fmars.2022.855192</u>
- Rune Ellemose, G. (2016). Connecting culture to creativity and innovation: how trust and other variations of corporate culture influence innovative behaviour [article]. *International Journal of Sustainable Economy*(4), 342.

https://search.ebscohost.com/login.aspx?direct=true&AuthType=shib&db=edsrep&AN=edsrep.a.ids.ijsuse.v8y2016i4p342.356&site=eds-live&custid=s1145751

- Russo, S. D., Antino, M., Zaniboni, S., Caetano, A., & Truxillo, D. M. (2020). The Effect of Age on Daily Positive Emotions and Work Behaviors. *Work Aging and Retirement*. <u>https://doi.org/10.1093/workar/waz026</u>
- Rutherford, A. (2011). ANOVA and ANCOVA : A GLM Approach. John Wiley & Sons, Incorporated. <u>http://ebookcentral.proquest.com/lib/vu/detail.action?docID=1011369</u>
- Sadia Tabassam, A., & Aman, A. (2018). Transformational Leadership and Organizational Performance. *Seisense Journal of Management*. <u>https://doi.org/10.33215/sjom.v1i3.28</u>
- Şahin, D. (2023). Mediating Effect of Creativity on How Burnout Affects Social Media Use: An Examination on Physicians. SAGE Open. https://doi.org/10.1177/21582440231218528
- Saini, G. K., & Jawahar, I. M. (2019). The Influence of Employer Rankings, Employment Experience, and Employee Characteristics on Employer Branding as an Employer of Choice. *Career Development International*. <u>https://doi.org/10.1108/cdi-11-2018-0290</u>
- Salaran, M. M., & Maritz, A. (2013). The Role of Social Ties in Fostering Innovation. Journal of Enterprising Culture. <u>https://doi.org/10.1142/s0218495813500106</u>
- Salazar, M. R., Feitosa, J., & Salas, E. (2017). Diversity and Team Creativity: Exploring Underlying Mechanisms. Group Dynamics Theory Research and Practice. <u>https://doi.org/10.1037/gdn0000073</u>
- Samani, S. A., Rasid, S. Z. B. A., & Sofian, S. (2015). Individual Control Over the Physical Work Environment to Affect Creativity. *Industrial Engineering & Management Systems*. <u>https://doi.org/10.7232/iems.2015.14.1.094</u>
- Sandoval-Reyes, J., Carlier, S. I., & Oliva, E. J. D. (2021). Remote Work, Work Stress, and Work–Life During Pandemic Times: A Latin America Situation. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph18137069</u>
- Santana, F. N., Wagner, C. H., Rubin, N. B., Bloomfield, L. S. P., Bower, E., Fischer, S. L., Santos, B. S., Smith, G., Muraida, C. T., & Wong Parodi, G. (2021). A Path Forward for Qualitative Research on Sustainability in the COVID-19 Pandemic. *Sustainability Science*, 16(3), 1061-1067. https://doi.org/10.1007/s11625-020-00894-8
- Santiago, H. (2023). From Idea Generation to Implementation: Transitioning Ideas Through Innovation Process Stages. *International Leadership Journal*, 15(2), 3-23.

- Sarbu, M. (2022, 2022/11/01/). Does telecommuting kill service innovation? Research in Transportation Economics, 95, 101206. <u>https://doi.org/https://doi.org/10.1016/j.retrec.2022.101206</u>
- Sari, D. K., Yudiarso, A., & Sinambela, F. C. (2021). Work Engagement and Innovative Work Behavior: Meta-Analysis Study. <u>https://doi.org/10.2991/assehr.k.210423.053</u>
- Saunders, M., Lewis, P., Thornhill, A., & Bristow, A. (2019). Research Methods for Business Students. In (Eighth Edition ed., pp. 128-171). Pearson Education Limited.
- Saurombe, M. D., Rayners, S. S., Mokgobu, K. A., & Manka, K. (2022). The Perceived Influence of Remote Working on Specific Human Resource Management Outcomes During the COVID-19 Pandemic. Sa Journal of Human Resource Management. <u>https://doi.org/10.4102/sajhrm.v20i0.2033</u>
- Savolainen, I., Oksa, R., Savela, N., Celuch, M., & Oksanen, A. (2021). COVID-19 Anxiety—A Longitudinal Survey Study of Psychological and Situational Risks Among Finnish Workers. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph18020794</u>
- Sawilowsky, S. S., & Blair, R. C. (1992). A More Realistic Look at the Robustness and Type II Error Properties of the T Test to Departures From Population Normality. *Psychological Bulletin*, 111(2), 352-360. <u>https://doi.org/10.1037/0033-</u> 2909.111.2.352
- Schindler, P. S. (2018). *Business Research Methods*. McGraw-Hill Higher Education. <u>http://ebookcentral.proquest.com/lib/vu/detail.action?docID=5834197</u>
- Schmidt, S. (2009). Shall We Really Do It Again? The Powerful Concept of Replication Is Neglected in the Social Sciences. *Review of General Psychology*. <u>https://doi.org/10.1037/a0015108</u>
- Schmoll, R., & Süß, S. (2019). Working Anywhere, Anytime: An Experimental Investigation of Workplace Flexibility's Influence on Organizational Attraction. *Management Revue*. <u>https://doi.org/10.5771/0935-9915-2019-1-40</u>
- Schumpeter, J. A. (1934). The theory of economic development: An inquiry into profits, capital, credit, interest, and the business cycle. Harvard University Press Cambridge, MA.
- Scotland, J. (2012). Exploring the Philosophical Underpinnings of Research: Relating Ontology and Epistemology to the Methodology and Methods of the Scientific, Interpretive, and Critical Research Paradigms. *English Language Teaching*, 5(9). <u>https://doi.org/10.5539/elt.v5n9p9</u>
- Scott, S. G., & Bruce, R. A. (1994). Determinants of innovative behavior: A path model of individual innovation in the workplace. *Academy of Management Journal*, 37(3), 580-607.

- Seddigh, A., Stenfors, C., Berntson, E., Bååth, R., Sikström, S., & Westerlund, H. (2015). The Association Between Office Design and Performance on Demanding Cognitive Tasks. *Journal of Environmental Psychology*. <u>https://doi.org/10.1016/j.jenvp.2015.05.001</u>
- Sener, I. N., & Bhat, C. R. (2011). A Copula-Based Sample Selection Model of Telecommuting Choice and Frequency. *Environment and Planning a Economy* and Space. <u>https://doi.org/10.1068/a43133</u>
- Shaed, M. M., Zainol, I. N. B. H., Yusof, M. B. M., & Bahrin, F. K. (2018). Types of Employee Participation in Decision Making (PDM) Amongst the Middle Management in the Malaysian Public Sector. *International Journal of Asian Social Science*. <u>https://doi.org/10.18488/journal.1.2018.88.603.613</u>
- Shafique, M. (2013). Thinking Inside the Box? Intellectual Structure of the Knowledge Base of Innovation Research (1988-2008). *Strategic Management Journal*, 34(1), 62-93. <u>https://doi.org/10.1002/smj.2002</u>
- Shahid, M., Chaudhry, S., Bilal, M., Amber, H., Aslam, S., Malik, S., & Shahzad, K. (2022). The Link Between Team Identification, Entrepreneurial Orientation, and Innovative Work Behavior and Its Dimensions in the Context of Pakistan. SAGE Open, 12(1), 21582440221079893. <u>https://doi.org/10.1177/21582440221079893</u>
- Shahzad, K., Arenius, P., Muller, A., Rasheed, M., & Bajwa, S. U. (2019). Unpacking the Relationship Between High-Performance Work Systems and Innovation Performance in SMEs. *Personnel Review*. <u>https://doi.org/10.1108/pr-10-2016-0271</u>
- Shai, O., Reich, Y., & Rubin, D. (2009). Creative Conceptual Design: Extending the Scope by Infused Design. Computer-Aided Design. <u>https://doi.org/10.1016/j.cad.2007.11.004</u>
- Shailja, S., Kumari, P., & Singla, H. (2023). Impact of Servant Leadership On innovative Work Behaviour: A moderation Mediation Analysis. *Leadership* & Organization Development Journal. <u>https://doi.org/10.1108/lodj-03-2022-0121</u>
- Shanker, R., Bhanugopan, R., Heijden, B. v. d., & Farrell, M. (2017). Organizational Climate for Innovation and Organizational Performance: The Mediating Effect of Innovative Work Behavior. *Journal of Vocational Behavior*. <u>https://doi.org/10.1016/j.jvb.2017.02.004</u>
- Sharifirad, M. S., & Ataei, V. (2012). Organizational Culture and Innovation Culture: Exploring the Relationships Between Constructs. *Leadership & Organization Development Journal*. <u>https://doi.org/10.1108/01437731211241274</u>
- Shatz, I. (2023). Assumption-Checking Rather Than (Just) Testing: The Importance of Visualization and Effect Size in Statistical Diagnostics. *Behavior Research Methods*, 56(2), 826-845. <u>https://doi.org/10.3758/s13428-023-02072-x</u>

- Shelton, C., Smith, A. F., & Mort, M. (2014). Opening Up the Black Box: An Introduction to Qualitative Research Methods in Anaesthesia. *Anaesthesia*, 69(3), 270-280. <u>https://doi.org/10.1111/anae.12517</u>
- Sheth, A., & Sinfield, J. V. (2022). An analytical framework to compare innovation strategies and identify simple rules. *Technovation*, 115.
- Shimura, A., Yokoi, K., Ishibashi, Y., Akatsuka, Y., & Inoue, T. (2021). Remote Work Decreases Psychological and Physical Stress Responses, but Full-Remote Work Increases Presenteeism. *Frontiers in Psychology*. https://doi.org/10.3389/fpsyg.2021.730969
- Shipton, H., Sparrow, P., Budhwar, P., & Brown, A. (2017). HRM and Innovation: Looking Across Levels. *Human Resource Management Journal*. <u>https://doi.org/10.1111/1748-8583.12102</u>
- Shipton, H., West, M., Dawson, J., Birdi, K., & Patterson, M. (2006). HRM as a Predictor of Innovation. *Human Resource Management Journal*. <u>https://doi.org/10.1111/j.1748-8583.2006.00002.x</u>
- Shockley, K. M., Allen, T. D., Dodd, H. R., & Waiwood, A. M. (2021). Remote Worker Communication During COVID-19: The Role of Quantity, Quality, and Supervisor Expectation-Setting. *Journal of Applied psychology*. <u>https://doi.org/10.1037/ap10000970</u>
- Shorten, A., & Smith, J. (2017). Mixed Methods Research: Expanding the Evidence Base. *Evidence-Based Nursing*, 20(3), 74-75. <u>https://doi.org/10.1136/eb-2017-102699</u>
- Sidi, Y., Glikson, E., & Cheshin, A. (2021). Do You Get What I Mean?!? The Undesirable Outcomes of (Ab)Using Paralinguistic Cues in Computer-Mediated Communication. *Frontiers in Psychology*. https://doi.org/10.3389/fpsyg.2021.658844
- Sijtsma, K. (2008). On the Use, the Misuse, and the Very Limited Usefulness Of Cronbach's Alpha. *Psychometrika*, 74(1), 107-120. <u>https://doi.org/10.1007/s11336-008-9101-0</u>
- Simon, P., Björklund, T., & Sheppard, S. (2019). Early-Career Engineers' Perceptions of Support for Innovation at the Workplace - What Seems to Matter. *Proceedings of the Design Society International Conference on Engineering Design*. <u>https://doi.org/10.1017/dsi.2019.181</u>
- Singh, A. S., & Masuku, M. B. (2014). Sampling techniques & determination of sample size in applied statistics research: An overview. *International Journal of economics, commerce and management, 2*(11), 1-22.

- Singh, H., Cascini, G., & McComb, C. (2022). Virtual and Face-to-Face Team Collaboration Comparison Through an Agent-Based Simulation. *Journal of Mechanical Design*. <u>https://doi.org/10.1115/1.4054242</u>
- Singh, M., & Sarkar, A. (2012). The relationship between psychological empowerment and innovative behavior. *Journal of Personnel Psychology*.
- Singh, Y., Bamel, U. K., & Vohra, V. (2020). The Mediating Effect of Meaningful Work Between Human Resource Practices and Innovative Work Behavior: A Study of Emerging Market. *Employee Relations*. <u>https://doi.org/10.1108/er-04-2020-0150</u>
- Škerlavaj, M., Černe, M., & Dysvik, A. (2014). I Get by With a Little Help From My Supervisor: Creative-Idea Generation, Idea Implementation, and Perceived Supervisor Support. *The leadership quarterly*. <u>https://doi.org/10.1016/j.leaqua.2014.05.003</u>
- Skountridaki, L. (2023). Missing Voices: Office Space Discontent as a Driving Force in Employee Hybrid Work Preferences. *Industrial Relations Journal*, 55(1), 54-77. <u>https://doi.org/10.1111/irj.12415</u>
- Slåtten, T., Mutonyi, B. R., & Lien, G. (2020). The Impact of Individual Creativity, Psychological Capital, and Leadership Autonomy Support on Hospital Employees' Innovative Behaviour. BMC Health Services Research. <u>https://doi.org/10.1186/s12913-020-05954-4</u>
- Smite, D., Moe, N. B., Hildrum, J., Gonzalez-Huerta, J., & Mendez, D. (2023, 2023/01/01/). Work-from-home is here to stay: Call for flexibility in postpandemic work policies. *Journal of Systems and Software*, 195, 111552. <u>https://doi.org/https://doi.org/10.1016/j.jss.2022.111552</u>
- Smith, K., Pickering, A. D., & Bhattacharya, J. (2022). The Creative Life: A Daily Diary Study of Creativity, Affect, and Well-Being in Creative Individuals. <u>https://doi.org/10.31234/osf.io/eaust</u>
- Sobaih, A. E. E., Coleman, P., Ritchie, C., & Jones, E. (2011). Part-Time Restaurant Employee Perceptions of Management Practices: An Empirical Investigation. Service Industries Journal. <u>https://doi.org/10.1080/02642069.2011.573659</u>
- Soewarno, N. N., Tjahjadi, B., & Fithrianti, F. (2019). Green Innovation Strategy and Green Innovation. *Management Decision*. <u>https://doi.org/10.1108/md-05-2018-0563</u>
- Soja, E., & Soja, P. (2020). Fostering ICT Use by Older Workers. *Journal of Enterprise* Information Management. <u>https://doi.org/10.1108/jeim-12-2018-0282</u>
- Sonenshein, S. (2014). How Organizations Foster the Creative Use of Resources. Academy of Management Journal. <u>https://doi.org/10.5465/amj.2012.0048</u>

- Sönmez, B., & Yıldırım, A. (2019). The Mediating Role of Autonomy in the Effect of Pro-Innovation Climate and Supervisor Supportiveness on Innovative Behavior of Nurses. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/ejim-05-2018-0088</u>
- Soomar, Z. (2020). A Framework for Building and Maintain Trust in Remote and Virtual Teams. *F1000research*. <u>https://doi.org/10.12688/f1000research.26626.1</u>
- Soputan, G. J., & Sumual, T. E. M. (2022). Employee' Innovative Work Behavior. SHS Web of Conferences, 149, 02014. <u>https://doi.org/10.1051/shsconf/202214902014</u>
- Sørensen, E., & Torfing, J. (2016). Co-Initiation of Collaborative Innovation in Urban Spaces. Urban Affairs Review. <u>https://doi.org/10.1177/1078087416651936</u>
- Spiegelaere, S. D., Gyes, G. V., Witte, H. D., Niesen, W., & Hootegem, G. V. (2014). On the Relation of Job Insecurity, Job Autonomy, Innovative Work Behaviour and the Mediating Effect of Work Engagement. *Creativity and innovation management*. <u>https://doi.org/10.1111/caim.12079</u>
- Stam, E., & Wennberg, K. (2009). The Roles of R&D in New Firm Growth. Small Business Economics. <u>https://doi.org/10.1007/s11187-009-9183-9</u>
- Stanisçuaski, F., Kmetzsch, L., Soletti, R. C., Reichert, F. M., Zandonà, E., Ludwig, Z. M. D. C., Lima, E. F., Neumann, A., Schwartz, I. V. D., Mello-Carpes, P. B., Tamajusuku, A. S. K., Werneck, F. P., Ricachenevsky, F. K., Infanger, C., Seixas, A., Staats, C. C., & Oliveira, L. d. (2021). Gender, Race and Parenthood Impact Academic Productivity During the COVID-19 Pandemic: From Survey to Action. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.663252</u>
- Stavrova, O., Spiridonova, T., Calseyde, P. v. d., Meyers, C. A., & Evans, A. M. (2023). Does Remote Work Erode Trust in Organizations? A Within - person Investigation in the COVID - 19 Context. Social and Personality Psychology Compass. <u>https://doi.org/10.1111/spc3.12762</u>
- Stentz, J. E., Clark, V. L. P., & Matkin, G. S. (2012). Applying mixed methods to leadership research: A review of current practices. *The leadership quarterly*, 23(6), 1173-1183.
- Sterling, A., & Boxall, P. (2013). Lean production, employee learning and workplace outcomes: a case analysis through the ability - motivation - opportunity framework. *Human Resource Management Journal*, 23(3), 227-240.
- Stewart, J. S., Oliver, E. G., Cravens, K. S., & Oishi, S. (2017). Managing millennials: Embracing generational differences. *Business Horizons*, 60(1), 45-54.
- Stoffers, J., & Heijden, B. I. J. M. v. d. (2018). An Innovative Work Behaviour-Enhancing Employability Model Moderated by Age. *European Journal of Training and Development*. <u>https://doi.org/10.1108/ejtd-10-2016-0074</u>

- Stonehouse, J. M., & Forrester, G. J. (1998, 1998/02/01). Robustness of the t and U tests under combined assumption violations. *Journal of Applied Statistics*, 25(1), 63-74. <u>https://doi.org/10.1080/02664769823304</u>
- Streiner, D. L. (2003). Being Inconsistent About Consistency: When Coefficient Alpha Does and Doesn't Matter. *Journal of Personality Assessment*, 80(3), 217-222. <u>https://doi.org/10.1207/s15327752jpa8003_01</u>
- Striukova, L., & Rayna, T. (2015). University-Industry Knowledge Exchange. European Journal of Innovation Management, 18(4), 471-492. <u>https://doi.org/10.1108/ejim-10-2013-0098</u>
- Sudeshna Basu, M., & Anjali, R. (2009). Innovative work behavior of managers: Implications regarding stressful challenges of modernized public and privatesector organizations. *Industrial Psychiatry Journal*, 18, 101-107. <u>https://doi.org/10.4103/0972-6748.62269</u>
- Sullivan, G. M., & Artino, A. R. (2013). Analyzing and Interpreting Data From Likert-Type Scales. *Journal of Graduate Medical Education*, 5(4), 541-542. <u>https://doi.org/10.4300/jgme-5-4-18</u>
- Sun, Y., Tuertscher, P., Majchrzak, A., & Malhotra, A. (2020). Pro-Socially Motivated Interaction for Knowledge Integration in Crowd-Based Open Innovation. *Journal of Knowledge Management*. <u>https://doi.org/10.1108/jkm-04-2020-0303</u>
- Suortti, C., & Sivunen, A. (2023). Communicative Tensions in Remote Work During the COVID-19 Pandemic. *Management Communication Quarterly*. <u>https://doi.org/10.1177/08933189231199052</u>
- Supriyanto, A., Ekowati, V. M., Rokhman, W., Ahamed, F., Munir, M., & Miranti, T. (2023). Empowerment Leadership as a Predictor of the Organizational Innovation in Higher Education. *International Journal of Professional Business Review*. <u>https://doi.org/10.26668/businessreview/2023.v8i2.1538</u>
- Swain, J. (2018). A Hybrid Approach to Thematic Analysis in Qualitative Research: Using a Practical Example. <u>https://doi.org/10.4135/9781526435477</u>
- Tabiu, A., Pangil, F., & Othman, S. Z. (2016). HRM practive and employee performance: mediation of ability, motivation and opportunity. *Asian Journal of Multidisciplinary Studies*, 4(6), 130-136.
- Takeuchi, R., Guo, N., Teschner, R. S., & Kautz, J. (2021). Reflecting on Death Amidst COVID-19 and Individual Creativity: Cross-Lagged Panel Data Analysis Using Four-Wave Longitudinal Data. *Journal of Applied psychology*. <u>https://doi.org/10.1037/ap10000949</u>
- Tams, S., & Dulipovici, A. (2019). The Creativity Model of Age and Innovation With IT: How to Counteract the Effects of Age Stereotyping on User Innovation. <u>https://doi.org/10.24251/hicss.2019.785</u>

- Tan, C.-Y., Chuah, C.-Q., & Lee, S.-T. (2021). Being Creative Makes You Happier: The Positive Effect of Creativity on Subjective Well-Being. International Journal of Environmental Research and Public Health. https://doi.org/10.3390/ijerph18147244
- Tan, J., & Sim, J. (2022). Driving Employee-Driven Innovation Through Workplace Learning: The Story of Singapore SMEs. *The Hungarian Educational Research Journal*, 12(4), 432-444. <u>https://doi.org/10.1556/063.2021.00103</u>
- Tang, Y., Shao, Y., & Chen, Y. (2019). Assessing the Mediation Mechanism of Job Satisfaction and Organizational Commitment on Innovative Behavior: The Perspective of Psychological Capital. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2019.02699</u>
- Taştan, S. B., & Davoudi, S. M. M. (2015, 2015/05/11/). An Examination of the Relationship between Leader-member Exchange and Innovative Work Behavior with the Moderating Role of Trust in Leader: A Study in the Turkish Context. *Procedia - Social and Behavioral Sciences*, 181, 23-32. <u>https://doi.org/https://doi.org/10.1016/j.sbspro.2015.04.862</u>
- Taylor, S. (2017). What Is Innovation? A Study of the Definitions, Academic Models and Applicability of Innovation to an Example of Social Housing in England. *Open Journal of Social Sciences*. <u>https://doi.org/10.4236/jss.2017.511010</u>
- Telstra. (2021). Busting the productivity myth: Hybrid working in Australia. Telstra. <u>https://exchange.telstra.com.au/wp-</u> <u>content/uploads/2021/10/Telstra_HybridWorkingReport.pdf</u>
- Terry, G., Hayfield, N., Clarke, V., & Braun, V. (2017). Thematic analysis. *The SAGE* handbook of qualitative research in psychology, 2(17-37), 25.
- Theurer, C. P., Tumasjan, A., & Welpe, I. M. (2018). Contextual Work Design and Employee Innovative Work Behavior: When Does Autonomy Matter? *Plos One*. <u>https://doi.org/10.1371/journal.pone.0204089</u>
- Thevanes, N., & Harikaran, S. (2020). Work-Life Balance and Organizational Performance: The Mediating Role of Organizational Citizenship Behavior. *Asian Journal of Social Sciences and Management Studies*, 7(4), 280-287. <u>https://doi.org/10.20448/journal.500.2020.74.280.287</u>
- Thi Minh Ly, P., Tien Thanh, P., Duy, L. T., Nghi, C. N. P., Giao, N. D. P., & Nghi, T. M. (2023). Online knowledge sharing and creativity in the context of working from home during the COVID-19 pandemic. *VINE: The Journal of Information & amp; Knowledge Management Systems*, 53(2), 292-314. https://doi.org/10.1108/VJIKMS-03-2022-0078
- Thiese, M. S., Ronna, B., & Ott, U. (2016). P Value Interpretations and Considerations. *Journal of Thoracic Disease*, 8(9), E928-E931. <u>https://doi.org/10.21037/jtd.2016.08.16</u>

- Thomas, V., Carr, B. B., Azmitia, M., & Whittaker, S. (2021). Alone and Online: Understanding the Relationships Between Social Media, Solitude, and Psychological Adjustment. *Psychology of Popular Media*. <u>https://doi.org/10.1037/ppm0000287</u>
- Thompson, V. A. (1965). Bureaucracy and Innovation. *Administrative science* quarterly, 10(1), 1-20. <u>https://doi.org/10.2307/2391646</u>
- Thoring, K., Desmet, P., & Badke-Schaub, P. (2018). Creative Environments for Design Education and Practice: A Typology of Creative Spaces. *Design Studies*. <u>https://doi.org/10.1016/j.destud.2018.02.001</u>
- Thrane, C. (2023). *Doing statistical analysis : a student's guide to quantitative research / Christer Thrane*. Routledge.
- Thunberg, S., & Arnell, L. (2021). Pioneering the Use of Technologies in Qualitative Research – A Research Review of the Use of Digital Interviews. *International Journal of Social Research Methodology*. <u>https://doi.org/10.1080/13645579.2021.1935565</u>
- Thurlings, M., Evers, A., & Vermeulen, M. (2015). Toward a Model of Explaining Teachers' Innovative Behavior. *Review of Educational Research*. <u>https://doi.org/10.3102/0034654314557949</u>
- Tidd, J., & Bessant, J. (2018). Innovation management challenges: From fads to fundamentals. *International Journal of Innovation Management*, 22(05), 1840007.
- Tkalich, A., Šmite, D., Andersen, N. H., & Moe, N. B. (2023). What Happens to Psychological Safety When Going Remote? *Ieee Software*. <u>https://doi.org/10.1109/ms.2022.3225579</u>
- Toh, T. L., Tee, N. E., Ping, K. J., & Yuan, C. K. (2022). Factors Affecting Remote Working Productivity: A Study at Higher Education Institution in Sibu, Sarawak. *The International Journal of Social Sciences and Humanities Invention*. <u>https://doi.org/10.18535/ijsshi/v9i04.05</u>
- Tomić, D., & Vizinger, K. (2023). Effects of Remote Business During the Covid-19 Pandemic - A Literature Review. American Journal of Economics and Business Innovation. <u>https://doi.org/10.54536/ajebi.v2i1.1088</u>
- Tønnessen, Ø. (2022). Employee Creativity in Coworking Spaces: Towards a Conceptual Framework. *European Journal of Workplace Innovation*. <u>https://doi.org/10.46364/ejwi.v7i1.891</u>
- Tønnessen, Ø., Dhir, A., & Flåten, B.-T. (2021, 2021/09/01/). Digital knowledge sharing and creative performance: Work from home during the COVID-19 pandemic. *Technological Forecasting and Social Change*, 170, 120866. https://doi.org/https://doi.org/10.1016/j.techfore.2021.120866

- Toscano, F., & Zappala, S. (2020). Social Isolation and Stress as Predictors of Productivity Perception and Remote Work Satisfaction During the COVID-19 Pandemic: The Role of Concern About the Virus in a Moderated Double Mediation. Sustainability. <u>https://doi.org/10.3390/su12239804</u>
- Tramontano, C., Grant, C. S., & Clarke, C. E. (2021). Development and Validation of the E-Work Self-Efficacy Scale to Assess Digital Competencies in Remote Working. *Computers in Human Behavior Reports*. <u>https://doi.org/10.1016/j.chbr.2021.100129</u>
- Treacy, S. H. (2022). Digitally Transforming Organisational Cultures: Ensuring Enhanced Innovation in a Remote Working World. *European Conference on Innovation and Entrepreneurship*. <u>https://doi.org/10.34190/ecie.17.1.370</u>
- Trenerry, B., Chng, S., Wang, Y., Suhaila, Z. S., Lim, S. S., Lü, H., & Oh, P. H. (2021). Preparing Workplaces for Digital Transformation: An Integrative Review and Framework of Multi-Level Factors. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.620766</u>
- Trochim, W. M. K., Donnelly, J. P., & Arora, K. (2016). *Research methods : the essential knowledge base* (Second edition. ed.). Cengage Learning.
- Turgut, E., & Sökmen, A. (2018). The Effects of Perceived Organizational Ethics on Innovative Work Behavior: Self Efficacy's Moderating and Mediating Role. *Turkish Journal of Business Ethics*, 11(1), 58-67. <u>https://doi.org/10.12711/tjbe.2018.11.1.0007</u>
- Turner, S., D'Lima, D., Hudson, E., Morris, S., Sheringham, J., Swart, N., & Fulop, N. (2017). Evidence Use in Decision-Making on Introducing Innovations: A Systematic Scoping Review With Stakeholder Feedback. *Implementation Science*. <u>https://doi.org/10.1186/s13012-017-0669-6</u>
- Tuzovic, S., & Kabadayi, S. (2020). The Influence of Social Distancing on Employee Well-Being: A Conceptual Framework and Research Agenda. *Journal of Service Management*. <u>https://doi.org/10.1108/josm-05-2020-0140</u>
- Ude, S. O., & Diala, I. (2015). Support for Creativity and Innovation, Resistance to Change, Organizational Commitment and Motivation. *International Journal of Computer & Organization Trends*. <u>https://doi.org/10.14445/22492593/ijcotv22p301</u>
- Umrani, W. A., Afsar, B., Khan, M., & Ahmed, U. (2019). Addressing the Issue of Job Performance Among Hospital Physicians in Pakistan: The Role of Job Security, Organizational Support, and Job Satisfaction. *Journal of Applied Biobehavioral Research*. <u>https://doi.org/10.1111/jabr.12169</u>
- Utama, D., & Soetjipto, B. (2022). Work-life Conflict and Innovative Behavior Examination in Telecommuting Era Perspective. Proceedings of the 4th International Conference on Economics, Business and Economic Education Science, ICE-BEES 2021, 27-28 July 2021, Semarang, Indonesia,

- Uzkurt, C., Kumar, R., Kimzan, H. S., & Eminoğlu, G. M. (2013). Role of Innovation in the Relationship Between Organizational Culture and Firm Performance. *European Journal of Innovation Management*. <u>https://doi.org/10.1108/14601061311292878</u>
- Van de Ven, A. H. (1986). Central problems in the management of innovation. *Management science*, 32(5), 590-607.
- van der Meulen, N., van Baalen, P., & van Heck, E. (2012). Please, do not disturb. Telework, distractions, and the productivity of the knowledge worker.
- Vandelanotte, C., Duncan, M. J., Short, C. E., Browne, M., Ronan, K. R., Happell, B., & Milia, L. D. (2013). Associations Between Occupational Indicators and Total, Work-Based and Leisure-Time Sitting: A Cross-Sectional Study. *BMC Public Health*. <u>https://doi.org/10.1186/1471-2458-13-1110</u>
- Veenendaal, A., & Bondarouk, T. (2015). Perceptions of HRM and their effect on dimensions of innovative work behaviour: Evidence from a manufacturing firm. *Management Revue*, 26(2), 138-160. <u>http://wallaby.vu.edu.au:2048/login?url=http://search.ebscohost.com/login.aspx?</u> <u>direct=true&db=edsjsr&AN=edsjsr.24570255&site=eds-live</u>
- Vega, R. P., Anderson, A. J., & Kaplan, S. A. (2015). A Within-Person Examination of the Effects of Telework. *Journal of Business and Psychology*, 30(2), 313-323. <u>http://www.jstor.org/stable/24634513</u>
- Vegt, G. S. v. d., & Janssen, O. (2003). Joint Impact of Interdependence and Group Diversity on Innovation. *Journal of Management*. <u>https://doi.org/10.1016/s0149-2063_03_00033-3</u>
- Velina, M., Valeinis, J., Greco, L., & Luta, G. (2016). Empirical Likelihood-Based ANOVA for Trimmed Means. *International Journal of Environmental Research* and Public Health, 13(10), 953. <u>https://doi.org/10.3390/ijerph13100953</u>
- Viererbl, B., Denner, N., & Koch, T. (2022). "You Don't Meet Anybody When Walking From the Living Room to the Kitchen": Informal Communication During Remote Work. *Journal of Communication Management*. <u>https://doi.org/10.1108/jcom-10-2021-0117</u>
- Vij, A., Souza, F. F., Barrie, H., Anilan, V., Sarmiento, S., & Washington, L. (2023, 2023/10/01/). Employee preferences for working from home in Australia. *Journal of Economic Behavior & Organization*, 214, 782-800. <u>https://doi.org/https://doi.org/10.1016/j.jebo.2023.08.020</u>
- Vincenzi, C. D., Pansini, M., Ferrara, B., Buonomo, I., & Benevene, P. (2022). Consequences of COVID-19 on Employees in Remote Working: Challenges, Risks and Opportunities an Evidence-Based Literature Review. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph191811672</u>

- Voordt, T. J. M. (2004, 2004/04/01). Productivity and employee satisfaction in flexible workplaces. *Journal of Corporate Real Estate*, 6(2), 133-148. https://doi.org/10.1108/14630010410812306
- Vukašinović, N., Benedičič, J., & Žavbi, R. (2020). Evolution of Communication Skills in Virtual Product Development Process: Experience From EGPR. <u>https://doi.org/10.5772/intechopen.90059</u>
- Wah, Y. B., & Sim, C. H. (2011). Comparisons of Various Types of Normality Tests. Journal of Statistical Computation and Simulation, 81(12), 2141-2155. <u>https://doi.org/10.1080/00949655.2010.520163</u>
- Wahyuni, W., Sutanto, B., & Supadi, S. (2021). The Mediating Role of Organizational Learning in the Relationship Between Organizational Commitment and Lecturer Innovative Behavior. *Jrti (Jurnal Riset Tindakan Indonesia)*. <u>https://doi.org/10.29210/3003673000</u>
- Waizenegger, L., McKenna, B., Cai, W., & Bendz, T. (2020). An Affordance Perspective of Team Collaboration and Enforced Working From Home During COVID-19. *European Journal of Information Systems*. <u>https://doi.org/10.1080/0960085x.2020.1800417</u>
- Walliman, N. (2018). *Research methods : the basics / Nicholas Walliman* (Second edition. ed.). Routledge.
- Wang, B., Liu, Y., Qian, J., & Parker, S. K. (2021, 2021/01/01/2021). Achieving Effective Remote Working During the COVID - 19 Pandemic: A Work Design Perspective. *Applied Psychology*, 70, 16-59.
- Wang, C., Change, C.-H., & Lee, Z. C. R. (2022). Business-to-Business Platform Ecosystem Practices and Their Impacts on Firm Performance: Evidence From High-Tech Manufacturing Firms. *Journal of Manufacturing Technology Management*. <u>https://doi.org/10.1108/jmtm-07-2021-0253</u>
- Wang, E. (2014). Moderators of the Relationship Between Social Bonding and Organizational Commitment. *Managing Service Quality*. <u>https://doi.org/10.1108/msq-08-2013-0157</u>
- Wang, J., Yang, J., & Xue, Y. (2017). Subjective Well-Being, Knowledge Sharing and Individual Innovation Behavior. *Leadership & Organization Development Journal*. <u>https://doi.org/10.1108/lodj-10-2015-0235</u>
- Wang, W. (2022). Toward Economic Growth and Value Creation Through Social Entrepreneurship: Modelling the Mediating Role of Innovation. Frontiers in Psychology. <u>https://doi.org/10.3389/fpsyg.2022.914700</u>
- Wang, W., Albert, L., & Sun, Q. (2020). Employee isolation and telecommuter organizational commitment. *Employee Relations: The International Journal*, 42(3), 609-625. <u>https://doi.org/10.1108/er-06-2019-0246</u>

- Wang, X., Fang, Y., Qureshi, I., & Janssen, O. (2015). Understanding Employee Innovative Behavior: Integrating the Social Network and Leader-Member Exchange Perspectives. *Journal of organizational behavior*. <u>https://doi.org/10.1002/job.1994</u>
- Wang, X., Zhang, Z., & Chun, D. (2021). How does mobile workplace stress affect employee innovative behavior? The role of work–family conflict and employee engagement. *Behavioral Sciences*, *12*(1), 2.
- Wang, Y., Gil, P. R. d., Chen, Y.-H., Kromrey, J. D., Kim, E. S., Pham, T. V., Nguyen, D., & Romano, J. (2016). Comparing the Performance of Approaches for Testing the Homogeneity of Variance Assumption in One-Factor ANOVA Models. *Educational and Psychological Measurement*, 77(2), 305-329. <u>https://doi.org/10.1177/0013164416645162</u>
- Wang, Y., Tang, C., Naumann, S. E., & Wang, Y. (2017). Paternalistic Leadership and Employee Creativity: A Mediated Moderation Model. *Journal of Management* & Organization. <u>https://doi.org/10.1017/jmo.2017.8</u>
- Wang, Y. X., Yang, Y. J., Wang, Y., Su, D., Li, S. W., Zhang, T., & Li, H. P. (2019). The mediating role of inclusive leadership: Work engagement and innovative behaviour among Chinese head nurses. *Journal of Nursing Management*, 27(4), 688-696. <u>https://doi.org/10.1111/jonm.12754</u>
- Wang, Z., Gao, M., & Panaccio, A. (2020). A Self Determination Approach to Understanding Individual Values as an Interaction Condition on Employees' Innovative Work Behavior in the High - Tech Industry. *The Journal of Creative Behavior*. <u>https://doi.org/10.1002/jocb.444</u>
- Weigt-Rohrbeck, J., & Linneberg, M. S. (2019). Democratizing Innovation Processes: Personal Initiative in Bottom-Up Eco-Innovation. *European Journal of Innovation Management*, 22(5), 821-844. <u>https://doi.org/10.1108/ejim-12-2018-0278</u>
- Weisberg, R. H. (2015). On the Usefulness of "Value" in the Definition of Creativity. *Creativity Research Journal*. <u>https://doi.org/10.1080/10400419.2015.1030320</u>
- Wen, J., Nasir, M., Yousaf, Z., Khattak, A., Yasir, M., Javed, A., & Shirazi, S. H. (2021). Innovation Performance in Digital Economy: Does Digital Platform Capability, Improvisation Capability and Organizational Readiness Really Matter? *European Journal of Innovation Management*. <u>https://doi.org/10.1108/ejim-10-2020-0422</u>

West, M. A., & Farr, J. L. (1990). Innovation at Work. John Wiley.

Wicaksono, A., & Pusparini, E. S. (2022). The Innovative Behavior During Work From Home in Indonesia: The Role of Job Autonomy and Work Engagement. <u>https://doi.org/10.2991/aebmr.k.220701.082</u>

- Wicherts, J. M., Veldkamp, C. L. S., Augusteijn, H. E. M., Bakker, M., Aert, R. C. M. v., & Assen, M. A. L. M. v. (2016). Degrees of Freedom in Planning, Running, Analyzing, and Reporting Psychological Studies: A Checklist to Avoid P-Hacking. *Frontiers in Psychology*, 7. <u>https://doi.org/10.3389/fpsyg.2016.01832</u>
- Widmann, A., & Mulder, R. H. (2018). Team learning behaviours and innovative work behaviour in work teams. *European Journal of Innovation Management*, 21(3), 501-520. <u>https://doi.org/10.1108/EJIM-12-2017-0194</u>
- Wilcox, R. (2017). Modern statistics for the social and behavioral sciences: A practical introduction (2nd ed.). CRC Press, Taylor & Francis Group.
- Williams, L. M., Brown, K. J., Palmer, D. M., Liddell, B. J., Kemp, A. H., Olivieri, G., Peduto, A., & Gordon, E. (2006). The Mellow Years?: Neural Basis of Improving Emotional Stability Over Age. *Journal of Neuroscience*. <u>https://doi.org/10.1523/jneurosci.0022-06.2006</u>
- Williamson, S., & Pearce, A. (2022). COVID-normal Workplaces: Should Working From Home Be a 'Collective Flexibility'? *Journal of Industrial Relations*. <u>https://doi.org/10.1177/00221856221094894</u>
- Wilson, J. (2014). Essentials of business research: A guide to doing your research project. Sage.
- Wipulanusat, W., Panuwatwanich, K., Stewart, R. A., & Sunkpho, J. (2019). Drivers and barriers to innovation in the Australian public service: a qualitative thematic analysis. *Engineering Management in Production and Services*, 11(1), 7-22.
- Wipulanusat, W., Sunkpho, J., & Stewart, R. A. (2021). Effect of Cross-Departmental Collaboration on Performance: Evidence From the Federal Highway Administration. *Sustainability*, 13(11), 6024. <u>https://doi.org/10.3390/su13116024</u>
- Wohlers, C., & Hertel, G. (2016). Choosing Where to Work at Work Towards a Theoretical Model of Benefits and Risks of Activity-Based Flexible Offices. *Ergonomics*. <u>https://doi.org/10.1080/00140139.2016.1188220</u>
- Wong, K., Chan, A. H. S., & Teh, P. L. (2020). How Is Work–Life Balance Arrangement Associated With Organisational Performance? A Meta-Analysis. International Journal of Environmental Research and Public Health, 17(12). https://doi.org/10.3390/ijerph17124446
- Wood, A. J., Graham, M., Lehdonvirta, V., & Hjorth, I. (2018). Good Gig, Bad Gig: Autonomy and Algorithmic Control in the Global Gig Economy. *Work Employment and Society*. <u>https://doi.org/10.1177/0950017018785616</u>
- Woods, S. A., Mustafa, M., Anderson, N., & Sayer, B. (2018). Innovative Work Behavior and Personality Traits. *Journal of Managerial Psychology*. <u>https://doi.org/10.1108/jmp-01-2017-0016</u>

- Wu, C.-H., Parker, S. K., & Jong, J. P. J. d. (2011). Need for Cognition as an Antecedent of Individual Innovation Behavior. *Journal of Management*. <u>https://doi.org/10.1177/0149206311429862</u>
- Wu, W.-Y., Bui, T. V., & Dao, T. A. (2022). The Influence of Cross-Cultural Stable and Dynamic Competencies on Expatriate Adaptation and Outcomes: The Case of Taiwan. Asia-Pacific Journal of Business Administration. <u>https://doi.org/10.1108/apjba-01-2021-0020</u>
- Xerri, M., & Brunetto, Y. (2013). Fostering Innovative Behaviour: The Importance of Employee Commitment and Organisational Citizenship Behaviour. *The International Journal of Human Resource Management*. <u>https://doi.org/10.1080/09585192.2013.775033</u>
- Xerri, M. J., & Reid, S. R. (2018). Human resources and innovative behaviour: Improving nursing performance. *International Journal of Innovation Management*, 22(02), 1850019.
- Xiu, L., Liang, X., Chen, Z., & Xu, W. (2017). Strategic Flexibility, Innovative HR Practices, and Firm Performance. *Personnel Review*. <u>https://doi.org/10.1108/pr-09-2016-0252</u>
- Xu, W., & Zammit, K. (2020). Applying thematic analysis to education: A hybrid approach to interpreting data in practitioner research. *International Journal of Qualitative Methods*, 19, 1609406920918810.
- Xu, Z., & Suntrayuth, S. (2022). Innovative Work Behavior in High-Tech Enterprises: Chain Intermediary Effect of Psychological Safety and Knowledge Sharing. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2022.1017121</u>
- Yagolkovskiy, S. (2013). Cognitive Stimulation of Individual Creativity in a Group Context: Mediated and Face-to-Face Idea Sharing. *SSRN Electronic Journal*. <u>https://doi.org/10.2139/ssrn.2286100</u>
- Yang, C.-L., Yamashita, N., Kuzuoka, H., Wang, H.-C., & Foong, E. (2022). Distance Matters to Weak Ties. *Proceedings of the Acm on Human-Computer Interaction*. <u>https://doi.org/10.1145/3492863</u>
- Yang, L., QI, L., & Zhang, B. (2022). Concepts and evaluation of saturation in qualitative research. Advances in Psychological Science, 30(3), 511.
- Yang, S., & Wu, H. (2021). The Performance Impact of New Ventures in Working Environment and Innovation Behavior From the Perspective of Personality Psychology. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.734014</u>
- Yokoi, T., Obwegeser, N., & Beretta, M. (2021). Crisis innovation: Leveraging virtual hackathons for rapid ideation. *IEEE Transactions on Engineering Management*.

- Yoo, H. S., Jung, Y. L., & Jun, S.-P. (2023). The effects of SMEs' R&D team diversity on project-level performances: evidence from South Korea's R&D subsidy program. *R and D Management*. <u>https://doi.org/10.1111/radm.12575</u>
- Yousaf, S. (2018). Influence of Servant Leadership on Project Success With Mediating Role of Team Building and Moderating Role of Trust in Pakistan. *Jinnah Business Review*. <u>https://doi.org/10.53369/ykbm8556</u>
- Yström, A., & Agogué, M. (2020). Exploring Practices in Collaborative Innovation: Unpacking Dynamics, Relations, and Enactment in In - between Spaces. *Creativity and innovation management*, 29(1), 141-145. <u>https://doi.org/10.1111/caim.12360</u>
- Yu, C., Yuan, Y., Fang, L., & Shen, W. (2021). The Influence of Trust on Creativity: A Review. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.706234</u>
- Yu, H., Yan, C., Zhang, L., Dong, Z., Ling, C., Zheng, X., & Zhao, Z.-Q. (2021). Does the Employability Paradox Exist or Not? An Inverted U-Shaped Model. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.588793</u>
- Yu, J., & Wu, Y. (2021). The Impact of Enforced Working From Home on Employee Job Satisfaction During COVID-19: An Event System Perspective. International Journal of Environmental Research and Public Health. https://doi.org/10.3390/ijerph182413207
- Yu, W., & Wang, Z. (2022). Dual Influencing Paths of Time Pressure on Employee Creativity. International Journal of Stress Management. <u>https://doi.org/10.1037/str0000267</u>
- Yuan, Y. (2021). Leader–Employee Congruence in Humor and Innovative Behavior: The Moderating Role of Dynamic Tenure. *Frontiers in Psychology*. <u>https://doi.org/10.3389/fpsyg.2021.579551</u>
- Zaltman, G., Duncan, R., & Holbek, J. (1973). *Innovations and organizations*. John Wiley & Sons.
- Zameer, H., Wang, Y., Yasmeen, H., & Mubarak, S. (2020). Green Innovation as a Mediator in the Impact of Business Analytics and Environmental Orientation on Green Competitive Advantage. *Management Decision*. <u>https://doi.org/10.1108/md-01-2020-0065</u>
- Zappalà, S., Toscano, F., & Topa, G. (2021). The Implementation of a Remote Work Program in an Italian Municipality Before COVID-19: Suggestions to HR Officers for the Post-Covid-19 Era. *European Journal of Investigation in Health Psychology and Education*. <u>https://doi.org/10.3390/ejihpe11030064</u>
- Zhang, C., & Liu, L. (2022). The Influence of Health-Promoting Leadership on Employees' Positive Workplace Outcomes: The Mediating Role of Employability and the Moderating Role of Workplace Civility. *International*

Journal of Environmental Research and Public Health. https://doi.org/10.3390/ijerph192215300

- Zheng, X., Nieberle, K. W., Braun, S., & Schyns, B. (2023). Is Someone Looking Over My Shoulder? An Investigation Into Supervisor Monitoring Variability, Subordinates' Daily Felt Trust, and Well - being. *Journal of organizational behavior*. <u>https://doi.org/10.1002/job.2699</u>
- Zikmund, W. G., Quinlan, C., & Griffin, M. (2019). *Business Research Methods*. Cengage. <u>http://ebookcentral.proquest.com/lib/vu/detail.action?docID=6335208</u>
- Zoonen, W. v., Sivunen, A., Blomqvist, K., Olsson, T., Ropponen, A., Henttonen, K., & Vartiainen, M. (2021). Factors Influencing Adjustment to Remote Work: Employees' Initial Responses to the COVID-19 Pandemic. *International Journal of Environmental Research and Public Health*. <u>https://doi.org/10.3390/ijerph18136966</u>

APPENDIX A - Ethics approval

Quest.noreply@vu.edu.au

To: Shahnaz.Naughton@vu.edu.au

Cc: snjezana.ahlgren@students.vu.edu.au; Wesley.Mcclendon@vu.edu.au

Mon 12/7/2021 12:05 PM

Dear DR SHAHNAZ NAUGHTON,

Your ethics application has been formally reviewed and finalised.

» Application ID: HRE21-055

» Chief Investigator: DR SHAHNAZ NAUGHTON

» Other Investigators:

» Application Title: Remote Working and Innovative Work Behaviours: A Case of Employees in Australian Professional Services Organisations

» Form Version: 13-07

The application has been accepted and deemed to meet the requirements of the National Health and Medical Research Council (NHMRC) 'National Statement on Ethical Conduct in Human Research (2007)' by the Victoria University Human Research Ethics Committee. Approval has been granted for two (2) years from the approval date; 12/07/2021.

Continued approval of this research project by the Victoria University Human Research Ethics Committee (VUHREC) is conditional upon the provision of a report within 12 months of the above approval date or upon the completion of the project (if earlier). A report proforma may be downloaded from the Office for Research website at: <u>http://research.vu.edu.au/hrec.php</u>.

Please note that the Human Research Ethics Committee must be informed of the following: any changes to the approved research protocol, project timelines, any serious events or adverse and/or unforeseen events that may affect continued ethical acceptability of the project. In these unlikely events, researchers must immediately cease all data collection until the Committee has approved the changes. Researchers are also reminded of the need to notify the approving HREC of changes to personnel in research projects via a request for a minor amendment. It should also be noted that it is the Chief Investigators' responsibility to ensure the research project is conducted in line with the recommendations outlined in the National Health and Medical Research Council (NHMRC) 'National Statement on Ethical Conduct in Human Research (2007).'

On behalf of the Committee, I wish you all the best for the conduct of the project.

Secretary, Human Research Ethics Committee Phone: 9919 4781 or 9919 4461 Email: researchethics@vu.edu.au

This is an automated email from an unattended email address. Do not reply to this address.

APPENDIX B – Questionnaire protocol

Part 1: Demographic Information

- 1) Gender Identity
 - 🗆 Male
 - Female
 - □ Non-binary
 - Prefer not to say
- 2) What is your age?
 - 🗆 18-22
 - 23-30
 - 🗆 31-40
 - 🗆 41-50
 - 0 51-60
 - 0 61+
- 3) What is your Employment Status?
 - 🗆 Full-time
 - Part-time
 - □ Casual/Temporary
 - □ Contract/Project based
- 4) What is your current employment level?
 - □ Administration
 - □ Not a manager or supervisor
 - □ Manager/supervisor without direct reports
 - □ Manager/supervisor with direct reports
 - □ Senior leadership/senior executive
 - CEO/GM/Owner
- 5) How long have you been at your current organisation?
 - Less than 2 years
 - Between 2-5 years
 - □ Between 6-10 years
 - \Box More than 10 years
- 6) Which of the following categories can your organisation be classified under?
 - □ Accounting Firm/Legal Firm
 - □ Arts/Design/Entertainment/Sports/Media
 - \Box Construction
 - □ Consulting
 - Education
 - Engineering
 - □ Farming, Fishing/Agriculture
 - □ Financial Services/Investment/Banking
 - □ Food and Beverage manufacturing/development
 - □ Government/public administration
 - □ Hospitality/events
Insurance

 \Box IT

- Manufacturing
- Medical/Pharma
- 🗆 Non-profit
- □ Other (please specify)
- Real Estate
- 🗆 Retail
- Sales
- □ Technical services/engineering services/architecture
- □ Telecommunications
- □ Transport/Aviation
- 7) How often do you work remotely e.g. from home?
 - \Box Once a week
 - \square 2-4 days per week
 - \Box Once a month
 - $\hfill\square$ twice a month
 - \Box 3 times per month
 - Always

Part 2: Innovative Work Behaviour and Remote Working

8) Idea Exploration

		1	2	3	4	5
When you are working AWAY from the office (e.g. at home, at a café, co-working space) how often do you:	Source of question (IWB Measurement Study)	Never	Rarely (once a month)	Sometimes (2 to 3 times per month)	Often (4 to 8 times per month)	Always
Wonder how things can be improved	De Jong and Den Hartog (2010)					
Pay attention to issues that are not part of your daily work	De Jong and Den Hartog (2010)					
Look for opportunities to improve an existing process	Kleysen and Street (2001)					
Look for opportunities to improve an existing product/service	Kleysen and Street (2001)					
Recognise opportunities to make a positive client difference	Kleysen and Street (2001)					

Keep myself	Messmann and			
informed about the	Mulder (2012)			
organisation's				
processes				
Keep myself	Messmann and			
informed about the	Mulder (2012)			
organisation's				
products/services				
Keep myself	Messmann and			
informed about	Mulder (2012)			
new work-related				
concepts				
Keep myself	Messmann and			
informed about	Mulder (2012)			
new industry-				
related				
developments				
Exchange ideas	Messmann and			
with my colleagues	Mulder (2012)			

9) Idea Generation

		1	2	3	4	5
When you are working AWAY from the office (e.g. at home, at a café, co-working space) how often do you:	Source of question (IWB Measurement Study)	Never	Rarely (once a month)	Sometimes (2 to 3 times per month)	Often (4 to 8 times per month)	Always
Try to generate solutions for problems	De Jong and Den Hartog (2010) & Kleysen and Street (2001)					
Find new approaches to execute tasks	De Jong and Den Hartog (2010)					
Generate new ideas	De Jong and Den Hartog (2010) & Kleysen and Street (2001)					
Ask critical questions about existing processes	Messmann and Mulder (2012)					
Ask critical questions about existing products/services	Messmann and Mulder (2012)					
Suggest improvements to ideas expressed by others	Messmann and Mulder (2012)					
Evaluate the strengths and weaknesses of new ideas	Kleysen and Street (2001)					

10) Idea Championing

		1	2	3	4	5
When you are	Source of	Never	Rarely	Sometimes	Often	Always
working AWAY	question (IWB		(once a	(2 to 3	(4 to 8	
from the office (e.g.	Measurement		month)	times per	times per	
at home, at a café,	Study)			month)	month)	
co-working space)						
how often do you:						
Promote new ideas	Messmann and					
to gain the support	Mulder (2012)					
of colleagues						
Promote new ideas	Messmann and					
to gain the support	Mulder (2012)					
of my managers						
Make	De Jong and Den					
organisational	Hartog (2010)					
members						
enthusiastic about						
an idea						
Persuade others of	De Jong and Den					
the importance of a	Hartog (2010) &					
new idea	Kleysen and					
Promoto now ideas	Messmann and					
in my own work	Mulder (2012)					
In my own work						

11) Idea Implementation

		1	2	3	4	5
When you are working AWAY from the office (e.g. at home, at a café, co-working space) how often do you:	Source of question (IWB Measurement Study)	Never	Rarely (once a month)	Sometimes (2 to 3 times per month)	Often (4 to 8 times per month)	Always
Contribute to implementing new ideas	De Jong and Den Hartog (2010)					
Put effort to developing new processes	De Jong and Den Hartog (2010)					
Put effort to developing new products/services	De Jong and Den Hartog (2010)					
Work out the bugs in new ideas and solutions	Kleysen and Street (2001)					
Implement beneficial changes	Kleysen and Street (2001)					

12) Would you like to participate in a one-on-one interview with the PhD student researcher regarding your experiences with working remotely?

The interview will be conducted via Zoom and will take 30 min of your time. You will be able to choose the time and date for the interview. You will be asked about your experiences and opinions of working away from the office. All information provided will be anonymous and confidential, and you will not be identified in any reports or the final thesis.

If you agree, you will be taken to a new window to enter your email address. This ensures your survey response remain anonymous.

□ Yes □ No

INFORMATION TO SURVEY PARTICIPANTS INVOLVED IN RESEARCH

You are invited to participate

You are invited to participate in a research project entitled *"Remote Working and Innovative Work Behaviours: A Case of Australian employees"*.

This project is being conducted by a student researcher Ms Snjezana Ahlgren as part of a PhD study at Victoria University under the supervision of Dr Shahnaz Naughton and Dr Wesley McClendon from College of Business, Victoria University, Australia.

Project explanation

The nature of how, where and when work gets done has fundamentally shifted over the past decade, with employees increasingly given the autonomy and flexibility to organise their work to suit the tasks they need to undertake. Another prominent development in modern economies is the ever-increasing importance of innovation for organizational success. Utilising different ways of working, such as working remotely away from the main office building, will influence employee behaviours, including innovative behaviours of employees. While the academic community has paid extensive attention to both concepts separately, not much is known about the relationship. This research study explores employee experiences in Australia, which utilise remote working practices to understand the relationship to their Innovative Work Behaviour (IWB). A mixed methods approach using employee interviews, and a questionnaire will be employed for the study, to provide deep insights into how employees experience the influence of working remotely on their innovative behaviour, and what factors may influence this relationship.

What will I be asked to do?

As a participant, you are being invited to answer questions in an online survey which will take from 10 to 15 minutes to complete. You will be asked for your experiences of working away from the main office building and how this relates to innovative behaviour. Your consent will be implied by submitting a completed questionnaire.

What will I gain from participating?

Your experiences and perceptions will provide valuable insights to the study. Your participation will make a significant contribution to the knowledge of remote working and innovative behaviour. The findings of this study will contribute to innovation management and assist practitioners to encourage employee innovation within their organisations.

How will the information I give be used?

The information provided by participants will be analysed and used for a Doctoral thesis, undertaken by Snjezana Ahlgren under the supervision of Dr Shahnaz Naughton and Dr Wesley McClendon. The raw data collected will be confidentially kept in a safe place during all stages of the project and only accessible

to the student and supervisors. Subsequent uses of records and data will be subject to standard data use policies which protect the anonymity of individuals and institutions. I understand that the researcher will not identify me by name in any reports using information obtained from this survey, and that my confidentiality as a participant in this study will remain secure.

The information may be used for the purpose of academic publication.

What are the potential risks of participating in this project?

There are no expected risks from participating in this project.

How will this project be conducted?

A mixed-methods approach will be used to investigate the aims of the study. First, an Online Questionnaire will be distributed to Australian based employees to test the hypothesis. Secondly, qualitative data will be collected from 15 semi-structured interviews. That data will be used to understand the experience of remote working and the relationship and influence this may have to their innovative work behaviour.

Who is conducting the study?

Dr Shahnaz Naughton, College of Business, Victoria University, Australia (shahnaz.naughton@vu.edu.au)

Dr Wesley McClendon, College of Business, Victoria University, Australia (wesley.mcclendon@vu.edu.au)

Snjezana Ahlgren, PhD Candidate, College of Business, Victoria University, Australia (snjezana.ahlgren@live.vu.edu.au)

Any queries about your participation in this project may be directed to the Chief Investigator listed above. If you have any queries or complaints about the way you have been treated, you may contact the Ethics Secretary, Victoria University Human Research Ethics Committee, Office for Research, Victoria University, PO Box 14428, Melbourne, VIC, 8001, email researchethics@vu.edu.au or phone (03) 9919 4781 or 4461.

INFORMATION TO PARTICIPANTS INVOLVED IN RESEARCH

You are invited to participate

You are invited to participate in a research project entitled "Remote Working and Innovative Work Behaviours: A Case of Australian employees".

This project is being conducted by a student researcher Ms Snjezana Ahlgren as part of a PhD study at Victoria University under the supervision of Dr Shahnaz Naughton and Dr Wesley McClendon from College of Business, Victoria University, Australia.

Project explanation

The nature of how, where and when work gets done has fundamentally shifted over the past decade, with employees increasingly given the autonomy and flexibility to organise their work to suit the tasks they need to undertake. Another prominent development in modern economies is the ever-increasing importance of innovation for organizational success. Utilising different ways of working, such as working remotely away from the main office building, will influence employee behaviours, including innovative behaviours of employees. While the academic community has paid extensive attention to both concepts separately, not much is known about the relationship. This research study explores employee experiences in Australia, which utilise remote working practices to understand the relationship to their Innovative Work Behaviour (IWB). A mixed methods approach using employee interviews, and a questionnaire will be employed for the study, to provide deep insights into how employees experience the influence of working remotely on their innovative behaviour, and what factors may influence this relationship.

What will I be asked to do?

Participation involves being interviewed by Snjezana Ahlgren from Victoria University, Australia. The interview will last approximately 45-60 minutes and will be conducted via video conferencing (Microsoft Teams or Zoom). Notes will be written during the interview.

The interview will only be audio recorded to protect your privacy and subsequently transcribed. You will be asked about your experiences of working away from the main office building and how this relates to innovative behaviours.

You will be required to sign and return a Consent Form for Interview Participants as evidence of your consent to participate in the interview prior to the interview date.

What will I gain from participating?

Your experiences and perceptions will provide valuable insights to the study. Your participation will make a significant contribution to the knowledge of remote working and innovative behaviour. The findings of this study will contribute to innovation management and assist practitioners to encourage employee innovation within their organisations.

How will the information I give be used?

The information provided by participants will be analysed and used for a Doctoral thesis, undertaken by Snjezana Ahlgren under the supervision of Dr Shahnaz Naughton and Dr Wesley McClendon. The raw

data collected will be confidentially kept in a safe place during all stages of the project and only accessible to the student and supervisors. Subsequent uses of records and data will be subject to standard data use policies which protect the anonymity of individuals and institutions. I understand that the researcher will not identify me by name in any reports using information obtained from this interview, and that my confidentiality as a participant in this study will remain secure. The information may be used for the purpose of academic publication.

What are the potential risks of participating in this project?

There may be potential low risk where employees may comment on their manager or colleague's behaviour during the interviews. However, the interview protocol is designed to ask the employee about their own experience and does not specifically ask them to comment on their manager or colleague's behaviour. Further, the participants can participate in the interview in their own time and away from their workplace, so there is a low probability their comments will be known to others.

How will this project be conducted?

A mixed-methods approach will be used to investigate the aims of the study. First, Questionnaires will be distributed to Australian based employees to test the hypothesis. Secondly, qualitative data will be collected from 15 semi-structured interviews. That data will be used to understand the experience of remote working and the relationship and influence this may have to their innovative work behaviour.

Who is conducting the study?

Dr Shahnaz Naughton, College of Business, Victoria University, Australia (shahnaz.naughton@vu.edu.au)

Dr Wesley McClendon, College of Business, Victoria University, Australia (wesley.mcclendon@vu.edu.au)

Snjezana Ahlgren, PhD Candidate, College of Business, Victoria University, Australia (snjezana.ahlgren@live.vu.edu.au)

Any queries about your participation in this project may be directed to the Chief Investigator listed above. If you have any queries or complaints about the way you have been treated, you may contact the Ethics Secretary, Victoria University Human Research Ethics Committee, Office for Research, Victoria University, PO Box 14428, Melbourne, VIC, 8001, email researchethics@vu.edu.au or phone (03) 9919 4781 or 4461.

CONSENT FORM FOR PARTICIPANTS INVOLVED IN RESEARCH

INFORMATION TO PARTICIPANTS:

We would like to invite you to be a part of a study into the relationship between remote working practices and innovative work behaviours, of Australian employees. The results of this study are expected to provide an employee perspective framework that describes the influence of remote working practices and innovative work behaviours.

CERTIFICATION BY PARTICIPANT

I,, of

certify that I am at least 18 years old* and that I am voluntarily giving my consent to participate in the study: *"Remote Working and Innovative Work Behaviours: A Case of Employees in Professional Services Organisation"* being conducted at Victoria University by Dr Shahnaz Naughton, Dr Wesley McClendon and PhD Candidate Snjezana Ahlgren.

I certify that the objectives of the study, together with any risks and safeguards associated with the procedures listed hereunder to be carried out in the research, have been fully explained to me by Snjezana Ahlgren and that I freely consent to participation involving the below mentioned procedures:

- My participation in this project is voluntary. I understand that I will not be paid for my participation.
- I may withdraw and discontinue participation at any time without penalty.
- Most participants will find the discussion throughout the interview interesting and thought provoking. If, however, I feel uncomfortable in any way during the interview session, I have the right to decline to answer any question or to end the interview.
- Participation involves being interviewed by Snjezana Ahlgren from Victoria University.
- The interview will last approximately 45-60 minutes. Notes may be written during the interview.
- Only an audio recording of the interview will be taken to protect my privacy. No video recording of the interview will be made. If I do not want to be audio taped, I will not be able to participate in the study.
- I understand that the researcher will not identify me by name in any reports using information obtained from this interview, and that my confidentiality as a participant in this study will remain secure. Subsequent uses of records and data will be subject to standard data use policies which protect the anonymity of individuals and organisations.
- Staff and other managers from my company will neither be present at the interview nor have access to raw notes or transcripts. This precaution will prevent my individual comments from having any negative repercussions.

I certify that I have had the opportunity to have any questions answered and that I understand that I can withdraw from this study at any time and that this withdrawal will not jeopardise me in any way.

I have been informed that the information I provide will be kept confidential.

Signed:

Date:

Any queries about your participation in this project may be directed to the researcher Dr Shahnaz Naughton +61 3 9919 1923 or Dr Wesley McClendon +61 (0) 432 419 116.

If you have any queries or complaints about the way you have been treated, you may contact the Ethics Secretary, Victoria University Human Research Ethics Committee, Office for Research, Victoria University, PO Box 14428, Melbourne, VIC, 8001, email Researchethics@vu.edu.au or phone (03) 9919 4781 or 4461.

APPENDIX F – Interview protocol

PART 1. PARTICIPANT'S BACKGROUND

1) To begin, please tell me about your work background (e.g. employment status; years of service at current organisation; highest education level; department, role level)

2) Tell me a little about your current role and what your responsibilities are?

PART 2: USAGE OF REMOTE WORKING

3) Where do you usually work when working remotely? Please think about your working routine prior to COVID-19 as well as currently e.g. home, cafe

4) How many days per week do you work remotely? Please think about your working routine prior to COVID-19 and currently.

5) What are the main reasons for working remotely when you do? (Aside from COVID-19 restrictions)

PART 3: ABILITY (A), MOTIVATION (M), OPPORTUNITY (O) FACTORS

6) (O) What resources are you provided with by your organisation to enable remote working? E.g. Tangible (laptop, internet access, mobile phone, home office equipment, home office furniture), Intangible (support, courses, mental health initiatives)

7) (O) What additional resources have you acquired yourself to enable remote working? Why did you purchase these items?

8) In terms of your role responsibilities, do you feel you can be innovative? Probe: why/why not? Is your experience the same in the office as working remotely?

9) (M) In terms of your approach to risk in your work life, would you say you are risk averse, a risk taker, or takes risk if you see an opportunity?

10) (M) Would you describe yourself as either an extrovert or introvert in your work life?

11) (M) How do you manage your well-being and balance when working remotely?

12) (O) In your view, what are some of the positives and negatives of working remotely?

13) (O) How important is it to you to have the option of working remotely? Why?

Idea Exploration

14) (O) Where do you normally do your" thinking" for work? probe: In the office or when working remotely, where are you sitting? Why there?

15) (O) What do you find yourself thinking about? probe: Do you find yourself thinking about new ideas, or looking at ways things can be improved/changed? How does this compare to when you are in the office? Why? What works/what doesn't?

16) (M) Where do you feel more "creative"? E.g., In the office or working remotely? Why?

17) Is this the same for your "operational" or "productive" type of work?

Idea Generation

18) (A) Collaboration refers to working jointly with one or more team members on an activity or project. When working remotely, how do you collaborate with your work colleagues? Probe: how does the collaboration experience compare to when you are working in the office? What works/doesn't work?

Idea Championing

19) (A) When you have new ideas/thoughts, do you share them with colleagues, managers etc? probe: why/why not? How do you share them? is this different to how you would share when you are in the office? Is it effective/not as effective?

20) (O) When you are not working in the office, are you hearing about new ideas from your workplace? probe: why/why not? Do you promote these new ideas to others or talk about them with others? Why/why not?

21) (A) What about your interactions with your teams/colleagues when working remotely? probe: interacting more/less? Barriers/challenges? Relationships? Ability to network?

22) (A) In your experience, how important is face time with your colleagues? Probe: which aspects? Any particular tasks?

Idea implementation

23) (A) As part of your role, would you contribute to the development or implementation of new ideas/changes? probe: Yes/No? Why? How does your remote experience compare to when you are in the office? What works/what doesn't? Barriers?

24) (M) Do you know of any incentives that your organisation provides to encourage employees to be innovative? E.g. programs, rewards, competitions probe: Is your experience the same when working remotely?